AN ORDINANCE 2016-10-20-0815

AUTHORIZING THE SUBMISSION OF A HEAD START GRANT APPLICATION TO THE U.S. DEPARTMENT OF HEALTH AND HUMAN SERVICES AND ACCEPTANCE UPON AWARD OF \$22,483,945.00 FOR THE PERIOD FEBRUARY 1, 2017 – JANUARY 31, 2018, A BUDGET OF \$28,104,932.00, WHICH INCLUDES AN IN-KIND MATCH OF \$5,620,987.00 AND A PERSONNEL COMPLEMENT OF 94 POSITIONS FOR THE DEPARTMENT OF HUMAN SERVICES; AUTHORIZING FOUR SERVICE PROVIDER AGREEMENTS IN A CUMULATIVE AMOUNT OF \$16,356,396.00 WITH RENEWAL OPTIONS UNTIL THE END OF THE GRANT TERM ON JULY 31, 2018.

* * * * *

WHEREAS, the Head Start program is a federally-funded early childhood education program administered by the U.S. Department of Health and Human Services and provides education and family support services to children between the ages of three and five; and

WHEREAS, the City, through the Department of Human Services, has served as a grantee for the Head Start Program for over 30 years and currently serves 3,020 children and their families in 26 Head Start centers located in the San Antonio and Edgewood Independent School Districts; and

WHEREAS, on August 11, 2016, the City received notification from HHS of available funding in the amount of \$22,483,945.00 for the continuation of Head Start program services through January 31, 2018; NOW THERFORE:

BE IT ORDAINED BY THE CITY COUNCIL OF THE CITY OF SAN ANTONIO:

SECTION 1. The City Manager or her designee, or the Director of the Department of Human Services (DHS) or her designee, is authorized to submit a grant application to the U.S. Department of Health and Human Services and to accept funds upon award in an amount of \$22,483,945.00 for the City of San Antonio Head Start Program for the period February 1, 2017 through January 31, 2018. The City Manager or her designee, or the Director of the DHS or her designee, is authorized to execute any and all necessary documents to effectuate the application and acceptance. A copy of the application is on file with DHS.

SECTION 2. The City Manager or her designee, or the Director of DHS or her designee, is further authorized to negotiate and execute, after review and approval of the City Attorney's Office, (1) the four service provider agreements attached hereto and incorporated herein in substantially final form for all purposes as **Attachments I-IV**, in a cumulative amount of \$16,356,396.00, (2) contract amendments based on program improvements or additional awards, so long as matching funds are available in City's budget, and (3) renewal agreements without further City Council approval based on additional award until the end of the grant term on July 31, 2018.

KD/vv 10/20/16 Item #12

SECTION 3. Payment not to exceed the budgeted amount up to \$16,356,396.00 is authorized for the following and should be encumbered with a purchase order.

	FY2016			
Provider	Allocation			
San Antonio ISD	\$12,053,245.00			
Edgewood ISD	\$4,171,411.00			
University Health System	\$20,000.00			
San Antonio Metropolitan Health Department	\$111,740.00			
Total	\$16,356,396,00			

SECTION 4. A proposed budget of \$28,104,932.00, including an in-kind matching budget, is attached hereto and incorporated herein for all purposes as **Attachment V**, and is hereby approved. Upon acceptance of this award, new funds and internal order numbers will be created, upon which the award amount of \$22,483.945.00 from the HHS will be appropriated in said fund. A formal final budget which will include a department specific fund, an Internal Order number, and General Ledger numbers will be submitted by the department upon award.

SECTION 5. The proposed personnel complement of ninety-four (94) positions, attached hereto and incorporated herein for all purposes as **Attachment VI**, is hereby approved.

SECTION 6. The financial allocations in this Ordinance are subject to approval by the Director of Finance, City of San Antonio. The Director of Finance, may, subject to concurrence by the City Manager or the City Manager's designee, correct allocations to specific SAP Fund Numbers, SAP Project Definitions, SAP WBS Elements, SAP Internal Orders, SAP Fund Centers, SAP Cost Centers, SAP Functional Areas, SAP Funds Reservation Document Numbers and SAP GL Accounts as necessary to carry out the purpose of this Ordinance.

SECTION 7. This ordinance is effective immediately upon the receipt of eight affirmative votes; otherwise, it is effective ten days after passage.

PASSED AND APPROVED this 20th day of October, 2016.

M A Y O R
Ivy R, Taylor

ATTEST:

ticia M. Vacek, City Clerk

APPROVED AS TO FORM:

City Attorne

Agenda Item:	12 (in consent v	ote: 5, 6, 7, 8,	9, 10, 11, 1	2, 14, 15,	16, 18, 19, 20, 23	3)					
Date:	0/20/2016										
Time:	09:27:27 AM	:27:27 AM									
Vote Type:	Motion to Approv	tion to Approve									
Description:	An Ordinance authorizing the submission of a Head Start grant application to the U.S. Department of Health and Human Services and acceptance upon award of an amount up to \$22,483,945.00 for the period of February 1, 2017 to January 31, 2018, a budget of \$28,104,932.00, which includes an in-kind match of up to \$5,620,987.00, and a personnel complement of 94 positions for the Department of Human Services; authorizing four service provider agreements in a cumulative amount of \$16,356,396.00 with renewal options until the end of the grant term on July 31, 2018. [María Villagómez, Assistant City Manager; Melody Woosley, Director, Human Services]										
Result:	Passed										
Voter	Group	Not Present	Yea	Nay	Abstain	Motion	Second				
Ivy R. Taylor	Mayor		x								
Roberto C. Treviño	District 1		x			x					
Alan Warrick	District 2		X				X				
Rebecca Viagran	District 3	х									
Rey Saldaña	District 4		X								
Shirley Gonzales	District 5		x								
Ray Lopez	District 6		х								
Cris Medina	District 7		x								
Ron Nirenberg	District 8		x								
Joe Krier	District 9		x								
Michael Gallagher	District 10		X								

INTERDEPARTMENTAL AGREEMENT

PROJECT NAME: Head Start Program Dental Services

This Interdepartmental Agreement (hereinafter referred to as "AGREEMENT") is hereby made and entered into by and between the Department of Human Services (hereinafter referred to as "DHS") of the City of San Antonio (hereinafter referred to as "CITY") and the San Antonio Metropolitan Health District, also referred to as the CITY's Health Department (hereinafter referred to as "DEPARTMENT").

WHEREAS, CITY, through DHS, has received certain funds from the U.S. Department of Health and Human Services (HHS) as Grantee for the Head Start Program serving the San Antonio and Edgewood Independent School Districts; and

WHEREAS, the City Council has adopted a budget for such funds and has included therein, pursuant to Ordinance No._______, dated _______, the allocation of up to One Hundred Eleven Thousan Seven Hundred Forty Dollars (\$111,740.00), for the provision of Dental Services (hereinafter referred to as "Project"); and

WHEREAS, the DHS wishes to engage DEPARTMENT to implement and manage said Project; and

WHEREAS, in consideration for the tasks to be performed by DEPARTMENT hereunder, such allocated funds shall be available for use by DEPARTMENT, as herein authorized; NOW THEREFORE:

The parties hereto severally and collectively agree, and by the execution hereof are bound, to the mutual obligations herein contained and to the performance and accomplishment of the tasks hereinafter described.

I. TERM

1.1 Except as otherwise provided for pursuant to the provisions hereof, this AGREEMENT shall commence on February 1, 2017 and shall terminate on January 31, 2018.

II. COMPLIANCE WITH FEDERAL, STATE, AND LOCAL LAWS

2.1 DEPARTMENT understands that funds provided to it pursuant to this AGREEMENT are funds which have been made available to CITY by the federal government under the Head Start Act and in accordance with CITY's HHS-approved Funding Application and with other specific assurances made and executed by CITY. DEPARTMENT, therefore, assures and certifies that it will comply with the requirements of the Head Start Program, with all regulations promulgated thereunder as codified at

Title 45 of the Code of Federal Regulations, and with any and all applicable amendments or revisions to said Head Start Act or regulations, as well as with the Special Provisions, affixed hereto and incorporated herein for all purposes as **Attachment III**.

2.2 DEPARTMENT shall observe and comply with all city, state and federal laws, regulations, ordinances, and codes affecting DEPARTMENT's operations pursuant to the AGREEMENT.

III. MAINTENANCE OF EFFORT

3.1 DEPARTMENT agrees that the funds and resources provided to it under the terms of this AGREEMENT shall in no way be substituted for funds and resources provided from other sources, nor shall such funds and resources in any way serve to reduce the funds, resources, services, or other benefits which would have been available to, or provided through, DEPARTMENT had this AGREEMENT not been executed.

IV. PERFORMANCE

- 4.1 DEPARTMENT shall manage, implement, perform, provide, and carry out in a timely manner all of the tasks, activities, and services set forth in the Scope of Work (attached hereto and incorporated herein for all purposes as **Attachment I**), and utilizing funds only in the manner allocated in the Project Budget (attached hereto and incorporated herein for all purposes as **Attachment II**).
- 4.2 Modifications or alterations to the Scope of Work or the Project Budget may be made only pursuant to the prior written approval of DHS.

V. FISCAL

- 5.1 Inasmuch as the City Council has designated to DHS the responsibility for fiscal oversight, control and monitoring of the Head Start Program, DEPARTMENT shall submit to DHS, for written approval prior to final processing, all Project matters fiscal in nature including, but not limited to, requests for payment, purchase orders, requisitions, budget adjustments and invoices, as required by Federal regulations and Head Start requirements. DEPARTMENT agrees and understands that funds are awarded only for the current fiscal year. DEPARTMENT is solely responsible for submitting any and all invoices incurred, through January 31, 2018 no later than 45 days from the expiration or early termination date of this AGREEMENT, unless the DEPARTMENT receives written authorization from the Director of DHS prior to such 45 day period allowing DEPARTMENT to submit a request for payment after such 45 day period.
- 5.2 The DEPARTMENT shall submit to DHS a full accounting of the Program Income and non-Federal Share funds received and total Program costs incurred. DEPARTMENT shall be further responsible for submitting all requests for payment for the period February 1, 2017 through January 31, 2018 no later than March 16, 2018.

- 5.3 Prior to commencement of this AGREEMENT, DEPARTMENT shall submit to DHS for its approval a monthly budget by line item for the entire term of this AGREEMENT along with its Program Budget, including detail by category alone. Additionally, throughout the term of this AGREEMENT, DEPARTMENT shall submit on or before the last day of each month a forecast of the projected monthly expenses for each month remaining in the AGREEMENT so that DHS may review and compare actual expenses to projected expenditures and address issues associated with DEPARTMENT'S expenditure rate (e.g., on or before March 31, 2017, DEPARTMENT shall submit the projected expenses by month for April 2017 through January 2018). DEPARTMENT'S budgeted development and administrative costs (as defined by 45 C.F.R. §1301.32) shall not exceed twelve percent (12%) of the Program Budget, unless the total Program Budget is modified in accordance with this AGREEMENT in which case this amount shall be reduced proportionately unless the Parties otherwise agree.
- 5.4 Approval required. DEPARTMENT shall seek and obtain DHS's Head Start Program Administrator and the Fiscal Administrator's prior written approval 30 calendar days before making budget modifications/revisions; in no event shall the total budget amount exceed the AGREEMENT amount in section 6.1 of this AGREEMENT. DHS may make exceptions to the 30-day notice requirement on a case by case basis, but DEPARTMENT must make the request in writing or via e-mail to the DHS Head Start Program Administrator. DEPARTMENT'S written request must be accompanied by a justification for the change and indicate which lines items are affected by such change

VI. BILLING AND PAYMENT

- 6.1 DHS will pay the DEPARTMENT a total not to exceed One Hundred Eleven Thousand Seven Hundred Forty Dollars (\$111,740.00) for completing objectives and activities listed in the attached Scope of Work, which shall also include reimbursement of outlined program costs. All obligations of DHS under this AGREEMENT shall be paid from Head Start funds.
- 6.2 The DEPARTMENT will bill DHS on a monthly basis for services provided, to include those services provided directly by DEPARTMENT, and any allowable costs or expenses from contractors or partners (including, but not limited to, the University of Texas Health Science Center San Antonio Dental School).
- 6.3 DHS shall remain liable for the payment of services rendered under this AGREEMENT until all such payments are made and received by DEPARTMENT.
- 6.4 DEPARTMENT'S Program Budget is comprised of the Federal and Non-Federal Share. Should DEPARTMENT fail to raise all of the Non-Federal Share funds (20% of the total Program Budget, or Twenty Seven Thousand Nine Hundred Thirty Five Dollars (\$27,935.00) it is required to raise for the operation of its Program, DHS reserves the right to limit its reimbursements to DEPARTMENT proportionately. To meet the requirements of this AGREEMENT, all claimed non-Federal Share must meet the requirements of 2 C.F.R Part 200 or 45 C.F.R. 75, as applicable.

VII. PROGRAM INCOME

- 7.1 For purposes of this AGREEMENT, "program income" shall mean earnings of DEPARTMENT realized from activities resulting from this AGREEMENT or from DEPARTMENT's management of funding provided or received hereunder. Such earnings shall include, but shall not be limited to, interest income, usage or rental/lease fees, income produced from contract-supported services of individuals or employees or from the use of equipment or facilities of DEPARTMENT provided as a result of this AGREEMENT, and payments from clients or third parties, if applicable, for services rendered by DEPARTMENT pursuant to this AGREEMENT. The parties agree that the DEPARTMENT will be required to return program income funds to CITY through DHS, as set out in Section VI above and in Attachment I.
- 7.2 On a monthly basis, DEPARTMENT shall report and return to DHS all program income received or accrued during the preceding month except for those proceeds set out above in section 6.1 and in Attachment I. Alternative arrangements to this requirement may be made only upon written request to and written approval by DHS.
- 7.3 Records of the receipt and disposition of program income shall be maintained by DEPARTMENT in the same manner as required from other AGREEMENT funds and shall be submitted to DHS in the format prescribed by DHS.
- 7.4 DEPARTMENT shall include this Article, in its entirety, in all of its subcontracts involving income-producing services or activities.
- 7.5 It shall be DEPARTMENT's responsibility to obtain from DHS a prior determination as to whether or not income arising directly or indirectly from this AGREEMENT, or from the performance thereof, constitutes program income, and unless otherwise approved in writing by DHS, DEPARTMENT shall be responsible to DHS for the repayment of any and all amounts determined by DHS to be program income.

VIII. MAINTENANCE AND ACCESSIBILITY OF RECORDS

- 8.1 DEPARTMENT agrees to maintain records that will provide accurate, current, separate, and complete disclosure of the status of any funds received pursuant to this AGREEMENT. DEPARTMENT further agrees:
- (A) That maintenance of said records shall be in compliance with all terms, provisions and requirements of this AGREEMENT and with all applicable federal and state regulations establishing standards for financial management; and
- (B) That DEPARTMENT's record system shall contain sufficient documentation to provide in detail full support and justification for each reimbursement.

8.2 As often, at such times, and in such form as DHS may require, DEPARTMENT shall, upon DHS' verbal or written request, make available and furnish to DHS any and all statements, reports, data, and information deemed by DHS to pertain to matters covered by this AGREEMENT.

IX. PERFORMANCE REPORTS

- 9.1 In addition to all those listed in this AGREEMENT, DEPARTMENT shall submit to DHS any and all reports as may be required by HHS or DHS. DEPARTMENT shall incorporate and use any DHS-approved tracking or information system (e.g., ChildPlus) for the delivery of comprehensive Head Start Services and collect, input and update all data in accordance with DHS' planned timeline to ensure the reporting of accurate and consistent information to HHS.
- 9.2 Additionally, DEPARTMENT shall maintain and furnish to DHS the following financial and programmatic information and reports, in such forms as DHS may prescribe, as required under the Head Start Act, as amended, and Title 45 of the Code of Federal Regulations, to include Part 75:
- (A) Head Start Service Provider report to Head Start Policy Council shall be submitted on a monthly basis;
- (B) Financial reports showing all actual and/or projected costs of the Project, an explanation of budgetary expenditures, Program Income, non-Federal Share amounts;
- (C) The results of the most recent financial audit;
- (D) The number of enrolled children that received oral health services by the DEPARTMENT;
- (E) Reports showing employee credentials and a list of personnel serving to satisfy the DEPARTMENT's in-kind non-Federal Share requirement;
- (F) Reports showing the wages of each employee; and
- (G) Any other information requested by DHS.

The DEPARTMENT shall maintain all applicable and appropriate supporting documentation of costs, including but not limited to, payroll records, invoices, contracts or vouchers, and make these available to DHS upon request.

X. CHANGES AND AMENDMENTS

10.1 Alterations, additions or deletions to the terms of this AGREEMENT shall be by written amendment executed by both DHS and DEPARTMENT, except those made by budget revisions as allowed in section 5.4 herein.

XI. SPECIAL CONDITIONS

- 11.1 DEPARTMENT agrees to comply with all laws, ordinances, codes, rules, regulations, policies, and procedures, including all licensing standards and all applicable accreditation standards, applicable to the funds received by DEPARTMENT hereunder as directed by DHS or as required in this AGREEMENT, including but not limited to:
- (A) The Head Start Act (42 U.S.C. §9831 et seq., as amended);
- (B) 45 C.F.R. Part 1301 et seq.;
- (C) The terms and conditions of HHS Grant Number _____ awarded to CITY, the document attached hereto and incorporated herein for all purposes as Attachment IV, as well as relevant Head Start information memorandum and publications issued by HHS;
- (D) As applicable, 45 C.F.R. Part 75 (Grants Administration regulations);

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

EXECUTED this	day of	
DEPARTMENT OF H SERVICES	IUMAN	SAN ANTONIO METROPOLITAN HEALTH DISTRICT
BY:		BY: Me 36
Melody Woosley Director		Vincent R. Nathan, PhD, MPH Interim Health Director

The Office of Management and Budget (OMB) Circular at 2 C.F.R. Part 200 et

ATTACHMENTS

(E)

Attachment I – Scope of Work Attachment II - Project Budget Attachment III - Special Provisions Attachment IV – HHS Notice of Award

Contract #:			
Attachment	I-Scor	e of	Work

SCOPE OF WORK

1. Summary

The San Antonio Metropolitan Health District (hereinafter referred to as DEPARTMENT), will provide oral health examinations and preventive care; including fluoride varnish treatments for children enrolled in the Head Start program ("enrollee(s)"). These examinations required by federal guidelines for the Head Start program, will be provided on site in each Head Start Center by a Metro Health dentist. These assessments are necessary to ensure that all children enrolled in the program gain access to quality dental care, and that those identified with unmet dental needs are referred to an appropriate dental provider for treatment. The Department of Human Services (hereinafter referred to as DHS), as the City department responsible for overseeing the Head Start Program, will cooperate with the DEPARTMENT to coordinate with Head Start contractors to ensure that enrollees receive dental screenings, recommended preventive services and designated follow-up care.

2. Program Scope of Services

DEPARTMENT agrees:

- 2.1 No later than 90 calendar days of the child's entry into the Program, and as needed throughout the program, to complete an initial oral health evaluation for each participating child. Dentists from Metro Health will travel to all Head Start centers to perform an oral health evaluation for each child. As appropriate, Metro Health dentists will prescribe fluoride varnish for participants, which will be applied by Metro Health dental staff and/or dental students and dental hygiene students from the UT Health Science Center School of Dentistry pursuant to section 2.15 of this Scope of Work under the supervision of Metro Health dental staff.
- 2.2 This Section intentionally left blank.
- 2.3 To input the individual evaluation results and document all preventive services provided by Metro Health into the Child Plus database within 3 days of an oral health exam, and to provide a monthly case management report of individual's identified as having Class 1 Urgent needs.
- 2.4 To input into Child Plus any pertinent documentation sent by local practitioners or Family Support Workers regarding participants' oral health status.
- 2.5 This section intentionally left blank.
- 2.6 To provide written referral and/or correspondence to the enrollee's parent explaining findings of the dental screenings. This document of the child's oral health status, along with contact information for the DEPARTMENT dental staff, will be given to the appropriate Head Start service provider(s) staff to be forwarded to parents/care givers.

Contract #:			
Attachment	I-Scope	of	Work

- 2.7 To provide individual case management services for children with urgent dental needs and children who are uninsured or underinsured for required dental services. Metro Health will serve as a resource for Family Support Workers to help families navigate the health care system.
- 2.8 To perform all intake and eligibility functions for Title V Maternal Child Health and Dental (MCHD) Grant funding for Head Start children who are not eligible for dental services through Medicaid or CHIP and to enroll families that qualify for services
- 2.9 To coordinate documentation and facilitate the referral and initial appointment with the University of Texas Health Science Center dental clinics and affiliated sites for comprehensive dental treatment.
- 2.10 To perform all billing and reporting functions for enrolled children and ensure that all copayments are waived for Title V eligible services.
- 2.11 To utilize available funding to support additional costs related to treatment including, but not limited to, facility fees and physician fees not covered by Title V Maternal Child Health Funding for dental services. In the event a Head Start child must be treated in a hospital setting, additional services will be coordinated through the University of Texas Health Science Center at San Antonio School of Dentistry, Department of Pediatric Dentistry (UT Health Science Center Dental School) in designated University-approved sites. Additional costs incurred by the UT Health Science Center may be included in reimbursement for services.
- 2.12 To provide ongoing training and technical assistance for Head Start teachers, staff and other oral health stakeholders during the 2016-2017 and 2017-2018 academic school year including development and production of training manuals for Head Start staff and community partners for CavityFree Kids. DEPARTMENT will provide at least two trainings per school year and on an as needed basis thereafter.
- 2.13 To comply with any and all other conditions, covenants, provisions and/or requirements contained herein requiring performance by Department.
- 2.14 To collaborate with the UT Health Science Center School of Dentistry and Head Start service providers for the purposes of providing service learning opportunities in Head Start including but not limited to dental students, dental hygiene students and dental public health residents.
- 2.15 To make best efforts to provide each participating enrollee with a minimum of two (2) fluoride varnish applications during the Head Start 2017-2018 grant year and will reassess the oral health status of children identified with untreated decay documented during the initial evaluation.
- 2.16 To conduct an ongoing monitoring of operations throughout the program year notifying the DHS when the DEPARTMENT identifies possible or actual lack of compliance with the Head Start Performance Standards, Head Start Act, DHS's program policies, or the terms of this AGREEMENT. DEPARTMENT will make copies of monitoring reports available to the DHS upon request.

Contract #:			
Attachment	I-Scope	of	Work

- 2.17 To seek and obtain the DHS's written approval before making any material revisions in DEPARTMENT's Head Start services that conflicts or violates (i) the City's Funding Application, including amendments, to the U.S. Department of Health and Human Services (HHS), (ii) the Terms of the Grant, and (iii) the terms of this agreement.
- 2.18 To make time and resources available to support: (i) participation by DEPARTMENT staff in meetings with Head Start staff for community assessment, self-assessment, strategic planning, development of training and technical assistance plan, communication and program development activities, Directors Meetings, and Health Coordinator Meetings (ii) participation in technical assistance trainings and service enhancements developed by DHS and the Head Start training and technical assistance service provider, as well as other Head Start trainings that may be developed by relevant federal or state agencies; and (iii) an appropriate level of attendance of DEPARTMENT's program management team at national, regional and/or State Head Start conferences/trainings.
- DHS and Head Start service providers collaboration and obligations

DHS agrees:

- 3.1 To inform Head Start service providers of available services and to encourage participation in the Departments' Oral Health Program for Head Start Children.
- 3.2 To provide access to the Child Plus database to DEPARTMENT and to ensure that Head Start service providers conduct basic administrative functions regarding enrollees including but not limited to providing monthly reports on follow-up of referrals and allowing regular audits of dental charts as determined to be necessary by the DEPARTMENT to verify that follow-ups were performed.
- 3.3 To instruct Head Start service providers that they must defer to the dentist for determination of the appropriate timeframe for follow-up as indicated in the referral information.
- 3.4 To inform Head Start service providers that the child must have written documentation of parental consent to access program services including dental examinations and fluoride varnish applications. DHS agrees and understands that the DEPARTMENT shall have the right to refuse to provide services or treatment to any child that does not have acceptable written documentation of parental consent.
- 3.5 To instruct Head Start service providers to obtain from each enrollee's parent or legal guardian pertinent medical history for the DEPARTMENT's use and reference at the time of exam or treatment, including record of any current medical conditions for which the child is being followed, and a record of any medications the child is currently taking prior to the time services are rendered by the DEPARTMENT.
- 3.6 To instruct Head Start service providers that they must provide Medicaid, CHIP or other third-party insurance information on enrollees to the DEPARTMENT, and to make every effort to encourage enrollment of potentially eligible children to third-party funding programs. For children who are not enrolled in Medicaid, CHIP, or other third-party

Contract #:			
Attachment	I-Scope	of	Work

- insurance, the DHS, or Head Start service providers, will provide social security numbers for identification purposes.
- 3.7 To assist uninsured or underinsured families in completing Title V MCHD Grant funding eligibility documentation required for coverage through the DEPARTMENT'S treatment program offered through the UT Health Science Center School of Dentistry.
- 3.8 To certify that all costs herein provided for reimbursement to the DEPARTMENT are allowable costs under the grant guidelines.
- 3.9 To pay for services rendered by the DEPARTMENT, within 30 calendar days of receiving a valid and approved Request for Payment, in accordance with the provisions contained in Article VI of the Interdepartmental Agreement between the parties.
- 3.10 To instruct Head Start service providers to designate a staff member of each organization to coordinate appointments with DEPARTMENT staff and to instruct Head Start service providers that they should notify DEPARTMENT staff at least 48 hours in advance of any cancellations or changes in scheduling.
- 3.11 To ensure Head Start service provider staff provides the appropriate assistance to Department staff and prepares clinical area for services prior to the arrival of the dental team.

Provider: Term: City of San Antonio Head Start Program San Antonio Metro Health, Support Services February 1, 2017 to January 31, 2018

Category	Federal Share	Non Federal Share	Total Budget	Federal Admin	Non Federal Admin	Total Admin Budget
Salaries	0	0	0	0	0	0
Fringes	0	0	0	0	0	0
Travel	2,500	0	2,500	0	0	0
Equipment	0	0	0	0	0	0
Supplies	17,240	0	17,240	0	0	0
Contractual	88,000	0	88,000	0	0	0
Other	4,000	27,935	31,935	0	0	0
otal Budget	111,740	27,935	139,676	0	0	0

Salel A 8/23/16
Signature and Date of Agency Authorized Representative

Section below for COSA use only	
Programmand Fiscal Review and Approval:	ima maket
MA Augus 31 Ang 2016	Mynahuan 08/31/10
Program Mobilitar Signature and Date Muca Danen 9/2/16	COSA Head Start Administrator Signature and Date
Fiscal Monitor Signature and Date 9/1/16	Fiscal Manager Signature and Date
DEGEIVED AUG 31 2016	

Э
_
-
=
-
2
-
•
-
-
-
-
-
•
-
-
-
•
-
-
- 1
-
-
-
•
_
c
-
٠,
-
-
Œ
-
6

美國 100年		The state of the s		-	1	AND STREET, STREET,					ALCOHOL: MARKET	The Real Property of the Parket
						ESTAB		Nex	Description and	Prom.d.	April Medical Control during the Control Medical Control	
CONTROL OF		JOSEPH SER ENGINEERS	Troutestant	SALES WAS DO	FORMS	TENNAM TENNAM	SAR SEE	HIGH TOPOLET TODGET	Bigggg	在 随时的	Ding	4 45 ENGLISH
Selaries						0						
Salaries Salaries			-			0					THE PERSON AND ADDRESS OF THE PERSON ADDRESS OF THE PERSON AND ADDRESS OF THE PERSON ADDRESS OF THE PERSON AND ADDRESS OF THE PERSON ADDRESS OF THE PERSON AND ADDRESS OF THE PERSON AND ADDRESS OF THE	
Salaries		***************************************				0					- A RIPECINA	AND IN A WARR OF STREET
Salaries						0						the section of the
Salaries			-			0					CONTRACTOR OF THE PARTY OF THE	The same of the sa
alarles			-			0						-
Salaries Salaries			-			0				-	Company of the Control of the Contro	THE RESERVE TO SHARE
Salaries						0						Profes Mere
Salaries						0					THE OWNER OF THE OWNER, WHEN	CAMPBELL AND THE
Salaries						0					A Tarage Charles	Personal Control
Salaries						0						Large Marting
Salaries		TWY COLL SELECTION OF THE SELECTION OF T	-			0					*********	Water Street
Salaries Salaries			-			0		-		-	Company Committee	A PROPERTY OF THE
Salaries			-			0				-	Charles Investment	STRUME SOFTIN
Salaries						0					THE SECOND	ATT WAS ASSETT
alarles						0						11-12-12-00
alaries						0						- Marinda to the
Salaries						0					THE REAL PROPERTY.	Charles and Charles
Salarles			-			0					MANUAL PROPERTY AND	THE THEOLOG
Salaries Salaries			-			0					PASTO OF FREE	A CHARLES
lalarios			+	-		0						
alarles	-					0					Sector Laboration	THE PERSON OF
alaries						0					70 Television	CALLED CONT.
Salaries						0						A MARKET MARK
Salarles	-		-			0					Marshorn Control	Will self
Salaries Salaries	-		+	-		0				-	CANCEL CONTRACTOR	A PROPERTY AND
Salaries	-		-			1				-	Water Sandanielle	The second
Salaries						0					THE PERSON NAMED IN	A CHICAGO
Salaries						0					Salvan Bergine	
Salaries					1111111				0		DESCRIPTION OF STREET, SAME	STATE OF STREET, STREE
Selaries						0			0		NAME AND THE	Europe Contract
Salaries Salaries			-			0			0	-		-
Salaries											CAN THE STREET	January .
Salaries				-		1				-		The same of
Salaries						0			0		The same of the	STATE OF THE PARTY
Salaries									0 0		MARIE THE DESIGNATIONS	O' Shield Married Street, Street, or other Con-
Salaries		apparents							0 0		SCHOOL SHOWS	
Salaries	-		-						0	A. Carrier and Car	CALL STRAIN	
Salaries Salaries	-		-		-				0 0		THE PERSON NAMED IN	diam's
Salariea			-	-		1			0 0		ted has to settle	and the same
Salaries					-	1			0 0		THE RESIDENCE THE	The state of the state of
Salarles									0		1000 1000	Section we seem
Salarles									0 0		The second second	The state of the s
Salaries									0 0		THE PERSON NAMED IN	
Salaries									0			

Upolated

В	C	D	E	F	G	H		J	K	L	M	N
Inge		FICA										
inge		Life Insurance; Dental; Long Term Disability										
inge		Health Insurance										
inge		Retirement										
ringe		Worker's Compensation										
ringe		Unemployment Ins.										
ravel	5207010		(Prior appro	volue bns lav	ting documents.	(hankupa		2,500	2,500			
quipment	3201010	Equipment >\$5,000 per unit cost, enter specific d					4	2,000	4000			
		Equipment >\$5,000 per unit cost, enter specific d	(Prior Appro	val a supports	o documents req	uled for each unit						
quipment		Equipment \$5,000 per unit cost, enter specific d	(Prior Appri	vai a supporti	ig documents req	ulted for each unit	4					
quipment		Equipment >\$5,000 per unit cost, enter specific d	(Prior Appri	wai & supportir	ig documents req	uired for each unit			_			
quipment		Equipment >\$5,000 per unit cost, enter specific d	(Phor Appro	oval & supportir	ig documents req	uired for each unit	9	-				
upplies	5302010	Office Supplies						2,500	2,500			
upplies		Classroom Supplies	1									
upplies	5304040	Medical and Dental Supplies	1					14,740	14,740			
upplies		Janitorial Supplies	1									
uppiles		Other Commodities	1.72									
aeiiqqu		Tools, Apparatus and Accessories (Under \$100 e	ach)									
opplies		Computer Software										
upplies		Cap <5000 - Computers	1									
upplies		Cap <5000 - Machinery & Equipment										
upplies		Cap <5000 - Furniture & Fixtures	1									
ontractual	5202010	Temporary Services (Tri-Starr/Stability)						80,000	80,000			
ontractual		UT Health Science Center School of Dentistry	1					8,000	8,000			
ontractual		(Specify)	1					5,555	5,555			
ontractual		(Specify)	1									
ontractual		(Specify)	1						-			
ontractual		(Specify)	1									
			4									
ontractual		(Specify)	-									
ontractual		(Specify)	_									
ther		Staff Development/Training/Seminars	-									
ther		Advertising and Publications	-									
other		Membership Dues and Licenses	1					1222				
other	5203060	Binding Printing and Reproduction	1					4,000	4,000			
ther		Subscriptions to Publications	1									
other		Transportation Fees-Staff Mileage	1									
Xher		Transportation Fees- Field Trips -Children	1									
Mher		Maint - Buildings and Improvements										
ther		Maint - Machinery & Equipment]									
ther		Maint & Rep- Automotive]									
ther		Mali and Parcel Post	1									
Kher		Rental of Office Equipment	1						7 - 1			
ther		Rental of Facilities	1									
Xther		Alarm and Security Services	1									
ther		M&R Parts Automotive	1									
ther		Food for children (NOT reimbursed by USDA)	1									
ther		Telephone and fax	1									
ther		Cellular Phone Service	1									-
ther		Motor Fuel and Lubricants	1									
ther		Software Licensing	1									
		Gas and Electricity	-									
ther			-									
ther		Water and Sewer	-									
ther		General Liability Insurance	4									
ther		Building Insurance	4									
ther		Vehicle Insurance	1									
ther		Equipment Insurance	1									
ther		Relocation Expense/Cost of Moving	1					and and	Endowe - F			
ther		In-Kind Parent Participation (Limited to Non-Federation)	aral Share)									
ther		In-Kind Volunteers (Limited to Non-Federal Shar	B)									
Other		In-Kind Other	1					27,935			27,935	
Other		in-Kind Other	1								2.,030	
0.07		TOTAL BUDGET						139,675	111,740	0	27,935	

Contract #:		
Attachment III	- Special	Provisions

ATTACHMENT III SPECIAL PROVISIONS — Program Year 2017-2018

I. RESTRICTIONS ON USE OF FUNDS OR PROPERTY

In addition to the other applicable restrictions on the use of Head Start funds provided under this Contract, the Contractor is prohibited from:

- using or transferring funds provided under this Contract for purposes other than authorized Head Start activities;
- using, pledging, granting a security interest in, or otherwise encumbering any right under this Contract
 or any property acquired with funds provided under this Contract as collateral or security for any loan,
 note debenture, bond or any other debt instrument;
- using any funds provided under this Contract for payment of principal or interest on any loan, note, debenture, bond or any other debt instrument, other than those approved in the 45 C.F.R. Part 74 and by the City

II. REQUIREMENTS FOR PARTICIPATION IN CHILD PLUS DATA SYSTEM

2.01 Child Plus is an electronic case management system managed and licensed by the City. This system maintains child files and an overall wait list and streamlines the process for program entry, qualification, position reservation and referrals. Child Plus enhances performance and improves the overall efficiency of data processing and automation systems in support of Head Start initiatives and is used to compile the annual Program Information Report (PIR).

2.02 Contractor shall:

- a) maintain and support Child Plus Data System
- b) provide a data entry specialist for Child Plus Data System that will be responsible for entering all required data into the system and who will be the designated contact person with regard to data entries;
- attend meetings with the City's Child Plus vendor and City staff to ensure continuity and commitment to the this system;
- support all design, development, testing and implementation protocols as established by the City by carrying out and complying therewith;
- e) participate in preliminary and final testing of the system using City protocols;
- f) provide the technical detail required for matching Contractor's system with the Child Plus Data System environment;
- allow City and its vendor to install data encryption software on the Child Care System Database network; and
- provide City and its vendor with access to Confidential Data with parental permission, as defined in Article 3.01 below, which data is critical for the Head Start project.

2.03 Both Parties agree to:

- use best efforts to cooperate and exchange information regarding all aspects of the Head Start project and comply with all reasonable requests of the other Party with respect to information concerning the system.
- b) Parties agree that nothing herein shall be construed as to control or in any way limit the right of parents to choose a Head Start provider.

III. CONFIDENTIAL DATA

3.01 The Parties to this Contract shall have access to the following data ("Confidential Data"), with parental permission in the case of the child:

Contract #:	
Attachment III -	- Special Provisions

Parent's Information:

Case Number First Name Middle Initial Last Name Street Address City Zip Code

Telephone Social Security Number (Optional)

Birth Date
Gender
Race
Disability
Yearly Income
Number of members in the Family
County of Residence

Employment and training status

Each child's Information:

Client Number First Name Middle Initial Last Name

Social Security Number (Optional)

Birth Date Gender Race Disability

- 3.02 Contractor understands that City intends to enter into additional agreements with other providers of child care services ("Additional Collaborators") in order to promote the success of the Head Start project. Confidential Data may be shared by City, Contractor, and any Additional Collaborator, except that all parties shall share such information in compliance with state and federal laws relating to confidentiality. All Additional Collaborators shall be required to enter into a written agreement with City containing the confidentiality requirements set forth in this Section III.
- 3.03 Each Party shall establish a method to secure the Confidential Data in accordance with the applicable federal, state, and local laws and regulations. This provision shall not be construed as limiting a Party to this Contract or an Additional Collaborator, or such Party's authorized representative's right of access to that Party's Confidential Data.
- 3.04 Neither Party shall disclose or publish Confidential Data or public school education data to any individual or organization that is not a Party to this Contract or an Additional Collaborator, unless required by law or a lawful order of a court of competent jurisdiction. Each Party shall take measures within its organization to ensure that Confidential Data or public school education data is accessible only by those persons working on the Head Start project, or directly providing other public school education / child care services, and only for the purpose of performing or assisting with services required by the Head Start project or other specific public school education / child care services.
- 3.05 Either Party may disclose Confidential Data to a third party ("Third Party") under contract or affiliated with that Party for the sole purpose of performing or assisting with services required in relation to the Head Start project or other specific child care services, and in compliance with state and federal laws relating to confidentiality. Confidential Data provided to a Third Party shall remain confidential and written confirmation by such Third Party that the Third Party will conform to the requirements of this section shall be provided to the Party prior to delivery of any information to the Third Party.

[INSERT THE HHS AWARD DOCUMENT]

Attachment II

Contract No.	
Contract 140.	

STATE OF TEXAS

HEAD START AGREEMENT FOR EDUCATION SERVICES BETWEEN THE CITY OF SAN ANTONIO

COUNTY OF BEXAR * BETWEEN

CITY OF SAN ANTONIO * EDGEWOOD INDEPENDENT SCHOOL DISTRICT (EISD)

This Agreemer	it is entered into by an	nd between the City of San Antonio (hereinafter referred to as "City"), a
Texas Municipal Corp	oration, acting by and	through its Director of the Department of Human Services pursuant to
Ordinance No.	dated	, and Edgewood Independent School District (EISD), a political
subdivision of the St	ate of Texas, acting	by and through its Board of Managers (hereinafter referred to as
"Contractor") (individu	ially "the Party" and co	collectively "the Parties") to set forth the objectives, understandings, and
agreements between the	ne Parties in connection	ion with the use of Head Start grant funds as described herein. This
Agreement is made and	d entered into by the Pa	Parties pursuant to the authority granted under the Interlocal Cooperation
Act. Texas Gov't Code	791 et sea.	

WITNESSETH:

WHEREAS, the City has received a grant pursuant to the Head Start Act (42 U.S.C. §9831 et seq., as amended) (the "Grant") for the purpose of providing Head Start services to children and families in the San Antonio and Edgewood Independent School Districts; and

WHEREAS, the City is authorized by the U.S. Department of Health and Human Services ("HHS"), Administration for Children and Families ("ACF"), and desires, through its Department of Human Services ("DHS")) to execute an agreement with Contractor to provide Head Start services to children residing in the Edgewood Independent School District (hereinafter referred to as the "Project" or "Program"); and

WHEREAS, Contractor desires, and is appropriately licensed and qualified, to enter into this Agreement with City and agrees to deliver the services described herein in accordance with applicable Head Start Performance Standards and other requirements more fully set forth below;

NOW THEREFORE, in consideration of the mutual promises and covenants herein contained and intending to be legally bound hereby. City and Contractor agree as follows:

I. SCOPE OF WORK

- 1.1 The Contractor will provide, oversee, administer, and carry out all activities and services in a manner satisfactory to the City and in compliance with the Contractor's Scope of Work, attached hereto and incorporated herein for all purposes as Attachment I, this Agreement, and the Terms of the Grant (hereinafter defined). If the terms of this Agreement are inconsistent or in conflict with applicable Terms of the Grant, the applicable Terms of the Grant shall be controlling, unless the inconsistency or conflict results from more stringent requirements set forth in this Agreement, in which case the terms imposing the most stringent requirements upon the Contractor shall control.
- 1.2 For purposes of this Agreement, the terms listed below shall have the following meanings:
 - (A) "Terms of the Grant" shall mean all requirements of the Grant, whether contained in the Head Start Act, as amended by the Improving Head Start for School Readiness Act of 2007 (42 U.S.C. §9831, et seq.), or other applicable statutes, implementing regulations (e.g., 45 C.F.R. §1301 et seq. (the "Head Start Performance Standards" or "Performance Standards") and 45 C.F.R. Part 75, as amended), rules, Executive Orders, the award document from HHS to the City, Relevant HHS Directives, or elsewhere, including, but not limited to circulars, Program Instructions, Information Memorandums and Policy Clarifications, the City's policies and procedures and the program design manual applicable to the

- Head Start Program, as such requirements exist as of the date of this Agreement and as such requirements may be established or modified (by amendment, deletion, addition or otherwise) during the period of the Agreement.
- (B) "Relevant HHS directives" shall mean regulations, manuals, guidelines, or other oral or written directives of HHS or any subdivision thereof, including the Administration for Children and Families, Head Start Bureau, the Program Operations Division and ACF Region VI, as such regulations, manuals, guidelines, or other oral or written directives shall be made applicable.
- 1.3 Contractor shall establish and implement policies and procedures governing personnel, financial management, and programmatic management, as specified more fully in 45 C.F.R Parts 1301 et. seq, and/or 45 C.F.R. Part 75. Such policies and procedures shall be consistent with the Terms of the Grant, the policies and procedures approved by the City's Head Start Policy Council and Governing Body, and content and service plans.
- City retains the authority to contract with third-parties for the delivery of other Head Start services in the San Antonio and Bexar County area. Contractor agrees to allow the City's other contractors access to the facilities leased and/or owned by Contractor in order to provide said services, so long as access would not cause disruption of Contractor's educational activities or purpose as an educational entity. Contractor agrees to cooperate with City and third-party Head Start contractors to establish, modify and comply with a set of policies and procedures and/or a program design manual governing the City's Head Start Program and the protocol for collaboration between Head Start service providers. Contractor agrees that, notwithstanding the fact that another Head Start service provider under the City of San Antonio Head Start program may be contracted to provide a category of service, Contractor, under the leadership of its Head Start Program Director, will be responsible for coordinating with Head Start service providers and working with the City to ensure provision of full array of services to which the children are entitled under the Terms of the Grant.

II. TERM

2.1 Except as otherwise provided, this Agreement shall begin on February 1, 2017 and shall terminate on January 31, 2018. The City, subject to its continued award of Head Start grant funds from HHS, shall have the option to renew this Agreement in an amount acceptable to Contractor's superintendent of schools for 1 additional 5-month term without the necessity of further City Council approval. In the event of a 5-month renewal under this option, appropriate revisions will be made to the budget, associated due dates, and Scope of Work to reflect the grant obligations and best meet the needs of children and families through Head Start Grant activities.

III. CONSIDERATION

- In consideration of the services to be delivered by Contractor, the City will reimburse Contractor a total amount not to exceed \$4,171,411.00 ("the Federal Share") during the period in which this Agreement is in effect for costs incurred in accordance with the Program Budget attached hereto and incorporated herein for all purposes as Attachment II. Contractor's Program Budget is comprised of the Federal Share and the Non-Federal Share. The Federal Share shall be no more than 80% of the total Program Budget. Should Contractor fail to raise all of the non-Federal Share funds (20% of the total Program Budget, or \$1,042,853.00) it is required to raise for the operation of its Program, City reserves the right to limit its reimbursements to Contractor proportionately. For instance, if Contractor succeeds in raising only eighty percent (80%) of its required non-Federal Share funds, City may limit its reimbursements to Contractor to eighty percent (80%) of City's total obligation to Contractor. Contractor may provide additional non-Federal share funds if Contractor, in its discretion, determines such funds are available. To meet the requirements of this Agreement, all claimed non-Federal Share must meet the requirements of 45 C.F.R. § 75.306, as applicable.
- 3.2 Prior to commencement of the Agreement, Contractor shall submit to City for its approval Contractor's proposed monthly budget by line item for the entire term of the Agreement along with its program Budget,

including detail by category alone. Until the City receives the initial proposed monthly budget for the entire term of the Agreement and prior to the City's approval, the City reserves the right to redirect the funding City has proposed be allocated to Contractor under this Agreement. City shall notify the Contractor of the amount redirected and the revised Agreement funding. Additionally, beginning February 1, 2017 and throughout the term of the Agreement, Contractor shall submit quarterly forecasts of the projected expenses for each month remaining in the Agreement so that the City may review and compare actual expenses to projected expenditures and address issues associated with Contractor's expenditure rate (e.g., for Q1, Contractor shall submit the projected expenses by month for February, April, May, and so on until January 31, 2018; for Q2, Contractor's budgeted development and administrative costs (as defined by 45 C.F.R. §1301.32) shall not exceed twelve percent (12%) of the Program Budget, unless the total Program Budget is modified in accordance with this Agreement in which case this amount shall be reduced proportionately unless the Parties otherwise agree.

- 3.3 Approval required. Contractor shall seek and obtain City's (City's Head Start Program Administrator and DHS's Fiscal Administrator) prior written approval 30 calendar days before making budget modifications. City may make exceptions to the 30-day notice requirement on a case by case basis, but otherwise Contractor must make request in writing or via email to the City's Head Start Program Administrator. Contractor's written request must be accompanied by a justification for the change and indicate which lines items are affected by such change
- 5.4 Contractor understands and agrees that should Contractor fail to meet or maintain its funded enrollment level as set forth in the Scope of Work which failure continues for a thirty (30) day period, City may reduce Contractor's funding by an amount equal to the difference between funded and actual enrollment at the per capita rate.
- 3.5 The funding level of this Agreement is based on an allocation from the following funding sources:
 - U.S. Department of Health and Human Services (HHS) Head Start Funds Catalog of Federal Domestic Assistance # 93,600
 - Consequently, Contractor agrees to comply with the Terms of the Grant and the Special Provisions, attached hereto and incorporated herein for all purposes as Attachment III.
- 3.6 It is expressly understood and agreed by the City and Contractor that the City's obligations under this Agreement are contingent upon the actual receipt of adequate grant funds from HHS to meet City's liabilities hereunder. This Agreement may be terminated by the City if HHS terminates the City as a grantee or reduces the amount granted to City, for any reason; provided that, if the reduction of grant funds does not result in complete unavailability of such funds, the Parties will use best efforts to amend this Agreement accordingly. City will promptly notify Contractor of any such HHS action.

IV. PAYMENT

Contractor agrees that this is a cost reimbursement contract and that the City's liability hereunder is limited to making reimbursements for allowable costs incurred as a direct result of services provided by the Contractor in accordance with the terms of this Agreement. Allowable costs are defined as those costs, which are necessary, reasonable and allowable under applicable federal, state, and local law, including but not limited to those laws referenced in Article XII hereof, for the proper administration and performance of the services to be provided under an agreement. All requested reimbursed costs must be consistent with the terms and provisions of the approved budgeted line items described in Attachment II of this Agreement, unless (a) a subsequent budget revision has been approved in accordance with the procedure set forth in Section 3.3 and signed by the Director of DHS in cases where the total Agreement Budget remains the same, or (b) an amendment has been approved and signed by the Director of DHS pursuant to Section 24.1 of this Agreement in cases where there is an increase or decrease to the total Agreement Budget. Approved budget revisions and amendments modify the Budget attached hereto, and in such cases Contractor's requested reimbursed costs must be consistent with the last revised, approved budget. Approved budget

revisions and amendments supersede prior conflicting or inconsistent agreements with regard to the referenced Project Budget, and all references in the Agreement to the budget shall mean the budget as revised through approved budget revisions or amendments. Budget revision requests shall be submitted in advance of anticipated expense(s). The City will not accept budget revision requests submitted after December 31, 2017, unless Contractor requests a reasonable extension before that date, which shall not be unreasonably denied by City. In no event shall the City be liable for any cost of Contractor not eligible for reimbursement as defined within the Agreement. Contractor shall remit to City within ten (10) Contractor business days after the City makes the request for remittance any funded amounts which were paid pursuant to this Article IV and used to cover disallowed costs. Any such amounts not remitted within ten (10) Contractor business days may, at City's option, be subject to offset against future funding obligations by City. For purposes of this Agreement, the term, "Contractor business day" shall mean every day of the week except all Saturdays, Sundays and those scheduled holidays officially adopted and approved by Contractor for Contractor employees.

- If specific circumstances require an advance payment on this Agreement, Contractor must submit to the Director of DHS a written request for such advance payment, including the specific reason for such request in the form prescribed by City. Contractor agrees that the City shall not be obligated to pay for any advances requested. In those instances in which advance payments are authorized, the Director of DHS may, in the Director's sole discretion, approve an advance payment on this Agreement. It is understood and agreed by the parties hereto that (a) each request requires submission to the Director of DHS no less than ten (10) business days prior to the actual ostensible cash need; (b) each request will be considered by the Director of DHS on a case-by-case basis, and (c) the decision by the Director of DHS whether or not to approve an advance payment is final. For purposes of this Agreement, the term "business day" shall mean every day of the week except all Saturdays, Sundays and those scheduled holidays officially adopted and approved by the San Antonio City Council for City of San Antonio employees, except in those instances where this Agreement expressly calls for "Contractor business days." In those instances in which advance payments are authorized:
 - (A) Contractor's payments to its vendors using funds advanced by the City shall be remitted to the vendors in a prompt and timely manner so long as services have been performed by the subject vendor, defined as not later than (10) business days after the Contractor is notified that an advance payment check is available from the City.
 - (B) The Contractor must deposit Agreement funds in an account in a bank insured with the Federal Deposit Insurance Corporation (FDIC). In those situations where Contractor's total deposits in said bank, including all Agreement funds deposited with said bank, exceed the FDIC insurance limit, the Contractor must arrange with said bank to automatically have the excess collaterally secured. A written copy of the collateral agreement must be obtained by Contractor from the Contractor's banking institution, maintained on file and be available for City monitoring reviews and audits. Advanced funds that cause the Contractor's account balance to exceed the FDIC limit shall be deposited in a manner consistent with the Public Funds Investment Act (Chapter 2256 of the Texas Government Code) as amended. Contractor shall maintain the FDIC insured bank account in which Agreement funds are deposited and its recordkeeping in a manner that will allow City to track expenditures made with Agreement funds.
 - (C) The City may deduct from monthly reimbursements amounts necessary to offset the amount advanced based upon the number of months remaining in the Agreement term, or from a single subsequent monthly reimbursement the full amount previously advanced to Contractor. The City may consider factors such as projected allowable costs and other indicators such as Contractor's financial stability. Contractor shall maintain a financial management system to account for periodic, or a lump sum, deduction from reimbursements.
- 4.3 Contractor shall submit to City not later than the forty-fifth (45th) day after the end of every month a monthly Request for Payment in the form prescribed by City, which details the specific costs (by category and by program account number) Contractor expensed in the previous month for the services delivered as described in Article I herein, including supporting documentation of such costs as may be required by the

Director of DHS. The Request for Payment shall also specify the Program Income (as defined herein) received or projected during the same time period. The Director of DHS may require the Contractor's submission of original or certified copies of invoices, cancelled checks, Contractor's general ledger and/or receipts to verify invoiced expenses.

- 4.4 City shall make reimbursement payments of eligible expenses to the Contractor of any undisputed amounts as determined by the Director of DHS in accordance with established procedures, so long as City receives a properly completed and documented Request for Payment. City shall make payment to Contractor as soon as reasonably practicable, but not later than 30 calendar days of receiving a valid and approved Request for Payment.
- 4.5 The Contractor shall submit to City a full accounting of the Program Income and non-Federal Share funds received and total Program costs incurred, along with all requests for payment for the period February 1, 2017 through January 31, 2018, no later than March 15, 2018. In addition, all purchase orders that have been encumbered by January 31, 2018 must be received and paid no later than March 15, 2018. In the event of early termination of this Agreement, Contractor shall submit the information 45 calendar days from the early termination date of the Agreement. These deadlines may be adjusted only if Contractor receives written authorization from the Director of DHS allowing Contractor to submit a request for payment at a later specified date.
- 4.6 Contractor agrees that the City shall not be obligated to any subcontractors or third party beneficiaries of the Contractor.
- 4.7 Contractor shall maintain a financial management system, and acceptable accounting records in accordance with 45 C.F.R. § 75.302 et seq., and that provide for:
 - (A) accurate, current, and complete disclosure of financial support from each federal, state and locally sponsored project and program in accordance with the reporting requirements set forth in Article VIII of this Agreement. If accrual basis reports are required, the Contractor shall develop accrual data for its reports based on an analysis of the documentation available;
 - (B) records that adequately identify the source and application of funds for City-sponsored activities. Such records shall contain information pertaining to City awards, authorizations, obligations, unobligated balances, assets, equity, outlays, and income;
 - (C) effective control over and accountability for all funds, property, and other assets. The Contractor shall adequately safeguard all such assets and shall ensure that they are used solely for authorized purposes. Contractor shall maintain a separate numbered account for all funds received and disbursed through this Agreement;
 - (D) comparison of actual outlays with budget amounts for each award. Whenever appropriate or required by the City, financial information should be related to performance and unit cost data;
 - (E) procedures to minimize the time elapsing between the transfer of funds from the City and the disbursement of said funds by the Contractor;
 - (F) procedures for determining reasonable, allowable, and allocable costs in accordance with the provisions of any and all applicable cost principles, including but not limited to the cost principles referenced in Article XII hereof, and the terms of the award, grant, or contract, with the City;
 - (G) accounting records that are supported by source documentation (i.e., timesheets, employee benefits, professional services agreements, purchases, and other documentation as required by City). Contractor shall maintain records and shall meet necessary requirements under Generally Accepted Accounting Principles [GAAP]; and

- (H) an accounting system based on generally acceptable accounting principles which accurately reflects all costs chargeable (paid and unpaid) to the Project. A Receipts and Disbursements Ledger must be maintained. A general ledger with an Income and Expense Account for each budgeted line item is necessary. Paid invoices revealing check number, date paid and evidence of goods or services received are to be filed according to the expense account to which they were charged.
- 4.8 Contractor agrees that Contractor costs or earnings claimed under this Agreement may not be claimed under another contract or grant from another agency, organization, business entity or governmental entity.
- 4.9 Contractor shall establish and abide by a cost allocation methodology and plan which ensures that the City is paying only its fair share of the costs for services, overhead, and staffing not solely devoted to the Project or funded pursuant to this Agreement. The Cost Allocation Plan is a plan that identifies and distributes the cost of services provided by staff and/or departments or functions. It is the means to substantiate and support how the costs of a program are charged to a particular cost category or to the Program so as to assure Head Start Grant funds provided hereunder do not subsidize other program(s). Contractor must ensure that costs allocated and charged to the Head Start grant are not charged to other Federal, State or Local awards to overcome fund deficiencies, to avoid restrictions imposed by law or terms of the Federal awards, or for other reasons. Contractor shall provide to City prior to the beginning of the Agreement term (i) a matrix identifying the shared use of such facilities and/or program services; and (ii) the Cost Allocation Plan and supporting documentation, along with its Budget, financial statements and audit that are applicable to the Contractor's Project. City shall have the right to approve the Cost Allocation Plan.
- 4.10 Contractor agrees to reimburse the City for any Contractor overpayment based upon reconciled adjustments resulting from Contractor's balance and/or Statement of Revenue and Expenditure sheet as of January 31, 2018, which balance or Statement sheet shall be due to the City no later than February 15, 2018. Reimbursement shall be made within 20 calendar days of written notification to Contractor of the need for reimbursement.
- 4.11 Upon expiration or early termination of this Agreement, or at any time during the term of this Agreement, all unused funds, rebates, advances exceeding allowable costs, or credits on-hand or collected thereafter relating to the Project, shall be returned by Contractor to the City within twenty (20) days of receipt of written notice.
- 4.12 Upon execution of this Agreement or at any time during the term of this Agreement, the City's Director of Finance, the City Auditor, or a person designated by the Director of DHS may review and approve all Contractor's systems of internal accounting and administrative controls prior to the release of funds hereunder.

V. PROGRAM INCOME

- 5.1 For purposes of this Agreement, "program income" shall mean earnings of Contractor realized from activities resulting from this Agreement or from Contractor's management of funding provided or received hereunder. Such earnings shall include, but shall not be limited to, interest income; usage or rental/lease fees; income produced from Agreement-supported services of individuals or employees or from the use of equipment or facilities of Contractor provided as a result of this Agreement; and if applicable, payments from clients or third parties for services rendered by Contractor pursuant to this Agreement. At the sole option of the Director of DHS, Contractor will either (a) be required to return program income funds to City through DHS, or (b) upon prior written approval by the Director of DHS, Contractor may be permitted to retain such funds to be:
 - (A) added to the Project and used to further eligible Project objectives, in which case proposed expenditures must first be approved by the City; or
 - (B) deducted from the total Project cost for the purpose of determining the net cost reimbursed by the City.

- 5.2 In any case where Contractor is required to return program income to DHS, Contractor must return such program income to City within the timeframe that may be specified by the Director of DHS. If the Director of DHS does not specify a timeframe for Contractor to return program income to City, then Contractor must return such program income to City on the same date that Contractor submits its statement of expenditures and revenues to DHS set forth in Article V, Section 5.4 of this Agreement. If the Director of DHS grants Contractor authority to retain program income, Contractor must submit all reports required by DHS within the timeframe specified in the Agreement.
- 5.3 Contractor shall provide DHS with thirty (30) calendar days written notice prior to the activity that generates program income. Such notice shall detail the type of activity, time, and place of all activities that generate program income.
- 5.4 The Contractor shall fully disclose and be accountable to the City for all program income. Contractor must submit a statement of expenditures and revenues to DHS within thirty (30) calendar days of the activity that generates program income. The statement is subject to audit verification by DHS. Failure by Contractor to report program income as required is grounds for suspension, cancellation, or termination of this Agreement.
- 5.5 Contractor is prohibited from charging fees or soliciting donations and is prohibited from inviting or contracting with vendors who shall charge fees or solicit donations from Head Start participants and their parents in any Agreement-funded project without the prior written approval of the Director of DHS. However, Contractor may engage in general school activity that is not specifically targeted at Head Start families.
- 5.6 Contractor shall include this Article V, in its entirety, in all of its subcontracts involving income-producing services or activities.

VI. ADMINISTRATION OF CONTRACT

- 6.1 The Contractor agrees to comply with all the terms and conditions that the City must comply with within its award document from HHS. A copy of said award document is attached hereto and incorporated herein for all purposes as Attachment IV. From time to time, the award document may be amended or supplemented, and these shall be incorporated into the Agreement collectively as Attachment IV.
- 6.2 In the event that any disagreement or dispute should arise between the Parties hereto pertaining to the interpretation or meaning of any part of this Agreement or its governing rules, regulations, laws, codes or ordinances, the City Manager or the Director of DHS, as representatives of the City and the parties ultimately responsible for all matters of compliance with HHS and City rules and regulations, shall have the final authority to render or secure an interpretation. In the event that a dispute arises with regard to interpretation of regulations or law as it applies to this Agreement, Contractor may request that the City secure, and the City shall request an interpretation or opinion from HHS in order to assist in resolution of the dispute.
- 6.3 Contractor shall not use funds awarded from this Agreement as matching funds for any federal, state or local grant without the prior written approval of the Director of DHS.
- 6.4 The City shall have the authority during normal business hours to make physical inspections of all operating facilities occupied by Contractor for the administration of this Agreement and to require such physical safeguarding devices as locks, alarms, security / surveillance systems, safes, fire extinguishers, sprinkler systems, etc. as reasonably necessary, to safeguard property and/or equipment authorized by this Agreement.
- 6.5 The Contractor Board of Directors or Board of Trustees, as applicable, and Contractor's management staff shall adopt and approve an Employee Integrity Policy, in the event that Contractor has none, and internal

program management procedures, and require all staff to abide by these and the Head Start standards as established in the HHS regulations, to preclude theft, embezzlement, improper inducement, obstruction of investigation or other criminal action, and to prevent fraud and program abuse. These policies and procedures shall require repayment of such erroneously received grant funds or property to the Contractor, or to the applicable Head Start service provider from whom such grant funds or property was received, if other than the Contractor, and shall specify any other consequences to Contractor's employees and vendors involved in such illegal activities and may include but not be limited to termination and prosecution where necessary. Said policies and procedures shall be provided to DHS upon request by DHS. In the event that DHS finds the policies and procedures to be lacking, DHS may recommend revision.

- 6.6 Contractor agrees to comply with the following check writing and handling procedures:
 - (A) No blank checks are to be signed in advance;
 - (B) No checks are to be made payable to cash or bearer with the exception of those for petty cash reimbursement, not to exceed a \$100.00 maximum per check. Contractor agrees that the aggregate amount of petty cash reimbursement shall not exceed \$500.00 for any given calendar month during the term of this Agreement unless Contractor receives prior written approval from DHS to exceed such limit. Such requests for petty cash must be supported by the submission to DHS of an original receipt.
 - (C) Checks issued by City to Contractor shall be deposited into the appropriate bank account immediately or by the next Contractor business day after Contractor's receipt of each such check, and shall never be cashed for purposes of receiving the face amount back. If such check(s) are not deposited within the next Contractor business day from the date of issue, such checks shall be investigated by City and stop-payment orders issued, as applicable. City agrees that stop payment orders shall not be issued due to the financial institution's posting policies that result in posting of checks for credit beyond the required timeframe so long as the checks are deposited within the required timeframe. Upon cancellation of any outstanding check, if deemed appropriate by City, such check may be reissued to the Contractor or if deemed by City not to be a valid expense, such check shall be immediately returned to the City.
 - (D) For checks other than petty cash reimbursement, Contractor shall adopt and comply with a policy requiring no less than two (2) signatures of authorized representatives of Contractor on each check. Said signatures may be computer generated so long as they are issued in accordance with Contractor's internal policies established to reduce the risk of fraud, theft, or embezzlement. Contractor understands and agrees that City's reimbursement is subject to compliance with this provision of the Agreement.
- 6.7 Wherever in this Agreement, Contractor is required to perform an action within a specified number of days or hours, Contractor may request additional time to perform. City shall give Contractor's request for additional time due consideration and shall grant Contractor's request whenever reasonable practicable, unless immediate compliance is required.
- 6.8 If the starting date of the Agreement term has already passed at the time of approval of the Agreement by Contractor, all deadlines imposed on Contractor for providing information to the City on or before the date of approval or within seven (7) Contractor business days of approval shall be extended in order for the Contractor to reasonably comply with the City's requirements; except and unless, the information has already been provided to the City. City and Contractor agree to confer and make best efforts to reasonably permit Contractor to comply with the extended deadlines.

VII. AUDIT

7.1 If Contractor expends \$750,000.00 or more of funds provided under this Agreement, or cumulative funds provided by or through City, and does not have to comply with the provisions of Section 7.2, then during the term of this Agreement, the Contractor shall have completed an independent audit of Contractor's Head Start program and the reporting required submitted to the City within the earlier of 30 calendar days after

receipt of the auditor's report(s), or nine (9) months after the end of Contractor's fiscal year, expiration or early termination of this Contract, whichever is earlier. Contractor understands and agrees to furnish DHS a copy of the audit report including the corrective action plan(s) of the Head Start program on all audit findings, a summary schedule of prior audit findings, management letter and/or conduct of audit letter, within thirty (30) calendar days upon receipt of said report or upon submission of said corrective action plan to the auditor.

Contractor agrees and understands that upon notification from federal, state, or local entities that have conducted program reviews and/or audits of the Contractor or its programs of any findings about accounting deficiencies, or violations of Contractor's financial operations, a copy of the notification, review, investigation, and audit violations report must be forwarded to DHS within a period of ten (10) calendar days upon the Contractor's receipt of the report.

7.2 Contractor agrees that if Contractor receives or expends more than \$750,000.00 in federal funds from the City, an audit shall be made in accordance with the Single Audit Act Amendments of 1996, the State of Texas Single Audit Circular, and the U.S. Office of Management and Budget Circular ("Uniform Guidance"). Contractor shall also be required to submit copies of its annual independent audit report, and all related reports issued by the independent certified public accountant within the earlier of 30 days after receipt of the auditor's report(s), or nine months after the end of the audit period, unless a longer period is agreed to in advance by the Federal cognizant or oversight agency for audit to the Federal Audit Clearinghouse. A copy of this report must also be provided to City within this same time period. Contractor may submit reports through the following website: http://harvester.census.gov/sac/ and may also contact the Clearinghouse by telephone at (301) 763-1551 (local), 1-888-222-9907 or 1-800-253-0696 (toll free).

Per 2 C.F.R. 200.36, upon completion of Form SF-SAC, Contractor may submit the completed report by mail to:

Federal Audit Clearinghouse Bureau of the Census 1201 E. 10th Street Jeffersonville, Indiana 47132

Contractor agrees to reimburse the City or supplement any disallowed costs with eligible and allowable expenses based upon reconciled adjustments resulting from Contractor's Single Audit. Reimbursement shall be made within thirty (30) calendar days of written notification regarding the need for reimbursement.

- 7.3 If Contractor expends less than \$750,000.00 of City dollars during the term of this Contract, then the Contactor shall complete and submit an unaudited financial statement(s) within a period not to exceed nine months immediately succeeding the end of Contractor's fiscal year, expiration or early termination of this Contract, whichever is earlier. Said financial statement shall include a balance sheet and income statement prepared by a bookkeeper and a cover letter signed by Contractor attesting to the correctness of said financial statement..
- 7.4 All financial statement(s) must include a schedule of receipts and disbursements by budgeted cost category for each program funded by or through the City.
- 7.5 The City reserves the right to conduct, or cause to be conducted, an audit or review of all funds received under this Agreement at any and all reasonable and necessary times required by City. The City Internal Audit Staff, a Certified Public Accounting (CPA) firm, or other personnel as designated by the City, may perform such audit(s) or reviews. The City reserves the right to determine the scope of every audit, so long as it is limited to Contractor's services for the Head Start program. In accordance herewith, Contractor agrees to make available to City all accounting and Project records.

Contractor shall during Contractor's normal business hours, and as often as deemed reasonably necessary by City and/or the applicable state or federal governing agency or any other auditing entity, make available the books, records, documents, reports, and evidence with respect to all matters covered by this Agreement and shall continue to be so available for a minimum period of four (4) years or whatever period is determined necessary based on the Records Retention guidelines, established by applicable law for this Agreement. Said records shall be maintained for the required period beginning immediately after Agreement termination, save and except there is litigation or if the audit report covering such agreement has not been accepted, then the Contractor shall retain the records until the resolution of such issues has satisfactorily occurred. The auditing entity shall have the authority to audit, examine and make excerpts, transcripts, and copies from all such books, records, documents and evidence, including all books and records used by Contractor in accounting for expenses incurred under this Agreement, all contracts, invoices, materials, payrolls, records of personnel, conditions of employment and other data relating to matters covered by this Agreement.

The City may, in its reasonable discretion, require the Contractor to use any and all of the City's accounting or administrative procedures that are in conformity with Generally Accepted Accounting Principles for state account standards in Texas in the planning, controlling, monitoring and reporting of all fiscal matters relating to this Agreement, and the Contractor shall abide by such requirements. Should a conflict exist between the parties' accounting procedures, Contractor shall use the stricter of the procedures.

7.6 When an audit or examination determines that the Contractor has expended funds or incurred costs which are questioned by the City and/or the applicable state or federal governing agency, the Contractor shall be notified and provided an opportunity to address the questioned expenditure or costs.

Should any expense or charge that has been reimbursed be subsequently disapproved or disallowed as a result of any site review or audit, the Contractor shall promptly refund such amount to the City no later than twenty (20) Contractor business days from the date of notification of such disapproval or disallowance by the City. At its sole option, DHS may deduct such claims from subsequent reimbursements and shall notify Contractor prior to exercising this option. Contractor shall provide to City a full refund of such amount no later than twenty (20) Contractor business days from the date of notification of such disapproval or disallowance by the City. If Contractor is obligated under the provision hereof to refund a disapproved or disallowed cost incurred, such refund shall be required and be made to City by check, cashiers check or money order. In the event that a dispute arises as to the allowability of an expenditure or charge pursuant to the federal regulations, then Contractor may request that the City secure, and the City shall request an interpretation or opinion from HHS in order to assist in resolution of the dispute. Should the City deduct such claims from subsequent reimbursements, the Contractor is forbidden from reducing Project expenditures and Contractor must use its own funds to maintain the Project.

Contractor agrees and understands that all expenses, fees, fines and penalties associated with the collection of delinquent debts owed by Contractor shall be the sole responsibility of the Contractor and shall not be paid from any Project funds received by the Contractor under this Agreement. Delinquent debts that would otherwise be identified as allowable costs may be paid with Project funds after written approval of DHS.

7.7 If the City determines that Contractor is in violation of the above requirements, the City shall have the right to dispatch auditors of its choosing to conduct the required audit and to have the Contractor pay for such audit from non-City resources. If after the audit is conducted it is determined that Contractor is in compliance with the contract then the cost of the audit, specifically the auditor's bill alone, shall be borne by the City. Contractor may recommend the hiring of alternate auditors, but the final decision on the selection of auditors shall rest with the City.

VIII. RECORDS, REPORTING, MONITORING AND INTELLECTUAL PROPERTY

8.1 In addition to those listed in this Agreement, the Contractor shall submit to DHS any and all reports as may be required of Contractor by HHS or as may be reasonably required of the City. Contractor shall incorporate and use a City approved tracking or information system, such as ChildPlus, for the delivery of comprehensive Head Start Services and collect, input and update all data required for the Program Information Report in accordance with the City's reasonable timeline to ensure the reporting of accurate and consistent information to HHS.

8.2 Additionally, Contractor shall maintain and furnish to City the appropriate financial and programmatic information and reports as listed in the Scope of Work,in such forms as the City may require or prescribe as required pursuant to the Head Start Act, as amended, or as may be required under federal regulations, such as 2 C.F.R. 200 st seq or other applicable regulations.

Contractor shall maintain all applicable and appropriate supporting documentation of costs, including but not limited to, payroll records, invoices, contracts or vouchers, and make these available to City upon request.

- 8.3 Contractor shall provide to DHS all information reasonably requested by DHS relating to the Contractor's Board functions, if applicable, including but not be limited to:
 - (A) Roster of current Board Members (name, title, and business address, telephone number, fax number and board e-mail address);
 - (B) Names and terms of Officers;
 - (C) Schedule of anticipated board meetings for current Fiscal Year;
 - (D) Board agendas relating to the Head Start Program to be submitted by electronic or facsimile transmission at the time of posting prior to each Board meeting. Prior to the time of posting, Contractor's administration shall attempt to notify City when a Head Start program item is anticipated to be placed on the agenda; and
 - (E) Minutes of every board meeting relating to the Head Start Program.
- 8.4 City reserves the right to reasonably request Contractor to provide additional records for travel expenses, long distance calls, faxes, internet service, cell phone calls, or other electronic communication devices charged to the budget associated with this Agreement.
- 8.5 If applicable, Contractor shall report all notices served, violations found or complaints filed with regard to licensing, or lack thereof, of Contractor's centers within one (1) Contractor business day of receipt of written notice from the State licensing, certifying or permit-issuing authority of a violation or complaint not counting weekends.
- 8.6 Contractor shall comply with federal Head Start Performance Standards (federal Head Start regulations), the Head Start Act and all applicable federal, state and local laws relating to child safety. Contractor shall establish and implement mutually agreeable administrative procedures to respond to medical, dental and other health emergencies with which all Head Start staff should be familiar and trained. These procedures shall include, among other things, methods of notifying parents in the event of a health emergency involving their children and established methods for handling cases of suspected or known child endangerment, abuse or neglect that are in compliance with applicable federal, state and local laws. If Contractor has knowledge of a claim or report of, or is aware of a program crisis related to, or suspects that media coverage would be negative due to, an incident of child endangerment, neglect or abuse of a Head Start child while in the Head Start program, Contractor shall contact the City's designated representative immediately, but not later than 24 hours, for the purpose of notification of the incident. Contractor shall contact the City's designated representative immediately even if the incident of child endangerment, neglect or abuse is not fully investigated by Contractor. If Contractor is unable to reach the City's designated representative, Contractor shall leave a verbal or written message via electronic mail notifying the City that Contractor is attempting to notify the City of an incident. Contractor further agrees to notify the parent of a Head Start child immediately, in any of the instances cited above, to include a situation in which the parent's child has received physical discipline, whether or not the instance may be characterized as suspected child abuse.
- 8.7 Within a period not to exceed forty-five (45) calendar days after the expiration or early termination date of the Agreement, Contractor shall submit all final client reports and all required deliverables to City. Contractor understands and agrees that in conjunction with the submission of the final report, the Contractor shall execute and deliver to City a receipt for all sums received and a release of all claims for said sums against the Project.

- Contractor shall maintain financial records, supporting documents, statistical records, and all other books, documents, papers or other records pertinent to this Agreement or the Grant in accordance with the official records retention schedules established within the Local Government Records Act of 1989 and any amendments thereto, or for such period as may be specifically required by 45 C.F.R §75.361, whichever is longer. Notwithstanding the foregoing, Contractor shall maintain all Agreement and Grant related documents for no less than four (4) years from the date of City's submission of the annual financial report covering the funds awarded hereunder. If an audit, litigation, or other action involving the records has been initiated before the end of the four (4) year period, Contractor agrees to maintain the records until the end of the four (4) year period or until the audit, litigation, or other action is completed, whichever is later. Records for real property and equipment acquired with Head Start funds shall be retained for four (4) years after final disposition.
- 8.9 Contractor shall make available to City, HHS, or any of their duly authorized representatives, upon appropriate notice and unless otherwise prohibited by law, such books, records, reports, documents, papers, policies and procedures as may be necessary for audit, examination, excerpt, transcription, and copy purposes, for as long as such records, reports, books, documents, and papers are retained. This right also includes timely and reasonable access to Contractor's facility and to Contractor's personnel for the purpose of interview and discussion related to such documents. Contractor shall, upon request, transfer certain records to the custody of City or HHS when City or HHS determines that the records possess long-term retention value unless otherwise prohibited by law, and subject to Contractor's right to use "educational records" as that term is understood under the Family Educational Rights and Privacy Act of 1974 ("FERPA").
- 8.10 The Contractor agrees to incorporate and use any City-approved tracking or information system for the delivery of comprehensive Head Start services. Contractor shall enter current, accurate and complete client data.
- 8.11 DHS is assigned monitoring, fiscal control, and evaluation of certain projects funded by the City with General or Grant Funds, including the Project covered by this Agreement. Therefore, Contractor agrees to permit City and/or HHS to evaluate, through monitoring, reviews, inspection or other means, the quality, appropriateness, and timeliness of services delivered under this Agreement and to assess Contractor's compliance with applicable legal and programmatic requirements. At such times and in such form as may be required by DHS, the Contractor shall make available to DHS and the Grantor of the Grant Funds, if applicable, such statements, reports, records, personnel files (including evidence of criminal background check as required by Head Start regulations), client files, data, all policies and procedures and information as may be requested by DHS and shall permit the City and Grantor of the Grant Funds, if applicable, to have interviews with its personnel, board members and program participants pertaining to the matters covered by this Agreement. Contractor agrees that the failure of the City to monitor, evaluate, or provide guidance and direction shall not relieve the Contactor of any liability to the City for failure to comply with the Terms of the Grant or the terms of this Agreement. Regarding evidence of criminal background check(s), the parties agree that City will accept a written statement that the checks have been conducted and that all persons who are employed have passed, so long as the statement includes the name(s) of the staff member(s) checked, and the date(s) performed.. If, at any time, HHS informs the City and / or Contractor that such written statement does not satisfy the requirements of the Terms of the Grant, Contractor agrees to provide additional information as may be legally permissible and as may be required by HHS or to cooperate with the City and HHS in order to resolve any conflict associated with provision of information related to criminal background checks.
- 8.12 City may, at its discretion, conduct periodic, announced monitoring visits to ensure program and administrative compliance with Head Start Performance Standards,, the Head Start Act and with Program goals and objectives for the agreement period. City reserves the right to make unannounced visits to Contractor Program sites when it is determined that such unannounced visits are in the interest of effective program management and service delivery. Contractor Program staff shall be informed by City representative(s) upon arrival at Contractor's site of the expected purpose and length of visit so that accommodations may be made. City's representative(s) shall provide proper identification to ensure the safety and security of all parties. City's representative(s) visiting or entering a campus site where students

are present shall comply with Contractor's rules and procedures under Contractor's Board policy series GKC ("Community Relations/Visitors to the Schools").

- 8.13 Contractor understands that the City will timely inform Contractor of the findings of any such review or monitoring, specifically any default under the Agreement or deficiencies in performance, and will inform Contractor in writing of Program strengths and weaknesses and specify a deadline for corrective action based upon contract or federal deadlines and the time needed for City to review, monitor and approve when necessary. The City will assist Contractor in finding solutions for Program improvement if and as appropriate.
- Unless otherwise provided herein, all reports, statements, records, data, policies and procedures or other information requested by DHS shall be submitted by Contractor to City within five (5) Contractor business days of the request made to Contractor via electronic communication or by other form of written correspondence. In the event that Contractor fails to deliver the required reports or information or delivers incomplete information within the prescribed time period, the City may, upon reasonable written notice, suspend reimbursements to Contractor until such reports are delivered to City. Furthermore, the Contractor ensures that all information contained in all required reports or information submitted to City is accurate.
- 8.15 (A) Unless disclosure is authorized by the City or is required by the Attorney General for the State of Texas or his duly authorized representative, Contractor agrees to maintain in confidence all information pertaining to the Project or other information and materials prepared for, provided by, or obtained from City including, without limitation, reports, information, project evaluation, project designs, data, other related information (collectively, the "Confidential Information") and to use the Confidential Information for the sole purpose of performing its obligations pursuant to this Agreement. Contractor shall protect the Confidential Information and shall take all reasonable steps to prevent the unauthorized disclosure, dissemination, or publication of the Confidential Information.
 - (B) If disclosure is permitted by law or required by order of a governmental agency or court of competent jurisdiction, Contractor shall give the Director of DHS prior written notice that such disclosure is required with a full and complete description regarding such requirement.
 - (C) Contractor shall establish specific procedures designed to meet the obligations of this Article VIII, Section 8.15, including, but not limited to execution of confidential disclosure agreements, regarding the Confidential Information with Contractor's employees and subcontractors prior to any disclosure of the Confidential Information to third parties. This Article VIII, Section 8.15 shall not be construed to limit HHS's, the City's or its authorized representatives' right to obtain copies, review and audit records or other information, confidential or otherwise, under this Agreement. Upon termination or expiration of this Agreement, Contractor shall return to City all copies of materials related to the Project, including the Confidential Information and subject to Contractor's right to use "Educational Records" as that term is understood under FERPA. All confidential obligations contained herein (including those pertaining to information transmitted orally) shall survive the termination of this Agreement. The Parties shall ensure that their respective employees, agents, and contractors are notified of the requirement to comply with the aforementioned obligations.
- 8.16 The Public Information Act, Government Code Section 552.021, requires the City to make public information available to the public. Under Government Code Section 552.002(a), public information means information that is written, produced, collected, assembled or maintained under a law or ordinance or in connection with the transaction of official business: 1) by a governmental body; or 2) for a governmental body and the governmental body owns the information, has a right of access to it, or has spent or contributed public money for the purpose of its writing, production, collection, assembly or maintenance. Therefore, if City receives a request under the Public Information Act (i.e., an open records request) for information within Contractor's possession pursuant to this Agreement, Contractor shall forward the requested documents to the City within two (2) Contractor business days of Contractor's designated liaison's receipt of the written request. If the requested information is confidential or may be kept confidential pursuant to state or federal law, the Contractor shall submit to City the list of specific statutory authority mandating and/or authorizing confidentiality no later than three (3) Contractor business days

Contractor's designated liaison's receipt of such request. For the purposes of communicating and coordinating with regard to public information requests, all communications shall be made to the designated public information liaison for each Party. Each Party shall designate in writing to the other Party the public information liaison for its organization and notice of a change in the designated liaison shall be made promptly to the other Party. The parties shall cooperate with each other to preserve confidential information or records that may be excluded from disclosure under the Family Educational Rights and Privacy Act (FERPA) (20 U.S.C. § 1232g; 34 CFR Part 99) and/or the Texas Public Information Act; and the parties shall coordinate efforts to seek any required Attorney General decision for the protection of such information from release.

8.17 In accordance with Texas law, Contractor acknowledges and agrees that all local government records as defined in Chapter 201, Section 201.003 (8) of the Texas Local Government Code created or received in the transaction of official business or the creation or maintenance of which were paid for with public funds are declared to be public property and subject to the provisions of Chapter 201 of the Texas Local Government Code and Subchapter J, Chapter 441 of the Texas Government Code. Thus, Contractor agrees that no such local government records produced by or on the behalf of Contractor pursuant to this Agreement shall be the subject of any copyright or proprietary claim by Contractor; however Contractor shall be entitled to maintain the confidentiality of "educational records" as that term is understood under FERPA and to use such records for educational purposes.

With the exception of student records, Contractor acknowledges and agrees that all local government records, as described herein, produced in the course of the work required by this Agreement, shall belong to and be the property of City and shall be made available to the City at any time unless otherwise prohibited by law. The parties agree that educational records created pursuant to Section 1.1 shall be maintained and utilized by Contractor as required by law. Contractor further agrees to turn over to City all such records upon termination of this Agreement, unless otherwise prohibited by law. Contractor agrees that it shall not, under any circumstances, release any records created during the course of performance of the Agreement to any entity without the written permission of the Director of DHS, unless required to do so by a court of competent jurisdiction or the Texas Attorney General or his designee, or as may be required or permitted by Contractor due to the record being an "educational record" as that term is understood under FERPA. DHS shall be notified of such request as set forth in Article VIII, Section 8.15 of this Agreement.

8.18 Ownership of Intellectual Property. Contractor and City agree that the Project shall be and remain the sole and exclusive proprietary property of City. The Project shall be deemed a "work for hire" within the meaning of the copyright laws of the United States, and ownership of the Project and all rights therein shall be solely vested in City. Contractor hereby grants, sells, assigns, and conveys to City all rights in and to the Project and the tangible and intangible property rights relating to or arising out of the Project, including, without limitation, any and all copyright, patent and trade secret rights. All intellectual property rights including, without limitation, patent, copyright, trade secret, trademark, brand names, color schemes, designs, screens, displays, user interfaces, data structures, organization, sequences of operation, trade dress, and other proprietary rights (the "Intellectual Property Rights") in the Project shall be solely vested in City. As owner of the tangible and intangible intellectual property, City shall have the right to reproduce, publish, authorize others to reproduce or publish, or otherwise use such material subject to confidentiality obligations as may be required by federal and /or state law for "educational records." Contractor agrees to execute all documents reasonably requested by City to perfect and establish City's right to the Intellectual Property Rights. In the event City shall be unable, after reasonable effort, to secure Contractor's signature on any documents relating to Intellectual Property Rights in the Project, including without limitation, any letters patent, copyright, or other protection relating to the Project, for any reason whatsoever, Contractor hereby irrevocably designates and appoints City and its duly authorized officers and agents as Contractor's agent and attorney-in-fact, to act for and in Contractor's behalf and stead to execute and file any such application or applications and to do all other lawfully permitted acts to further the prosecution and issuance of letters patent, copyright or other analogous protection thereon with the same legal force and effect as if executed by Contractor. Provided, however, nothing herein contained is intended nor shall it be construed to require Contractor to transfer any ownership interest in Contractor's best practice and benchmarking information to the City.

- 8.19 In the event that Contractor desires to copyright material or to permit any third-party to do so, Contractor must obtain City's prior written approval to do so and must appropriately acknowledge City's support in any such materials.
- Subject to obligations to maintain confidentiality under the Health Insurance Portability and Accountability Act of 1996 ("HIPAA") and the HIPAA Business Associate Agreement, attached hereto and incorporated herein for all purposes as Attachment V, and subject to the requirements of the Family Educational Rights and Privacy Act ("FERPA") and the limitations imposed under law regarding transfer of information, any and all writings, documents or information in whatsoever form and character produced by Contractor pursuant to the provisions of this Agreement is the exclusive property of City; and no such writing, document or information shall be the subject of any copyright or proprietary claim by Contractor. Contractor understands and acknowledges that as the exclusive owner of any and all such writings, documents and information, City has the right to use all such writings, documents and information as City desires, without restriction; subject, however to Contractor's continuing rights regarding "educational records" as that term is understood under FERPA. City agrees that it will not release to the public "educational records" that come into its possession unless otherwise authorized by law.

IX. INSURANCE

9.1 Contractor and the City each maintain adequate general liability insurance coverage and self-insurance for worker's compensation claims and causes of action to meet their statutory obligations to each party's employees.

X. LIMITED LIABILITY

10.1 CONTRACTOR AND THE CITY ACKNOWLEDGE THEY ARE POLITICAL SUBDIVISIONS OF THE STATE OF TEXAS AND ARE SUBJECT TO, AND COMPLY WITH THE APPLICABLE PROVISIONS OF THE TEXAS TORT CLAIMS ACT, AS SET OUT IN THE CIVIL PRACTICE AND REMEDIES CODE, SECTION 101.001, ET. SEQ., AND THE REMEDIES AUTHORIZED THEREIN REGARDING CLAIMS OR CAUSES OF ACTION THAT MAY BE ASSERTED BY THIRD PARTIES FOR ACCIDENT, INJURY OR DEATH.

XI. THIS ARTICLE INTENTIONALLY LEFT BLANK

XII. APPLICABLE LAWS

- 12.1 Contractor, and all of the work performed under this Agreement, shall comply with all applicable laws, rules, regulations and codes of the United States and the State of Texas and with the charter, ordinances, bond ordinances, and rules and regulations of the City of San Antonio and Bexar County. Contractor agrees to abide by any and all future amendments or additions to such laws, rules, regulations, policies and procedures as they may be promulgated. Contractor may request additional time to come into and demonstrate compliance, unless immediate compliance is required by the applicable Head Start regulations, the Head Start Act, or Public Information Act request requirements. Should the City need to abide by some other law, rule, regulation, policy or procedure, such requirement will be made known to Contractor upon consideration of Contractor's request for additional time.
- 12.2 The Contractor understands that certain funds provided it pursuant to this Agreement are funds which have been made available by the City's General Operating Budget and/or by federal, state, or other granting entities. Contractor agrees to comply with all laws, ordinances, codes, rules, regulations, policies, and procedures, including licensing and accreditation standards applicable to the funds received by Contractor hereunder as directed by the City or as otherwise required in this Agreement, including but not limited to:

- (A) The Head Start Act (42 U.S.C. §9831 et seq., as amended);
- (B) 45 C.F.R. Part 1301 et seq .;
- (C) The Terms of the Grant;
- (D) As applicable, 45 C.F.R. Part 75 ("Uniform Administrative Requirements, Cost Principles, and Audit Requirements for HHS Awards");
- (E) Texas Child Care Licensing laws;
- (F) The Office of Management and Budget (OMB) Circular at 2 C.F.R. 200 et. al. titled "Uniform Administrative Requirements, Cost Principles, and Audit Requirements for Federal Awards," ("Uniform Guidance"), as applicable to the funds received by Contractor hereunder;
- (G) Official record retention schedules as established by the Local Government Records Act of 1989; and
- (H) The Texas Public Information Act, at Chapter 552, The Texas Government Code.

12.3 Contractor further understands and agrees:

- (A) Contractor shall comply with all standards, orders or regulations issued pursuant to the Clean Air Act (42 U.S.C. §§7401-7671q) and the Federal Water Pollution Control Act (33 U.S.C. §§1251-1387), as amended and as applicable. Contractor agrees to report each violation to the City and understands that the City will, in turn, report each violation as required to the federal agency providing funds for this Contract and the appropriate EPA Regional Office. Additionally, Contractor agrees to include these requirements in each subcontract to this Contract exceeding \$150,000.00 financed in whole or in part with federal funds
- (B) to make positive efforts to utilize small businesses, minority-owned firms and women's business enterprises in connection with the work performed hereunder, whenever possible.
- (C) to provide for the rights of the Federal Government in any invention resulting from the work performed hereunder, in accordance with 37 C.F.R. Part 401 and any applicable implementing regulations.
- (D) to include a provision requiring compliance with the Copeland "Anti-Kickback" Act (40 U.S.C 3145), as applicable under Appendix II of the OMB Uniform Guidance and as supplemented by Department of Labor regulations at 29 C.F.R. Part 3, "Contractors and Subcontractors on Public Building or Public Work Financed in Whole or In Part by Loans or Grants from the United States".
- (E) to comply with the Davis-Bacon Act, as amended (40 U.S.C. 3141-3144, and 3146-3148) and as applicable under Appendix II of the OMB Uniform Guidance, and as supplemented by Department of Labor regulations at 29 C.F.R. Part 5, implementing regulations, and the relevant Additional OMB Provisions attached hereto and incorporated herein for all purposes as Attachment VI, and to include a provision requiring compliance with the each in any construction contracts of more than \$2,000.00, and report all suspected or reported violations to HHS.
- (F) to comply with the certification and disclosure requirements of the Byrd Anti-Lobbying Amendment (31 U.S.C. § 1352), and any applicable implementing regulations. Contractor verifies it has tendered said Certificate to the City.
- (G) to comply with the applicable standards under the McKinney-Vento Homeless Assistance Act (42 U.S.C. §§11431 et seq.), and any applicable implementing regulations, as may be applicable.

- (H) to comply with the Contract Work Hours and Safety Standards Act (40 USC 3701-3708) as applicable under Appendix II of the OMB Uniform Guidance, and as supplemented by Department of Labor regulations (29 CFR Part 5), relating to all contracts that involve the employment of mechanics or laborers, and the relevant provisions in Attachment VI, which provides, in part, that each contractor must be required to compute the wages of every mechanic and laborer on the basis of a standard work week of 40 hours, that work in excess of the standard work week be compensated at a rate at least one and a half times the basic rate of pay, and that no laborer or mechanic must be required to work under conditions which are unsanitary, hazardous or dangerous.
- (I) to comply with the prohibitions contained in the Pro-Children Act of 1994 (20 U.S.C §6081-84), relating to no smoking within any indoor facility (or portion thereof) owned or leased or contracted for by Contractor for the provision of regular or routine health care or day care or early childhood development (Head Start) services to children or for the use of the employees of the City or Contractor who provide such services.
- (J) Contractor and its subcontractors shall comply with section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act, including, but not limited to, the regulatory provisions of 40 CFR Part 247, and Executive Order 12873, as they apply to the procurement of the items designated in Subpart B of 40 CFR Part 247.
- 12.4 The Contractor certifies that it will provide a drug-free workplace in compliance with the Drug-Free Workplace Act of 1988 (41 U.S.C. §§ 701-707 and 8101-8106, as amended). Failure to comply with the above-referenced law could subject the Contractor to suspension of payments, termination of Agreement, and debarment and suspension actions.
- 12.5 Contractor shall comply with all federal, state, or local laws, rules, and orders prohibiting discrimination, and shall not engage in employment practices which have the effect of discriminating against any employee or applicant for employment, and will take affirmative steps to ensure that applicants are employed and employees are treated during employment without regard to their race, color, religion, national origin, sex, sexual orientation, gender identity, veteran status, age, disability, or political belief or affiliation, unless exempted by state or federal law, or as otherwise established herein. Consistent with the foregoing, Contractor agrees to comply with Executive Order 11246, entitled "Equal Employment Opportunity", as amended by Executive Orders 13665 and 11375, and as supplemented by regulations at 41 C.F.R. Part 60. Additionally, Contractor certifies that it will comply fully with the following nondiscrimination, minimum wage and equal opportunity provisions, including but not limited to:
 - (A) Title VII of the Civil Rights Act of 1964, as amended;
 - (B) Section 504 of the Rehabilitation Act of 1973, as amended;
 - (C) The Age Discrimination Act of 1975, as amended
 - (D) Title IX of the Education Amendments of 1972, as amended; (Title 20 USC sections 1681-1688)
 - (E) Fair Labor Standards Act of 1938, as amended
 - (F) Equal Pay Act of 1963, P.L. 88-38; and
 - (G) All applicable regulations implementing the above laws.
 - 12.5.1 Further, if Contractor engages in any contract for construction, as defined by 41 C.F.R. Part 60-1.3, using Head Start funds that, except as otherwise provided under 41 C.F.R. Part 60, meets the definition of "federally assisted construction contract" in 41 C.F.R. Part 60-1.3, Contractor must comply with the Equal Employment Opportunity provisions in Attachment VI and all of the Executive Order and Code of Federal Regulations provisions previously cited in Section 12.5, and must include the provisions in any of its subcontracts.
- 12.6 The Contractor warrants that any and all taxes that the Contractor may be obligated for, including but not limited to, federal, state, and local taxes, fees, special assessments, Federal and State payroll and income taxes, personal property, real estate, sales and franchise taxes, are current, and paid to the fullest extent

liable as of the execution date of the Agreement. The Contractor shall comply with all applicable local, State, and Federal laws including, but not limited to, related to:

- (A) worker's compensation;
- (B) unemployment insurance
- (C) timely deposits of payroll deductions;
- (D) filing of Information on Tax Return form 990 or 990T, Quarterly Tax Return Form 941, W-2's Form 1099 on individuals who received compensation other than wages, such as car allowance, Forms 1099 and 1096 for contract or consultant work, non-employee compensation, etc;
- (E) Occupational Safety and Health Act regulations; and
- (F) Employee Retirement Income Security Act of 1974, P.L. 93-406.
- 12.7 Contractor agrees to comply with the Americans with Disabilities Act of 1990, 42 U.S.C. 12101 et seq., and all regulations thereunder.
- 12.8 All expenditures by the Contractor or any of its subcontractors must be made in accordance with all applicable federal, state and local laws, rules and regulations.
- 12.9 If applicable, Contractor shall submit to DHS its most recent form 990 or 990T and also submit any that are filed with the Internal Revenue Service subsequent to its last submission to the City if filed during the term of the Agreement.

XIII. NO SOLICITATION/CONFLICT OF INTEREST

- 13.1 The Contractor warrants that no person or selling agency or other organization has been employed or retained to solicit or secure this Agreement upon a contract or understanding for a commission, percentage, brokerage, or contingent fee and further that no such understanding or agreement exists or has existed with any employee of the Contractor or the City. For breach or violation of this warrant, the City shall have the right to terminate this Agreement without liability or, at its discretion, to deduct from the Agreement or otherwise recover the full amount of such commission, percentage, brokerage, or contingent fee, or to seek such other remedies as legally may be available.
- 13.2 Contractor covenants that neither it nor any member of its governing body or of its staff presently has any interest, direct or indirect, which would conflict in any manner or degree with the performance of services required to be performed under this Agreement. Contractor further covenants that in the performance of this Agreement, no persons having such interest shall be employed or appointed as a member of its governing body or of its staff.
- 13.3 Contractor further covenants that no member of its governing body or of its staff shall possess any interest in, or use their position for, a purpose that is or gives the appearance of being motivated by desire for private gain for themselves or others, particularly those with which they have family, business, or other ties.
- 13.4 No member of City's governing body or of its staff who exercises any function or responsibility in the review or approval of the undertaking or carrying out of this Agreement shall:
 - (A) Participate in any decision relating to this Agreement which may affect his or her personal interest or the interest of any corporation, partnership, or association in which he or she has a direct or indirect interest; or
 - (B) Have any direct or indirect interest in this Agreement or the proceeds thereof.
- 13.5 Contractor acknowledges that it is informed that Charter of the City of San Antonio and its Ethics Code prohibit a City officer or employee, as those terms are defined in Section 2-52 of the Ethics Code, from having a financial interest in any contract with the City. An officer or employee has "prohibited financial interest" in a contract with the City or in the sale to the City of land, materials, supplies or service, if any of

the following individual(s) or entities is a party to the contract or sale: A City officer or employee; his parent, child or spouse; anentity in which the officer or employee, or his parent, child or spouse owns ten (10) percent or more of the voting stock or shares of the entity, or ten (10) percent or more of the fair market value of the entity; an entity in which any individual or entity above listed is subcontractor on a City contract, a partner or a parent or subsidiary entity.

13.6 Contractor warrants and certifies, and this Agreement is made in reliance thereon, that neither the Contractor nor his or her spouse, parent, child, sibling or first-degree relative is a City officer or employee as defined by Section 2-52 (e) of the City Ethics Code. If Contractor is a business entity, the Contractor representative further warrants and certifies that no City officer or employee nor any spouse, parent, child, sibling of first-degree relative of a City officer or employee owns ten (10) percent or more of the voting stock or shares of the business entity, or ten (10) percent or more of the fair market value of the business entity). Contractor further warrants and certifies that is has tendered to the City a Discretionary Contracts Disclosure Statement in compliance with the City's Ethics Code.

XIV. TERMINATION

- 14.1 (a) Termination for Cause Upon written notice in accordance with Article XXVI, Official Communication, CITY may terminate this Contract as of the date provided in the notice in whole or in part, upon the occurrence of either;
 - (i) Contractor's failure to fulfill, in a timely and proper manner, obligations under this Agreement to include performance standards established by the City, or if the Contractor should violate any of the covenants, conditions, or stipulations of the Contract, or
 - (ii) Notification by a local, state, or federal agency of a formal charge, probation, deferred adjudication, or conviction involving fraud, theft, or the commission of a felony by Contractor or Contractor's employee working in the Head Start Program. In the case of a Contractor's employee being the subject of the notification, Contractor shall have the opportunity to cure via the immediate termination and/or removal of the employee from the Head Start Program.
 - (b) Termination for Convenience This Contract may be terminated in whole or in part upon providing notice in accordance with Article XXVI, Official Communication, which shall specify a date, not sooner than 120 days following the day on which notice is sent but not later than the end of Contractor's fiscal year, unless earlier terminated under any other provision herein.
- 14.2 The Contractor shall be entitled to receive just and equitable compensation for any work satisfactorily completed prior to such termination date. The question of satisfactory completion of such work shall be reasonably determined by the City, and its decision shall be final. To the extent that compliance with an Agreement matter falls under HHS authority to review, or in the event that a dispute arises with regard to interpretation of regulations or law as it applies to this Agreement, Contractor may request that the City secure, and the City shall request an interpretation or opinion from HHS in order to assist in resolution of the dispute. It is further expressly understood and agreed by the Parties that Contractor's performance upon which final payment is conditioned shall include, but not be limited to, the Contractor's complete and satisfactory performance, of its obligations for which final payment is sought.
- Notwithstanding any other remedy contained herein or provided by law, the City may delay, suspend, limit, or cancel funds, rights or privileges herein given the Contractor for failure to comply with the terms and provisions of this Agreement. Specifically, at the option of the City, the Contractor may be placed on probation during which time the City may withhold reimbursements in cases where it reasonably determines that the Contractor is not in compliance with this Agreement. The Contractor shall not be relieved of liability to the City for damages sustained by the City by virtue of any breach of this Agreement, and the City may withhold funds otherwise due as damages, in addition to retaining and utilizing any other remedies available to City.

- 14.4 If an employee of Contractor is discharged or otherwise leaves employment with Contractor, then the Contractor shall pay in full to such employee all of such employee's earned salaries and wages, within the timeframe specified by law.
- 14.5 Should the Contractor be debarred by federal government or the City pursuant to a debarment policy currently existing or hereafter adopted, said debarment may be grounds for termination.
- 14.6 This Agreement is subject to the availability of federal grant funds to City and may be terminated by the City if HHS terminates the City as a grantee or reduces the amount granted to City, for any reason; provided that, if the reduction of grant funds does not result in complete unavailability of such funds, the Parties will use best efforts to amend this Agreement accordingly. City will promptly notify Contractor of any such HHS action.
- In all instances of termination, Contractor shall not incur new obligations after the effective date of termination, and shall cancel as many outstanding obligations as possible. Contractor shall submit to City all required reports including a final financial statement which shall be a statement of all expenditures incurred by Contractor under this Agreement. City shall pay Contractor the full cost of obligations that City determines were not subject to cancellation if such costs are properly documented, allowable, within the approved budget, and unavoidably incurred by Contractor prior to termination or expiration. The foregoing shall constitute full and complete reimbursement for all of Contractor's performance under this Agreement.

XV. PROHIBITION OF POLITICAL ACTIVITIES

- 15.1 Contractor agrees that no funds provided from or through the City shall be contributed or used to conduct political activities for the benefit of any candidate for elective public office, political party, organization or cause, whether partisan or non-partisan, nor shall the personnel involved in the administration of the Project provided for in this Agreement be assigned to work for or on behalf of any partisan or non-partisan political activity.
- 15.2 Contractor agrees that no funds provided under this Agreement may be used in any way to attempt to influence, in any manner, a member of Congress or any other State or local elected or appointed official.
- 15.3 The prohibitions set forth in Article XV, Sections 15.1 and 15.2 of this Agreement include, but are not limited to, the following:
 - (A) an activity to further the election or defeat of any candidate for public office or for any activity undertaken to influence the passage, defeat or final content of local, state or federal legislation;
 - (B) working or directing other personnel to work on any political activity during time paid for with City funds, including, but not limited to activities such as taking part in voter registration drives, voter transportation activities, lobbying, collecting contributions, making speeches, organizing or assisting at meetings or rallies, or distributing political literature;
 - (C) coercing personnel, whether directly or indirectly, to work on political activities on their personal time, including activities such as taking part in voter registration drives, voter transportation activities, lobbying, collecting contributions, making speeches, organizing or assisting at meetings or rallies, or distributing political literature; and
 - (D) using facilities or equipment paid for, in whole or in part with City funds for political purposes including physical facilities such as office space, office equipment or supplies, such as telephones, computers, fax machines, during and after regular business hours.
- 15.4 To ensure that the above policies are complied with, Contractor shall provide every member of its personnel paid out of Agreement funds with a statement provided by Contractor of the above prohibitions and have each said individual sign a statement acknowledging receipt of the policy. Such statement shall

include a paragraph that directs any staff person who has knowledge of violations or feels that he or she has been pressured to violate the above policies to call and report the same to DHS. Contractor shall list the name and number of a contact person from DHS on the statement that Contractor's personnel can call to report said violations.

- 15.5 Contractor agrees that in any instance where an investigation of the above is ongoing or has been confirmed based upon then current reasonable evidence of impropriety, salaries paid to the Contractor under this Agreement may, at the City's discretion, be withheld until the situation is resolved.
- 15.6 This Article shall not be construed to prohibit any person from exercising his or her right to express his or her opinion or to limit any individual's right to vote. Further, Contractor and staff members are not prohibited from participating in political activities on their own volition, if done during time not paid for with Agreement funds.

XVI. PERSONNEL

- 16.1 Contractor shall establish and maintain an organizational structure that supports the accomplishment of Program objectives, addresses the major functions and responsibilities assigned to each staff position and provides evidence of adequate mechanisms for staff supervision and support to ensure the effective oversight of the Head Start program operations. Contractor shall ensure that, at a minimum, the program management functions listed in the Scope of Work are assigned to and adopted by staff within the Program.
- 16.2 Contractor acknowledges and agrees that Head Start guidelines and City policy require that Contractor shall staff each Head Start classroom with two teachers, or one teacher and a teacher's aide, at all times. To the fullest extent possible, Contractor shall staff each classroom with a volunteer in addition to the two paid staff persons. Additionally, Contractor will have adequate staff to fully implement all areas of program operations, including those having the specific management functions and expertise set forth in the Scope of Work.
- 16.3 At the beginning of this Agreement period, Contractor shall submit to City a report which specifically (a) lists the number and percentage of classroom personnel in its center based program having child development associate credentials or associate, baccalaureate or advanced degrees; and (b) describes to City Contractor's compliance with the goals described in Sections 16.3, 16.4 and 16.5 of this Agreement.
- 16.4 Contractor understands that the Head Start Act requires that grantees and their contractors, if any, ensure, and demonstrate upon request, that all of the teachers staffing its Head Start center-based program classrooms have (a) a baccalaureate or advanced degree in early childhood education; or (b) a baccalaureate or advanced degree and course work equivalent to a major relating to early childhood education, with experience teaching preschool-age children. Contractor agrees that during the term of this Agreement, Contractor will only employ teachers meeting the necessary qualifications.
- 16.5 Contractor understands that the Head Start Act requires that grantees and their contractors, if any, ensure, and demonstrate upon request, that all teaching assistants staffing its Head Start center-based program classrooms (a) have at least a child development associate credential; or (b) are enrolled in a program leading to an associate or baccalaureate degree. Contractor agrees that during the term of this Agreement, Contractor will only employ teaching assistants meeting the necessary qualifications.
- 16.6 Contractor understands that the Head Start Act requires that grantees and their contractors, if any, ensure, and demonstrate upon request, that all Head Start education coordinators in its center-based programs, including those who serve as curriculum specialists: (a) have the capacity to offer assistance to other teachers in the implementation and adaptation of curricula to the group and individual needs of children in a Head Start classroom; and (b) have (i) a baccalaureate or advanced degree in early childhood education or (ii) a baccalaureate or advanced degree and coursework equivalent to a major relating to early childhood education, with experience teaching preschool-age children. Contractor agrees that during the term of this Agreement, Contractor will only employ education coordinators meeting the necessary qualifications.

- 16.7 Contractor agrees and acknowledges that each of its Head Start teachers shall attend not less than 15 clock hours of professional development per Contractor fiscal year. The term "professional development" means high-quality activities that will improve the knowledge and skills of Head Start teachers and staff, as relevant to their roles and functions, in program administration and the provision of services and instruction, as appropriate, in a manner that has a positive and lasting impact and improves service delivery to enrolled children and their families.
- 16.8 Contractor understands that the Head Start Act requires grantees and their contractors, if any, ensure and demonstrate upon request that all Head Start staff have the knowledge, skills, and experience they need to perform their assigned functions responsibly. Therefore, in addition to all other provisions in this article, Contractor agrees that Contractor's Head Start Director and all Program center site directors, site managers, content coordinators and other administrators must have education and/or training in the area of early childhood education and family support. To meet these obligations, Contractor will:
 - (A) only employ a Head Start Director, and Program center site directors, site managers, content coordinators and other Program administrators meeting the necessary qualifications;
 - (B) promptly notify City of any transfers and/or disciplinary actions affecting Head Start personnel referred to in subsection (A) of this section 16.8; and
 - (C) ensure, in the event a Head Start position referred to in subsection (A) of this section 16.8 is vacated for any reason, that a replacement meeting the necessary qualifications is hired within 60 calendar days or, following notification to City of a delay, as soon as reasonably practical, but not later than 90 days after the position first became vacant unless the parties come to a new mutually acceptable deadline.
- 16.9 Contractor shall create, in consultation with each of its employees, and implement a professional development plan for all Head Start employees who provide direct services to children. Contractor shall regularly evaluate such plans to determine their impact on teacher and staff effectiveness.
- 16.10 Contractor understands that the City shall periodically perform its own wage and salary comparison and shall issue such results to Contractor. Contractor understands and agrees that City shall have no obligation to reimburse Contractor in excess of wages to an employee that exceed the average rate of compensation paid to persons providing substantially comparable services in the area. For purposes of this Agreement, the City shall accept the wage information set forth in the most recent study commissioned by, or issued by, the Texas Association of School Boards, in determining the rate of reimbursement. Although the City may consider factors such as training and experience as affecting compensation levels, the City shall have the sole and absolute authority to determine the rate of City's reimbursement under the Agreement, which determination shall not be arbitrary or unreasonable, and its decision shall be final due to the City's obligation of ensuring that wage comparability studies meet the requirements of the Head Start Act and implementing regulations. Subject to the restriction set forth in 16.10, Contractor may compensate its employees above the rate the City will reimburse, so long as the additional compensation is not charged to the Agreement budget.
- 16.11 Contractor expressly understands and agrees that in accordance with 42 U.S.C. §9848, no portion of the Agreement funds provided hereunder may be used to pay its employee if compensation (including non-federal funds) to that employee exceeds \$179,700.00, or the maximum authorized compensation as may be adjusted from time to time. Furthermore, Contractor agrees that all employees must devote to the Head Start Program the time proportionate to the percentage of their compensation funded through the Head Start program grant (e.g., employees who are one hundred percent (100%) funded through the Head Start grant must devote one hundred percent (100%) of their time and effort to support the Head Start program). Contractor agrees to submit employee certifications if requested by the City or HHS.
- 16.12 The Contractor agrees to establish internal procedures that assure employees of an established complaint and grievance process. The grievance process will include procedures to receive, investigate, and resolve complaints and grievances in an expeditious manner.
- 16.13 Contractor agrees to comply with all applicable federal regulations regarding the setting of, and maximum amount allowable for, salaries of Contractor's employees.

- 16.14 Contractor agrees to continue to work on placing written job descriptions for Head Start personnel in individual personnel folders for each position in the organization, or to have them placed online, until completed, but that all descriptions shall be filed or online no later than the expiration date of the Agreement. Contractor agrees to provide to the City specific job description(s) upon request.
- 16.15 The Contractor agrees to provide the City with the names and license registration of any employees of Contractor regulated by State law whose activities contribute towards, facilitate, or coordinate the performance of this Agreement.
- 16.16 Chief Executive Officers (CEOs), directors and other supervisory personnel of Contractor may not supervise a spouse, parents, children, brothers, sisters, and in-laws standing in the same relationship, (hereinafter referred to as "Relatives") who are involved in any capacity with program delivery supported through Agreement funds. Relatives, however, may be co-workers in the same Project in a non-supervisory position.

XVII. ADVERSARIAL PROCEEDINGS

- 17.1 Except in circumstances where the following is in conflict with federal law or regulations pertaining to the Head Start Program, Contractor agrees to comply with the following special provisions:
 - (A) Under no circumstances will the funds received under this Agreement be used, either directly or indirectly, to pay costs or attorney fees incurred in any adversarial proceeding against the City or any other public entity; and
 - (B) Contractor, at the City's option, could be ineligible for consideration to receive any future funding while any adversarial proceeding against the City remains unresolved.

XVIII. FEDERAL AND CITY-SUPPORTED PROJECT

- 18.1 This Section is applicable to all publicity, public presentations, signs, press releases, public notices, flyers, brochures, marketing materials, and other informational material prepared, created, posted and/or disseminated during the term of the Agreement by Contractor. Contractor shall obtain City's prior approval of the language and logo, as applicable, to be used, and the Parties agree that all publicity regarding the establishment or operation of the Head Start Program affiliation between City and Contractor described herein shall be planned and implemented as mutually agreed to in advance by the Parties. Contractor agrees that all public notices and any publicity, signs and/or marketing materials regarding any program which is funded by this Agreement shall provide a written statement acknowledging the role of the Federal funds provided by HHS through City hereunder, which shall read as follows: "The City of San Antonio Head Start Program services provided by the San Antonio Independent School District are funded by the City of San Antonio, Department of Human Services, through a federal grant received from the U.S. Department of Health and Human Services." These public notices or signs include, but are not limited to, signs identifying the facilities from which these programs are provided, and electronic media. In addition, all publicity related to Contractor's Head Start services shall note that the Program is operated on a nondiscriminatory basis.
- 18.2 Contractor further agrees to provide City with a copy of all proposed official communications to the public, Head Start parents, and employees as it may relate to the City's implementation of the City's Head Start Program model, the transition of Head Start contracts or transition of the Head Start Program, and to obtain the City's approval prior to dissemination.

XIX. PROPERTY, EQUIPMENT AND SUPPLIES

19.1 The City retains ownership of all equipment/property purchased with funds received through the City and such equipment/property shall, at the City's sole option, revert to the City at Agreement's termination, for

whatever reason. The Contractor agrees to relinquish and transfer possession of and, if applicable, title to said property without the requirement of a court order upon termination of this Agreement. Equipment that has reverted to the Contractor through a City-paid lease agreement with option to buy will be considered the same as though the equipment was purchased outright with Agreement funds. It is understood that the terms, "equipment" and "property", as used herein, means tangible, nonexpendable, personal property having a useful life of more than one year and an acquisition cost of \$5,000 or more per unit and shall include not only furniture and other durable property, but also vehicles, but shall not include supplies and consumables.

- 19.2 Contractor agrees that no equipment purchased with Agreement funds may be disposed of without receiving prior written approval from DHS. In cases of theft and/or loss of equipment, it is the responsibility of the Contractor to replace it with like equipment. Agreement funds cannot be used to replace equipment in those instances. All replacement equipment will be treated in the same manner as equipment purchased with Agreement funds.
- 19.3 Contractor shall maintain accurate and complete records on all equipment and property obtained with Agreement funds to include:
 - (A) A description of the equipment, including the model and serial number or other identification number, if applicable;
 - (B) The date of acquisition, cost and procurement source, purchase order number, and vendor number;
 - (C) An indication of whether the equipment is new or used;
 - (D) The vendor's name (or transferred from);
 - (E) The location of the property;
 - (F) The property number shown on the property tag; ("City of San Antonio Head Start Program"); and
 - (G) A list of disposed items and disposition
- 19.4 The Contractor is fully and solely responsible for the insuring, safeguarding, maintaining, and reporting of lost, stolen, missing, damaged, or destroyed equipment/property purchased or leased with Agreement funds. Contractor shall inform the City of incidents of loss, theft, damage or destruction of equipment (as defined in 19.1 above) or property, excluding supplies and consumables, purchased or leased with Agreement funds.
- 19.5 All equipment purchased under this Agreement shall be fully insured against fire, loss and theft. Contractor shall, at a minimum, provide the equivalent insurance for real property and equipment acquired with Agreement funds as provided to other property acquired or owned by the Contractor.
- 19.6 Upon request, Contractor shall provide an annual inventory of assets purchased with funds received through the City to DHS.
- 19.7 Contractor shall fully comply with the property and equipment requirements of 45 C.F.R Part 75, including but not limited to Sections 75.316 through 75.323, related to the following:
 - (A) Insurance Coverage
 - (B) Real Property
 - (C) Federally-owned and exempt property
 - (D) Equipment
 - (E) Supplies
 - (F) Intangible property
 - (G) Property trust relationship
- 19.8 Relative to property, equipment and supplies purchased with Head Start grant funds: For individual equipment purchases in the amount of \$5,000.00 or greater or cumulative purchases in the amount of \$100,000.00 or greater, Contractor shall obtain prior approval from DHS. Contractor shall not split the purchase of a line item with a value greater than the preceding thresholds in order to avoid obtaining approval from DHS.

- 19.9 Contractor will maintain a system for tracking, on an ongoing basis, inventory of equipment and supplies purchased with Head Start funds that either (i) has a purchase price of \$5,000.00 or greater; or (ii) meets such other criteria as City may prescribe (and which City shall notify Contractor as appropriate). Upon request, Contractor will provide City a status report of the current inventory of equipment and supplies meeting these requirements. City shall have the right to review and approve Contractor's inventory tracking system.
- 19.10 City reserves the right to require transfer of property acquired with funds awarded under this Agreement as provided in 45 C.F.R. Part 75, including but not limited to §75.316 et seq.
- 19.11 Contractor acknowledges and agrees that City is the intended third-party beneficiary of any and all facility leases with third-parties to which Contractor is or becomes a party in connection with the approved Program sites listed in Exhibit I-A to Attachment I, or as a consequence of this Agreement. As such, Contractor shall use its best efforts to cause the lessor of any such lease to execute an acknowledgment prepared by City that City is an intended third-party beneficiary of such lease. Contractor shall honor all of its material obligations under any and all such leases. Contractor shall stay in good standing under any and all such leases and Contractor shall immediately notify City in writing in the event of any breach or alleged breach of any such lease that could result in the termination of such lease. Contractor shall submit to City for review and approval all non-disturbance, subordination and similar agreements it is requested to execute in connection with any such lease. In the event of an event giving rise to a right of first refusal in favor of Contractor under any such lease, Contractor shall promptly notify City of such event and allow City to step into Contractor's shoes at tenant under such lease in order to exercise such right.

XX. TRAVEL

- 20.1 The costs associated with budgeted travel for business, either in-town or out-of-town, are allowable costs provided documentation of expenses is present.
 - (A) Contractor agrees that mileage reimbursement paid to Contractor's employees shall be reimbursed at a rate no more liberal than the City's policy for mileage reimbursement, which is consistent with Internal Revenue Service (IRS) rules. Contractor further agrees that in order for its employees to be eligible for mileage reimbursement, the employees 1) shall be required to possess a valid Texas Driver's License and liability insurance as required by law, and 2) must fully comply with its own established mileage recording policies. Mileage records are subject to spot-checks by City auditors and monitors. Contractor shall strongly encourage the participation by its employees in an approved defensive driving course. Evidence of the required driver's license and liability insurance must be kept on file with the Contractor.
 - (B) Contractor agrees that in order to obtain reimbursement of the costs associated with budgeted out of town travel for business in connection with this Agreement, Contractor shall:
 - obtain City's prior approval, unless such expense has been previously submitted and approved
 as part of Contractor's budget, and provide City with detailed documentation of such business
 travel expense(s), (any amount over approved budgeted amounts must be pre-approved by
 City or such overage will not be paid),
 - ensure that any and all costs associated with out-of-town travel (including per diem rates) shall not be more liberal than the City's travel policies which conform with the reimbursement rates established by the United States General Services Administration,
 - 3) purchase all business travel at economy class rates and shall document such, and
 - maintain supporting documentation for conferences to include itineraries and documentation certifying conference attendance and provide such documentation to City upon request.

XXI. NO USE OF FUNDS FOR RELIGIOUS ACTIVITIES

21.1 Contractor agrees that none of the performance rendered hereunder shall involve, and no portion of the funds received hereunder shall be used, directly or indirectly, for the construction, operations, maintenance or administration of any sectarian or religious facility or activity, nor shall said performance rendered or funds received be utilized so as to benefit, directly or indirectly, any such sectarian or religious facility or activity.

XXII. DEBARMENT

- 22.1 Contractor certifies that neither it nor its principals are presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from participation in any state or federal program.
- 22.2 Contractor shall provide immediate written notice to City, in accordance with the notice requirements of Article XXVI herein, if, at any time during the term of the Agreement, including any renewals hereof, Contractor learns that its certification was erroneous when made or have become erroneous by reason of changed circumstances.

XXIII. ASSIGNMENT

23.1 Contractor shall not assign or transfer Contractor's interest in this Agreement or any portion thereof without the written consent of the City, and if applicable, the Grantor of the Grant Funds. Any attempt to transfer, pledge or otherwise assign shall be void ab initio and shall confer no rights upon any third person or party.

XXIV. AMENDMENT

- 24.1 Any alterations, additions or deletions to the terms hereof shall be by amendment in writing executed by both City and Contractor and evidenced by passage of a subsequent City ordinance, as to City's approval; provided, however, the Director of DHS shall have the authority to execute an amendment of this Agreement without the necessity of seeking any further approval by the City Council of the City of San Antonio if permitted by all applicable local, state and federal laws, and in the following circumstances:
 - A. an increase in funding (the cause of which is unrelated to the reason set forth in Section 24.1(E) below) of this Agreement in an amount not exceeding (a) twenty-five percent (25%) of the total amount of this Agreement or (b) \$25,000.00, whichever is the lesser amount; provided, however, that the cumulative total of all amendments executed without City Council approval pursuant to this subsection and increasing Agreement funding during the term of this Agreement shall not exceed the foregoing amount;
 - B. modifications to the Scope of Work set forth in Attachment I hereto, so long as the terms of the amendment stay within the substantive parameters set forth in the original Scope of Work, also set forth in Attachment I hereto;
 - C. budget line item shifts of funds, so long as the total dollar amount of the budget set forth in Section 3.1 of this Agreement remains unchanged (these modifications may be accomplished through Budget Revisions);
 - D. modifications to the insurance provisions described in Article IX of this Agreement that receive the prior written approval of the City of San Antonio's Risk Manager and the Director of DHS.
 - E. decreases (and increases if the City agrees to allocate additional enrollment slots to Contractor) in Agreement funding based upon Head Start Program enrollment levels, and modifications to Agreement terms related to enrollment; provided, however, that the cumulative total of all Head Start Program contracts, as amended, shall not exceed the City's total budget for the Head Start Program budget for the applicable grant year. Contractor shall execute any and all amendments to this Agreement that are required as a result of a modification made pursuant to this Section 24.1(E).
- 24.2 Contractor's Superintendent shall have the authority to execute an amendment of this Agreement to the same extent as the Director of DHS under Section 24.1 of this Agreement without the necessity of seeking

any further approval by Contractor's Board of Trustees, if not otherwise prohibited by federal or state law or regulation or prohibited by Contractor's policies, rules or Board directives.

24.3 Contractor further agrees that except when the terms of this Agreement expressly provide otherwise, any alterations, additions or deletions to the terms hereof shall be by amendment in writing and approved by HHS.

XXV. SUBCONTRACTING

- 25.1 None of the work or services covered by this Agreement shall be sub-contracted without the prior written consent of the City and the Grantor of the Grant Funds, if so required by said Grantor.
- 25.2 Contractor must comply with all applicable local, State and Federal procurement standards, rules, regulations and laws in all its sub-contracts related to the work or funds herein. It is further agreed by the Parties hereto that the City has the authority to monitor, audit, examine, and make copies and transcripts of all sub-contracts, as often as deemed appropriate by the City. If, in the sole determination of the City, it is found that all applicable local, State and Federal procurement standards, rules, regulations and laws have not been met by Contractor with respect to any of its sub-contracts, then the Contractor will be deemed to be in default of this Agreement, and as such, this Agreement will be subject to termination in accordance with the provisions hereof.
- 25.3 Any work or services for sub-contracting hereunder, shall be sub-contracted only by written Agreement, and unless specific waiver is granted in writing by City, shall be subject by its terms to each and every provision of this Agreement. Compliance by sub-contractors with this Agreement shall be the responsibility of Contractor. Contractor agrees that payment for services of any sub-contractor shall be submitted through Contractor, and Contractor shall be responsible for all payments to sub-contractors.
- 25.4 Contractor certifies that its subcontractors are not presently debarred, suspended or proposed for debarment, declared ineligible or voluntarily excluded from participation in any state or federal program.
- 25.5 Contractor understands and agrees that all subcontracts for performance of this Agreement in excess of \$10,000.00 must address termination for cause and for convenience, including the manner by which the termination will be effected and the basis for settlement.

XXVI. OFFICIAL COMMUNICATIONS

Except where the terms of this Agreement expressly provide otherwise, any election, notice or communication required or permitted to be given under this Agreement shall be in writing and deemed to have been duly given if and when delivered personally (with receipt acknowledged), or three (3) days after depositing same in the U.S. mail, first class, with proper postage prepaid, or upon receipt if sending the same by certified mail, return receipt requested, or upon receipt when sent by a commercial courier service (such as Federal Express or DHL Worldwide Express) for expedited delivery to be confirmed in writing by such courier, at the addresses set forth below or to such other address as either Party may from time to time designate in writing.

City:
Director
Department of Human Services
106 S. St. Mary's Street, Suite 700
San Antonio, TX 78205

Contractor: Superintendent Edgewood Independent School District (EISD) 5358 W. Commerce Street San Antonio, TX 78237

Notices of changes of address by either Party must be made in writing delivered to the other Party's last known address within five (5) business days of the change.

XXVII. VENUE

27.1 Contractor and City agree that this Agreement shall be governed by and construed in accordance with the laws of the State of Texas. Any action or proceeding brought to enforce the terms of this Agreement or adjudicate any dispute arising out of this Agreement shall be bought in a court of competent jurisdiction in San Antonio, Bexar County, Texas.

XXVIII, GENDER

28.1 Words of any gender used in this Agreement shall be held and construed to include any other gender, and words in the singular number shall be held to include the plural, unless the context otherwise requires.

XXIX. REPRESENTATIONS AND OTHER OBLIGATIONS

- 29.1 The signer of this Agreement for Contractor represents, warrants, assures and guarantees that (s)he has full legal authority to execute this Agreement on behalf of Contractor and to bind Contractor to all of the terms, conditions, provisions and obligations herein contained. Whether a non-profit or public entity, Contractor must be authorized to do business in the State of Texas and be formed under and operating in accordance with all applicable laws of the State of Texas. Upon request by the City, Contractor shall provide DHS verification of the foregoing requirements.
- 29.2 This Agreement is based on the representation of Contractor that it is financially viable, solvent, and accountable for its expenditures; that it has the continuing capability to furnish the non-Federal Share of the cost of operating its approved Contractor Program; and that Program funds disbursed to Contractor will be expended only for allowable costs in the implementation of the Contractor Program which is the subject of this Agreement. Contractor represents that there are no financial limitations or impediments that would make it not viable, solvent and accountable such that the flow of Program funds might be diverted away from the operation and maintenance of the Program which is the subject of this Agreement.
- 29.3 In the event that circumstances arise which might result in interference with Contractor's ability to provide the services which are the subject of this Agreement, Contractor agrees to inform City of those circumstances immediately upon their discovery. Contractor agrees that reimbursement to Contractor, upon reasonable notice, may be suspended by City until such financial circumstances giving rise to the possible interference with the operation of the Program have been eliminated, provided, however, that authorized expenditures made prior to the suspension, and approved by City shall be disbursed pursuant to the terms of this Agreement.

XXX, LICENSES AND TRAINING

30.1 Contractor warrants and certifies that Contractor's employees and its subcontractors have the requisite training, license or certification to provide said services, and meet all competence standards promulgated by all other authoritative bodies, as applicable to the services provided herein.

XXXI. INDEPENDENT CONTRACTOR

31.1 It is expressly understood and agreed that the Contractor is and shall be deemed to be an independent contractor, responsible for its respective acts or omissions and that the City shall in no way be responsible therefor, and that neither Party hereto has authority to bind the other nor to hold out to third parties that it has the authority to bind the other.

- 31.2 Nothing contained herein shall be deemed or construed by the parties hereto or by any third party as creating the relationship of employer-employee, principal-agent, partners, joint venture, or any other similar such relationship, between the Parties hereto.
- 31.3 Any and all of the employees of the Contractor, wherever located, while engaged in the performance of any work required by the City under this Agreement shall be considered employees of the Contractor only, and not of the City, and any and all claims that may arise from the Workers' Compensation Act on behalf of said employees while so engaged shall be the sole obligation and responsibility of the Contractor.

XXXII. SEVERABILITY

32.1 If any clause or provision of this Agreement is held invalid, illegal or unenforceable under present or future federal, state or local laws, including but not limited to the City Charter, City Code, or ordinances of City, then and in that event it is the intention of the Parties hereto that such invalidity, illegality or unenforceability shall not affect any other clause or provision hereof and that the remainder of this Agreement shall be construed as if such invalid, illegal or unenforceable clause or provision was never contained herein; it is also the intention of the Parties hereto that in lieu of each clause or provision of this Agreement that is invalid, illegal or unenforceable, there be added as a part of this Agreement a clause or provision as similar in terms to such invalid, illegal or unenforceable clause or provision as may be possible, legal, valid and enforceable.

XXXIII. NON-WAIVER OF PERFORMANCE

33.1 No waiver by City of a breach of any of the terms, conditions, covenants or guarantees of this Agreement shall be construed or held to be a waiver of any succeeding or preceding breach of the same or any other term, condition, covenant or guarantee herein contained. Further, any failure of City to insist in any one or more cases upon the strict performance of any of the covenants of this Agreement, or to exercise any option herein contained, shall in no event be construed as a waiver or relinquishment for the future of such covenant or option.

XXXIV. CONTRIBUTION PROHIBITIONS

- 34.1 Contractor acknowledges that City Code Section 2-309 provides that any person acting as a legal signatory to a proposed contractual relationship that applies for a "high profile" discretionary contract, as defined by the City of San Antonio Contracting Policy and Process Manual, may not make a campaign contribution to any councilmember or candidate at any time from the tenth business day after the Request for Proposal (RFP) or Request for Qualifications (RFQ) or other solicitation is released, or for a contract for which no competitive solicitation has been issued by the City from the time the City begins discussions or negotiations, and ending on the 30th calendar day following the contract award. Contractor understands that if the legal signatory entering the Agreement has made such a contribution, the City may not award the Agreement to that contributor or to that contributor's business entity. Any legal signatory for a proposed high-risk contract must be identified within the response to the RFP or RFQ, if the identity of the signatory will be different from the individual submitting the response.
- 34.2 Contractor acknowledges that the City has identified this Agreement as high profile.
- 34.3 Contractor warrants and certifies, and this Agreement is made in reliance thereon, that the individual signing this Agreement has not made any contributions in violation of City Code section 2-309, and will not do so for 30 calendar days following the award of this Agreement. Should the signor of this Agreement violate this provision, the City Council may, in its discretion, declare the Agreement void.

XXXV. ENTIRE AGREEMENT

35.1 This Agreement and its attachments, if any, constitute the entire and integrated Agreement between the Parties hereto and contain all of the terms and conditions agreed upon, and supersede all prior negotiations, representations, or agreements, either oral or written. No such other negotiations or representations may be enforced by either Party nor may they be employed for interpretation purposes in any dispute involving this Agreement.

In witness of which this Agreement has been executed the <u>AD</u> day of <u>SEPHEMBLY 2014</u> and is effective for the term hereof.

CITY OF SAN ANTONIO:	CONTRACTOR: Edgewood Independent School District (EISD)	
Melody Woosley, Director Department of Human Services	Dr. Sylvester Perez, Interim Superintendent	
APPROVED AS TO FORM:		
Assistant City Atterney	Board President (if required by Contractor)	

ATTACHMENTS

Attachment I - Scope of Work

Attachment II - Program Budget

Attachment III - Special Provisions

Attachment IV - HHS Award Document

Attachment V - HIPAA Business Associate Agreement, if applicable

Attachment VI - Additional OMB Provisions

ATTACHMENT I [INSERT SCOPE OF WORK HERE]

SCOPE OF WORK

1. Summary

The Contractor shall ensure full enrollment and shall serve the number of income and age eligible children as indicated below, and perform Head Start services in accordance with the Head Start Performance Standards as provided in Head Start Regulations, 45 CFR Part 1301 et seq with the Head Start Act, as amended, 42 U.S.C. 9831 et seq and with the terms of this Contract. The Contractor will operate full day classroom(s) at the City approved sites for a minimum of 177 days of planned class operations for the Head Start grant period.

Number of children in full day care on the first day of the program:	777	
Minimum number of children with disabilities:	78	
Service Area:	Children who reside in San Antonio and are served in the Edgewood Independent School District	

2. Enrollment, Recruitment, Selection, Eligibility and Attendance (ERSEA)

- A. Contractor shall have primary responsibility for the verification of eligibility, recruitment, enrollment, selection of children, and attendance, in accordance with the ERSEA policies established by the City. Contractor shall work jointly with Family and Community Support Service staff conducting recruitment activities, receiving applications, determining eligibility, and completing enrollment forms for Contractor's service area. Contractor shall ensure that it is in compliance with the eligibility, recruitment, selection, enrollment and attendance provisions of the Head Start Act, the Head Start Performance Standards and the City's program policies and other federal guidance related to the Head Start program.
- B. Contractor shall establish and maintain full funded enrollment of 777 Head Start eligible children in the Contractor's Program from the first day of operations. Funded enrollment means the number of children which the Contractor is to serve, as indicated by the City and terms of this Contract. To maintain full enrollment, Contractor must ensure that once it determines that a vacancy exists, no more than 30 calendar days may elapse before the vacancy is filled. Contractor will document for the City its efforts to fill all vacancies within 30 calendar days.
- C. If full funded enrollment is not maintained for a period of 30 days the City may reduce Contractor's funding at the established per capita rate based on the difference between funded enrollment slots and actual enrollment. Contractor shall develop at the beginning of each enrollment year and maintain during the year a waiting list that ranks children according to the program's selection criteria and ERSEA policies. Children on this waiting list may include children who are currently enrolled in the State Pre-Kindergarten program. If Contractor is under enrolled and does not have a waiting list, Contractor will fill any vacancies in the federal Head Start classrooms, and in doing so, may recruit from the state Pre-K classes. City shall have the right to place additional children in Contractor's Program, in coordination with the Contractor, up to the point where Contractor is fully enrolled.
- D. Contractor shall develop a plan that includes recruitment, selection and enrollment services in accordance with City's established ERSEA policies and, in a manner that will promote children receiving two (2) years of Head Start services or more if appropriate. Contractor shall submit to the City its August Program Design that establishes intact and separate 3-year old and 4-year old classrooms prior to the first day of school. If there is a change in the Program Design at any point during the school year, Contractor shall notify the designated Head Start staff in writing within 2 business days thate such change was made. If Contractor is unable to implement the required Program Design, Contractor shall inform the City and receive City approval before establishing an alternative Program Design.

- E. In accordance with the Head Start Act, Contractor shall ensure that a minimum of 10% of the children enrolled in Contractor's Program are children with disabilities who are determined to be eligible for special education and related services. Contractor must meet this 10% requirement by mid-point of the program year, as determined by the City.
 - (i) If Contractor recognizes that it is having difficulty in meeting the percentage with disabilities requirement during its enrollment, recruitment and selection period, Contractor must provide the City with the following:
 - a. a written description of specific steps the Contractor has taken to meet the requirement in the current program year to include efforts made to collaborate with the local agencies providing services under Section 619 and Part C of IDEA, and an explanation of why, despite these efforts, it was unable to meet the 10% requirement
 - a written confirmation from the local IDEA agencies of Contractor's efforts to
 actively collaborate with them to promote the enrollment of children with
 disabilities and specific efforts to recruit and enroll eligible children with
 disabilities; and
 - c. a written description of the Contractor's proposed approach to improve the enrollment of children with disabilities to reach the 10% requirement to include a description of how the IDEA agencies in the community will work with the Contractor to implement this approach
 - (ii) Additionally, if Contractor is unable to meet the 10% requirement, the City shall have the right to place additional children with disabilities in Contractor's Program up to the point where Contractor achieves a 10% enrollment of children with disabilities. City and Contractor shall coordinate placement of additional children with disabilities.
- F. Contractor agrees that when the monthly average daily attendance rate in a center-based program falls below 85 percent, Contractor shall initiate and collaborate with the City in the provision of absentee services in accordance with §1305.8 of the Head Start Performance standards. Absentee services include analyzing the cause of absenteeism and the provision of additional support, which may include, but is not limited to, face-to-face meeting(s) with the family and other direct contact with the child's parents/legal guardian. Contractor shall establish procedures for the removal of children from the program due to the lack of attendance in accordance with the City's policies.

3. Program Services

- A. Contractor shall provide Head Start Program Services (hereinafter defined) to eligible children in the Service Area. Such services shall be provided to eligible children within Contractor's Service Area(s) without regard to age, race, creed, color, sex, or national origin. Any proposal to extend or modify the Service Area(s) or the sites at which services are to be delivered shall be formally submitted in writing to City for approval.
- B. Contractor shall provide (i) Education and Early Childhood Development Services; (ii) Child Disability Services; and (iii) Child Nutrition Services and shall collaborate with City to ensure the provision of Family and Community Support Services, Child Mental Health, Medical Health Services and Dental Health Services, to meet the needs of the children and families served by Contractor's Program ("Head Start Program Services"). Contractor agrees that, notwithstanding the fact that another Head Start service provider under the City of San Antonio Head Start program may be contracted to provide a category of service, Contractor, under the leadership of its Head Start Program Director will be responsible for coordinating with Head Start service providers and working with the City to ensure provision of full array of services to which the children are entitled under the Terms of the Grant. Contractor shall also coordinate with other Independent School Districts, health service providers and agencies in the community in order to provide comprehensive services to the children and families served by the Program and to leverage community resources.

- C. Contractor shall establish and maintain an organizational structure that supports the accomplishment of Program objectives, addresses the major functions and responsibilities assigned to each staff position and provides evidence of adequate mechanisms for staff supervision and support to ensure the effective oversight of the Head Start program operations. Contractor shall ensure that, at a minimum, the following program management functions are assigned to and adopted by staff within the Program:
 - i. Program management must be assigned to an individual serving in the role of the Head Start Director. This individual must have demonstrated skills and abilities in a management capacity relevant to human services program management and must have ultimate responsibility for ensuring the children enrolled in Contractor's Head Start program are provided (by Contractor or a collaborating Head Start service provider) the full array of services to which the children are entitled under the Terms of the Grant.
 - ii. Management of education and early childhood development must be assigned to an individual serving in the role of an Education and Early Childhood Development Coordinator. This individual and any additional supporting staff or consultants must have training and experience in areas that include theories and principles of child growth and development, early childhood education, and family support. Staff and consultants must meet the qualifications for classroom teachers, as specified in the Head Start Act, Section 648A, and any subsequent amendments.
 - iii. Management of health services, including medical and dental health, must be assigned to an individual serving in the role of a Health Services Coordinator. The City recommends that this individual and any additional supporting staff or consultants have training and experience in public health, nursing, health education, maternal and child health, or health administration. Contractor must also coordinate with health service providers and follow established protocols to facilitate health care providers' ability to ensure children receive all the required Head Start health services.
 - iv. Management of child nutrition services must be assigned to an individual serving in the role of a Nutrition Services Coordinator. This individual and any supporting staff or consultants must be supported by staff or consultants who are registered dictitians or nutritionists. This person must also coordinate with City and City's nutrition services provider(s).
 - v. Coordination with mental health services provider(s) must be assigned to an individual serving in the role of an Education and Early Childhood Development Coordinator, Health Services Coordinator or Disabilities Coordinator. The City recommends that this individual have experience in serving children and families with mental health needs. This person must also coordinate with City and City's mental health services provider(s).
 - vi. Coordination with family and community support services staff, including parent activities, must be assigned to an individual serving in the role of a Family and Community Support Services Liaison. The City recommends that this individual and any supporting staff or consultants have training and experience in field(s) related to social, human, or family services, and shall coordinate with City. It is also recommended that parent involvement services be supported by staff or consultants with training, experience, and skills in assisting the parents of young children in advocating and decision-making for families.
 - vii. Management of disabilities services must be assigned to an individual serving in the role of Disabilities Coordinator. This individual and any supporting staff or consultants must have training and experience in securing and individualizing needed services for children with disabilities, and shall coordinate with City.
 - viii. Management of ERSEA responsibilities must be assigned to an individual serving in the role of ERSEA Coordinator. This individual and any supporting staff or consultants must have knowledge, training, and experience in the enrollment, recruitment, selection, eligibility and attendance provisions of the Head Start Act, the Head Start Performance Standards; and the City's program policies and other federal guidance related to the Head Start program.
 - ix. Management of monitoring responsibilities must be assigned to an individual serving as the Monitoring Coordinator. This individual and any supporting staff or consultants must have knowledge, training and experience with the Head Start Performance Standards, the Head Start Act, and monitoring of the Head Start Program. Contractor shall ensure these individuals further develop their knowledge about the various Head Start content areas.

Contractor shall ensure that members of program's management team and any other necessary staff provide uninterrupted Head Start Program management services throughout the Agreement term, including during the summer months so as to ensure adequate planning, coordination and performance of critical program activities. Critical program activities include, but are not limited to ongoing recruitment activities and services, determination of eligibility, development of a waitlist for the upcoming school year, and completion of enrollment.

- D. Contractor shall provide the following education and early childhood development services in compliance with the requirements of 45 C.F.R Part 1304.21, et. seq.:
 - i. Contractor must help children gain the skills and confidence necessary to be prepared to succeed in their present environment and with later responsibilities in school and life; the Contractor's approach to child development and education must be developmentally and linguistically appropriate, recognize the child's rate of development, language, cultural background and learning style; be inclusive of children with disabilities; provide an environment of acceptance that supports and respects gender, culture, language, ethnicity and family composition; provide a balanced daily program of child-initiated and adult-directed activities; and allow and enable children to independently use toilet facilities when it is developmentally appropriate and the efforts are supported by the parents.
 - ii. Contractor shall encourage and support parents in their efforts to become involved in the development of the program's curriculum and approach to child development and education; provide opportunities to increase parents' child observation skills and to share assessments with staff that will help plan the learning experience; and encourage parents to participate in staff-parent conferences and home visits to discuss their child's development and education.
 - iii. Contractor must support social and emotional development by encouraging development which enhances child's strengths by building trust; fostering independence; encouraging self-control and respect for feelings and rights of others; supporting and respecting the home language, culture, and family composition in ways that support the child's health and wellbeing; and planning for routines and transitions.
 - iv. Contractor must provide for the development of each child's cognitive and language skills by supporting each child's learning, using various strategies including experimentation, inquiry, observation, play and exploration; ensuring opportunities for creative self-expression through activities such as art, music, movement, and dialogue; promoting interaction and language use among children and between children and adults; and supporting emerging literacy and numeracy developments through materials and activities according to the child's developmental level.
 - v. Contractor must promote each child's physical development by providing sufficient time, indoor and outdoor space, equipment, materials and adult guidance for active play and movement that support the development of gross motor skills and of fine motor skills according to the child's developmental level; and providing an appropriate environment and adult guidance for the participation of children with special needs.
 - vi. Contractor, in collaboration with the parents, must implement a researched-based curriculum that aligns with the Head Start Early Learning Outcomes Framework (HSELOF, 2015), and state Pre-kindergarten Guidelines that supports each child's individual pattern of development and learning; provides for the development of cognitive skills; integrates all educational aspects of the health, nutrition, and mental health services into program activities; helps children develop emotional security, facility in social relationships, feelings of competence, self-esteem, and positive attitudes toward learning; enhances each child's understanding of self as an individual and as a member of a group; and provides individual and small group experiences both indoors and outdoors.
 - vii. Contractor must implement the plan of action to achieve school readiness goals that are age appropriate and will address at a minimum the domains of language, literacy, cognition, motor and physical well-being, social and emotional development and approaches to learning; and achieve the integration with the Parent, Family, and Community Engagement Framework. Contractor will participate in updating school readiness goals on an annual basis, analyze achievement, and identify areas for improvement. As part of this plan of action, the Contractor must align the program's curriculum with school readiness goals to include improving the quality of teacher-child interactions,

evidence-based teaching practices and curriculum and inclusion of parents to support school readiness.

- viii. Contractor will assess child progress on an ongoing basis and conduct data aggregation and submit analysis of child progress assessments to the City at least three times per year (at the beginning, midpoint and end of school year) to demonstrate program-level progress. Contractor will also provide the City direct access to the assessment data management system for the purposes of aggregating, analyzing, and producing program-wide reports. Contractor will make mid-year adjustments in instruction and/or professional development support should patterns or trends be identified in the analysis and coordinate with City as needed. Contractor must maintain and use the information from ongoing child assessment data to inform teachers and survey parents how best to individualize each child's learning and progress across domains.
- ix. Starting with the 2017-2018 school year, Contractor shall have in place a formal, written teacher coaching plan that is research or practice-based to address professional development of Head Start teachers. Once such a plan is created or if such a plan currently exists, the Contractor shall provide a written plan to the City by September 2017.
- E. Contractor shall perform the following services as it relates to the specific service listed below in compliance with the requirements of 45 C.F.R Parts 1304.20, 1304.22, 1304.23, 1304.24 and 1304.40:

i. Child Health and Developmental Services

- a. Contractor shall be responsible for performing the following no later than 90 calendar days from the child's entry into the program: (1) make a determination as to whether or not each child has an ongoing source of continuous, accessible health care. If the child does not, Contractor must assist the parents in accessing a source of care; and (2) obtain from a health care professional a determination as to whether the child is up-to-date on a schedule of age appropriate preventive and primary health care, which includes medical, dental and mental health. This schedule must incorporate the requirements for the schedule utilized by the Early and Periodic Screening, Diagnosis, and Treatment (EPSDT) program of the Medicaid agency of the State, and the latest immunization recommendations issued by the Centers for Disease Control and Prevention, as well as any additional recommendations from the local Health Services Advisory Committee. Contractor shall collaborate with health service providers and parents to facilitate and expedite the provision of these services.
- b. Contractor shall collaborate with the Head Start health service providers and Family Community support staff to develop and implement procedures for ongoing care to identify any new or recurring medical, dental, or developmental concerns so that it can quickly make the appropriate referrals.
- c. Contractor, in collaboration with the parents, and within 45 calendar days of the child's entry into the program, must perform or obtain linguistically and age appropriate screening procedures to identify concerns regarding the child's developmental, sensory (visual and auditory), behavioral, motor, language, social, cognitive, perceptual, and emotional skills. The annual behavioral and developmental screenings of all three (3) and four (4) year old children must be performed in accordance with Texas Health Steps' EPSDT.
- d. Contractor must use the information from the screening for developmental, sensory, and behavioral concerns, and ongoing observations, medical and dental evaluation and treatments, and insights from the child's parents to help staff and parents determine how the program can best respond to each child's individual characteristics, strengths and needs.

ii. Child Health and Safety

a. Contractor must establish and implement policies and procedures to respond to medical and dental health emergencies with which all staff are familiar and trained, to include posted policies and plans of action for emergencies that require rapid response from staff; posted locations and telephone numbers of emergency response systems and up-to-date family contact information and authorization for emergency care; posted emergency evacuation routes and other safety procedures for emergencies; methods of notifying parents in the event of an emergency; and established methods for handling cases of suspected or known child abuse and neglect.

- b. Contractor must temporarily exclude a child with a short-term injury or an acute or short-term injury or an acute or short-term contagious illness, that cannot be readily accommodated, from program participation in center-based activities or group experiences, but only for that generally short-term period when keeping the child in care poses a significant risk to the health and safety of the child or anyone in contact with the child.
- c. Contractor must not deny admission to any child, nor exclude any enrolled child from program participation for a long-term period, solely on the basis of his or her health care needs or medication requirements, unless the child poses a significant risk to the health and safety of the child or anyone in contact with the child and the risk cannot be eliminated or reduced to an acceptable level through reasonable modifications in the Contractor's policies or by providing appropriate auxiliary aids.
- Contractor must establish and maintain written procedures regarding the administration, handling, and storage of medication for every child.
- e. Contractor must ensure staff and volunteers can demonstrate safety practices; foster safety awareness among children and parents by incorporating it into child and parent activities; and ensure staff, volunteers and children follow the appropriate hygiene requirements.
- f. Contractor must have and maintain well-supplied first aid kits, appropriate for all ages served and the program size, at each facility and available on outings away from the site.

iii. Child Nutrition

- a. Contractor must work with families to identify each child's nutritional needs, taking into account staff and family discussions regarding any relevant nutrition-related assessment data; information about family eating patterns, including cultural preference, special dietary requirements, and feeding requirements of each child with disabilities; and information about major community nutritional issues.
- b. Contractor must design and implement a nutrition program that meets the nutritional needs and feeding requirements of each child, and takes into account the length of the program day.
- c. Contractor must ensure that nutritional services in center-based settings contribute to the development and socialization of enrolled children and ensure that at all meals are served family style.
- d. Contractor must use funds from USDA Food and Consumer Services Child Nutrition Programs as the primary source of payment for meal services.
- Contractor shall participate in the USDA School Lunch or Child and Adult Care Food Program.
 - Pursuant to City policies, Contractor must report amount of meals and snacks served to Head Start children on a monthly basis to City.

iv. Child Mental Health

- a. Contractor must work collaboratively with parents to solicit information, observation, and concerns about their child's mental health; share staff observations of their child and discuss with parents their child's behavior and development.
- b. Contractor must secure or refer, when appropriate, the services of mental health professionals on a schedule of sufficient frequency to enable the timely and effective identification of and intervention in family and staff concerns about a child's mental health. Contractor may work with a variety of mental health agencies subject to compliance with Section 38.010 of the Texas Education Code when applicable. Contractor may utilize other community mental health resources, as needed.

v. Family and Community Support

- a. Contractor, in collaboration with family and community support staff, must provide parent involvement and education activities that are responsive to the ongoing and expressed needs of the parents.
- b. Contractor, in collaboration with family and community support staff, must provide opportunities to include parents in the development of the program's curriculum and approach to child development and education.

- c. Contractor shall collaborate with the City, Head Start service providers and other community agencies to provide opportunities to enhance parenting skills, knowledge and understanding of the educational and developmental needs and activities of their children.
- d. Contractor must conduct two (2) home visits, unless parents expressly forbid such visits, and at least two (2) staff-parent conferences per child per program year to enhance the knowledge and understanding of the educational and developmental progress and activities of children in the program. Contractor must not require that parents permit home visits as a condition of the child's participation in the program.
- e. Contractor, in collaboration with family and community support staff, must establish and maintain procedures to support successful transitions for enrolled children and families from previous child care programs into Head Start and from Head Start to elementary school; and assist parents in becoming their children's advocates as they transition.
- F. Contractor shall encourage parent participation and attendance in center parent meetings, Parent Connection Committee meetings, Head Start Policy Council, volunteer events, and other parent activities.
- G. Contractor shall participate in developing an annual program School Readiness Plan of Action according to the timeline and requirements provided by the City. Such School Readiness Plan must be developed in compliance with Program Instruction ACF-PS-11-04 and ACF-IM-HS-11-06. Each Provider shall maintain a School Readiness Leadership Team that includes, at a minimum, teachers, site administrators, parents, and Education and Family Support staff. The School Readiness Leadership Teams meet no less than twice per year to review and discuss outcomes as applicable to the defined school readiness goals.
- H. The Contractor shall submit to City all Contractor eligibility, recruitment, selection, enrollment, attendance, performance and staffing plans, which must be in accordance with the City's policies.
- I. Contractor shall provide a substantial portion of the substantive programmatic work provided for under this Scope of Work, Attachment I, directly. However, should Contractor desire to perform this work through one or more sub awards to third parties, it shall seek and obtain City's written approval prior to doing so. In all such sub awards, Contractor shall ensure that it passes through all of the requirements that apply to Contractor hereunder to the sub recipient of such sub awarded funds.
- J. Contractor shall establish an ongoing monitoring system and conduct internal monitoring of operations throughout the program year, notifying the City when the Contractor identifies possible or actual lack of compliance with the Head Start Performance Standards, Head Start Act, City's program policies or terms of this Contract. Contractor's monitoring system shall be a Contractor-developed monitoring instrument or checklist prepared no less than quarterly, and must include monitoring of all fiscal matters relating to the Head Start Program and this Agreement. Contractor will make copies of monitoring reports available to the City upon request.

4. Program Governance

- A. Contractor's Governing Board shall be in full compliance with Head Start requirements regarding governance, management and programmatic operations applicable to recipients of Head Start grant funds, including those set forth at 45 C.F.R. §1301 et. seq. Contractor's Governing Board members or representative of the Board shall be offered the opportunity to participate in Board education activities arranged by City. Contractor shall also offer the Governing Board members or representative of the Board the opportunity to engage in a cooperative strategic planning process with respect to the Program and shall submit any final strategic plans developed through such process to City for approval.
- B. Contractor shall assure that City is kept fully apprised of the composition and actions of Contractor's Governing Board to the extent such actions affect Contractor's Head Start Program.
- C, Contractor shall seek and obtain the City's written approval before making any material revisions in Contractor's Head Start Program that conflicts with or violates (i) the City's refunding application, as

amended, to the U.S. Department of Health and Human Services (HHS), (ii) the Terms of the Grant, (iii) the terms of this Contract; or (iv) any changes to the locations of Contractor's Head Start centers.

D. Contractor shall collaborate with the City and Family and Community Support service staff to elect one or more parent members from center locations to serve on City's Head Start Policy Council in accordance with the Policy Council By-laws.

5. Financial and Program Reports

Additionally, Contractor shall maintain and furnish to City the following financial and programmatic information and reports, in such forms as the City may prescribe, and any others as may be required pursuant to the Head Start Act, as amended, and 45 C.F.R. Part 75. If a following information or report due date falls on a weekend or holiday, then it shall be due by the following business day.

- End of the Month Enrollment (EOM) Report containing the information for the preceding month, submitted upon request;
- Classroom Attendance Report, submitted by the 3rd business day of every month;
- Classroom/Site Set up Information Report (Program Design) submitted at the beginning of each program year or every time there are changes to the classroom/site set up;
- D. Disability Services Monthly Report, submitted by the 15th day of every month;
- E. Head Start Service Provider total amount of United States Department of Agriculture (USDA) reimbursement received for Head Start children and staff shall be submitted on a monthly basis;
- F. The total amount of public and private funds received by Contractor and the amount from each source;
- G. Financial reports showing all actual and/or projected costs of the Program, an explanation of budgetary expenditures, Program Income, non-Federal Share amounts;
- H. The results of the most recent financial audit involving the Head Start Program;
- Reports showing employee credentials and a list of personnel serving to satisfy Contractor's inkind non-Federal Share requirement;
- J. Reports showing the wages of each employee;
- K. Transition Activity Report, submitted by the 5th business day of every month;
- Behavioral Services Activity Report, by the 5th business day of every month;
- M. Student Assessment Data Analysis Report, as periodically required by the City;
- N. Recruitment Activity Report, by the 15th of every month;
- O. Parent Engagement Activity Report, by the 10th of every month; but should the 10th fall on a weekend or holiday, the report shall be due the preceding business day;
- P. Health Activities Report due by the due by the 10th of every month:
- Q. Training and Technical Assistance Report, by the 5th business day of every month;

- School Readiness Plan of Action update, due bi-annually by the 5th business days of December and June;
- Child Development Outcomes Progress Report based on Contractor's assessment tool, submitted as directed or when requested by City;
- T. Academic Progress Data, by June following each school year, and in a form and scope mutually agreed upon by the parties, on the academic performance of Head Start children who will remain in Contractor's Program, with the understanding that Contractor may utilize its own data assessment tool to evaluate the student progress; and
- U. Any other information reasonably requested by City.

6. Licensure/Staffing

- A. Contractor shall obtain and maintain all necessary and appropriate State licenses, permits, certifications, and approvals required for the operation of Contractor's facilities and programs including those supported by this Contract, unless exempt from such requirement. Upon commencement of the Contract, Contractor shall notify the City that it is in compliance with this provision. If at any time Contractor is out of compliance with this provision, Contractor shall notify the City within one (1) Contractor business day of receipt of written notice of violation or complaint from the state licensing, certifying or permit-issuing authority indicating lack of licensure, permitting or certification, as the case may be, and shall take all necessary steps to cure such violation. Contractor further agrees that all personnel, either employed or contracted, assigned by Contractor to perform the Head Start Program Services set forth above shall, as appropriate or required by law, be fully qualified and authorized under applicable law, to perform such Head Start Program Services.
- B. Contractor will ensure that staffing will comply with the Head Start Performance Standards, this Agreement, and other applicable law.

7. Facilities.

All Contractor facilities at the approved sites shall meet applicable federal, state and local safety standards. Contractor shall at all times during the term of this Contract, keep the facility, including play grounds, campus and classroom furniture and equipment, kitchens, restrooms and other areas within the building in good, clean, safe, and reasonable operating condition and repair, and in compliance with Head Start standards. If facilities are found to be out of compliance with Head Start Health and Safety standards, then Contractor shall take corrective action in accordance with subsection B. City shall be allowed to inspect the facilities and the licensing and/or certification documents relating to the facility during Contractor's normal operating time as long as the said inspection does not disrupt the delivery of Head Start Program Services, or at any other time as agreed by the Contractor. Upon commencement of this Agreement and upon request, Contractor shall provide City an upto-date copy of any leases or other agreements (as well as any amendments or modifications thereto) for facilities used to provide services to children enrolled in the Head Start program and funded pursuant to this Agreement.

- A. In the case of non compliance with the Head Start Health and Safety standards, the City will notify the contractor via a monitoring report with photographs when possible.
- B. When any work order request is submitted for necessary internal repairs at Head Start Centers, Contractor shall address such problem within 10 Contractor business days from the submission of the work order or from the time the need for repair is brought to the attention of Contractor whichever is earlier. For larger and/or more expensive projects, the Contractor Head Start Director shall notify City in writing of the timeline for projected completion as soon as reasonably practicable.

C. City will follow-up on all Health and Safety non-compliances immediately following the 10 days for corrective actions to be completed and submit a follow-up report stating either completed or not completed.

8. Participation.

Contractor shall make time and resources available to support: (i) participation by Contractor in meetings with City staff for community assessment, self-assessment, strategic planning, development of training and technical assistance plan, communication and program development activities; (ii) participation in technical assistance trainings and service enhancements developed by City and the Head Start training and technical assistance provider, as well as other Head Start trainings that may be developed by relevant federal or state agencies; and (iii) an appropriate level of attendance of Contractor's program management team and parent committee members at national, regional and/or State Head Start conferences/trainings.

Contractor, in collaboration with Family and Community Support Service staff, shall ensure that resources are made available through reasonable accommodations to low-income parents to participate, and attend any Parent Connection Committee meetings.

9. Transportation Services/Vehicles.

Contractor shall provide transportation services in compliance with the requirements of 45 CFR part 1310, et. seq., subject to any waiver that may be obtained by City and/or Contractor.

- A. Contractor shall successfully complete an annual inspection of all school buses and other vehicles used for the transportation of children under this Contract.
- B. Contractor shall ensure that all drivers shall be certified for the operation of said vehicles, and that all bus monitors assigned to vehicles used to provide such services receive the appropriate training. Contractor shall otherwise be in compliance with all applicable federal, State, and local requirements governing the transportation of children.
- C. Contractor shall, upon request, submit to City a complete set of such documents regarding the vehicles used for the transportation of children and the drivers operating the vehicles.

10. Submission of Center Information and Program Design

- A. Prior to execution of the Contract, Contractor shall submit to the City for its approval a Center list, which shall include the number of centers and the name and address for each center operated by Contractor. If at any time Contractor proposes a change to the Center list, Contractor shall notify and receive the City's approval prior to making said change.
- B. Prior to execution of the Contract, Contractor will submit to the City the Contractor's Program Design, which shall include, the total number of children enrolled, number of sites, number of classrooms, classroom age group designation, language designation, and hours of operation, among other things. If at any time Contractor proposes a change to the Program Design, Contractor shall request and receive the City's approval prior to making said change.

ATTACHMENT II [INSERT PROGRAM BUDGET HERE]

ATTACHMENT III [INSERT SPECIAL PROVISIONS HERE]

Contract #:		
Attachment III -	-Special	Provisions

ATTACHMENT III SPECIAL PROVISIONS — Program Year 2017-2018

I. RESTRICTIONS ON USE OF FUNDS OR PROPERTY

In addition to the other applicable restrictions on the use of Head Start funds provided under this Contract, the Contractor is prohibited from:

- using or transferring funds provided under this Contract for purposes other than authorized Head Start activities;
- using, pledging, granting a security interest in, or otherwise encumbering any right under this Contract
 or any property acquired with funds provided under this Contract as collateral or security for any loan,
 note debenture, bond or any other debt instrument;
- using any funds provided under this Contract for payment of principal or interest on any loan, note, debenture, bond or any other debt instrument, other than those approved in the 45 C.F.R. Part 74 and by the City

II. REQUIREMENTS FOR PARTICIPATION IN CHILD PLUS DATA SYSTEM

2.01 Child Plus is an electronic case management system managed and licensed by the City. This system maintains child files and an overall wait list and streamlines the process for program entry, qualification, position reservation and referrals. Child Plus enhances performance and improves the overall efficiency of data processing and automation systems in support of Head Start initiatives and is used to compile the annual Program Information Report (PIR).

2.02 Contractor shall:

- a) maintain and support Child Plus Data System
- b) provide a data entry specialist for Child Plus Data System that will be responsible for entering all required data into the system and who will be the designated contact person with regard to data entries;
- attend meetings with the City's Child Plus vendor and City staff to ensure continuity and commitment to the this system;
- d) support all design, development, testing and implementation protocols as established by the City by carrying out and complying therewith;
- e) participate in preliminary and final testing of the system using City protocols;
- f) provide the technical detail required for matching Contractor's system with the Child Plus Data System environment;
- g) allow City and its vendor to install data encryption software on the Child Care System Database network; and
- provide City and its vendor with access to Confidential Data with parental permission, as defined in Article 3.01 below, which data is critical for the Head Start project.

2.03 Both Parties agree to:

- a) use best efforts to cooperate and exchange information regarding all aspects of the Head Start project and comply with all reasonable requests of the other Party with respect to information concerning the system.
- b) Parties agree that nothing herein shall be construed as to control or in any way limit the right of parents to choose a Head Start provider.

III, CONFIDENTIAL DATA

3.01 - The Parties to this Contract shall have access to the following data ("Confidential Data"), with parental permission in the case of the child:

Contract #:		
Attachment III	- Special	Provisions

Parent's Information:

Case Number First Name Middle Initial Last Name Street Address City Zip Code

Social Security Number (Optional)

Birth Date
Gender
Race
Disability
Yearly Income
Number of memi

Telephone

Number of members in the Family

County of Residence

Employment and training status

Each child's Information:

Client Number First Name Middle Initial Last Name

Social Security Number (Optional)

Birth Date Gender Race Disability

- 3.02 Contractor understands that City intends to enter into additional agreements with other providers of child care services ("Additional Collaborators") in order to promote the success of the Head Start project. Confidential Data may be shared by City, Contractor, and any Additional Collaborator, except that all parties shall share such information in compliance with state and federal laws relating to confidentiality. All Additional Collaborators shall be required to enter into a written agreement with City containing the confidentiality requirements set forth in this Section III.
- 3.03 Each Party shall establish a method to secure the Confidential Data in accordance with the applicable federal, state, and local laws and regulations. This provision shall not be construed as limiting a Party to this Contract or an Additional Collaborator, or such Party's authorized representative's right of access to that Party's Confidential Data.
- 3.04 Neither Party shall disclose or publish Confidential Data or public school education data to any individual or organization that is not a Party to this Contract or an Additional Collaborator, unless required by law or a lawful order of a court of competent jurisdiction. Each Party shall take measures within its organization to ensure that Confidential Data or public school education data is accessible only by those persons working on the Head Start project, or directly providing other public school education / child care services, and only for the purpose of performing or assisting with services required by the Head Start project or other specific public school education / child care services.
- 3.05 Either Party may disclose Confidential Data to a third party ("Third Party") under contract or affiliated with that Party for the sole purpose of performing or assisting with services required in relation to the Head Start project or other specific child care services, and in compliance with state and federal laws relating to confidentiality. Confidential Data provided to a Third Party shall remain confidential and written confirmation by such Third Party that the Third Party will conform to the requirements of this section shall be provided to the Party prior to delivery of any information to the Third Party.

ATTACHMENT IV [INSERT HHS AWARD DOCUMENT HERE]

ATTACHMENT V [INSERT HIPAA BAA HERE]

WITNESSETH:

HIPAA BUSINESS ASSOCIATE AGREEMENT

This HIPAA Business Associate Agreement is entered into by and between the City of San Antonio ("Covered Entity"), and Edgewood Independent School District, a Business Associate ("BA").

WHEREAS, the City of San Antonio and BA have entered into an agreement to provide Head Start services ("Service Contract"), effective February 1, 2017; and

WHEREAS, Covered Entity and BA may need to use, disclose and/or make available certain information pursuant to the terms of the Service Contract, some of which may constitute Protected Health Information ("PHI"); and

WHEREAS, Covered Entity and BA intend to protect the privacy and provide for the security of PHI disclosed to each other pursuant to the Service Contract in compliance with the Health Insurance Portability and Accountability Act of 1996, Public Law 104-191 ("HIPAA") and regulations promulgated thereunder by the U.S. Department of Health and Human Services (the "HIPAA Regulations"), Health Information Technology for Economic and Clinical Health Act ("HITECH Act") and other applicable laws; and

WHEREAS, the purpose of this Agreement is to satisfy certain standards and requirements of HIPAA and the HIPAA Regulations, including, but not limited to, Title 45, Section 164.504(e) of the Code of Federal Regulations ("C.F.R."), as the same may be amended from time to time;

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, hereby agree as follows:

- A. <u>Definitions</u>. For the purposes of this Agreement, the following terms have the meanings ascribed to them:
 - (1) "Disclosure" with respect to PHI, shall mean the release, transfer, provision of access to or divulging in any other manner of PHI outside the entity holding the PHI.
 - (2) "Individual" shall have the same meaning as the term "Individual" in 45 C.F.R. 164.501 and shall include a person who qualifies as a personal representative in accordance with 45 C.F.R. 164.502(g).
 - (3) "Parties" shall mean Covered Entity and BA. "Business Associate" shall generally have the same meaning as the term "business associate" at 45 CFR 160.103. "Covered Entity" shall generally have the same meaning as the term "covered entity" at 45 CFR 160.103.
 - (4) "Privacy Rule" shall mean the Standards for Privacy of Individually Identifiable Health Information at 45 C.F.R. part 160 and Part 164, subparts A and E.
 - (5) "Security Rule" shall mean the HIPAA regulation that is codified at 45 C.F.R. Part 164.

- (6) "Protected Health Information" or "PHI" shall have the same meaning as the term "protected health information" in 45 C.F.R. 164.501, limited to the information created or received by BA from or on behalf of Covered Entity. PHI includes "Electronic Protected Health Information" or "EPHI" and shall have the meaning given to such term under the HIPAA Rule, including but not limited to 45 CFR Parts 160, 162, 164, and under HITECH.
- (6) "Required By Law" shall have the same meaning as the term "required by law" in 45 CFR § 164.501.
- (7) "Secretary" shall mean the Secretary of the Department of Health and Human Services or his designee.
- (8) "PHI Breach" shall mean an acquisition, access, use, or disclosure of PHI in a manner not permitted by the Privacy Rules and such action compromises the security or privacy of the PHI.
- (9) The Health Information Technology for Economic and Clinical Health ("HITECH") Act shall mean Division A, Title XII of the American Recovery and Reinvestment Act of 2009 (Pub. L. 111-5).
- B. BA Obligations and Activities. BA agrees that it shall:
 - (1) Not use or disclose the PHI other than as permitted or required by this Agreement or as Required by Law;
 - (2) Establish and maintain appropriate administrative, physical, and technical safeguards that reasonably and appropriately protect, consistent with the services provided under this Agreement, the confidentiality, integrity, and availability of the electronic protected health information that it creates, receives, maintains, or transmits on behalf of covered entity;
 - (3) Mitigate, to the extent practicable, any harmful effect that is known to BA of a use or disclosure of PHI by BA in violation of the requirements of this Agreement;
 - (4) Report to Covered Entity any use or disclosure of PHI of which BA is aware or becomes aware that is not provided for or allowed by this Agreement as well as any security incident that BA becomes of aware of;
 - (5) Ensure that a business associate agreement is in place with any of its agents or subcontractors with which BA does business and to whom it provides PHI received from, created or received by BA on behalf of Covered Entity are aware of and agree to the same restrictions and conditions that apply through this Agreement to BA with respect to such information, and further agree to implement reasonable and appropriate administrative, physical and technical safeguards that render such PHI unusable, unreadable and indecipherable to individuals unauthorized to acquire or otherwise have access to such PHI;
 - (6) Provide access, at the request of Covered Entity, and in a reasonable time and manner as agreed by the Parties, to PHI in a Designated Record Set to Covered Entity or, as

directed by Covered Entity, to an Individual in order to meet the requirements 45 C.F.R. §164.524;

- (7) Make any amendment(s) to PHI in a Designated Record Set that the Covered Entity directs or agrees to pursuant to 45 C.F.R. 164.526 at the request of the Covered Entity or an Individual, and in a reasonable time and manner agreed to by the Parties;
- (8) Make available to the Covered Entity or to the Secretary all internal practices, books and records, including policies and procedures, relating to the use and disclosure of PHI received from, or created or received by the BA on behalf of the Covered Entity, for purposes of the Secretary in determining Covered Entity's compliance with the Privacy Rule;
- (9) Document such disclosures of PHI, and information related to such disclosures, as would be required for Covered Entity to respond to a request from an Individual for an accounting of disclosures of PHI in accordance with 45 C.F.R. 164.528;
- (10) Provide Covered Entity or an Individual, in a reasonable time and manner as agreed to by the Parties, information collected in accordance with Section B(9) of this Agreement, to permit Covered Entity to respond to a request by an Individual for an accounting of disclosures of PHI in accordance with 45 C.F.R. 164.528;
- (11) Will immediately, and in no event later than three days from discovery, notify Covered Entity of any breach of PHI, including ePHI, and will coordinate with Covered Entity to identify, record, investigate, and report to an affected individual and US Department of Health and Human Services, as required, any covered PHI breach. Breach notification to Covered Entity must include: names of individuals with contact information for those who were or may have been impacted by the HIPAA Breach; a brief description of the circumstances of the HIPAA Breach, including the dated of the breach and date of discover; a description of the types of unsecured PHI involved in the breach; a brief description of what the BA has done or is doing to investigate the breach and mitigate harm. BA will appoint a breach liaison and provide contact information to provide information and answer questions Covered Entity may have concerning the breach:
- (12) Comply with all HIPAA Security Rule requirements;
- (11) Comply with the provisions of HIPAA Privacy Rule for any obligation Covered Entity delegates to BA;
- (12) Under no circumstances may BA sell PHI in such a way as to violate Texas Health and Safety Code, Chapter 181.153, effective September 1, 2012, nor shall BA use PHI for marketing purposes in such a manner as to violate Texas Health and Safety Code Section 181.152, or attempt to re-identify any information in violation of Texas Health and Safety Code Section 181.151, regardless of whether such action is on behalf of or permitted by the Covered Entity.

C. Permitted Uses and Disclosures by BA

(1) Except as otherwise limited in this Agreement, BA may use or disclose PHI to perform functions, activities, or services for, or on behalf of, Covered Entity as specified in the

Service Contract, provided that such use or disclosure would not violate the Privacy Rule if done by Covered Entity.

- (2) Except as otherwise limited in this Agreement, BA may disclose PHI for the proper management and administration of the BA, provided that disclosures are Required By Law, or BA obtains reasonable assurances from the person to whom the information is disclosed that it will remain confidential and used or further disclosed only as Required By Law or for the purpose for which it was disclosed to the person, and the person notifies the BA of any instances of which it is aware in which the confidentiality of the information has been breached.
- (3) Except as otherwise limited in this Agreement, BA may use PHI to provide Data Aggregation Services to Covered Entity as permitted by 45 C.F.R. 164.504(e)(2)(i)(B).
- (4) BA may use PHI to report violations of law to appropriate Federal and State authorities, consistent with 45 C.F,R. 502(j)(1).
- D. <u>Obligations of Covered Entity</u>. Covered Entity shall inform BA of its privacy practices and restrictions as follows. Covered Entity shall:
 - (1) notify BA of any limitations in its notice of privacy practices in accordance with 45 C.F.R. 164.520, to the extent that such limitation may affect BA's use or disclosure of PHI;
 - (2) notify BA of any changes in, or revocation of, permission by any Individual to use or disclose PHI, to the extent that such changes may affect BA's use or disclosure of PHI;
 - (3) notify BA of any restriction to the use or disclosure of PHI that Covered Entity has agreed to in accordance with 45 C.F.R. 164.522 to the extent that such changes may affect BA's use or disclosure of PHI.
 - (4) coordinate with BA regarding any PHI breach and make timely notification to affected individuals within 60 days of discovery.

Permissible Requests by Covered Entity.

Covered Entity shall not request BA to use or disclose PHI in any manner that would not be permissible under the Privacy Rule if done by Covered Entity, except that the BA may use or disclose PHI for data aggregation or management and administrative activities of the BA.

F. Term and Termination,

(1) The term of this Agreement shall commence on February 1, 2017. This Agreement shall terminate when all PHI encompassed by this Agreement is destroyed or returned to Covered Entity or, if it is infeasible to return or destroy the PHI, protections are extended to such information in accordance with the termination provisions in this Section.

(2) Termination for Cause. Upon Covered Entity's knowledge of a material breach by BA, Covered Entity shall either (a) provide an opportunity for BA to cure the breach in accordance with the terms of the Service Contract or, if the BA does not cure the breach or end the violation within the time for cure specified in the Service Contract, end the violation and terminate this Agreement and the Contract; or (b) immediately terminate this Agreement and the Service Contract if BA has breached a material term of this Agreement and cure is not possible. If neither termination nor cure is feasible, Covered Entity shall report the violation to the Secretary of the U.S. Department of Health and Human Services.

(3) Effect of Termination.

- (a) Except as provided below in paragraph (b) of this Section F(3), upon termination of this Agreement for any reason, BA shall return or destroy all PHI received from the Covered Entity, or created or received by BA on behalf of Covered Entity. This provision shall apply to PHI that is in the possession of BA or its subcontractors or agents. BA shall not retain any copies of PHI, except as required by law.
- (b) In the event that BA determines that returning or destroying PHI is infeasible, BA shall provide to Covered Entity written notification of the condition that makes the return or destruction of PHI infeasible. Upon BA's conveyance of such written notification, BA shall extend the protections of this Agreement to such PHI and limit further uses and disclosures of such PHI to those purposes that make its return or destruction infeasible, for so long as BA maintains such PHI.
- G. Amendment to Comply with Law. The Parties agree to take written action as is necessary to amend this Agreement to comply with any Privacy Rules and HIPAA legal requirements for Covered Entity without the need for additional council action.
- H. <u>Survival</u>. The respective rights and obligations of the BA under Sections B, C (2) and (4), and F(3) shall survive the termination of this Agreement.
- Interpretation. Any ambiguity in this Agreement shall be interpreted to permit Covered Entity to comply with the Privacy Rule.
- J. <u>Regulatory References</u>. A reference in this Agreement to a section in the Privacy Rule means the section as in effect or amended.
- K. No Third Party Beneficiaries. Nothing express or implied in this Agreement is intended to confer, nor shall anything herein confer upon any person other than Covered Entity, BA, and their respective successors or assigns, any rights, remedies, obligations, or liabilities whatsoever.
- L. <u>LIMITED LIABILITY</u>, CONTRACTOR AND THE CITY ACKNOWLEDGE THEY ARE POLITICAL SUBDIVISIONS OF THE STATE OF TEXAS AND ARE SUBJECT TO, AND COMPLY WITH THE APPLICABLE PROVISIONS OF THE TEXAS TORT CLAIMS ACT, AS SET OUT IN THE CIVIL PRACTICE AND REMEDIES CODE, SECTION 101.001, ET. SEO., AND THE REMEDIES AUTHORIZED THEREIN

REGARDING CLAIMS OR CAUSES OF ACTION THAT MAY BE ASSERTED BY THIRD PARTIES FOR ACCIDENT, INJURY OR DEATH.

- M. Reimbursement. BA will reimburse Covered Entity for reasonable costs incurred responding to a PHI breach by BA or any of BA's subcontractors.
- N. <u>Waiver</u>. No provision of this Agreement or any breach thereof shall be deemed waived unless such waiver is in writing and signed by the party claimed to have waived such provision or breach. No waiver of a breach shall constitute a waiver of or excuse any different or subsequent breach.
- O. <u>Assignment</u>. Neither party may assign (whether by operation or law or otherwise) any of its rights or delegate or subcontract any of its obligations under this Agreement without the prior written consent of the other party. Notwithstanding the foregoing, Covered Entity shall have the right to assign its rights and obligations hereunder to any entity that is an affiliate or successor of Covered Entity, without the prior approval of BA.
- P. Entire Agreement. This Agreement constitutes the complete agreement between BA and Covered Entity relating to the matters specified in this Agreement, and supersedes all prior representations or agreements, whether oral or written, with respect to such matters. In the event of any conflict between the terms of this Agreement and the terms of the Service Contracts or any such later agreement(s), the terms of this Agreement shall control unless the terms of such Service Contract comply with the federal law and regulations commonly referred to as the Privacy Standards and the Security Standards. No oral modification or waiver of any of the provisions of this Agreement shall be binding on either party. This Agreement is for the benefit of, and shall be binding upon the Parties, their affiliates and respective successors and assigns. No third party shall be considered a third-party beneficiary under this Agreement, nor shall any third party have any rights as a result of this Agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

Q. Governing Law. This Agreement shall be governed by and interpreted in accordance with the laws of the State of Texas.

EXECUTED to be effective on February 1, 2017.

COVERED ENTITY City of San Antonio BUSINESS ASSOCIATE: Edgewood Independent School District

Melody Woosley, Director Department of Human Services Dr. Sylvester Perez, Interim Superintendent

APPROVED AS TO FORM:

Assistant City Attorney

ATTACHMENT VI [INSERT ADDITIONAL OMB PROVISIONS HERE]

Additional OMB Provisions

from Appendix II to Part 200—Contract Provisions for Non-Federal Entity Contracts Under Federal Awards

In addition to other provisions required by HHS or the City, all contracts made by the City under the Federal award must contain provisions covering the following, as applicable (2 C.F.R. 200, Appendix II).

Hereinafter in this Attachment VI, Center shall be referred to as "contractor."

Provision	Page Number		
Equal Employment Opportunity	2		
Davis Bacon Act	4		
Contract Work Hours and Safety Standards Act	9		

EQUAL EMPLOYMENT OPPORTUNITY provisions (60 C.F.R. 1.4(b).

During the performance of this contract, contractor agrees:

- (1) The contractor will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. The contractor will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.
- (2) The contractor will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive considerations for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.
- (3) The contractor will send to each labor union or representative of workers with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of the contractor's commitments under this section, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- (4) The contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- (5) The contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to CONTRACTOR's books, records, and accounts by the U.S. Department of Health and Human Services ("HHS") and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- (6) In the event of the contractor's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations, or orders, this contract may be canceled, terminated, or suspended in whole or in part and the contractor may be declared ineligible for further Government contracts or federally-assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.

(7) The contractor will include the portion of the sentence immediately preceding paragraph (1) and the provisions of paragraphs (1) through (7) in every subcontract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The contractor will take such action with respect to any subcontract or purchase order as HHS may direct as a means of enforcing such provisions, including sanctions for noncompliance; provided, however, that in the event a contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the administering agency the contractor may request HHS to enter into such litigation to protect the interests of the United States.

DAVIS BACON ACT provisions (29 C.F.R. § 5.5(a))

For any contract or subcontract in excess of \$2,000 which is entered into for the actual construction, alteration and/or repair, including painting and decorating, of a public building or public work, or building or work financed in whole or in part from federal funds, and which is subject to the labor standards provisions of any of the acts listed in 29 C.F.R. §5.1, the following § 5.5(a) must be included and complied with:

- (1) Minimum wages.
- (i) All laborers and mechanics employed or working upon the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor, which is incorporated herein by reference, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of paragraph (a)(1)(iv) of this section; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in §5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein: Provided, that the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under paragraph (a)(1)(ii) of this section) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

- (ii)
- (A) The contracting officer shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The contracting officer shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:
 - (1) The work to be performed by the classification requested is not performed by a classification in the wage determination; and
 - (2) The classification is utilized in the area by the construction industry; and
 - (3) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.
- (B) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the contracting officer agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the contracting officer to the Administrator of the Wage and Hour Division, Employment Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an

authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

- (C) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the contracting officer do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the contracting officer shall refer the questions, including the views of all interested parties and the recommendation of the contracting officer, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.
- (D) The wage rate (including fringe benefits where appropriate) determined pursuant to paragraphs (a)(1)(ii) (B) or (C) of this section, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.
- (iii) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.
- (iv) If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, Provided, That the Secretary of Labor has found, upon the written request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.
- (2) Withholding. HHS or the City shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld from the contractor under this contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), all or part of the wages required by the contract, HHS may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.
- (3) Payrolls and basic records.
- (i) Payrolls and basic records relating thereto shall be maintained by the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work (or under the United States Housing Act of 1937, or under the Housing Act of 1949, in the construction or development of the project). Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected,

and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

(ii)

- (A) The contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to HHS if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit the payrolls to the applicant, sponsor, or owner, as the case may be, for transmission to HHS. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on weekly transmittals. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from the Wage and Hour Division Web site at http://www.dol.goy/esa/whd/forms/wh347instr.htm or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to HHS if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit them to the applicant, sponsor, or owner, as the case may be, for transmission to HHS, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of this section for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the sponsoring government agency (or the applicant, sponsor, or owner).
- (B) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:
 - (1) That the payroll for the payroll period contains the information required to be provided under §5.5 (a)(3)(ii) of Regulations, 29 CFR part 5, the appropriate information is being maintained under §5.5 (a)(3)(i) of Regulations, 29 CFR part 5, and that such information is correct and complete;
 - (2) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in Regulations, 29 CFR part 3;
 - (3) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.
- (C) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH-347 shall satisfy the requirement for submission of the "Statement of Compliance" required by paragraph (a)(3)(ii)(B) of this section.
- (D) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under section 1001 of title 18 and section 231 of title 31 of the United States Code.
- (iii) The contractor or subcontractor shall make the records required under paragraph (a)(3)(i) of this section available for inspection, copying, or transcription by authorized representatives of HHS or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the Federal agency may, after written notice to the contractor, sponsor, applicant, or owner,

take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

(4) Apprentices and trainees-

- (i) Apprentices. Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (ii) Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration, Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program,

- the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (iii) Equal employment opportunity. The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR part 30.
- (5) Compliance with Copeland Act requirements. The contractor shall comply with the requirements of 29 CFR part 3, which are incorporated by reference in this contract.
- (6) Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses contained in 29 CFR 5.5(a)(1) through (10) and such other clauses as the (write in the name of the Federal agency) may by appropriate instructions require, and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in 29 CFR 5.5.
- (7) Contract termination: debarment. A breach of the contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in 29 CFR 5.12.
- (8) Compliance with Davis-Bacon and Related Act requirements. All rulings and interpretations of the Davis-Bacon and Related Acts contained in 29 CFR parts 1, 3, and 5 are herein incorporated by reference in this contract.
- (9) Disputes concerning labor standards. Disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and the contracting agency, the U.S. Department of Labor, or the employees or their representatives.
- (10) Certification of eligibility.
- (i) By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (ii) No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (iii) The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

CONTRACT WORK HOURS AND SAFETY STANDARDS ACT provisions (29 C.F.R. § 5.5(b))

(1) Overtime requirements.

No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek.

- (2) Violation; liability for unpaid wages; liquidated damages.
- In the event of any violation of the clause set forth in paragraph (b)(1) of this section the contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in paragraph (b)(1) of this section, in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in paragraph (b)(1) of this section.
- (3) Withholding for unpaid wages and liquidated damages.

The U.S. Department of Health and Human Services or the City shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in paragraph (b)(2) of this section.

(4) Subcontracts.

The contractor or subcontractor shall insert in any subcontracts the clauses set forth in paragraph (b)(1) through (4) of this section and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in paragraphs (b)(1) through (4) of this section.

EDGEWOOD INDEPENDENT SCHOOL DISTRICT

Catagoni	1	Description	BUDGET		
Category Personnel/Salaries	Agency GL 11-6119.00	Teachers (25 program / 21 ln Kind)	1,366,004		
Personnel/Salaries		Teachers's Aides (50)	851,53		
Personnel/Salaries		Head Start Director			
Personnel/Salaries		Program Managers			
-		Family Development (Services) Coordinator/ERSEA	126,293 56,391		
Personnel/Salaries			57,109		
Personnel/Salaries Personnel/Salaries		Administrative Assistants (4)	79,804		
		Data Entry Clerk Bus Drivers (3)	28,097		
Personnel/Salaries		Bus Monitors (3)	20,181		
Personnel/Salaries		Custodians (7)	112,369		
Personnel/Salaries		The state of the s	70,000		
Personnel/Salaries		The state of the s			
Personnel/Salaries		Cafeteria Monitors	31,798		
Personnel/Salaries		Health Assistant/Dietician (LVN)	25,959		
Personnel/Salaries		Program Compliance Monitor	57,752		
Personnel/Salaries		Accountant	30,901		
Personnel/Salaries	23-6129.01	Bookkeeper	20,476		
Personnel/Salaries	23-6129.03	ERSEA Specialist	28,506		
Personnel/Salaries		AIT Behavior Specialist	56,999		
Personnel/Salaries	31-6119.00	Counselor	57,190		
Personnel/Salaries	61-6129	Parent Laison	19,201		
Fringe Benefits	6141	FICA	42,642		
Fringe Benefits	6142	Health Insurance	279,824		
Fringe Benefits	6146	Retirement	302,241 114,899		
Fringe Benefits	6143	Worker's Compensation			
Travel	23-6411	Official Travel (Out of Town Only)	5,800		
Supplies	23-6399	Office Supplies	8,000		
Supplies	51-6319	Janitorial Supplies	5,000		
Supplies	33-6399	Medical & Dental Supplies	8,000		
Supplies	6005	Child & Education Supplies (Classroom)	50,000		
Supplies	6395	**Cap <5000 - Computer Equipment	5,000		
Supplies	6395.01	**Cap <5000 - Machinery/Equipment	2,000		
Supplies	6395.02	**Cap <5000 - Furniture & Fixture	3,000		
Contractual		Fees to Professional (CPR Training)	5,000		
Other Other		Education - Classes Staff Development Adv and Publications	5,000		
Other			1,000		
Other		Membership Dues Binding & Printing	1,000		
Other		Transportation Fees (Staff local mileage)	1,500		
Other			500		
Other	51-6249	Transportation Fees (Field Trips) Mail & Parcel Post	1,000		
Other	61-6499.05	Food (child/teacher meals not reimbursed by USDA)	115,000		
Other	51-6257-6258		40,000		
		Water and Sewer			
Other	51-6255 TOTAL FEDERAL		12,500		
	TOTAL FEDERAL	Non Federal	4,171,410		
	TOTAL FEDERAL	& NON FEDERAL	1,042,853		
	TOTAL PEDEKAL	& NON FEDERAL	5,214,263		

Attachment III

Contract	No.		
Conduct		 	 _

STATE OF TEXAS

HEAD START AGREEMENT FOR EDUCATION SERVICES

COUNTY OF BEXAR *

BETWEEN THE CITY OF SAN ANTONIO

CITY OF SAN ANTONIO * SAN ANTONIO INDEPENDENT SCHOOL DISTRICT (SAISD)

WITNESSETH:

WHEREAS, the City has received a grant pursuant to the Head Start Act (42 U.S.C. §9831 et seq., as amended) (the "Grant") for the purpose of providing Head Start services to children and families in the San Antonio and Edgewood Independent School Districts; and

WHEREAS, the City is authorized by the U.S. Department of Health and Human Services ("HHS"), Administration for Children and Families ("ACF"), and desires, through its Department of Human Services ("DHS")) to execute an agreement with Contractor to provide Head Start services to children residing in the San Antonio Independent School District (hereinafter referred to as the "Project" or "Program"); and

WHEREAS, Contractor desires, and is appropriately licensed and qualified, to enter into this Agreement with City and agrees to deliver the services described herein in accordance with applicable Head Start Performance Standards and other requirements more fully set forth below;

NOW THEREFORE, in consideration of the mutual promises and covenants herein contained and intending to be legally bound hereby, City and Contractor agree as follows:

I. SCOPE OF WORK

- 1.1 The Contractor will provide, oversee, administer, and carry out all activities and services in a manner satisfactory to the City and in compliance with the Contractor's Scope of Work, attached hereto and incorporated herein for all purposes as Attachment I, this Agreement, and the Terms of the Grant (hereinafter defined). If the terms of this Agreement are inconsistent or in conflict with applicable Terms of the Grant shall be controlling, unless the inconsistency or conflict results from more stringent requirements set forth in this Agreement, in which case the terms imposing the most stringent requirements upon the Contractor shall control.
- 1.2 For purposes of this Agreement, the terms listed below shall have the following meanings:
 - (A) "Terms of the Grant" shall mean all requirements of the Grant, whether contained in the Head Start Act, as amended by the Improving Head Start for School Readiness Act of 2007 (42 U.S.C. §9831, et seq.), or other applicable statutes, implementing regulations (e.g., 45 C.F.R. §1301 et seq. (the "Head Start Performance Standards" or "Performance Standards") and 45 C.F.R. Part 75, as amended), rules, Executive Orders, the award document from HHS to the City, Relevant HHS Directives, or elsewhere, including, but not limited to circulars, Program Instructions, Information Memorandums and Policy Clarifications, the City's policies and procedures and the program design manual applicable to the

- Head Start Program, as such requirements exist as of the date of this Agreement and as such requirements may be established or modified (by amendment, deletion, addition or otherwise) during the period of the Agreement.
- (B) "Relevant HHS directives" shall mean regulations, manuals, guidelines, or other oral or written directives of HHS or any subdivision thereof, including the Administration for Children and Families, Head Start Bureau, the Program Operations Division and ACF Region VI, as such regulations, manuals, guidelines, or other oral or written directives shall be made applicable.
- 1.3 Contractor shall establish and implement policies and procedures governing personnel, financial management, and programmatic management, as specified more fully in 45 C.F.R Parts 1301 et. seq. and/or 45 C.F.R. Part 75. Such policies and procedures shall be consistent with the Terms of the Grant, the policies and procedures approved by the City's Head Start Policy Council and Governing Body, and content and service plans.
- City retains the authority to contract with third-parties for the delivery of other Head Start services in the San Antonio and Bexar County area. Contractor agrees to allow the City's other contractors access to the facilities leased and/or owned by Contractor in order to provide said services, so long as access would not cause disruption of Contractor's educational activities or purpose as an educational entity. Contractor agrees to cooperate with City and third-party Head Start contractors to establish, modify and comply with a set of policies and procedures and/or a program design manual governing the City's Head Start Program and the protocol for collaboration between Head Start service providers. Contractor agrees that, notwithstanding the fact that another Head Start service provider under the City of San Antonio Head Start program may be contracted to provide a category of service, Contractor, under the leadership of its Head Start Program Director, will be responsible for coordinating with Head Start service providers and working with the City to ensure provision of full array of services to which the children are entitled under the Terms of the Grant.

II. TERM

2.1 Except as otherwise provided, this Agreement shall begin on February 1, 2017 and shall terminate on January 31, 2018. The City, subject to its continued award of Head Start grant funds from HHS, shall have the option to renew this Agreement in an amount acceptable to Contractor's superintendent of schools for 1 additional 5-month term without the necessity of further City Council approval. In the event of a 5-month renewal under this option, appropriate revisions will be made to the budget, associated due dates, and Scope of Work to reflect the grant obligations and best meet the needs of children and families through Head Start Grant activities.

III. CONSIDERATION

- In consideration of the services to be delivered by Contractor, the City will reimburse Contractor a total amount not to exceed \$12,053,245.00 ("the Federal Share") during the period in which this Agreement is in effect for costs incurred in accordance with the Program Budget attached hereto and incorporated herein for all purposes as Attachment II. Contractor's Program Budget is comprised of the Federal Share and the Non-Federal Share. The Federal Share shall be no more than 80% of the total Program Budget. Should Contractor fail to raise all of the non-Federal Share funds (20% of the total Program Budget, or \$3,013,311.00) it is required to raise for the operation of its Program, City reserves the right to limit its reimbursements to Contractor proportionately. For instance, if Contractor succeeds in raising only eighty percent (80%) of its required non-Federal Share funds, City may limit its reimbursements to Contractor to eighty percent (80%) of City's total obligation to Contractor. Contractor may provide additional non-Federal share funds if Contractor, in its discretion, determines such funds are available. To meet the requirements of this Agreement, all claimed non-Federal Share must meet the requirements of 45 C.F.R. § 75.306, as applicable.
- 3.2 Prior to commencement of the Agreement, Contractor shall submit to City for its approval Contractor's proposed monthly budget by line item for the entire term of the Agreement along with its program Budget.

including detail by category alone. Until the City receives the initial proposed monthly budget for the entire term of the Agreement and prior to the City's approval, the City reserves the right to redirect the funding City has proposed be allocated to Contractor under this Agreement. City shall notify the Contractor of the amount redirected and the revised Agreement funding. Additionally, beginning February 1, 2017 and throughout the term of the Agreement, Contractor shall submit quarterly forecasts of the projected expenses for each month remaining in the Agreement so that the City may review and compare actual expenses to projected expenditures and address issues associated with Contractor's expenditure rate (e.g., for Q1, Contractor shall submit the projected expenses by month for February, April, May, and so on until January 31, 2018; for Q2, Contractor shall submit the projected expenses by month for May, June and so on until January 31, 2018). Contractor's budgeted development and administrative costs (as defined by 45 C.F.R. §1301.32) shall not exceed twelve percent (12%) of the Program Budget, unless the total Program Budget is modified in accordance with this Agreement in which case this amount shall be reduced proportionately unless the Parties otherwise agree.

- 3.3 Approval required. Contractor shall seek and obtain City's (City's Head Start Program Administrator and DHS's Fiscal Administrator) prior written approval 30 calendar days before making budget modifications. City may make exceptions to the 30-day notice requirement on a case by case basis, but otherwise Contractor must make request in writing or via email to the City's Head Start Program Administrator. Contractor's written request must be accompanied by a justification for the change and indicate which lines items are affected by such change
- 3.4 Contractor understands and agrees that should Contractor fail to meet or maintain its funded enrollment level as set forth in the Scope of Work which failure continues for a thirty (30) day period, City may reduce Contractor's funding by an amount equal to the difference between funded and actual enrollment at the per capita rate.
- 3.5 The funding level of this Agreement is based on an allocation from the following funding sources:
 - U.S. Department of Health and Human Services (HHS) Head Start Funds Catalog of Federal Domestic Assistance # 93.600
 - Consequently, Contractor agrees to comply with the Terms of the Grant and the Special Provisions, attached hereto and incorporated herein for all purposes as Attachment III.
- 3.6 It is expressly understood and agreed by the City and Contractor that the City's obligations under this Agreement are contingent upon the actual receipt of adequate grant funds from HHS to meet City's liabilities hereunder. This Agreement may be terminated by the City if HHS terminates the City as a grantee or reduces the amount granted to City, for any reason; provided that, if the reduction of grant funds does not result in complete unavailability of such funds, the Parties will use best efforts to amend this Agreement accordingly. City will promptly notify Contractor of any such HHS action.

IV. PAYMENT

4.1 Contractor agrees that this is a cost reimbursement contract and that the City's liability hereunder is limited to making reimbursements for allowable costs incurred as a direct result of services provided by the Contractor in accordance with the terms of this Agreement. Allowable costs are defined as those costs, which are necessary, reasonable and allowable under applicable federal, state, and local law, including but not limited to those laws referenced in Article XII hereof, for the proper administration and performance of the services to be provided under an agreement. All requested reimbursed costs must be consistent with the terms and provisions of the approved budgeted line items described in Attachment II of this Agreement, unless (a) a subsequent budget revision has been approved in accordance with the procedure set forth in Section 3.3 and signed by the Director of DHS in cases where the total Agreement Budget remains the same, or (b) an amendment has been approved and signed by the Director of DHS pursuant to Section 24.1 of this Agreement in cases where there is an increase or decrease to the total Agreement Budget. Approved budget revisions and amendments modify the Budget attached hereto, and in such cases Contractor's requested reimbursed costs must be consistent with the last revised, approved budget. Approved budget

revisions and amendments supersede prior conflicting or inconsistent agreements with regard to the referenced Project Budget, and all references in the Agreement to the budget shall mean the budget as revised through approved budget revisions or amendments. Budget revision requests shall be submitted in advance of anticipated expense(s). The City will not accept budget revision requests submitted after December 31, 2017, unless Contractor requests a reasonable extension before that date, which shall not be unreasonably denied by City. In no event shall the City be liable for any cost of Contractor not eligible for reimbursement as defined within the Agreement. Contractor shall remit to City within ten (10) Contractor business days after the City makes the request for remittance any funded amounts which were paid pursuant to this Article IV and used to cover disallowed costs. Any such amounts not remitted within ten (10) Contractor business days may, at City's option, be subject to offset against future funding obligations by City. For purposes of this Agreement, the term, "Contractor business day" shall mean every day of the week except all Saturdays, Sundays and those scheduled holidays officially adopted and approved by Contractor for Contractor employees.

- 4.2 If specific circumstances require an advance payment on this Agreement, Contractor must submit to the Director of DHS a written request for such advance payment, including the specific reason for such request in the form prescribed by City. Contractor agrees that the City shall not be obligated to pay for any advances requested. In those instances in which advance payments are authorized, the Director of DHS may, in the Director's sole discretion, approve an advance payment on this Agreement. It is understood and agreed by the parties hereto that (a) each request requires submission to the Director of DHS no less than ten (10) business days prior to the actual ostensible cash need; (b) each request will be considered by the Director of DHS on a case-by-case basis, and (c) the decision by the Director of DHS whether or not to approve an advance payment is final. For purposes of this Agreement, the term "business day" shall mean every day of the week except all Saturdays, Sundays and those scheduled holidays officially adopted and approved by the San Antonio City Council for City of San Antonio employees, except in those instances where this Agreement expressly calls for "Contractor business days." In those instances in which advance payments are authorized:
 - (A) Contractor's payments to its vendors using funds advanced by the City shall be remitted to the vendors in a prompt and timely manner so long as services have been performed by the subject vendor, defined as not later than (10) business days after the Contractor is notified that an advance payment check is available from the City.
 - (B) The Contractor must deposit Agreement funds in an account in a bank insured with the Federal Deposit Insurance Corporation (FDIC). In those situations where Contractor's total deposits in said bank, including all Agreement funds deposited with said bank, exceed the FDIC insurance limit, the Contractor must arrange with said bank to automatically have the excess collaterally secured. A written copy of the collateral agreement must be obtained by Contractor from the Contractor's banking institution, maintained on file and be available for City monitoring reviews and audits. Advanced funds that cause the Contractor's account balance to exceed the FDIC limit shall be deposited in a manner consistent with the Public Funds Investment Act (Chapter 2256 of the Texas Government Code) as amended. Contractor shall maintain the FDIC insured bank account in which Agreement funds are deposited and its recordkeeping in a manner that will allow City to track expenditures made with Agreement funds.
 - (C) The City may deduct from monthly reimbursements amounts necessary to offset the amount advanced based upon the number of months remaining in the Agreement term, or from a single subsequent monthly reimbursement the full amount previously advanced to Contractor. The City may consider factors such as projected allowable costs and other indicators such as Contractor's financial stability. Contractor shall maintain a financial management system to account for periodic, or a lump sum, deduction from reimbursements.
- 4.3 Contractor shall submit to City not later than the forty-fifth (45th) day after the end of every month a monthly Request for Payment in the form prescribed by City, which details the specific costs (by category and by program account number) Contractor expensed in the previous month for the services delivered as described in Article I herein, including supporting documentation of such costs as may be required by the

Director of DHS. The Request for Payment shall also specify the Program Income (as defined herein) received or projected during the same time period. The Director of DHS may require the Contractor's submission of original or certified copies of invoices, cancelled checks, Contractor's general ledger and/or receipts to verify invoiced expenses.

- 4.4 City shall make reimbursement payments of eligible expenses to the Contractor of any undisputed amounts as determined by the Director of DHS in accordance with established procedures, so long as City receives a properly completed and documented Request for Payment. City shall make payment to Contractor as soon as reasonably practicable, but not later than 30 calendar days of receiving a valid and approved Request for Payment.
- 4.5 The Contractor shall submit to City a full accounting of the Program Income and non-Federal Share funds received and total Program costs incurred, along with all requests for payment for the period February 1, 2017 through January 31, 2018, no later than March 15, 2018. In addition, all purchase orders that have been encumbered by January 31, 2018 must be received and paid no later than March 15, 2018. In the event of early termination of this Agreement, Contractor shall submit the information 45 calendar days from the early termination date of the Agreement. These deadlines may be adjusted only if Contractor receives written authorization from the Director of DHS allowing Contractor to submit a request for payment at a later specified date.
- 4.6 Contractor agrees that the City shall not be obligated to any subcontractors or third party beneficiaries of the Contractor.
- 4.7 Contractor shall maintain a financial management system, and acceptable accounting records in accordance with 45 C.F.R. § 75.302 et seq., and that provide for:
 - (A) accurate, current, and complete disclosure of financial support from each federal, state and locally sponsored project and program in accordance with the reporting requirements set forth in Article VIII of this Agreement. If accrual basis reports are required, the Contractor shall develop accrual data for its reports based on an analysis of the documentation available;
 - (B) records that adequately identify the source and application of funds for City-sponsored activities. Such records shall contain information pertaining to City awards, authorizations, obligations, unobligated balances, assets, equity, outlays, and income;
 - (C) effective control over and accountability for all funds, property, and other assets. The Contractor shall adequately safeguard all such assets and shall ensure that they are used solely for authorized purposes. Contractor shall maintain a separate numbered account for all funds received and disbursed through this Agreement;
 - (D) comparison of actual outlays with budget amounts for each award. Whenever appropriate or required by the City, financial information should be related to performance and unit cost data;
 - (E) procedures to minimize the time elapsing between the transfer of funds from the City and the disbursement of said funds by the Contractor;
 - (F) procedures for determining reasonable, allowable, and allocable costs in accordance with the provisions of any and all applicable cost principles, including but not limited to the cost principles referenced in Article XII hereof, and the terms of the award, grant, or contract, with the City;
 - (G) accounting records that are supported by source documentation (i.e., timesheets, employee benefits, professional services agreements, purchases, and other documentation as required by City). Contractor shall maintain records and shall meet necessary requirements under Generally Accepted Accounting Principles [GAAP]; and

- (H) an accounting system based on generally acceptable accounting principles which accurately reflects all costs chargeable (paid and unpaid) to the Project. A Receipts and Disbursements Ledger must be maintained. A general ledger with an Income and Expense Account for each budgeted line item is necessary. Paid invoices revealing check number, date paid and evidence of goods or services received are to be filed according to the expense account to which they were charged.
- 4.8 Contractor agrees that Contractor costs or earnings claimed under this Agreement may not be claimed under another contract or grant from another agency, organization, business entity or governmental entity.
- 4.9 Contractor shall establish and abide by a cost allocation methodology and plan which ensures that the City is paying only its fair share of the costs for services, overhead, and staffing not solely devoted to the Project or funded pursuant to this Agreement. The Cost Allocation Plan is a plan that identifies and distributes the cost of services provided by staff and/or departments or functions. It is the means to substantiate and support how the costs of a program are charged to a particular cost category or to the Program so as to assure Head Start Grant funds provided hereunder do not subsidize other program(s). Contractor must ensure that costs allocated and charged to the Head Start grant are not charged to other Federal, State or Local awards to overcome fund deficiencies, to avoid restrictions imposed by law or terms of the Federal awards, or for other reasons. Contractor shall provide to City prior to the beginning of the Agreement term (i) a matrix identifying the shared use of such facilities and/or program services; and (ii) the Cost Allocation Plan and supporting documentation, along with its Budget, financial statements and audit that are applicable to the Contractor's Project. City shall have the right to approve the Cost Allocation Plan.
- 4.10 Contractor agrees to reimburse the City for any Contractor overpayment based upon reconciled adjustments resulting from Contractor's balance and/or Statement of Revenue and Expenditure sheet as of January 31, 2018, which balance or Statement sheet shall be due to the City no later than February 15, 2018. Reimbursement shall be made within 20 calendar days of written notification to Contractor of the need for reimbursement.
 - 4.11 Upon expiration or early termination of this Agreement, or at any time during the term of this Agreement, all unused funds, rebates, advances exceeding allowable costs, or credits on-hand or collected thereafter relating to the Project, shall be returned by Contractor to the City within twenty (20) days of receipt of written notice.
 - 4.12 Upon execution of this Agreement or at any time during the term of this Agreement, the City's Director of Finance, the City Auditor, or a person designated by the Director of DHS may review and approve all Contractor's systems of internal accounting and administrative controls prior to the release of funds hereunder.

V. PROGRAM INCOME

- 5.1 For purposes of this Agreement, "program income" shall mean earnings of Contractor realized from activities resulting from this Agreement or from Contractor's management of funding provided or received hereunder. Such earnings shall include, but shall not be limited to, interest income; usage or rental/lease fees; income produced from Agreement-supported services of individuals or employees or from the use of equipment or facilities of Contractor provided as a result of this Agreement, and if applicable, payments from clients or third parties for services rendered by Contractor pursuant to this Agreement. At the sole option of the Director of DHS, Contractor will either (a) be required to return program income funds to City through DHS, or (b) upon prior written approval by the Director of DHS, Contractor may be permitted to retain such funds to be:
 - (A) added to the Project and used to further eligible Project objectives, in which case proposed expenditures must first be approved by the City; or
 - (B) deducted from the total Project cost for the purpose of determining the net cost reimbursed by the City.

- 5.2 In any case where Contractor is required to return program income to DHS, Contractor must return such program income to City within the timeframe that may be specified by the Director of DHS. If the Director of DHS does not specify a timeframe for Contractor to return program income to City, then Contractor must return such program income to City on the same date that Contractor submits its statement of expenditures and revenues to DHS set forth in Article V, Section 5.4 of this Agreement. If the Director of DHS grants Contractor authority to retain program income, Contractor must submit all reports required by DHS within the timeframe specified in the Agreement.
- 5.3 Contractor shall provide DHS with thirty (30) calendar days written notice prior to the activity that generates program income. Such notice shall detail the type of activity, time, and place of all activities that generate program income.
- 5.4 The Contractor shall fully disclose and be accountable to the City for all program income. Contractor must submit a statement of expenditures and revenues to DHS within thirty (30) calendar days of the activity that generates program income. The statement is subject to audit verification by DHS. Failure by Contractor to report program income as required is grounds for suspension, cancellation, or termination of this Agreement.
- 5.5 Contractor is prohibited from charging fees or soliciting donations and is prohibited from inviting or contracting with vendors who shall charge fees or solicit donations from Head Start participants and their parents in any Agreement-funded project without the prior written approval of the Director of DHS. However, Contractor may engage in general school activity that is not specifically targeted at Head Start families.
- 5.6 Contractor shall include this Article V, in its entirety, in all of its subcontracts involving income-producing services or activities.

VI. ADMINISTRATION OF CONTRACT

- 6.1 The Contractor agrees to comply with all the terms and conditions that the City must comply with within its award document from HHS. A copy of said award document is attached hereto and incorporated herein for all purposes as Attachment IV. From time to time, the award document may be amended or supplemented, and these shall be incorporated into the Agreement collectively as Attachment IV.
- 6.2 In the event that any disagreement or dispute should arise between the Parties hereto pertaining to the interpretation or meaning of any part of this Agreement or its governing rules, regulations, laws, codes or ordinances, the City Manager or the Director of DHS, as representatives of the City and the parties ultimately responsible for all matters of compliance with HHS and City rules and regulations, shall have the final authority to render or secure an interpretation. In the event that a dispute arises with regard to interpretation of regulations or law as it applies to this Agreement, Contractor may request that the City secure, and the City shall request an interpretation or opinion from HHS in order to assist in resolution of the dispute.
- 6.3 Contractor shall not use funds awarded from this Agreement as matching funds for any federal, state or local grant without the prior written approval of the Director of DHS.
- 6.4 The City shall have the authority during normal business hours to make physical inspections of all operating facilities occupied by Contractor for the administration of this Agreement and to require such physical safeguarding devices as locks, alarms, security / surveillance systems, safes, fire extinguishers, sprinkler systems, etc. as reasonably necessary, to safeguard property and/or equipment authorized by this Agreement.
- 6.5 The Contractor Board of Directors or Board of Trustees, as applicable, and Contractor's management staff shall adopt and approve an Employee Integrity Policy, in the event that Contractor has none, and internal

program management procedures, and require all staff to abide by these and the Head Start standards as established in the HHS regulations, to preclude theft, embezzlement, improper inducement, obstruction of investigation or other criminal action, and to prevent fraud and program abuse. These policies and procedures shall require repayment of such erroneously received grant funds or property to the Contractor, or to the applicable Head Start service provider from whom such grant funds or property was received, if other than the Contractor, and shall specify any other consequences to Contractor's employees and vendors involved in such illegal activities and may include but not be limited to termination and prosecution where necessary. Said policies and procedures shall be provided to DHS upon request by DHS. In the event that DHS finds the policies and procedures to be lacking, DHS may recommend revision.

- 6.6 Contractor agrees to comply with the following check writing and handling procedures:
 - (A) No blank checks are to be signed in advance;
 - (B) No checks are to be made payable to cash or bearer with the exception of those for petty cash reimbursement, not to exceed a \$100.00 maximum per check. Contractor agrees that the aggregate amount of petty cash reimbursement shall not exceed \$500.00 for any given calendar month during the term of this Agreement unless Contractor receives prior written approval from DHS to exceed such limit. Such requests for petty cash must be supported by the submission to DHS of an original receipt.
 - (C) Checks issued by City to Contractor shall be deposited into the appropriate bank account immediately or by the next Contractor business day after Contractor's receipt of each such check, and shall never be cashed for purposes of receiving the face amount back. If such check(s) are not deposited within the next Contractor business day from the date of issue, such checks shall be investigated by City and stop-payment orders issued, as applicable. City agrees that stop payment orders shall not be issued due to the financial institution's posting policies that result in posting of checks for credit beyond the required timeframe so long as the checks are deposited within the required timeframe. Upon cancellation of any outstanding check, if deemed appropriate by City, such check may be reissued to the Contractor or if deemed by City not to be a valid expense, such check shall be immediately returned to the City.
 - (D) For checks other than petty cash reimbursement, Contractor shall adopt and comply with a policy requiring no less than two (2) signatures of authorized representatives of Contractor on each check. Said signatures may be computer generated so long as they are issued in accordance with Contractor's internal policies established to reduce the risk of fraud, theft, or embezzlement. Contractor understands and agrees that City's reimbursement is subject to compliance with this provision of the Agreement.
- 6.7 Wherever in this Agreement, Contractor is required to perform an action within a specified number of days or hours, Contractor may request additional time to perform. City shall give Contractor's request for additional time due consideration and shall grant Contractor's request whenever reasonable practicable, unless immediate compliance is required.
- 6.8 If the starting date of the Agreement term has already passed at the time of approval of the Agreement by Contractor, all deadlines imposed on Contractor for providing information to the City on or before the date of approval or within seven (7) Contractor business days of approval shall be extended in order for the Contractor to reasonably comply with the City's requirements; except and unless, the information has already been provided to the City. City and Contractor agree to confer and make best efforts to reasonably permit Contractor to comply with the extended deadlines.

VII. AUDIT

7.1 If Contractor expends \$750,000.00 or more of funds provided under this Agreement, or cumulative funds provided by or through City, and does not have to comply with the provisions of Section 7.2, then during the term of this Agreement, the Contractor shall have completed an independent audit of Contractor's Head Start program and the reporting required submitted to the City within the earlier of 30 calendar days after

receipt of the auditor's report(s), or nine (9) months after the end of Contractor's fiscal year, expiration or early termination of this Contract, whichever is earlier. Contractor understands and agrees to furnish DHS a copy of the audit report including the corrective action plan(s) of the Head Start program on all audit findings, a summary schedule of prior audit findings, management letter and/or conduct of audit letter, within thirty (30) calendar days upon receipt of said report or upon submission of said corrective action plan to the auditor.

Contractor agrees and understands that upon notification from federal, state, or local entities that have conducted program reviews and/or audits of the Contractor or its programs of any findings about accounting deficiencies, or violations of Contractor's financial operations, a copy of the notification, review, investigation, and audit violations report must be forwarded to DHS within a period of ten (10) calendar days upon the Contractor's receipt of the report.

7.2 Contractor agrees that if Contractor receives or expends more than \$750,000.00 in federal funds from the City, an audit shall be made in accordance with the Single Audit Act Amendments of 1996, the State of Texas Single Audit Circular, and the U.S. Office of Management and Budget Circular ("Uniform Guidance"). Contractor shall also be required to submit copies of its annual independent audit report, and all related reports issued by the independent certified public accountant within the earlier of 30 days after receipt of the auditor's report(s), or nine months after the end of the audit period, unless a longer period is agreed to in advance by the Federal cognizant or oversight agency for audit to the Federal Audit Clearinghouse. A copy of this report must also be provided to City within this same time period. Contractor may submit reports through the following website: http://harvester.census.gov/sac/ and may also contact the Clearinghouse by telephone at (301) 763-1551 (local), 1-888-222-9907 or 1-800-253-0696 (toll free).

Per 2 C.F.R. 200.36, upon completion of Form SF-SAC, Contractor may submit the completed report by mail to:

Federal Audit Clearinghouse Bureau of the Census 1201 E. 10th Street Jeffersonville, Indiana 47132

Contractor agrees to reimburse the City or supplement any disallowed costs with eligible and allowable expenses based upon reconciled adjustments resulting from Contractor's Single Audit. Reimbursement shall be made within thirty (30) calendar days of written notification regarding the need for reimbursement.

- 7.3 If Contractor expends less than \$750,000.00 of City dollars during the term of this Contract, then the Contactor shall complete and submit an unaudited financial statement(s) within a period not to exceed nine months immediately succeeding the end of Contractor's fiscal year, expiration or early termination of this Contract, whichever is earlier. Said financial statement shall include a balance sheet and income statement prepared by a bookkeeper and a cover letter signed by Contractor attesting to the correctness of said financial statement.
- 7.4 All financial statement(s) must include a schedule of receipts and disbursements by budgeted cost category for each program funded by or through the City.
- 7.5 The City reserves the right to conduct, or cause to be conducted, an audit or review of all funds received under this Agreement at any and all reasonable and necessary times required by City. The City Internal Audit Staff, a Certified Public Accounting (CPA) firm, or other personnel as designated by the City, may perform such audit(s) or reviews. The City reserves the right to determine the scope of every audit, so long as it is limited to Contractor's services for the Head Start program. In accordance herewith, Contractor agrees to make available to City all accounting and Project records.

Contractor shall during Contractor's normal business hours, and as often as deemed reasonably necessary by City and/or the applicable state or federal governing agency or any other auditing entity, make available the books, records, documents, reports, and evidence with respect to all matters covered by this Agreement

and shall continue to be so available for a minimum period of four (4) years or whatever period is determined necessary based on the Records Retention guidelines, established by applicable law for this Agreement. Said records shall be maintained for the required period beginning immediately after Agreement termination, save and except there is litigation or if the audit report covering such agreement has not been accepted, then the Contractor shall retain the records until the resolution of such issues has satisfactorily occurred. The auditing entity shall have the authority to audit, examine and make excerpts, transcripts, and copies from all such books, records, documents and evidence, including all books and records used by Contractor in accounting for expenses incurred under this Agreement, all contracts, invoices, materials, payrolls, records of personnel, conditions of employment and other data relating to matters covered by this Agreement.

The City may, in its reasonable discretion, require the Contractor to use any and all of the City's accounting or administrative procedures that are in conformity with Generally Accepted Accounting Principles for state account standards in Texas in the planning, controlling, monitoring and reporting of all fiscal matters relating to this Agreement, and the Contractor shall abide by such requirements. Should a conflict exist between the parties' accounting procedures, Contractor shall use the stricter of the procedures.

7.6 When an audit or examination determines that the Contractor has expended funds or incurred costs which are questioned by the City and/or the applicable state or federal governing agency, the Contractor shall be notified and provided an opportunity to address the questioned expenditure or costs.

Should any expense or charge that has been reimbursed be subsequently disapproved or disallowed as a result of any site review or audit, the Contractor shall promptly refund such amount to the City no later than twenty (20) Contractor business days from the date of notification of such disapproval or disallowance by the City. At its sole option, DHS may deduct such claims from subsequent reimbursements and shall notify Contractor prior to exercising this option. Contractor shall provide to City a full refund of such amount no later than twenty (20) Contractor business days from the date of notification of such disapproval or disallowance by the City. If Contractor is obligated under the provision hereof to refund a disapproved or disallowed cost incurred, such refund shall be required and be made to City by check, cashiers check or money order. In the event that a dispute arises as to the allowability of an expenditure or charge pursuant to the federal regulations, then Contractor may request that the City secure, and the City shall request an interpretation or opinion from HHS in order to assist in resolution of the dispute. Should the City deduct such claims from subsequent reimbursements, the Contractor is forbidden from reducing Project expenditures and Contractor must use its own funds to maintain the Project.

Contractor agrees and understands that all expenses, fees, fines and penalties associated with the collection of delinquent debts owed by Contractor shall be the sole responsibility of the Contractor and shall not be paid from any Project funds received by the Contractor under this Agreement. Delinquent debts that would otherwise be identified as allowable costs may be paid with Project funds after written approval of DHS.

7.7 If the City determines that Contractor is in violation of the above requirements, the City shall have the right to dispatch auditors of its choosing to conduct the required audit and to have the Contractor pay for such audit from non-City resources. If after the audit is conducted it is determined that Contractor is in compliance with the contract then the cost of the audit, specifically the auditor's bill alone, shall be borne by the City. Contractor may recommend the hiring of alternate auditors, but the final decision on the selection of auditors shall rest with the City.

VIII. RECORDS, REPORTING, MONITORING AND INTELLECTUAL PROPERTY

8.1 In addition to those listed in this Agreement, the Contractor shall submit to DHS any and all reports as may be required of Contractor by HHS or as may be reasonably required of the City. Contractor shall incorporate and use a City approved tracking or information system, such as ChildPlus, for the delivery of comprehensive Head Start Services and collect, input and update all data required for the Program Information Report in accordance with the City's reasonable timeline to ensure the reporting of accurate and consistent information to HHS.

8.2 Additionally, Contractor shall maintain and furnish to City the appropriate financial and programmatic information and reports as listed in the Scope of Work, in such forms as the City may require or prescribe as required pursuant to the Head Start Act, as amended, or as may be required under federal regulations, such as 2 C.F.R. 200 st seq or other applicable regulations.

Contractor shall maintain all applicable and appropriate supporting documentation of costs, including but not limited to, payroll records, invoices, contracts or vouchers, and make these available to City upon request.

- 8.3 Contractor shall provide to DHS all information reasonably requested by DHS relating to the Contractor's Board functions, if applicable, including but not be limited to:
 - (A) Roster of current Board Members (name, title, and business address, telephone number, fax number and board e-mail address);
 - (B) Names and terms of Officers;
 - (C) Schedule of anticipated board meetings for current Fiscal Year;
 - (D) Board agendas relating to the Head Start Program to be submitted by electronic or facsimile transmission at the time of posting prior to each Board meeting. Prior to the time of posting, Contractor's administration shall attempt to notify City when a Head Start program item is anticipated to be placed on the agenda; and
 - (E) Minutes of every board meeting relating to the Head Start Program.
- 8.4 City reserves the right to reasonably request Contractor to provide additional records for travel expenses, long distance calls, faxes, internet service, cell phone calls, or other electronic communication devices charged to the budget associated with this Agreement.
- 8.5 If applicable, Contractor shall report all notices served, violations found or complaints filed with regard to licensing, or lack thereof, of Contractor's centers within one (1) Contractor business day of receipt of written notice from the State licensing, certifying or permit-issuing authority of a violation or complaint not counting weekends.
- 8.6 Contractor shall comply with federal Head Start Performance Standards (federal Head Start regulations), the Head Start Act and all applicable federal, state and local laws relating to child safety. Contractor shall establish and implement mutually agreeable administrative procedures to respond to medical, dental and other health emergencies with which all Head Start staff should be familiar and trained. These procedures shall include, among other things, methods of notifying parents in the event of a health emergency involving their children and established methods for handling cases of suspected or known child endangerment, abuse or neglect that are in compliance with applicable federal, state and local laws. If Contractor has knowledge of a claim or report of, or is aware of a program crisis related to, or suspects that media coverage would be negative due to, an incident of child endangerment, neglect or abuse of a Head Start child while in the Head Start program, Contractor shall contact the City's designated representative immediately, but not later than 24 hours, for the purpose of notification of the incident. Contractor shall contact the City's designated representative immediately even if the incident of child endangerment, neglect or abuse is not fully investigated by Contractor. If Contractor is unable to reach the City's designated representative, Contractor shall leave a verbal or written message via electronic mail notifying the City that Contractor is attempting to notify the City of an incident. Contractor further agrees to notify the parent of a Head Start child immediately, in any of the instances cited above, to include a situation in which the parent's child has received physical discipline, whether or not the instance may be characterized as suspected child abuse.
- 8.7 Within a period not to exceed forty-five (45) calendar days after the expiration or early termination date of the Agreement, Contractor shall submit all final client reports and all required deliverables to City. Contractor understands and agrees that in conjunction with the submission of the final report, the Contractor shall execute and deliver to City a receipt for all sums received and a release of all claims for said sums against the Project.

- 8.8 Contractor shall maintain financial records, supporting documents, statistical records, and all other books, documents, papers or other records pertinent to this Agreement or the Grant in accordance with the official records retention schedules established within the Local Government Records Act of 1989 and any amendments thereto, or for such period as may be specifically required by 45 C.F.R §75.361, whichever is longer. Notwithstanding the foregoing, Contractor shall maintain all Agreement and Grant related documents for no less than four (4) years from the date of City's submission of the annual financial report covering the funds awarded hereunder. If an audit, litigation, or other action involving the records has been initiated before the end of the four (4) year period, Contractor agrees to maintain the records until the end of the four (4) year period or until the audit, litigation, or other action is completed, whichever is later. Records for real property and equipment acquired with Head Start funds shall be retained for four (4) years after final disposition.
- 8.9 Contractor shall make available to City, HHS, or any of their duly authorized representatives, upon appropriate notice and unless otherwise prohibited by law, such books, records, reports, documents, papers, policies and procedures as may be necessary for audit, examination, excerpt, transcription, and copy purposes, for as long as such records, reports, books, documents, and papers are retained. This right also includes timely and reasonable access to Contractor's facility and to Contractor's personnel for the purpose of interview and discussion related to such documents. Contractor shall, upon request, transfer certain records to the custody of City or HHS when City or HHS determines that the records possess long-term retention value unless otherwise prohibited by law, and subject to Contractor's right to use "educational records" as that term is understood under the Family Educational Rights and Privacy Act of 1974 ("FERPA").
- 8.10 The Contractor agrees to incorporate and use any City-approved tracking or information system for the delivery of comprehensive Head Start services. Contractor shall enter current, accurate and complete client data.
- 8.11 DHS is assigned monitoring, fiscal control, and evaluation of certain projects funded by the City with General or Grant Funds, including the Project covered by this Agreement. Therefore, Contractor agrees to permit City and/or HHS to evaluate, through monitoring, reviews, inspection or other means, the quality, appropriateness, and timeliness of services delivered under this Agreement and to assess Contractor's compliance with applicable legal and programmatic requirements. At such times and in such form as may be required by DHS, the Contractor shall make available to DHS and the Grantor of the Grant Funds, if applicable, such statements, reports, records, personnel files (including evidence of criminal background check as required by Head Start regulations), client files, data, all policies and procedures and information as may be requested by DHS and shall permit the City and Grantor of the Grant Funds, if applicable, to have interviews with its personnel, board members and program participants pertaining to the matters covered by this Agreement. Contractor agrees that the failure of the City to monitor, evaluate, or provide guidance and direction shall not relieve the Contactor of any liability to the City for failure to comply with the Terms of the Grant or the terms of this Agreement. Regarding evidence of criminal background check(s), the parties agree that City will accept a written statement that the checks have been conducted and that all persons who are employed have passed, so long as the statement includes the name(s) of the staff member(s) checked, and the date(s) performed.. If, at any time, HHS informs the City and / or Contractor that such written statement does not satisfy the requirements of the Terms of the Grant, Contractor agrees to provide additional information as may be legally permissible and as may be required by HHS or to cooperate with the City and HHS in order to resolve any conflict associated with provision of information related to criminal background checks.
- 8.12 City may, at its discretion, conduct periodic, announced monitoring visits to ensure program and administrative compliance with Head Start Performance Standards,, the Head Start Act and with Program goals and objectives for the agreement period. City reserves the right to make unannounced visits to Contractor Program sites when it is determined that such unannounced visits are in the interest of effective program management and service delivery. Contractor Program staff shall be informed by City representative(s) upon arrival at Contractor's site of the expected purpose and length of visit so that accommodations may be made. City's representative(s) shall provide proper identification to ensure the safety and security of all parties. City's representative(s) visiting or entering a campus site where students

- are present shall comply with Contractor's rules and procedures under Contractor's Board policy series GKC ("Community Relations/Visitors to the Schools").
- 8.13 Contractor understands that the City will timely inform Contractor of the findings of any such review or monitoring, specifically any default under the Agreement or deficiencies in performance, and will inform Contractor in writing of Program strengths and weaknesses and specify a deadline for corrective action based upon contract or federal deadlines and the time needed for City to review, monitor and approve when necessary. The City will assist Contractor in finding solutions for Program improvement if and as appropriate.
- Unless otherwise provided herein, all reports, statements, records, data, policies and procedures or other information requested by DHS shall be submitted by Contractor to City within five (5) Contractor business days of the request made to Contractor via electronic communication or by other form of written correspondence. In the event that Contractor fails to deliver the required reports or information or delivers incomplete information within the prescribed time period, the City may, upon reasonable written notice, suspend reimbursements to Contractor until such reports are delivered to City. Furthermore, the Contractor ensures that all information contained in all required reports or information submitted to City is accurate.
 - 8.15 (A) Unless disclosure is authorized by the City or is required by the Attorney General for the State of Texas or his duly authorized representative, Contractor agrees to maintain in confidence all information pertaining to the Project or other information and materials prepared for, provided by, or obtained from City including, without limitation, reports, information, project evaluation, project designs, data, other related information (collectively, the "Confidential Information") and to use the Confidential Information for the sole purpose of performing its obligations pursuant to this Agreement. Contractor shall protect the Confidential Information and shall take all reasonable steps to prevent the unauthorized disclosure, dissemination, or publication of the Confidential Information.
 - (B) If disclosure is permitted by law or required by order of a governmental agency or court of competent jurisdiction, Contractor shall give the Director of DHS prior written notice that such disclosure is required with a full and complete description regarding such requirement.
 - (C) Contractor shall establish specific procedures designed to meet the obligations of this Article VIII, Section 8.15, including, but not limited to execution of confidential disclosure agreements, regarding the Confidential Information with Contractor's employees and subcontractors prior to any disclosure of the Confidential Information to third parties. This Article VIII, Section 8.15 shall not be construed to limit HHS's, the City's or its authorized representatives' right to obtain copies, review and audit records or other information, confidential or otherwise, under this Agreement. Upon termination or expiration of this Agreement, Contractor shall return to City all copies of materials related to the Project, including the Confidential Information and subject to Contractor's right to use "Educational Records" as that term is understood under FERPA. All confidential obligations contained herein (including those pertaining to information transmitted orally) shall survive the termination of this Agreement. The Parties shall ensure that their respective employees, agents, and contractors are notified of the requirement to comply with the aforementioned obligations.
 - The Public Information Act, Government Code Section 552.021, requires the City to make public information available to the public. Under Government Code Section 552.002(a), public information means information that is written, produced, collected, assembled or maintained under a law or ordinance or in connection with the transaction of official business: 1) by a governmental body; or 2) for a governmental body and the governmental body owns the information, has a right of access to it, or has spent or contributed public money for the purpose of its writing, production, collection, assembly or maintenance. Therefore, if City receives a request under the Public Information Act (i.e., an open records request) for information within Contractor's possession pursuant to this Agreement, Contractor shall forward the requested documents to the City within two (2) Contractor business days of Contractor's designated liaison's receipt of the written request. If the requested information is confidential or may be kept confidential pursuant to state or federal law, the Contractor shall submit to City the list of specific statutory authority mandating and/or authorizing confidentiality no later than three (3) Contractor business days

Contractor's designated liaison's receipt of such request. For the purposes of communicating and coordinating with regard to public information requests, all communications shall be made to the designated public information liaison for each Party. Each Party shall designate in writing to the other Party the public information liaison for its organization and notice of a change in the designated liaison shall be made promptly to the other Party. The parties shall cooperate with each other to preserve confidential information or records that may be excluded from disclosure under the Family Educational Rights and Privacy Act (FERPA) (20 U.S.C. § 1232g; 34 CFR Part 99) and/or the Texas Public Information Act; and the parties shall coordinate efforts to seek any required Attorney General decision for the protection of such information from release.

In accordance with Texas law, Contractor acknowledges and agrees that all local government records as defined in Chapter 201, Section 201.003 (8) of the Texas Local Government Code created or received in the transaction of official business or the creation or maintenance of which were paid for with public funds are declared to be public property and subject to the provisions of Chapter 201 of the Texas Local Government Code and Subchapter J, Chapter 441 of the Texas Government Code. Thus, Contractor agrees that no such local government records produced by or on the behalf of Contractor pursuant to this Agreement shall be the subject of any copyright or proprietary claim by Contractor; however Contractor shall be entitled to maintain the confidentiality of "educational records" as that term is understood under FERPA and to use such records for educational purposes.

With the exception of student records, Contractor acknowledges and agrees that all local government records, as described herein, produced in the course of the work required by this Agreement, shall belong to and be the property of City and shall be made available to the City at any time unless otherwise prohibited by law. The parties agree that educational records created pursuant to Section 1.1 shall be maintained and utilized by Contractor as required by law. Contractor further agrees to turn over to City all such records upon termination of this Agreement, unless otherwise prohibited by law. Contractor agrees that it shall not, under any circumstances, release any records created during the course of performance of the Agreement to any entity without the written permission of the Director of DHS, unless required to do so by a court of competent jurisdiction or the Texas Attorney General or his designee, or as may be required or permitted by Contractor due to the record being an "educational record" as that term is understood under FERPA. DHS shall be notified of such request as set forth in Article VIII, Section 8.15 of this Agreement.

8.18 Ownership of Intellectual Property. Contractor and City agree that the Project shall be and remain the sole and exclusive proprietary property of City. The Project shall be deemed a "work for hire" within the meaning of the copyright laws of the United States, and ownership of the Project and all rights therein shall be solely vested in City. Contractor hereby grants, sells, assigns, and conveys to City all rights in and to the Project and the tangible and intangible property rights relating to or arising out of the Project, including, without limitation, any and all copyright, patent and trade secret rights. All intellectual property rights including, without limitation, patent, copyright, trade secret, trademark, brand names, color schemes, designs, screens, displays, user interfaces, data structures, organization, sequences of operation, trade dress, and other proprietary rights (the "Intellectual Property Rights") in the Project shall be solely vested in City. As owner of the tangible and intangible intellectual property, City shall have the right to reproduce, publish, authorize others to reproduce or publish, or otherwise use such material subject to confidentiality obligations as may be required by federal and /or state law for "educational records." Contractor agrees to execute all documents reasonably requested by City to perfect and establish City's right to the Intellectual Property Rights. In the event City shall be unable, after reasonable effort, to secure Contractor's signature on any documents relating to Intellectual Property Rights in the Project, including without limitation, any letters patent, copyright, or other protection relating to the Project, for any reason whatsoever, Contractor hereby irrevocably designates and appoints City and its duly authorized officers and agents as Contractor's agent and attorney-in-fact, to act for and in Contractor's behalf and stead to execute and file any such application or applications and to do all other lawfully permitted acts to further the prosecution and issuance of letters patent, copyright or other analogous protection thereon with the same legal force and effect as if executed by Contractor. Provided, however, nothing herein contained is intended nor shall it be construed to require Contractor to transfer any ownership interest in Contractor's best practice and benchmarking information to the City.

- 8.19 In the event that Contractor desires to copyright material or to permit any third-party to do so, Contractor must obtain City's prior written approval to do so and must appropriately acknowledge City's support in any such materials.
- Subject to obligations to maintain confidentiality under the Health Insurance Portability and Accountability Act of 1996 ("HIPAA") and the HIPAA Business Associate Agreement, attached hereto and incorporated herein for all purposes as Attachment V, and subject to the requirements of the Family Educational Rights and Privacy Act ("FERPA") and the limitations imposed under law regarding transfer of information, any and all writings, documents or information in whatsoever form and character produced by Contractor pursuant to the provisions of this Agreement is the exclusive property of City; and no such writing, document or information shall be the subject of any copyright or proprietary claim by Contractor. Contractor understands and acknowledges that as the exclusive owner of any and all such writings, documents and information, City has the right to use all such writings, documents and information as City desires, without restriction; subject, however to Contractor's continuing rights regarding "educational records" as that term is understood under FERPA. City agrees that it will not release to the public "educational records" that come into its possession unless otherwise authorized by law.

IX. INSURANCE

9.1 Contractor and the City each maintain adequate general liability insurance coverage and self-insurance for worker's compensation claims and causes of action to meet their statutory obligations to each party's employees.

X. LIMITED LIABILITY

10.1 CONTRACTOR AND THE CITY ACKNOWLEDGE THEY ARE POLITICAL SUBDIVISIONS OF THE STATE OF TEXAS AND ARE SUBJECT TO, AND COMPLY WITH THE APPLICABLE PROVISIONS OF THE TEXAS TORT CLAIMS ACT, AS SET OUT IN THE CIVIL PRACTICE AND REMEDIES CODE, SECTION 101.001, ET. SEQ., AND THE REMEDIES AUTHORIZED THEREIN REGARDING CLAIMS OR CAUSES OF ACTION THAT MAY BE ASSERTED BY THIRD PARTIES FOR ACCIDENT, INJURY OR DEATH.

XI. THIS ARTICLE INTENTIONALLY LEFT BLANK

XII. APPLICABLE LAWS

- 12.1 Contractor, and all of the work performed under this Agreement, shall comply with all applicable laws, rules, regulations and codes of the United States and the State of Texas and with the charter, ordinances, bond ordinances, and rules and regulations of the City of San Antonio and Bexar County. Contractor agrees to abide by any and all future amendments or additions to such laws, rules, regulations, policies and procedures as they may be promulgated. Contractor may request additional time to come into and demonstrate compliance, unless immediate compliance is required by the applicable Head Start regulations, the Head Start Act, or Public Information Act request requirements. Should the City need to abide by some other law, rule, regulation, policy or procedure, such requirement will be made known to Contractor upon consideration of Contractor's request for additional time.
- 12.2 The Contractor understands that certain funds provided it pursuant to this Agreement are funds which have been made available by the City's General Operating Budget and/or by federal, state, or other granting entities. Contractor agrees to comply with all laws, ordinances, codes, rules, regulations, policies, and procedures, including licensing and accreditation standards applicable to the funds received by Contractor hereunder as directed by the City or as otherwise required in this Agreement, including but not limited to:

- (A) The Head Start Act (42 U.S.C. §9831 et seq., as amended);
- (B) 45 C.F.R. Part 1301 et seg .;
- (C) The Terms of the Grant;
- (D) As applicable, 45 C.F.R. Part 75 ("Uniform Administrative Requirements, Cost Principles, and Audit Requirements for HHS Awards");
- (E) Texas Child Care Licensing laws;
- (F) The Office of Management and Budget (OMB) Circular at 2 C.F.R. 200 et. al. titled "Uniform Administrative Requirements, Cost Principles, and Audit Requirements for Federal Awards," ("Uniform Guidance"), as applicable to the funds received by Contractor hereunder;
- (G) Official record retention schedules as established by the Local Government Records Act of 1989; and
- (H) The Texas Public Information Act, at Chapter 552, The Texas Government Code.

12.3 Contractor further understands and agrees:

- (A) Contractor shall comply with all standards, orders or regulations issued pursuant to the Clean Air Act (42 U.S.C. §§7401-7671q) and the Federal Water Pollution Control Act (33 U.S.C. §§1251-1387), as amended and as applicable. Contractor agrees to report each violation to the City and understands that the City will, in turn, report each violation as required to the federal agency providing funds for this Contract and the appropriate EPA Regional Office. Additionally, Contractor agrees to include these requirements in each subcontract to this Contract exceeding \$150,000.00 financed in whole or in part with federal funds
- (B) to make positive efforts to utilize small businesses, minority-owned firms and women's business enterprises in connection with the work performed hereunder, whenever possible.
- (C) to provide for the rights of the Federal Government in any invention resulting from the work performed hereunder, in accordance with 37 C.F.R. Part 401 and any applicable implementing regulations.
- (D) to include a provision requiring compliance with the Copeland "Anti-Kickback" Act (40 U.S.C 3145), as applicable under Appendix II of the OMB Uniform Guidance and as supplemented by Department of Labor regulations at 29 C.F.R. Part 3, "Contractors and Subcontractors on Public Building or Public Work Financed in Whole or In Part by Loans or Grants from the United States".
- (E) to comply with the Davis-Bacon Act, as amended (40 U.S.C. 3141-3144, and 3146-3148) and as applicable under Appendix II of the OMB Uniform Guidance, and as supplemented by Department of Labor regulations at 29 C.F.R. Part 5, implementing regulations, and the relevant Additional OMB Provisions attached hereto and incorporated herein for all purposes as Attachment VI, and to include a provision requiring compliance with the each in any construction contracts of more than \$2,000.00, and report all suspected or reported violations to HHS.
- (F) to comply with the certification and disclosure requirements of the Byrd Anti-Lobbying Amendment (31 U.S.C. § 1352), and any applicable implementing regulations. Contractor verifies it has tendered said Certificate to the City.
- (G) to comply with the applicable standards under the McKinney-Vento Homeless Assistance Act (42 U.S.C. §§11431 et seq.), and any applicable implementing regulations, as may be applicable.

- (H) to comply with the Contract Work Hours and Safety Standards Act (40 USC 3701-3708) as applicable under Appendix II of the OMB Uniform Guidance, and as supplemented by Department of Labor regulations (29 CFR Part 5), relating to all contracts that involve the employment of mechanics or laborers, and the relevant provisions in Attachment VI, which provides, in part, that each contractor must be required to compute the wages of every mechanic and laborer on the basis of a standard work week of 40 hours, that work in excess of the standard work week be compensated at a rate at least one and a half times the basic rate of pay, and that no laborer or mechanic must be required to work under conditions which are unsanitary, hazardous or dangerous.
- (I) to comply with the prohibitions contained in the Pro-Children Act of 1994 (20 U.S.C §6081-84), relating to no smoking within any indoor facility (or portion thereof) owned or leased or contracted for by Contractor for the provision of regular or routine health care or day care or early childhood development (Head Start) services to children or for the use of the employees of the City or Contractor who provide such services.
- (J) Contractor and its subcontractors shall comply with section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act, including, but not limited to, the regulatory provisions of 40 CFR Part 247, and Executive Order 12873, as they apply to the procurement of the items designated in Subpart B of 40 CFR Part 247.
- 12.4 The Contractor certifies that it will provide a drug-free workplace in compliance with the Drug-Free Workplace Act of 1988 (41 U.S.C. §§ 701-707 and 8101-8106, as amended). Failure to comply with the above-referenced law could subject the Contractor to suspension of payments, termination of Agreement, and debarment and suspension actions.
- 12.5 Contractor shall comply with all federal, state, or local laws, rules, and orders prohibiting discrimination, and shall not engage in employment practices which have the effect of discriminating against any employee or applicant for employment, and will take affirmative steps to ensure that applicants are employed and employees are treated during employment without regard to their race, color, religion, national origin, sex, sexual orientation, gender identity, veteran status, age, disability, or political belief or affiliation, unless exempted by state or federal law, or as otherwise established herein. Consistent with the foregoing, Contractor agrees to comply with Executive Order 11246, entitled "Equal Employment Opportunity", as amended by Executive Orders 13665 and 11375, and as supplemented by regulations at 41 C.F.R. Part 60. Additionally, Contractor certifies that it will comply fully with the following nondiscrimination, minimum wage and equal opportunity provisions, including but not limited to:
 - (A) Title VII of the Civil Rights Act of 1964, as amended:
 - (B) Section 504 of the Rehabilitation Act of 1973, as amended;
 - (C) The Age Discrimination Act of 1975, as amended
 - (D) Title IX of the Education Amendments of 1972, as amended; (Title 20 USC sections 1681-1688)
 - (E) Fair Labor Standards Act of 1938, as amended
 - (F) Equal Pay Act of 1963, P.L. 88-38; and
 - (G) All applicable regulations implementing the above laws.
 - 12.5.1 Further, if Contractor engages in any contract for construction, as defined by 41 C.F.R. Part 60-1.3, using Head Start funds that, except as otherwise provided under 41 C.F.R. Part 60, meets the definition of "federally assisted construction contract" in 41 C.F.R. Part 60-1.3, Contractor must comply with the Equal Employment Opportunity provisions in Attachment VI and all of the Executive Order and Code of Federal Regulations provisions previously cited in Section 12.5, and must include the provisions in any of its subcontracts.
- 12.6 The Contractor warrants that any and all taxes that the Contractor may be obligated for, including but not limited to, federal, state, and local taxes, fees, special assessments, Federal and State payroll and income taxes, personal property, real estate, sales and franchise taxes, are current, and paid to the fullest extent

liable as of the execution date of the Agreement. The Contractor shall comply with all applicable local, State, and Federal laws including, but not limited to, related to:

- (A) worker's compensation;
- (B) unemployment insurance
- (C) timely deposits of payroll deductions;
- (D) filing of Information on Tax Return form 990 or 990T, Quarterly Tax Return Form 941, W-2's Form 1099 on individuals who received compensation other than wages, such as car allowance, Forms 1099 and 1096 for contract or consultant work, non-employee compensation, etc;
- (E) Occupational Safety and Health Act regulations; and
- (F) Employee Retirement Income Security Act of 1974, P.L. 93-406.
- 12.7 Contractor agrees to comply with the Americans with Disabilities Act of 1990, 42 U.S.C. 12101 et seq., and all regulations thereunder.
- 12.8 All expenditures by the Contractor or any of its subcontractors must be made in accordance with all applicable federal, state and local laws, rules and regulations.
- 12.9 If applicable, Contractor shall submit to DHS its most recent form 990 or 990T and also submit any that are filed with the Internal Revenue Service subsequent to its last submission to the City if filed during the term of the Agreement.

XIII. NO SOLICITATION/CONFLICT OF INTEREST

- 13.1 The Contractor warrants that no person or selling agency or other organization has been employed or retained to solicit or secure this Agreement upon a contract or understanding for a commission, percentage, brokerage, or contingent fee and further that no such understanding or agreement exists or has existed with any employee of the Contractor or the City. For breach or violation of this warrant, the City shall have the right to terminate this Agreement without liability or, at its discretion, to deduct from the Agreement or otherwise recover the full amount of such commission, percentage, brokerage, or contingent fee, or to seek such other remedies as legally may be available.
- 13.2 Contractor covenants that neither it nor any member of its governing body or of its staff presently has any interest, direct or indirect, which would conflict in any manner or degree with the performance of services required to be performed under this Agreement. Contractor further covenants that in the performance of this Agreement, no persons having such interest shall be employed or appointed as a member of its governing body or of its staff.
- 13.3 Contractor further covenants that no member of its governing body or of its staff shall possess any interest in, or use their position for, a purpose that is or gives the appearance of being motivated by desire for private gain for themselves or others, particularly those with which they have family, business, or other ties.
- 13.4 No member of City's governing body or of its staff who exercises any function or responsibility in the review or approval of the undertaking or carrying out of this Agreement shall:
 - (A) Participate in any decision relating to this Agreement which may affect his or her personal interest or the interest of any corporation, partnership, or association in which he or she has a direct or indirect interest; or
 - (B) Have any direct or indirect interest in this Agreement or the proceeds thereof.
- 13.5 Contractor acknowledges that it is informed that Charter of the City of San Antonio and its Ethics Code prohibit a City officer or employee, as those terms are defined in Section 2-52 of the Ethics Code, from having a financial interest in any contract with the City. An officer or employee has "prohibited financial interest" in a contract with the City or in the sale to the City of land, materials, supplies or service, if any of

the following individual(s) or entities is a party to the contract or sale: A City officer or employee; his parent, child or spouse; anentity in which the officer or employee, or his parent, child or spouse owns ten (10) percent or more of the voting stock or shares of the entity, or ten (10) percent or more of the fair market value of the entity; an entity in which any individual or entity above listed is subcontractor on a City contract, a partner or a parent or subsidiary entity.

13.6 Contractor warrants and certifies, and this Agreement is made in reliance thereon, that neither the Contractor nor his or her spouse, parent, child, sibling or first-degree relative is a City officer or employee as defined by Section 2-52 (e) of the City Ethics Code. If Contractor is a business entity, the Contractor representative further warrants and certifies that no City officer or employee nor any spouse, parent, child, sibling of first-degree relative of a City officer or employee owns ten (10) percent or more of the voting stock or shares of the business entity, or ten (10) percent or more of the fair market value of the business entity). Contractor further warrants and certifies that is has tendered to the City a Discretionary Contracts Disclosure Statement in compliance with the City's Ethics Code.

XIV. TERMINATION

- 14.1 (a) Termination for Cause Upon written notice in accordance with Article XXVI, Official Communication, CITY may terminate this Contract as of the date provided in the notice in whole or in part, upon the occurrence of either:
 - (i) Contractor's failure to fulfill, in a timely and proper manner, obligations under this Agreement to include performance standards established by the City, or if the Contractor should violate any of the covenants, conditions, or stipulations of the Contract, or
 - (ii) Notification by a local, state, or federal agency of a formal charge, probation, deferred adjudication, or conviction involving fraud, theft, or the commission of a felony by Contractor or Contractor's employee working in the Head Start Program. In the case of a Contractor's employee being the subject of the notification, Contractor shall have the opportunity to cure via the immediate termination and/or removal of the employee from the Head Start Program.
 - (b) Termination for Convenience This Contract may be terminated in whole or in part upon providing notice in accordance with Article XXVI, Official Communication, which shall specify a date, not sooner than 120 days following the day on which notice is sent but not later than the end of Contractor's fiscal year, unless earlier terminated under any other provision herein.
- 14.2 The Contractor shall be entitled to receive just and equitable compensation for any work satisfactorily completed prior to such termination date. The question of satisfactory completion of such work shall be reasonably determined by the City, and its decision shall be final. To the extent that compliance with an Agreement matter falls under HHS authority to review, or in the event that a dispute arises with regard to interpretation of regulations or law as it applies to this Agreement, Contractor may request that the City secure, and the City shall request an interpretation or opinion from HHS in order to assist in resolution of the dispute. It is further expressly understood and agreed by the Parties that Contractor's performance upon which final payment is conditioned shall include, but not be limited to, the Contractor's complete and satisfactory performance, of its obligations for which final payment is sought.
- 14.3 Notwithstanding any other remedy contained herein or provided by law, the City may delay, suspend, limit, or cancel funds, rights or privileges herein given the Contractor for failure to comply with the terms and provisions of this Agreement. Specifically, at the option of the City, the Contractor may be placed on probation during which time the City may withhold reimbursements in cases where it reasonably determines that the Contractor is not in compliance with this Agreement. The Contractor shall not be relieved of liability to the City for damages sustained by the City by virtue of any breach of this Agreement, and the City may withhold funds otherwise due as damages, in addition to retaining and utilizing any other remedies available to City.

- 14.4 If an employee of Contractor is discharged or otherwise leaves employment with Contractor, then the Contractor shall pay in full to such employee all of such employee's earned salaries and wages, within the timeframe specified by law.
- 14.5 Should the Contractor be debarred by federal government or the City pursuant to a debarment policy currently existing or hereafter adopted, said debarment may be grounds for termination.
- 14.6 This Agreement is subject to the availability of federal grant funds to City and may be terminated by the City if HHS terminates the City as a grantee or reduces the amount granted to City, for any reason; provided that, if the reduction of grant funds does not result in complete unavailability of such funds, the Parties will use best efforts to amend this Agreement accordingly. City will promptly notify Contractor of any such HHS action.
- 14.7 In all instances of termination, Contractor shall not incur new obligations after the effective date of termination, and shall cancel as many outstanding obligations as possible. Contractor shall submit to City all required reports including a final financial statement which shall be a statement of all expenditures incurred by Contractor under this Agreement. City shall pay Contractor the full cost of obligations that City determines were not subject to cancellation if such costs are properly documented, allowable, within the approved budget, and unavoidably incurred by Contractor prior to termination or expiration. The foregoing shall constitute full and complete reimbursement for all of Contractor's performance under this Agreement.

XV. PROHIBITION OF POLITICAL ACTIVITIES

- 15,1 Contractor agrees that no funds provided from or through the City shall be contributed or used to conduct political activities for the benefit of any candidate for elective public office, political party, organization or cause, whether partisan or non-partisan, nor shall the personnel involved in the administration of the Project provided for in this Agreement be assigned to work for or on behalf of any partisan or non-partisan political activity.
- 15.2 Contractor agrees that no funds provided under this Agreement may be used in any way to attempt to influence, in any manner, a member of Congress or any other State or local elected or appointed official.
- 15.3 The prohibitions set forth in Article XV, Sections 15.1 and 15.2 of this Agreement include, but are not limited to, the following:
 - (A) an activity to further the election or defeat of any candidate for public office or for any activity undertaken to influence the passage, defeat or final content of local, state or federal legislation;
 - (B) working or directing other personnel to work on any political activity during time paid for with City funds, including, but not limited to activities such as taking part in voter registration drives, voter transportation activities, lobbying, collecting contributions, making speeches, organizing or assisting at meetings or rallies, or distributing political literature;
 - (C) coercing personnel, whether directly or indirectly, to work on political activities on their personal time, including activities such as taking part in voter registration drives, voter transportation activities, lobbying, collecting contributions, making speeches, organizing or assisting at meetings or rallies, or distributing political literature; and
 - (D) using facilities or equipment paid for, in whole or in part with City funds for political purposes including physical facilities such as office space, office equipment or supplies, such as telephones, computers, fax machines, during and after regular business hours.
- 15.4 To ensure that the above policies are complied with, Contractor shall provide every member of its personnel paid out of Agreement funds with a statement provided by Contractor of the above prohibitions and have each said individual sign a statement acknowledging receipt of the policy. Such statement shall

- include a paragraph that directs any staff person who has knowledge of violations or feels that he or she has been pressured to violate the above policies to call and report the same to DHS. Contractor shall list the name and number of a contact person from DHS on the statement that Contractor's personnel can call to report said violations.
- 15.5 Contractor agrees that in any instance where an investigation of the above is ongoing or has been confirmed based upon then current reasonable evidence of impropriety, salaries paid to the Contractor under this Agreement may, at the City's discretion, be withheld until the situation is resolved.
- 15.6 This Article shall not be construed to prohibit any person from exercising his or her right to express his or her opinion or to limit any individual's right to vote. Further, Contractor and staff members are not prohibited from participating in political activities on their own volition, if done during time not paid for with Agreement funds.

XVI. PERSONNEL

- 16.1 Contractor shall establish and maintain an organizational structure that supports the accomplishment of Program objectives, addresses the major functions and responsibilities assigned to each staff position and provides evidence of adequate mechanisms for staff supervision and support to ensure the effective oversight of the Head Start program operations. Contractor shall ensure that, at a minimum, the program management functions listed in the Scope of Work are assigned to and adopted by staff within the Program.
- 16.2 Contractor acknowledges and agrees that Head Start guidelines and City policy require that Contractor shall staff each Head Start classroom with two teachers, or one teacher and a teacher's aide, at all times. To the fullest extent possible, Contractor shall staff each classroom with a volunteer in addition to the two paid staff persons. Additionally, Contractor will have adequate staff to fully implement all areas of program operations, including those having the specific management functions and expertise set forth in the Scope of Work.
- 16.3 At the beginning of this Agreement period, Contractor shall submit to City a report which specifically (a) lists the number and percentage of classroom personnel in its center based program having child development associate credentials or associate, baccalaureate or advanced degrees; and (b) describes to City Contractor's compliance with the goals described in Sections 16.3, 16.4 and 16.5 of this Agreement.
- 16.4 Contractor understands that the Head Start Act requires that grantees and their contractors, if any, ensure, and demonstrate upon request, that all of the teachers staffing its Head Start center-based program classrooms have (a) a baccalaureate or advanced degree in early childhood education; or (b) a baccalaureate or advanced degree and course work equivalent to a major relating to early childhood education, with experience teaching preschool-age children. Contractor agrees that during the term of this Agreement, Contractor will only employ teachers meeting the necessary qualifications.
- 16.5 Contractor understands that the Head Start Act requires that grantees and their contractors, if any, ensure, and demonstrate upon request, that all teaching assistants staffing its Head Start center-based program classrooms (a) have at least a child development associate credential; or (b) are enrolled in a program leading to an associate or baccalaureate degree. Contractor agrees that during the term of this Agreement, Contractor will only employ teaching assistants meeting the necessary qualifications.
- 16.6 Contractor understands that the Head Start Act requires that grantees and their contractors, if any, ensure, and demonstrate upon request, that all Head Start education coordinators in its center-based programs, including those who serve as curriculum specialists: (a) have the capacity to offer assistance to other teachers in the implementation and adaptation of curricula to the group and individual needs of children in a Head Start classroom; and (b) have (i) a baccalaureate or advanced degree in early childhood education or (ii) a baccalaureate or advanced degree and coursework equivalent to a major relating to early childhood education, with experience teaching preschool-age children. Contractor agrees that during the term of this Agreement, Contractor will only employ education coordinators meeting the necessary qualifications.

- 16.7 Contractor agrees and acknowledges that each of its Head Start teachers shall attend not less than 15 clock hours of professional development per Contractor fiscal year. The term "professional development" means high-quality activities that will improve the knowledge and skills of Head Start teachers and staff, as relevant to their roles and functions, in program administration and the provision of services and instruction, as appropriate, in a manner that has a positive and lasting impact and improves service delivery to enrolled children and their families.
- 16.8 Contractor understands that the Head Start Act requires grantees and their contractors, if any, ensure and demonstrate upon request that all Head Start staff have the knowledge, skills, and experience they need to perform their assigned functions responsibly. Therefore, in addition to all other provisions in this article, Contractor agrees that Contractor's Head Start Director and all Program center site directors, site managers, content coordinators and other administrators must have education and/or training in the area of early childhood education and family support. To meet these obligations, Contractor will:
 - (A) only employ a Head Start Director, and Program center site directors, site managers, content coordinators and other Program administrators meeting the necessary qualifications;
 - (B) promptly notify City of any transfers and/or disciplinary actions affecting Head Start personnel referred to in subsection (A) of this section 16.8; and
 - (C) ensure, in the event a Head Start position referred to in subsection (A) of this section 16.8 is vacated for any reason, that a replacement meeting the necessary qualifications is hired within 60 calendar days or, following notification to City of a delay, as soon as reasonably practical, but not later than 90 days after the position first became vacant unless the parties come to a new mutually acceptable deadline.
- 16.9 Contractor shall create, in consultation with each of its employees, and implement a professional development plan for all Head Start employees who provide direct services to children. Contractor shall regularly evaluate such plans to determine their impact on teacher and staff effectiveness.
- 16.10 Contractor understands that the City shall periodically perform its own wage and salary comparison and shall issue such results to Contractor. Contractor understands and agrees that City shall have no obligation to reimburse Contractor in excess of wages to an employee that exceed the average rate of compensation paid to persons providing substantially comparable services in the area. For purposes of this Agreement, the City shall accept the wage information set forth in the most recent study commissioned by, or issued by, the Texas Association of School Boards, in determining the rate of reimbursement. Although the City may consider factors such as training and experience as affecting compensation levels, the City shall have the sole and absolute authority to determine the rate of City's reimbursement under the Agreement, which determination shall not be arbitrary or unreasonable, and its decision shall be final due to the City's obligation of ensuring that wage comparability studies meet the requirements of the Head Start Act and implementing regulations. Subject to the restriction set forth in 16.10, Contractor may compensate its employees above the rate the City will reimburse, so long as the additional compensation is not charged to the Agreement budget.
- 16.11 Contractor expressly understands and agrees that in accordance with 42 U.S.C. §9848, no portion of the Agreement funds provided hereunder may be used to pay its employee if compensation (including non-federal funds) to that employee exceeds \$179,700.00, or the maximum authorized compensation as may be adjusted from time to time. Furthermore, Contractor agrees that all employees must devote to the Head Start Program the time proportionate to the percentage of their compensation funded through the Head Start program grant (e.g., employees who are one hundred percent (100%) funded through the Head Start grant must devote one hundred percent (100%) of their time and effort to support the Head Start program). Contractor agrees to submit employee certifications if requested by the City or HHS.
- 16.12 The Contractor agrees to establish internal procedures that assure employees of an established complaint and grievance process. The grievance process will include procedures to receive, investigate, and resolve complaints and grievances in an expeditious manner.
- 16.13 Contractor agrees to comply with all applicable federal regulations regarding the setting of, and maximum amount allowable for, salaries of Contractor's employees.

- 16.14 Contractor agrees to continue to work on placing written job descriptions for Head Start personnel in individual personnel folders for each position in the organization, or to have them placed online, until completed, but that all descriptions shall be filed or online no later than the expiration date of the Agreement. Contractor agrees to provide to the City specific job description(s) upon request.
- 16.15 The Contractor agrees to provide the City with the names and license registration of any employees of Contractor regulated by State law whose activities contribute towards, facilitate, or coordinate the performance of this Agreement.
- 16.16 Chief Executive Officers (CEOs), directors and other supervisory personnel of Contractor may not supervise a spouse, parents, children, brothers, sisters, and in-laws standing in the same relationship, (hereinafter referred to as "Relatives") who are involved in any capacity with program delivery supported through Agreement funds. Relatives, however, may be co-workers in the same Project in a non-supervisory position.

XVII. ADVERSARIAL PROCEEDINGS

- 17.1 Except in circumstances where the following is in conflict with federal law or regulations pertaining to the Head Start Program, Contractor agrees to comply with the following special provisions:
 - (A) Under no circumstances will the funds received under this Agreement be used, either directly or indirectly, to pay costs or attorney fees incurred in any adversarial proceeding against the City or any other public entity; and
 - (B) Contractor, at the City's option, could be ineligible for consideration to receive any future funding while any adversarial proceeding against the City remains unresolved.

XVIII. FEDERAL AND CITY-SUPPORTED PROJECT

- This Section is applicable to all publicity, public presentations, signs, press releases, public notices, flyers, 18.1 brochures, marketing materials, and other informational material prepared, created, posted and/or disseminated during the term of the Agreement by Contractor. Contractor shall obtain City's prior approval of the language and logo, as applicable, to be used, and the Parties agree that all publicity regarding the establishment or operation of the Head Start Program affiliation between City and Contractor described herein shall be planned and implemented as mutually agreed to in advance by the Parties. Contractor agrees that all public notices and any publicity, signs and/or marketing materials regarding any program which is funded by this Agreement shall provide a written statement acknowledging the role of the Federal funds provided by HHS through City hereunder, which shall read as follows: "The City of San Antonio Head Start Program services provided by the San Antonio Independent School District are funded by the City of San Antonio, Department of Human Services, through a federal grant received from the U.S. Department of Health and Human Services." These public notices or signs include, but are not limited to, signs identifying the facilities from which these programs are provided, and electronic media. In addition, all publicity related to Contractor's Head Start services shall note that the Program is operated on a nondiscriminatory basis.
- 18.2 Contractor further agrees to provide City with a copy of all proposed official communications to the public, Head Start parents, and employees as it may relate to the City's implementation of the City's Head Start Program model, the transition of Head Start contracts or transition of the Head Start Program, and to obtain the City's approval prior to dissemination.

XIX, PROPERTY, EQUIPMENT AND SUPPLIES

19.1 The City retains ownership of all equipment/property purchased with funds received through the City and such equipment/property shall, at the City's sole option, revert to the City at Agreement's termination, for

whatever reason. The Contractor agrees to relinquish and transfer possession of and, if applicable, title to said property without the requirement of a court order upon termination of this Agreement. Equipment that has reverted to the Contractor through a City-paid lease agreement with option to buy will be considered the same as though the equipment was purchased outright with Agreement funds. It is understood that the terms, "equipment" and "property", as used herein, means tangible, nonexpendable, personal property having a useful life of more than one year and an acquisition cost of \$5,000 or more per unit and shall include not only furniture and other durable property, but also vehicles, but shall not include supplies and consumables.

- 19.2 Contractor agrees that no equipment purchased with Agreement funds may be disposed of without receiving prior written approval from DHS. In cases of theft and/or loss of equipment, it is the responsibility of the Contractor to replace it with like equipment. Agreement funds cannot be used to replace equipment in those instances. All replacement equipment will be treated in the same manner as equipment purchased with Agreement funds.
- 19.3 Contractor shall maintain accurate and complete records on all equipment and property obtained with Agreement funds to include:
 - (A) A description of the equipment, including the model and serial number or other identification number, if applicable;
 - (B) The date of acquisition, cost and procurement source, purchase order number, and vendor number;
 - (C) An indication of whether the equipment is new or used;
 - (D) The vendor's name (or transferred from);
 - (E) The location of the property;
 - (F) The property number shown on the property tag; ("City of San Antonio Head Start Program"); and
 - (G) A list of disposed items and disposition
- 19.4 The Contractor is fully and solely responsible for the insuring, safeguarding, maintaining, and reporting of lost, stolen, missing, damaged, or destroyed equipment/property purchased or leased with Agreement funds. Contractor shall inform the City of incidents of loss, theft, damage or destruction of equipment (as defined in 19.1 above) or property, excluding supplies and consumables, purchased or leased with Agreement funds.
- 19.5 All equipment purchased under this Agreement shall be fully insured against fire, loss and theft. Contractor shall, at a minimum, provide the equivalent insurance for real property and equipment acquired with Agreement funds as provided to other property acquired or owned by the Contractor.
- 19.6 Upon request, Contractor shall provide an annual inventory of assets purchased with funds received through the City to DHS.
- 19.7 Contractor shall fully comply with the property and equipment requirements of 45 C.F.R Part 75, including but not limited to Sections 75.316 through 75.323, related to the following:
 - (A) Insurance Coverage
 - (B) Real Property
 - (C) Federally-owned and exempt property
 - (D) Equipment
 - (E) Supplies
 - (F) Intangible property
 - (G) Property trust relationship
- 19.8 Relative to property, equipment and supplies purchased with Head Start grant funds: For individual equipment purchases in the amount of \$5,000.00 or greater or cumulative purchases in the amount of \$100,000.00 or greater, Contractor shall obtain prior approval from DHS. Contractor shall not split the purchase of a line item with a value greater than the preceding thresholds in order to avoid obtaining approval from DHS.

- 19.9 Contractor will maintain a system for tracking, on an ongoing basis, inventory of equipment and supplies purchased with Head Start funds that either (i) has a purchase price of \$5,000.00 or greater; or (ii) meets such other criteria as City may prescribe (and which City shall notify Contractor as appropriate). Upon request, Contractor will provide City a status report of the current inventory of equipment and supplies meeting these requirements. City shall have the right to review and approve Contractor's inventory tracking system.
- 19.10 City reserves the right to require transfer of property acquired with funds awarded under this Agreement as provided in 45 C.F.R. Part 75, including but not limited to §75.316 et seq.
- 19.11 Contractor acknowledges and agrees that City is the intended third-party beneficiary of any and all facility leases with third-parties to which Contractor is or becomes a party in connection with the approved Program sites listed in Exhibit I-A to Attachment I, or as a consequence of this Agreement. As such, Contractor shall use its best efforts to cause the lessor of any such lease to execute an acknowledgment prepared by City that City is an intended third-party beneficiary of such lease. Contractor shall honor all of its material obligations under any and all such leases. Contractor shall stay in good standing under any and all such leases and Contractor shall immediately notify City in writing in the event of any breach or alleged breach of any such lease that could result in the termination of such lease. Contractor shall submit to City for review and approval all non-disturbance, subordination and similar agreements it is requested to execute in connection with any such lease. In the event of an event giving rise to a right of first refusal in favor of Contractor under any such lease, Contractor shall promptly notify City of such event and allow City to step into Contractor's shoes at tenant under such lease in order to exercise such right.

XX. TRAVEL

- 20.1 The costs associated with budgeted travel for business, either in-town or out-of-town, are allowable costs provided documentation of expenses is present.
 - (A) Contractor agrees that mileage reimbursement paid to Contractor's employees shall be reimbursed at a rate no more liberal than the City's policy for mileage reimbursement, which is consistent with Internal Revenue Service (IRS) rules. Contractor further agrees that in order for its employees to be eligible for mileage reimbursement, the employees 1) shall be required to possess a valid Texas Driver's License and liability insurance as required by law, and 2) must fully comply with its own established mileage recording policies. Mileage records are subject to spot-checks by City auditors and monitors. Contractor shall strongly encourage the participation by its employees in an approved defensive driving course. Evidence of the required driver's license and liability insurance must be kept on file with the Contractor.
 - (B) Contractor agrees that in order to obtain reimbursement of the costs associated with budgeted out of town travel for business in connection with this Agreement, Contractor shall:
 - obtain City's prior approval, unless such expense has been previously submitted and approved
 as part of Contractor's budget, and provide City with detailed documentation of such business
 travel expense(s), (any amount over approved budgeted amounts must be pre-approved by
 City or such overage will not be paid),
 - 2) ensure that any and all costs associated with out-of-town travel (including per diem rates) shall not be more liberal than the City's travel policies which conform with the reimbursement rates established by the United States General Services Administration,
 - 3) purchase all business travel at economy class rates and shall document such, and
 - 4) maintain supporting documentation for conferences to include itineraries and documentation certifying conference attendance and provide such documentation to City upon request.

XXI. NO USE OF FUNDS FOR RELIGIOUS ACTIVITIES

21.1 Contractor agrees that none of the performance rendered hereunder shall involve, and no portion of the funds received hereunder shall be used, directly or indirectly, for the construction, operations, maintenance or administration of any sectarian or religious facility or activity, nor shall said performance rendered or funds received be utilized so as to benefit, directly or indirectly, any such sectarian or religious facility or activity.

XXII. DEBARMENT

- 22.1 Contractor certifies that neither it nor its principals are presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from participation in any state or federal program.
- 22.2 Contractor shall provide immediate written notice to City, in accordance with the notice requirements of Article XXVI herein, if, at any time during the term of the Agreement, including any renewals hereof, Contractor learns that its certification was erroneous when made or have become erroneous by reason of changed circumstances.

XXIII. ASSIGNMENT

23.1 Contractor shall not assign or transfer Contractor's interest in this Agreement or any portion thereof without the written consent of the City, and if applicable, the Grantor of the Grant Funds. Any attempt to transfer, pledge or otherwise assign shall be void ab initio and shall confer no rights upon any third person or party.

XXIV. AMENDMENT

- 24.1 Any alterations, additions or deletions to the terms hereof shall be by amendment in writing executed by both City and Contractor and evidenced by passage of a subsequent City ordinance, as to City's approval; provided, however, the Director of DHS shall have the authority to execute an amendment of this Agreement without the necessity of seeking any further approval by the City Council of the City of San Antonio if permitted by all applicable local, state and federal laws, and in the following circumstances:
 - A. an increase in funding (the cause of which is unrelated to the reason set forth in Section 24.1(E) below) of this Agreement in an amount not exceeding (a) twenty-five percent (25%) of the total amount of this Agreement or (b) \$25,000.00, whichever is the lesser amount; provided, however, that the cumulative total of all amendments executed without City Council approval pursuant to this subsection and increasing Agreement funding during the term of this Agreement shall not exceed the foregoing amount;
 - B. modifications to the Scope of Work set forth in Attachment I hereto, so long as the terms of the amendment stay within the substantive parameters set forth in the original Scope of Work, also set forth in Attachment I hereto;
 - C. budget line item shifts of funds, so long as the total dollar amount of the budget set forth in Section 3.1 of this Agreement remains unchanged (these modifications may be accomplished through Budget Revisions);
 - D. modifications to the insurance provisions described in Article IX of this Agreement that receive the prior written approval of the City of San Antonio's Risk Manager and the Director of DHS.
 - E. decreases (and increases if the City agrees to allocate additional enrollment slots to Contractor) in Agreement funding based upon Head Start Program enrollment levels, and modifications to Agreement terms related to enrollment; provided, however, that the cumulative total of all Head Start Program contracts, as amended, shall not exceed the City's total budget for the Head Start Program budget for the applicable grant year. Contractor shall execute any and all amendments to this Agreement that are required as a result of a modification made pursuant to this Section 24.1(E).
- 24.2 Contractor's Superintendent shall have the authority to execute an amendment of this Agreement to the same extent as the Director of DHS under Section 24.1 of this Agreement without the necessity of seeking

- any further approval by Contractor's Board of Trustees, if not otherwise prohibited by federal or state law or regulation or prohibited by Contractor's policies, rules or Board directives.
- 24.3 Contractor further agrees that except when the terms of this Agreement expressly provide otherwise, any alterations, additions or deletions to the terms hereof shall be by amendment in writing and approved by HHS.

XXV. SUBCONTRACTING

- 25.1 None of the work or services covered by this Agreement shall be sub-contracted without the prior written consent of the City and the Grantor of the Grant Funds, if so required by said Grantor.
- 25.2 Contractor must comply with all applicable local, State and Federal procurement standards, rules, regulations and laws in all its sub-contracts related to the work or funds herein. It is further agreed by the Parties hereto that the City has the authority to monitor, audit, examine, and make copies and transcripts of all sub-contracts, as often as deemed appropriate by the City. If, in the sole determination of the City, it is found that all applicable local, State and Federal procurement standards, rules, regulations and laws have not been met by Contractor with respect to any of its sub-contracts, then the Contractor will be deemed to be in default of this Agreement, and as such, this Agreement will be subject to termination in accordance with the provisions hereof.
- 25.3 Any work or services for sub-contracting hereunder, shall be sub-contracted only by written Agreement, and unless specific waiver is granted in writing by City, shall be subject by its terms to each and every provision of this Agreement. Compliance by sub-contractors with this Agreement shall be the responsibility of Contractor. Contractor agrees that payment for services of any sub-contractor shall be submitted through Contractor, and Contractor shall be responsible for all payments to sub-contractors.
- 25.4 Contractor certifies that its subcontractors are not presently debarred, suspended or proposed for debarment, declared ineligible or voluntarily excluded from participation in any state or federal program.
- 25.5 Contractor understands and agrees that all subcontracts for performance of this Agreement in excess of \$10,000.00 must address termination for cause and for convenience, including the manner by which the termination will be effected and the basis for settlement.

XXVI. OFFICIAL COMMUNICATIONS

26.1 Except where the terms of this Agreement expressly provide otherwise, any election, notice or communication required or permitted to be given under this Agreement shall be in writing and deemed to have been duly given if and when delivered personally (with receipt acknowledged), or three (3) days after depositing same in the U.S. mail, first class, with proper postage prepaid, or upon receipt if sending the same by certified mail, return receipt requested, or upon receipt when sent by a commercial courier service (such as Federal Express or DHL Worldwide Express) for expedited delivery to be confirmed in writing by such courier, at the addresses set forth below or to such other address as either Party may from time to time designate in writing.

City:
Director
Department of Human Services
106 S. St. Mary's Street, Suite 700
San Antonio, TX 78205

Contractor: Superintendent San Antonio Independent School District (SAISD) 141 Lavaca Street San Antonio, TX 78210

Notices of changes of address by either Party must be made in writing delivered to the other Party's last known address within five (5) business days of the change.

XXVII. VENUE

27.1 Contractor and City agree that this Agreement shall be governed by and construed in accordance with the laws of the State of Texas. Any action or proceeding brought to enforce the terms of this Agreement or adjudicate any dispute arising out of this Agreement shall be bought in a court of competent jurisdiction in San Antonio, Bexar County, Texas.

XXVIII. GENDER

Words of any gender used in this Agreement shall be held and construed to include any other gender, and words in the singular number shall be held to include the plural, unless the context otherwise requires.

XXIX. REPRESENTATIONS AND OTHER OBLIGATIONS

- 29.1 The signer of this Agreement for Contractor represents, warrants, assures and guarantees that (s)he has full legal authority to execute this Agreement on behalf of Contractor and to bind Contractor to all of the terms, conditions, provisions and obligations herein contained. Whether a non-profit or public entity, Contractor must be authorized to do business in the State of Texas and be formed under and operating in accordance with all applicable laws of the State of Texas. Upon request by the City, Contractor shall provide DHS verification of the foregoing requirements.
- 29.2 This Agreement is based on the representation of Contractor that it is financially viable, solvent, and accountable for its expenditures; that it has the continuing capability to furnish the non-Federal Share of the cost of operating its approved Contractor Program; and that Program funds disbursed to Contractor will be expended only for allowable costs in the implementation of the Contractor Program which is the subject of this Agreement. Contractor represents that there are no financial limitations or impediments that would make it not viable, solvent and accountable such that the flow of Program funds might be diverted away from the operation and maintenance of the Program which is the subject of this Agreement.
- 29.3 In the event that circumstances arise which might result in interference with Contractor's ability to provide the services which are the subject of this Agreement, Contractor agrees to inform City of those circumstances immediately upon their discovery. Contractor agrees that reimbursement to Contractor, upon reasonable notice, may be suspended by City until such financial circumstances giving rise to the possible interference with the operation of the Program have been eliminated, provided, however, that authorized expenditures made prior to the suspension, and approved by City shall be disbursed pursuant to the terms of this Agreement.

XXX. LICENSES AND TRAINING

30.1 Contractor warrants and certifies that Contractor's employees and its subcontractors have the requisite training, license or certification to provide said services, and meet all competence standards promulgated by all other authoritative bodies, as applicable to the services provided herein.

XXXI. INDEPENDENT CONTRACTOR

31.1 It is expressly understood and agreed that the Contractor is and shall be deemed to be an independent contractor, responsible for its respective acts or omissions and that the City shall in no way be responsible therefor, and that neither Party hereto has authority to bind the other nor to hold out to third parties that it has the authority to bind the other.

- 31.2 Nothing contained herein shall be deemed or construed by the parties hereto or by any third party as creating the relationship of employer-employee, principal-agent, partners, joint venture, or any other similar such relationship, between the Parties hereto.
- Any and all of the employees of the Contractor, wherever located, while engaged in the performance of any work required by the City under this Agreement shall be considered employees of the Contractor only, and not of the City, and any and all claims that may arise from the Workers' Compensation Act on behalf of said employees while so engaged shall be the sole obligation and responsibility of the Contractor.

XXXII. SEVERABILITY

32.1 If any clause or provision of this Agreement is held invalid, illegal or unenforceable under present or future federal, state or local laws, including but not limited to the City Charter, City Code, or ordinances of City, then and in that event it is the intention of the Parties hereto that such invalidity, illegality or unenforceability shall not affect any other clause or provision hereof and that the remainder of this Agreement shall be construed as if such invalid, illegal or unenforceable clause or provision was never contained herein; it is also the intention of the Parties hereto that in lieu of each clause or provision of this Agreement that is invalid, illegal or unenforceable, there be added as a part of this Agreement a clause or provision as similar in terms to such invalid, illegal or unenforceable clause or provision as may be possible, legal, valid and enforceable.

XXXIII. NON-WAIVER OF PERFORMANCE

33.1 No waiver by City of a breach of any of the terms, conditions, covenants or guarantees of this Agreement shall be construed or held to be a waiver of any succeeding or preceding breach of the same or any other term, condition, covenant or guarantee herein contained. Further, any failure of City to insist in any one or more cases upon the strict performance of any of the covenants of this Agreement, or to exercise any option herein contained, shall in no event be construed as a waiver or relinquishment for the future of such covenant or option.

XXXIV. CONTRIBUTION PROHIBITIONS

- Contractor acknowledges that City Code Section 2-309 provides that any person acting as a legal signatory to a proposed contractual relationship that applies for a "high profile" discretionary contract, as defined by the City of San Antonio Contracting Policy and Process Manual, may not make a campaign contribution to any councilmember or candidate at any time from the tenth business day after the Request for Proposal (RFP) or Request for Qualifications (RFQ) or other solicitation is released, or for a contract for which no competitive solicitation has been issued by the City from the time the City begins discussions or negotiations, and ending on the 30th calendar day following the contract award. Contractor understands that if the legal signatory entering the Agreement has made such a contribution, the City may not award the Agreement to that contributor or to that contributor's business entity. Any legal signatory for a proposed high-risk contract must be identified within the response to the RFP or RFQ, if the identity of the signatory will be different from the individual submitting the response.
- 34.2 Contractor acknowledges that the City has identified this Agreement as high profile.
- 34.3 Contractor warrants and certifies, and this Agreement is made in reliance thereon, that the individual signing this Agreement has not made any contributions in violation of City Code section 2-309, and will not do so for 30 calendar days following the award of this Agreement. Should the signor of this Agreement violate this provision, the City Council may, in its discretion, declare the Agreement void.

XXXV. ENTIRE AGREEMENT

35.1	This Agreement and its attachments, if any, cons Parties hereto and contain all of the terms and cond representations, or agreements, either oral or writte enforced by either Party nor may they be employed Agreement.	ditions agreed upon, and sup en. No such other negotiation	ersede all prior negotiations, ns or representations may be
	ness of which this Agreement has been executed the _ve for the term hereof.	day of	and is
сіту о	F SAN ANTONIO:	CONTRACTOR: San Antonio Independent	School District (SAISD)
	Woosley, Director ent of Human Services	Pedro Martinez, Superior	téndent
B	VED AS TO FORM:	Board President (if requi	a Corrector)

ATTACHMENTS

Attachment I – Scope of Work

Attachment II – Program Budget

Attachment III – Special Provisions

Attachment IV – HHS Award Document

Attachment V – HIPAA Business Associate Agreement, if applicable

Attachment VI – Additional OMB Provisions

ATTACHMENT I [INSERT SCOPE OF WORK HERE]

SCOPE OF WORK

Summary

The Contractor shall ensure full enrollment and shall serve the number of income and age eligible children as indicated below, and perform Head Start services in accordance with the Head Start Performance Standards as provided in Head Start Regulations, 45 CFR Part 1301 et seq with the Head Start Act, as amended, 42 U.S.C. 9831 et seq and with the terms of this Contract. The Contractor will operate full day classroom(s) at the City approved sites for a minimum of 177 days of planned class operations for the Head Start grant period.

Number of children in full day care on the first day of the program:	2,243
Minimum number of children with disabilities:	224
Service Area:	Children who reside in San Antonio and are served in the San Antonio Independent School District

2. Enrollment, Recruitment, Selection, Eligibility and Attendance (ERSEA)

- A. Contractor shall have primary responsibility for the verification of eligibility, recruitment, enrollment, selection of children, and attendance, in accordance with the ERSEA policies established by the City. Contractor shall work jointly with Family and Community Support Service staff conducting recruitment activities, receiving applications, determining eligibility, and completing enrollment forms for Contractor's service area. Contractor shall ensure that it is in compliance with the eligibility, recruitment, selection, enrollment and attendance provisions of the Head Start Act, the Head Start Performance Standards and the City's program policies and other federal guidance related to the Head Start program.
- B. Contractor shall establish and maintain full funded enrollment of 2,243 Head Start eligible children in the Contractor's Program from the first day of operations. Funded enrollment means the number of children which the Contractor is to serve, as indicated by the City and terms of this Contract. To maintain full enrollment, Contractor must ensure that once it determines that a vacancy exists, no more than 30 calendar days may elapse before the vacancy is filled. Contractor will document for the City its efforts to fill all vacancies within 30 calendar days.
- C. If full funded enrollment is not maintained for a period of 30 days the City may reduce Contractor's funding at the established per capita rate based on the difference between funded enrollment slots and actual enrollment. Contractor shall develop at the beginning of each enrollment year and maintain during the year a waiting list that ranks children according to the program's selection criteria and ERSEA policies. Children on this waiting list may include children who are currently enrolled in the State Pre-Kindergarten program. If Contractor is under enrolled and does not have a waiting list, Contractor will fill any vacancies in the federal Head Start classrooms, and in doing so, may recruit from the state Pre-K classes. City shall have the right to place additional children in Contractor's Program, in coordination with the Contractor, up to the point where Contractor is fully enrolled.
- D. Contractor shall develop a plan that includes recruitment, selection and enrollment services in accordance with City's established ERSEA policies and, in a manner that will promote children receiving two (2) years of Head Start services or more if appropriate. Contractor shall submit to the City its August Program Design that establishes intact and separate 3-year old and 4-year old classrooms prior to the first day of school. If there is a change in the Program Design at any point during the school year, Contractor shall notify the designated Head Start staff in writing within 2 business days that such change was made. If Contractor is unable to implement the required Program Design, Contractor shall inform the City and receive City approval before establishing an alternative Program Design.

- E. In accordance with the Head Start Act, Contractor shall ensure that a minimum of 10% of the children enrolled in Contractor's Program are children with disabilities who are determined to be eligible for special education and related services. Contractor must meet this 10% requirement by mid-point of the program year, as determined by the City.
 - (i) If Contractor recognizes that it is having difficulty in meeting the percentage with disabilities requirement during its enrollment, recruitment and selection period, Contractor must provide the City with the following:
 - a. a written description of specific steps the Contractor has taken to meet the requirement in the current program year to include efforts made to collaborate with the local agencies providing services under Section 619 and Part C of IDEA, and an explanation of why, despite these efforts, it was unable to meet the 10% requirement
 - a written confirmation from the local IDEA agencies of Contractor's efforts to actively collaborate with them to promote the enrollment of children with disabilities and specific efforts to recruit and enroll eligible children with disabilities; and
 - c. a written description of the Contractor's proposed approach to improve the enrollment of children with disabilities to reach the 10% requirement to include a description of how the IDEA agencies in the community will work with the Contractor to implement this approach
 - (ii) Additionally, if Contractor is unable to meet the 10% requirement, the City shall have the right to place additional children with disabilities in Contractor's Program up to the point where Contractor achieves a 10% enrollment of children with disabilities. City and Contractor shall coordinate placement of additional children with disabilities.
- F. Contractor agrees that when the monthly average daily attendance rate in a center-based program falls below 85 percent, Contractor shall initiate and collaborate with the City in the provision of absentee services in accordance with §1305.8 of the Head Start Performance standards. Absentee services include analyzing the cause of absenteeism and the provision of additional support, which may include, but is not limited to, face-to-face meeting(s) with the family and other direct contact with the child's parents/legal guardian. Contractor shall establish procedures for the removal of children from the program due to the lack of attendance in accordance with the City's policies.

3. Program Services

- A. Contractor shall provide Head Start Program Services (hereinafter defined) to eligible children in the Service Area. Such services shall be provided to eligible children within Contractor's Service Area(s) without regard to age, race, creed, color, sex, or national origin. Any proposal to extend or modify the Service Area(s) or the sites at which services are to be delivered shall be formally submitted in writing to City for approval.
- B. Contractor shall provide (i) Education and Early Childhood Development Services; (ii) Child Disability Services; and (iii) Child Nutrition Services and shall collaborate with City to ensure the provision of Family and Community Support Services, Child Mental Health, Medical Health Services and Dental Health Services, to meet the needs of the children and families served by Contractor's Program ("Head Start Program Services"). Contractor agrees that, notwithstanding the fact that another Head Start service provider under the City of San Antonio Head Start program may be contracted to provide a category of service, Contractor, under the leadership of its Head Start Program Director will be responsible for coordinating with Head Start service providers and working with the City to ensure provision of full array of services to which the children are entitled under the Terms of the Grant. Contractor shall also coordinate with other Independent School Districts, health service providers and agencies in the community in order to provide comprehensive services to the children and families served by the Program and to leverage community resources.

- C. Contractor shall establish and maintain an organizational structure that supports the accomplishment of Program objectives, addresses the major functions and responsibilities assigned to each staff position and provides evidence of adequate mechanisms for staff supervision and support to ensure the effective oversight of the Head Start program operations. Contractor shall ensure that, at a minimum, the following program management functions are assigned to and adopted by staff within the Program:
 - i. Program management must be assigned to an individual serving in the role of the Head Start Director. This individual must have demonstrated skills and abilities in a management capacity relevant to human services program management and must have ultimate responsibility for ensuring the children enrolled in Contractor's Head Start program are provided (by Contractor or a collaborating Head Start service provider) the full array of services to which the children are entitled under the Terms of the Grant.
 - ii. Management of education and early childhood development must be assigned to an individual serving in the role of an Education and Early Childhood Development Coordinator. This individual and any additional supporting staff or consultants must have training and experience in areas that include theories and principles of child growth and development, early childhood education, and family support. Staff and consultants must meet the qualifications for classroom teachers, as specified in the Head Start Act, Section 648A, and any subsequent amendments:
 - iii. Management of health services, including medical and dental health, must be assigned to an individual serving in the role of a Health Services Coordinator. The City recommends that this individual and any additional supporting staff or consultants have training and experience in public health, nursing, health education, maternal and child health, or health administration. Contractor must also coordinate with health service providers and follow established protocols to facilitate health care providers' ability to ensure children receive all the required Head Start health services.
 - iv. Management of child nutrition services must be assigned to an individual serving in the role of a Nutrition Services Coordinator. This individual and any supporting staff or consultants must be supported by staff or consultants who are registered dietitians or nutritionists. This person must also coordinate with City and City's nutrition services provider(s).
 - v. Coordination with mental health services provider(s) must be assigned to an individual serving in the role of an Education and Early Childhood Development Coordinator, Health Services Coordinator or Disabilities Coordinator. The City recommends that this individual have experience in serving children and families with mental health needs. This person must also coordinate with City and City's mental health services provider(s).
 - vi. Coordination with family and community support services staff, including parent activities, must be assigned to an individual serving in the role of a Family and Community Support Services Liaison. The City recommends that this individual and any supporting staff or consultants have training and experience in field(s) related to social, human, or family services, and shall coordinate with City. It is also recommended that parent involvement services be supported by staff or consultants with training, experience, and skills in assisting the parents of young children in advocating and decision-making for families.
 - vii. Management of disabilities services must be assigned to an individual serving in the role of Disabilities Coordinator. This individual and any supporting staff or consultants must have training and experience in securing and individualizing needed services for children with disabilities, and shall coordinate with City.
 - viii. Management of ERSEA responsibilities must be assigned to an individual serving in the role of ERSEA Coordinator. This individual and any supporting staff or consultants must have knowledge, training, and experience in the enrollment, recruitment, selection, eligibility and attendance provisions of the Head Start Act, the Head Start Performance Standards; and the City's program policies and other federal guidance related to the Head Start program.
 - ix. Management of monitoring responsibilities must be assigned to an individual serving as the Monitoring Coordinator. This individual and any supporting staff or consultants must have knowledge, training and experience with the Head Start Performance Standards, the Head Start Act, and monitoring of the Head Start Program. Contractor shall ensure these individuals further develop their knowledge about the various Head Start content areas.

Contractor shall ensure that members of program's management team and any other necessary staff provide uninterrupted Head Start Program management services throughout the Agreement term, including during the summer months so as to ensure adequate planning, coordination and performance of critical program activities. Critical program activities include, but are not limited to ongoing recruitment activities and services, determination of eligibility, development of a waitlist for the upcoming school year, and completion of enrollment.

- D. Contractor shall provide the following education and early childhood development services in compliance with the requirements of 45 C.F.R Part 1304.21, et. seq.:
 - i. Contractor must help children gain the skills and confidence necessary to be prepared to succeed in their present environment and with later responsibilities in school and life; the Contractor's approach to child development and education must be developmentally and linguistically appropriate, recognize the child's rate of development, language, cultural background and learning style; be inclusive of children with disabilities; provide an environment of acceptance that supports and respects gender, culture, language, ethnicity and family composition; provide a balanced daily program of child-initiated and adult-directed activities; and allow and enable children to independently use toilet facilities when it is developmentally appropriate and the efforts are supported by the parents.
 - ii. Contractor shall encourage and support parents in their efforts to become involved in the development of the program's curriculum and approach to child development and education; provide opportunities to increase parents' child observation skills and to share assessments with staff that will help plan the learning experience; and encourage parents to participate in staff-parent conferences and home visits to discuss their child's development and education.
 - iii. Contractor must support social and emotional development by encouraging development which enhances child's strengths by building trust; fostering independence; encouraging self-control and respect for feelings and rights of others; supporting and respecting the home language, culture, and family composition in ways that support the child's health and wellbeing; and planning for routines and transitions.
 - iv. Contractor must provide for the development of each child's cognitive and language skills by supporting each child's learning, using various strategies including experimentation, inquiry, observation, play and exploration; ensuring opportunities for creative self-expression through activities such as art, music, movement, and dialogue; promoting interaction and language use among children and between children and adults; and supporting emerging literacy and numeracy developments through materials and activities according to the child's developmental level.
 - v. Contractor must promote each child's physical development by providing sufficient time, indoor and outdoor space, equipment, materials and adult guidance for active play and movement that support the development of gross motor skills and of fine motor skills according to the child's developmental level; and providing an appropriate environment and adult guidance for the participation of children with special needs.
 - vi. Contractor, in collaboration with the parents, must implement a researched-based curriculum that aligns with the Head Start Early Learning Outcomes Framework (HSELOF, 2015), and state Pre-kindergarten Guidelines that supports each child's individual pattern of development and learning; provides for the development of cognitive skills; integrates all educational aspects of the health, nutrition, and mental health services into program activities; helps children develop emotional security, facility in social relationships, feelings of competence, self-esteem, and positive attitudes toward learning; enhances each child's understanding of self as an individual and as a member of a group; and provides individual and small group experiences both indoors and outdoors.
 - vii. Contractor must implement the plan of action to achieve school readiness goals that are age appropriate and will address at a minimum the domains of language, literacy, cognition, motor and physical well-being, social and emotional development and approaches to learning, and achieve the integration with the Parent, Family, and Community Engagement Framework. Contractor will participate in updating school readiness goals on an annual basis, analyze achievement, and identify areas for improvement. As part of this plan of action, the Contractor must align the program's curriculum with school readiness goals to include improving the quality of teacher-child interactions,

- evidence-based teaching practices and curriculum and inclusion of parents to support school readiness.
- viii. Contractor will assess child progress on an ongoing basis and conduct data aggregation and submit analysis of child progress assessments to the City at least three times per year (at the beginning, midpoint and end of school year) to demonstrate program-level progress. Contractor will also provide the City direct access to the assessment data management system for the purposes of aggregating, analyzing, and producing program-wide reports. Contractor will make mid-year adjustments in instruction and/or professional development support should patterns or trends be identified in the analysis and coordinate with City as needed. Contractor must maintain and use the information from ongoing child assessment data to inform teachers and survey parents how best to individualize each child's learning and progress across domains.
- ix. Starting with the 2017-2018 school year, Contractor shall have in place a formal, written teacher coaching plan that is research or practice-based to address professional development of Head Start teachers. Once such a plan is created or if such a plan currently exists, the Contractor shall provide a written plan to the City by September 2017.
- E. Contractor shall perform the following services as it relates to the specific service listed below in compliance with the requirements of 45 C.F.R Parts 1304.20, 1304.22, 1304.23, 1304.24 and 1304.40:
 - i. Child Health and Developmental Services
 - a. Contractor shall be responsible for performing the following no later than 90 calendar days from the child's entry into the program: (1) make a determination as to whether or not each child has an ongoing source of continuous, accessible health care. If the child does not, Contractor must assist the parents in accessing a source of care; and (2) obtain from a health care professional a determination as to whether the child is up-to-date on a schedule of age appropriate preventive and primary health care, which includes medical, dental and mental health. This schedule must incorporate the requirements for the schedule utilized by the Early and Periodic Screening, Diagnosis, and Treatment (EPSDT) program of the Medicaid agency of the State, and the latest immunization recommendations issued by the Centers for Disease Control and Prevention, as well as any additional recommendations from the local Health Services Advisory Committee. Contractor shall collaborate with health service providers and parents to facilitate and expedite the provision of these services.
 - b. Contractor shall collaborate with the Head Start health service providers and Family Community support staff to develop and implement procedures for ongoing care to identify any new or recurring medical, dental, or developmental concerns so that it can quickly make the appropriate referrals.
 - c. Contractor, in collaboration with the parents, and within 45 calendar days of the child's entry into the program, must perform or obtain linguistically and age appropriate screening procedures to identify concerns regarding the child's developmental, sensory (visual and auditory), behavioral, motor, language, social, cognitive, perceptual, and emotional skills. The annual behavioral and developmental screenings of all three (3) and four (4) year old children must be performed in accordance with Texas Health Steps' EPSDT.
 - d. Contractor must use the information from the screening for developmental, sensory, and behavioral concerns, and ongoing observations, medical and dental evaluation and treatments, and insights from the child's parents to help staff and parents determine how the program can best respond to each child's individual characteristics, strengths and needs.

ii. Child Health and Safety

a. Contractor must establish and implement policies and procedures to respond to medical and dental health emergencies with which all staff are familiar and trained, to include posted policies and plans of action for emergencies that require rapid response from staff; posted locations and telephone numbers of emergency response systems and up-to-date family contact information and authorization for emergency care; posted emergency evacuation routes and other safety procedures for emergencies; methods of notifying parents in the event of an emergency; and established methods for handling cases of suspected or known child abuse and neglect.

- b. Contractor must temporarily exclude a child with a short-term injury or an acute or short-term injury or an acute or short-term contagious illness, that cannot be readily accommodated, from program participation in center-based activities or group experiences, but only for that generally short-term period when keeping the child in care poses a significant risk to the health and safety of the child or anyone in contact with the child.
- c. Contractor must not deny admission to any child, nor exclude any enrolled child from program participation for a long-term period, solely on the basis of his or her health care needs or medication requirements, unless the child poses a significant risk to the health and safety of the child or anyone in contact with the child and the risk cannot be eliminated or reduced to an acceptable level through reasonable modifications in the Contractor's policies or by providing appropriate auxiliary aids.
- d. Contractor must establish and maintain written procedures regarding the administration, handling, and storage of medication for every child.
- e. Contractor must ensure staff and volunteers can demonstrate safety practices; foster safety awareness among children and parents by incorporating it into child and parent activities; and ensure staff, volunteers and children follow the appropriate hygiene requirements.
- Contractor must have and maintain well-supplied first aid kits, appropriate for all ages served and the program size, at each facility and available on outings away from the site.

iii. Child Nutrition

- a. Contractor must work with families to identify each child's nutritional needs, taking into account staff and family discussions regarding any relevant nutrition-related assessment data; information about family eating patterns, including cultural preference, special dietary requirements, and feeding requirements of each child with disabilities; and information about major community nutritional issues.
- b. Contractor must design and implement a nutrition program that meets the nutritional needs and feeding requirements of each child, and takes into account the length of the program day.
- c. Contractor must ensure that nutritional services in center-based settings contribute to the development and socialization of enrolled children and ensure that at all meals are served family style.
- d. Contractor must use funds from USDA Food and Consumer Services Child Nutrition Programs as the primary source of payment for meal services.
- Contractor shall participate in the USDA School Lunch or Child and Adult Care Food Program.
 - Pursuant to City policies, Contractor must report amount of meals and snacks served to Head Start children on a monthly basis to City.

iv. Child Mental Health

- a. Contractor must work collaboratively with parents to solicit information, observation, and concerns about their child's mental health; share staff observations of their child and discuss with parents their child's behavior and development.
- b. Contractor must secure or refer, when appropriate, the services of mental health professionals on a schedule of sufficient frequency to enable the timely and effective identification of and intervention in family and staff concerns about a child's mental health. Contractor may work with a variety of mental health agencies subject to compliance with Section 38.010 of the Texas Education Code when applicable. Contractor may utilize other community mental health resources, as needed.

v. Family and Community Support

- a. Contractor, in collaboration with family and community support staff, must provide parent involvement and education activities that are responsive to the ongoing and expressed needs of the parents.
- b. Contractor, in collaboration with family and community support staff, must provide opportunities to include parents in the development of the program's curriculum and approach to child development and education.

- c. Contractor shall collaborate with the City, Head Start service providers and other community agencies to provide opportunities to enhance parenting skills, knowledge and understanding of the educational and developmental needs and activities of their children.
- d. Contractor must conduct two (2) home visits, unless parents expressly forbid such visits, and at least two (2) staff-parent conferences per child per program year to enhance the knowledge and understanding of the educational and developmental progress and activities of children in the program. Contractor must not require that parents permit home visits as a condition of the child's participation in the program.
- e. Contractor, in collaboration with family and community support staff, must establish and maintain procedures to support successful transitions for enrolled children and families from previous child care programs into Head Start and from Head Start to elementary school; and assist parents in becoming their children's advocates as they transition.
- F. Contractor shall encourage parent participation and attendance in center parent meetings, Parent Connection Committee meetings, Head Start Policy Council, volunteer events, and other parent activities.
- G. Contractor shall participate in developing an annual program School Readiness Plan of Action according to the timeline and requirements provided by the City. Such School Readiness Plan must be developed in compliance with Program Instruction ACF-PS-11-04 and ACF-IM-HS-11-06. Each Provider shall maintain a School Readiness Leadership Team that includes, at a minimum, teachers, site administrators, parents, and Education and Family Support staff. The School Readiness Leadership Teams meet no less than twice per year to review and discuss outcomes as applicable to the defined school readiness goals.
- H. The Contractor shall submit to City all Contractor eligibility, recruitment, selection, enrollment, attendance, performance and staffing plans, which must be in accordance with the City's policies.
- I. Contractor shall provide a substantial portion of the substantive programmatic work provided for under this Scope of Work, Attachment I, directly. However, should Contractor desire to perform this work through one or more sub awards to third parties, it shall seek and obtain City's written approval prior to doing so. In all such sub awards, Contractor shall ensure that it passes through all of the requirements that apply to Contractor hereunder to the sub recipient of such sub awarded funds.
- J. Contractor shall establish an ongoing monitoring system and conduct internal monitoring of operations throughout the program year, notifying the City when the Contractor identifies possible or actual lack of compliance with the Head Start Performance Standards, Head Start Act, City's program policies or terms of this Contract. Contractor's monitoring system shall be a Contractor-developed monitoring instrument or checklist prepared no less than quarterly, and must include monitoring of all fiscal matters relating to the Head Start Program and this Agreement. Contractor will make copies of monitoring reports available to the City upon request.

4. Program Governance

- A. Contractor's Governing Board shall be in full compliance with Head Start requirements regarding governance, management and programmatic operations applicable to recipients of Head Start grant funds, including those set forth at 45 C.F.R. §1301 et. seq. Contractor's Governing Board members or representative of the Board shall be offered the opportunity to participate in Board education activities arranged by City. Contractor shall also offer the Governing Board members or representative of the Board the opportunity to engage in a cooperative strategic planning process with respect to the Program and shall submit any final strategic plans developed through such process to City for approval.
- B. Contractor shall assure that City is kept fully apprised of the composition and actions of Contractor's Governing Board to the extent such actions affect Contractor's Head Start Program.
- C. Contractor shall seek and obtain the City's written approval before making any material revisions in Contractor's Head Start Program that conflicts with or violates (i) the City's refunding application, as

amended, to the U.S. Department of Health and Human Services (HHS), (ii) the Terms of the Grant, (iii) the terms of this Contract; or (iv) any changes to the locations of Contractor's Head Start centers.

D. Contractor shall collaborate with the City and Family and Community Support service staff to elect one or more parent members from center locations to serve on City's Head Start Policy Council in accordance with the Policy Council By-laws.

5. Financial and Program Reports

Additionally, Contractor shall maintain and furnish to City the following financial and programmatic information and reports, in such forms as the City may prescribe, and any others as may be required pursuant to the Head Start Act, as amended, and 45 C.F.R. Part 75. If a following information or report due date falls on a weekend or holiday, then it shall be due by the following business day.

- A. End of the Month Enrollment (EOM) Report containing the information for the preceding month, submitted upon request;
- B. Classroom Attendance Report, submitted by the 3rd business day of every month;
- Classroom/Site Set up Information Report (Program Design) submitted at the beginning of each program year or every time there are changes to the classroom/site set up;
- D. Disability Services Monthly Report, submitted by the 15th day of every month;
- Head Start Service Provider total amount of United States Department of Agriculture (USDA) reimbursement received for Head Start children and staff shall be submitted on a monthly basis;
- F. The total amount of public and private funds received by Contractor and the amount from each source;
- G. Financial reports showing all actual and/or projected costs of the Program, an explanation of budgetary expenditures, Program Income, non-Federal Share amounts;
- H. The results of the most recent financial audit involving the Head Start Program;
- Reports showing employee credentials and a list of personnel serving to satisfy Contractor's inkind non-Federal Share requirement;
- Reports showing the wages of each employee;
- K. Transition Activity Report, submitted by the 5th business day of every month;
- L. Behavioral Services Activity Report, by the 5th business day of every month;
- M. Student Assessment Data Analysis Report, as periodically required by the City;
- N. Recruitment Activity Report, by the 15th of every month:
- O. Parent Engagement Activity Report, by the 10th of every month; but should the 10th fall on a weekend or holiday, the report shall be due the preceding business day;
- P. Health Activities Report due by the due by the 10th of every month:
- Q. Training and Technical Assistance Report, by the 5th business day of every month;

- R. School Readiness Plan of Action update, due bi-annually by the 5th business days of December and June;
- S. Child Development Outcomes Progress Report based on Contractor's assessment tool, submitted as directed or when requested by City;
- T. Academic Progress Data, by June following each school year, and in a form and scope mutually agreed upon by the parties, on the academic performance of Head Start children who will remain in Contractor's Program, with the understanding that Contractor may utilize its own data assessment tool to evaluate the student progress; and
- U. Any other information reasonably requested by City.

6. Licensure/Staffing

- A. Contractor shall obtain and maintain all necessary and appropriate State licenses, permits, certifications, and approvals required for the operation of Contractor's facilities and programs including those supported by this Contract, unless exempt from such requirement. Upon commencement of the Contract, Contractor shall notify the City that it is in compliance with this provision. If at any time Contractor is out of compliance with this provision, Contractor shall notify the City within one (1) Contractor business day of receipt of written notice of violation or complaint from the state licensing, certifying or permit-issuing authority indicating lack of licensure, permitting or certification, as the case may be, and shall take all necessary steps to cure such violation. Contractor further agrees that all personnel, either employed or contracted, assigned by Contractor to perform the Head Start Program Services set forth above shall, as appropriate or required by law, be fully qualified and authorized under applicable law, to perform such Head Start Program Services.
- B. Contractor will ensure that staffing will comply with the Head Start Performance Standards, this Agreement, and other applicable law.

7. Facilities.

All Contractor facilities at the approved sites shall meet applicable federal, state and local safety standards. Contractor shall at all times during the term of this Contract, keep the facility, including play grounds, campus and classroom furniture and equipment, kitchens, restrooms and other areas within the building in good, clean, safe, and reasonable operating condition and repair, and in compliance with Head Start standards. If facilities are found to be out of compliance with Head Start Health and Safety standards, then Contractor shall take corrective action in accordance with subsection B. City shall be allowed to inspect the facilities and the licensing and/or certification documents relating to the facility during Contractor's normal operating time as long as the said inspection does not disrupt the delivery of Head Start Program Services, or at any other time as agreed by the Contractor. Upon commencement of this Agreement and upon request, Contractor shall provide City an upto-date copy of any leases or other agreements (as well as any amendments or modifications thereto) for facilities used to provide services to children enrolled in the Head Start program and funded pursuant to this Agreement.

- A. In the case of non compliance with the Head Start Health and Safety standards, the City will notify the contractor via a monitoring report with photographs when possible.
- B. When any work order request is submitted for necessary internal repairs at Head Start Centers, Contractor shall address such problem within 10 Contractor business days from the submission of the work order or from the time the need for repair is brought to the attention of Contractor whichever is earlier. For larger and/or more expensive projects, the Contractor Head Start Director shall notify City in writing of the timeline for projected completion as soon as reasonably practicable.

C. City will follow-up on all Health and Safety non-compliances immediately following the 10 days for corrective actions to be completed and submit a follow-up report stating either completed or not completed.

8. Participation.

Contractor shall make time and resources available to support: (i) participation by Contractor in meetings with City staff for community assessment, self-assessment, strategic planning, development of training and technical assistance plan, communication and program development activities; (ii) participation in technical assistance trainings and service enhancements developed by City and the Head Start training and technical assistance service provider, as well as other Head Start trainings that may be developed by relevant federal or state agencies; and (iii) an appropriate level of attendance of Contractor's program management team and parent committee members at national, regional and/or State Head Start conferences/trainings.

Contractor, in collaboration with Family and Community Support Service staff, shall ensure that resources are made available through reasonable accommodations to low-income parents to participate, and attend any Parent Connection Committee meetings.

9. Transportation Services/Vehicles.

Contractor shall provide transportation services in compliance with the requirements of 45 CFR part 1310, et. seq., subject to any waiver that may be obtained by City and/or Contractor.

- A. Contractor shall successfully complete an annual inspection of all school buses and other vehicles used for the transportation of children under this Contract.
- B. Contractor shall ensure that all drivers shall be certified for the operation of said vehicles, and that all bus monitors assigned to vehicles used to provide such services receive the appropriate training. Contractor shall otherwise be in compliance with all applicable federal, State, and local requirements governing the transportation of children.
- C. Contractor shall, upon request, submit to City a complete set of such documents regarding the vehicles used for the transportation of children and the drivers operating the vehicles.

10. Submission of Center Information and Program Design

- A. Prior to execution of the Contract, Contractor shall submit to the City for its approval a Center list, which shall include the number of centers and the name and address for each center operated by Contractor. If at any time Contractor proposes a change to the Center list, Contractor shall notify and receive the City's approval prior to making said change.
- B. Prior to execution of the Contract, Contractor will submit to the City the Contractor's Program Design, which shall include, the total number of children enrolled, number of sites, number of classrooms, classroom age group designation, language designation, and hours of operation, among other things. If at any time Contractor proposes a change to the Program Design, Contractor shall request and receive the City's approval prior to making said change.

ATTACHMENT II [INSERT PROGRAM BUDGET HERE]

City of San Antonio Head Start Program San Antonio ISD

Provider: Term:

February 1, 2017 to January 31, 2018

Category	Federal Share	Non Federal Share	Total Budget	Federal Admin	Non Federal Admin	Total Admin Budget	
Salaries	9,107,529	3,013,311	12,120,840	632,643	439,384	1,072,027	
Fringes	2,813,716	0	2,813,716	210,809	0	210,809	
Travel	2,000	0	2,000	2,000	0	2,000	
Equipment	0	0	0	0	0	0	
Supplies	122,000	0	122,000	0	0	0	
Contractual	0	0	0	0	0	0	
Other	8,000	0	8,000	0	0	0	
stal Budget /	12,053,245	3,013,311	15,066,556	845,452	439,384	1,284,836	
Zied	Salu	8/22/10	0				
Signature and Da	to of Agency Authorized Represe	ntative					

Program and Fiscal Review and Approval:

Program Monitor Signature and Date

COSA Head Start Administrator Signature and Date

Fiscal Monitor Signature and Date

Fiscal Manager Signature and Date

B	C	D	E	F-	G	Н	Tien	J	K	L	M	N
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1						ExFxG		HxJ	Federal enter under app	From J,	Non Federal/Mat	
Calegory	Contractor GL	Job Title or Description	# of Positions	Salary/Wage per Pay Period	# of Pay Periods	Total Annual Salary	% Allocated to Head Start	BUDGET	Direct	Admin	Direct	Admin
Salaries	205-11-6119-0	Teacher- BA Degree	100	4,445	12	5,334,000	100%	5,334,000	5,334,000		1 1 1 1 1 1 1	
alaries		Instructional Assistants	115	1,433	12	1,977,540	100%	1,977,540	1,977,540	4	200.20	16 474 , 244
Salaries		Head Start Senior Director	1	9,365	12	112,380	75%	84,285		84,285		4 1747
Salaries	205-21-6119-3	HS Program Design Coordinator	2	5,714	12	137,138	100%	137,136		137,138	Section of the sectio	
Salaries	205-21-6119-3	HS Compliance Monitors	5	4,377	12	262,620	100%	282,620		262,620	1.00	
Salarles	205-21-6119-3	HS Education Specialist Coor II	2	5,164	12	123,936	75%	92,952	92,952		1 - 1	(\$15.90) etc. 1
Salaries	205-21-6119-3	HS Disability Coordinator	1	5,625	12	67,500	100%	87,500	87,500		1	Tige the Title
Salarles		HS Behavior Mgrnt Specialist	2	5,081	12	121,944	100%	121,944	121,944		W. Z. T. C. (A. L.)	SOUTH STREET
Salaries		1HS Nutritionist Coordinator	1	4,859	12	58,308	100%	58,308	58,308		DAME NAME:	24.4
Selaries		Admin Asst. III - Admin	1	3,214	12	38,568	75%	28,926		28,928	12 14 36 11 12 11 11 11 11 11 11 11 11 11 11 11	Transfer of the
Salarles	205-21-8129-3	Administrative Asst. III - Dept	1	3,114	12	37,368	50%	18,684		18,884	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	* 560 T. T.
Salarles	205-21-6119-3	HS Health/Transportation Coordinator	1	4,687	12	56,004	100%	56,004	56,004		10 NY 1071	H_54. //
Salaries	205-81-8119-3	7 HS Family Services Coor.	1	5,344	12	64,128	75%	48,096	48,096		\$436	A CONTRACT
Salaries		GHS Parent Facilitators	2	1,894	12	45,458	75%	34,092	34,092		adding the	State of the
Salaries		(Administrative Asst, I Data Clerk)	2	2,452	12	58,848	75%	44,136		44,138		AND STATE
Salaries		(Admin. Asst, I (Health Data Clerk)	1	2,459	12	29,508	100%	29,508		29,508		
Salaries	205-41-8119-0	ERSEA Coordinator	1	4,558	12	54,696	100%	54,696		27,348	1000 1000	Sugar State S
Salaries		Registered Nurses	4	4,388	12	210,624	100%	210,624			TO DESCRIPTION	CALCULA
Salaries		04HS Custodians	9	945	26	221,130	100%	221,130			POPULACE.	der grent det.
Salaries	205-51-6121-0		1	17,129	12	205,548	100%	205,548			ALTONOMY TO	Safarving?
Salaries	205-11/33-611	ZExtra Duty	1	750	12	9,000	100%	9,000	9,000		**************************************	A Shipping of
Salaries	205-21-6124-0	3 Part-Time Administrative Support (2)	2	450	12	10,800	100%	10,800	10,800		Strain Control	Oktobrandiken.
Salaries	199-23-6119-0	Of Center Principals	4	8,574	11	289,256	100%	289,256			144,628	144,628
Salaries	199-23-6129-0	1 Admin. Asst II - Center Secretary	4	2,534	11	111,498	100%	111,496			· · · · · · · · · · · · · · · · · · ·	111,498
Salaries	199-23-6129-0	Admin. Asst II - Center Data Clerk	4	2,700	11	118,800	100%	118,800		1.	10 to 400 to 000 to 10	118,800
Salaries	199-23-6129-3	38 Admin. Asst I - Center Office Clerk	4	1,485	11	64,460	100%	84,460			Auto angerbas	64,460
Salaries	199-51-6129-6	Of Center Head Custodian	4	2,616	11	115,104	100%	115,104			115,104	1556146487811
Salarles	183-11-8119-	Of Bilingual Teachers	24	4,559	9	984,744		984,744			984,744	William Shield
Salaries	164-11-6129-	Of Teacher Assistants (Floaters)	55	1,680	10	924,000	100%	924,000			924,000	CERTIFICACES.
Salaries		OCPPCD Teachers	6	4,795	10	239,750		239,750			239,750	Protein Beller
Salaries		00PPCD Teacher Assistants	10	1,857	10	165,700	100%	165,700			185,700	TOP TO GET AND A
Salaries						0					Visited Visit	NEGOTORS 1
Salaries						0		1			L. Pressymm	THE WAS A
Salaries						0					PENSEMBER	2-10年間の
Salaries						0		(1000 1 P. P. St. Co. Land	Contract Contract
Salaries	1			1		0		(the state of the	THE PROPERTY.
Salaries						- O		(71, 25(b) (12)	A. 14 MARCO
Salaries					-	0				1	The division of	NATIONAL SECTION
Salaries		The second secon				0			-		17.50 14.43	15444112450
Salaries	1		1			0		1			NO WELLIAM	**********
Salaries	*					1 0		1		-	Tagair ide	THE PROPERTY OF
Salaries					-	1 0					The Walles	THAT HAND IN
Salaries	-			1	-	0					A. Armendan	12 1 1 1 1 1 1 1 1 1
Salaries		-	1	-	-	1 0					-5 (8) Miles	- १९५ स्ट्रा
Salaries	-		-	1	-	1 0					77 909656	
Salaries	-		1			- 0			0		+12: 2:00 (4:01)	200.000
Salaries	-			1	-	1					# # # # # # # # # # # # # # # # # # #	-1-1-1-1-1-1
	+		-	-	-	1		-		A Company of the Comp	TOTAL STATE	1 1 1 1 1 1 1 1
Salaries	-		1	-		- 0			0 0		11-1-7-7-7-7	1 vis. 1
Salaries	-		-	-		- 0					1111	4, 14, 14, 14, 14
Salaries	-		-		-							1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1
Salaries	-		-	-					0 0		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	The Paris
Salaries					1				0 0		1,000	April 10 to the

BUDGET.SIGNATURE

Page 2

В	C	D	E	F	G	Н	1	J	K	L	M	N
inge	205-XX-8141-X	FICA	200					718,668	663,485	53,183		
nge	A TYPE	Life Insurance; Dental; Long Term Disability								7117		
inge	205-XX-6142-X	Health Insurance						1,244,838	1,149,458	95,380		
inge	205-XX-8146-X	Retirement						759,927	704.401	55,528		
inge		Worker's Compensation						86,915	, 80,465	8,450		
inge		Unemployment Ins.						5,368	5,098	270		-
ravel			/Drior approx	al and eupporti	ng documents, req	ukad		2,000	0,000	2,000		
							-	2,000		2,000		
quipment		Equipment >\$5,000 per unit cost, enter specific da	(Prior Appro	val & supporting	documents requir	ed for each unity					71.	
quipment		Equipment >\$5,000 per unit cost, enter specific da	Prior Appro	val & supporting	documents requir	ed for each unity		-				
quipment		Equipment >\$5,000 per unit cost, enter specific da									1,000,000	
quipment		Equipment >\$5,000 per unit cost, enter specific da	(Prior Appro	val & supporting	documents requir	ed for each unit)					**** 1.1.1.	1000
upplies		Office Supplies									1 11 11 14	17.
upplies		Classroom Supplies						42,000	42,000		47.74	
upplies	33-6399.00	Medical and Dental Supplies						10,000	10,000		12.4	
upplies		Janitorial Supplies									1. (1.1	
upplies		Other Commodities									(25%)	
upplies		Tools, Apparalus and Accessories (Under \$100 eac	ch)									
upplies		Computer Software	191							197.	AND PARTY	
upplies		Cap <5000 - Computers									Ad armer	
Supplies		Cap <5000 - Machinery & Equipment										1000
Supplies	xx.6399.01	Cap <5000 - Furniture & Fixtures						70,000	70,000		Com a Annahan	10
	AA.0000.01	(Specify)						10,000	. 0,000			-
ontractual	-										11 14 1 1 1 1 1 1 1	-
ontractual		(Specify)									manufact 15	
contractual		(Specify)								-	7.504 A	
Contractual		(Specify)									*******	
Contractual		(Specify)									47 14 75 THE F.	
contractual		(Specify)										
Contractual	A NEW C	(Specify)									A. A. A. P. P. C. L. S.	- Y4
Contractual		(Specify)									1. T. T. P 1. C.	
Other		Staff Development/Training/Seminars						8,000	8,000		****	4
Other		Advertising and Publications								44-24-4-11	77.45W	TAR
Other		Membership Dues and Licenses									Trible 1111	
Other		Binding Printing and Reproduction									4354 4 11 1813	E PER TO
Other		Subscriptions to Publications									The little tree	** *******
Other		Transportation Fees-Staff Mileage									11.1 70.00 11.00	1. 226
Other	-	Transportation Fees- Field Trips - Children									7 *1 * 1 1 1 1 1 mm 1 mm	10, 0. 4 junto
Other		Maint - Buildings and Improvements							-		414 194 5 194	12. 2 · 14
Other	-	Maint - Machinery & Equipment								-	13 30 11 6 10 10	
		Maint & Rep- Automotive							-		10000000	1 1 1
Other		Mall and Parcel Post									E	-
Other								-			W-Y	
Other		Rental of Office Equipment									20 3 417	
Other		Rental of Facilities						-			1 1444	1
Other		Alarm and Security Services									11,000	
Other		M&R Parts Automotive									TOP, Lynn	-
Other		Food for children (NOT reimbursed by USDA)								9.0		
Other		Telephone and fax									Tallian et and	1111
Other		Cellular Phone Service									1.74 m. 1.1 -	
Other		Motor Fuel and Lubricants									*****	
Other		Software Licensing										1 3 DE
Other	1	Gas and Electricity									350.00	
Other	1	Water and Sewer									4. 2	
		General Liability Insurance	1							-		7
Other	-							-		-	3.51	-
Other	-	Building Insurance									1 1 1	-
Other		Vehicle Insurance										
Other		Equipment Insurance									19.	
Other		Relocation Expense/Cost of Moving									strategy a	
Other		In-Kind Parent Participation (Limited to Non-Feder	al Share)									
Other		In-Kind Volunteers (Limited to Non-Federal Share)									alumin a "	
Other	-	In-Kind Other	1									
Other	-	In-Kind Other	1					-			N 70 10	1
	1	prisma Color						A series and a second			And the second second	

ATTACHMENT III [INSERT SPECIAL PROVISIONS HERE]

Contract #:		
Attachment III	- Special	Provisions

ATTACHMENT III SPECIAL PROVISIONS — Program Year 2017-2018

1. RESTRICTIONS ON USE OF FUNDS OR PROPERTY

In addition to the other applicable restrictions on the use of Head Start funds provided under this Contract, the Contractor is prohibited from:

- using or transferring funds provided under this Contract for purposes other than authorized Head Start activities;
- using, pledging, granting a security interest in, or otherwise encumbering any right under this Contract
 or any property acquired with funds provided under this Contract as collateral or security for any loan,
 note debenture, bond or any other debt instrument;
- 3) using any funds provided under this Contract for payment of principal or interest on any loan, note, debenture, bond or any other debt instrument, other than those approved in the 45 C.F.R. Part 74 and by the City

II. REQUIREMENTS FOR PARTICIPATION IN CHILD PLUS DATA SYSTEM

2.01 Child Plus is an electronic case management system managed and licensed by the City. This system maintains child files and an overall wait list and streamlines the process for program entry, qualification, position reservation and referrals. Child Plus enhances performance and improves the overall efficiency of data processing and automation systems in support of Head Start initiatives and is used to compile the annual Program Information Report (PIR).

2.02 Contractor shall:

- a) maintain and support Child Plus Data System
- provide a data entry specialist for Child Plus Data System that will be responsible for entering all required data into the system and who will be the designated contact person with regard to data entries;
- c) attend meetings with the City's Child Plus vendor and City staff to ensure continuity and commitment to the this system;
- d) support all design, development, testing and implementation protocols as established by the City by carrying out and complying therewith;
- e) participate in preliminary and final testing of the system using City protocols;
- f) provide the technical detail required for matching Contractor's system with the Child Plus Data System environment;
- allow City and its vendor to install data encryption software on the Child Care System Database network; and
- provide City and its vendor with access to Confidential Data with parental permission, as defined in Article 3.01 below, which data is critical for the Head Start project.

2.03 Both Parties agree to:

- a) use best efforts to cooperate and exchange information regarding all aspects of the Head Start project and comply with all reasonable requests of the other Party with respect to information concerning the system.
- b) Parties agree that nothing herein shall be construed as to control or in any way limit the right of parents to choose a Head Start provider.

III. CONFIDENTIAL DATA

3.01 The Parties to this Contract shall have access to the following data ("Confidential Data"), with parental permission in the case of the child:

Contract #:	
Attachment III	- Special Provisions

Parent's Information:

Case Number First Name Middle Initial Last Name Street Address

City Zip Code Telephone

Social Security Number (Optional)

Birth Date
Gender
Race
Disability
Yearly Income
Number of members in the Family
County of Residence

Employment and training status

Each child's Information:

Client Number First Name Middle Initial Last Name

Social Security Number (Optional)

Birth Date Gender Race Disability

- 3.02 Contractor understands that City intends to enter into additional agreements with other providers of child care services ("Additional Collaborators") in order to promote the success of the Head Start project. Confidential Data may be shared by City, Contractor, and any Additional Collaborator, except that all parties shall share such information in compliance with state and federal laws relating to confidentiality. All Additional Collaborators shall be required to enter into a written agreement with City containing the confidentiality requirements set forth in this Section III.
- 3.03 Each Party shall establish a method to secure the Confidential Data in accordance with the applicable federal, state, and local laws and regulations. This provision shall not be construed as limiting a Party to this Contract or an Additional Collaborator, or such Party's authorized representative's right of access to that Party's Confidential Data.
- 3.04 Neither Party shall disclose or publish Confidential Data or public school education data to any individual or organization that is not a Party to this Contract or an Additional Collaborator, unless required by law or a lawful order of a court of competent jurisdiction. Each Party shall take measures within its organization to ensure that Confidential Data or public school education data is accessible only by those persons working on the Head Start project, or directly providing other public school education / child care services, and only for the purpose of performing or assisting with services required by the Head Start project or other specific public school education / child care services.
- 3.05 Either Party may disclose Confidential Data to a third party ("Third Party") under contract or affiliated with that Party for the sole purpose of performing or assisting with services required in relation to the Head Start project or other specific child care services, and in compliance with state and federal laws relating to confidentiality. Confidential Data provided to a Third Party shall remain confidential and written confirmation by such Third Party that the Third Party will conform to the requirements of this section shall be provided to the Party prior to delivery of any information to the Third Party.

ATTACHMENT IV [INSERT HHS AWARD DOCUMENT HERE]

SAI NUMBER:

DEPARTMENT OF HEALTH AND HUMAN SERVICES ADMINISTRATION FOR CHILDREN AND FAMILIES NOTICE OF AWARD

PMS DOCUMENT NUMBER:

1. AWARDING OFFICE:				ASSISTANCE TYP	PE: 3. A	WARD NO.:	4. AMEND. NO.		
Office of	f Head Start		Di	Discretionary Grant 0		H7074-04-00	0		
				PE OF ACTION: 7. AWARD AUTH					
Service			Non-Comp	eting Continuation	42	USC 9801 ET S			
	SET PERIOD:	100	, PROJECT			10. CAT NO.:			
02/0	01/2016 THRU 01/	31/2017	07/01/2013	THRU 08		93,600			
CITY OF 111 Sole San Anti	FIPIENT ORGANIZATION: F SAN ANTONIO edad St Ste 500 onio, TX 78205-2288 Authorizing Official: Ivy Tay	lor , Mayor			12. PROJ Head Sta	ECT / PROGRAM	M TITLE:		
13. COU	INTV:	14, CONGR.	DIST:	15 PRINCI	PAL INVEST	GATOR OR PRO	OGRAM DIRECTOR:		
Bexar		20		Willette C			m Head Start Administ		
	16. APPROVED BU	DGET:			17. AWARD	COMPUTATION	V:		
Perso	nnel	\$ 3,645,855	5.00 A. N	A. NON-FEDERAL SHARE \$ 5,522,538.00 20%					
Fringe Benefits \$ 1,367,409.00		B.F	B. FEDERAL SHARE\$ 22,090,151.00 80%						
	l			A, TOTAL FEDERAL SHARE\$ 22,090,151.00					
	ment	45.	00.0						
	ies		В, О	B. UNOBLIGATED BALANCE FEDERAL SHARE\$ 0.00 C. FED. SHARE AWARDED THIS BUDGET PERIOD\$ 0.00					
							1		
	actual	D 1818 818 8	3.00 19. 4	AMOUNT AWARD	ED THIS ACT	ION:	\$ 22,090,151.00		
	ies/Construction	4.		20. FEDERAL \$ AWARDED THIS PROJECT PERIOD: 77.244 000					
			1.00	\$ 77,341,929.					
Direct	Costs	\$ 22,090,151	.00	AUTHORIZED TRE	ATMENT OF	PROGRAM INC	OMF:		
Indired	ct Costs% of \$	\$ 0	00	ductive					
In Kind	d Contributions	\$ 0	.00 22.	22. APPLICANT EIN:		. PAYEE EIN:	24. OBJECT CLASS		
Total Approved Budget \$ 22,090,151.00			.00 746	002070	17	46002070A1	41.51		
			5. FINANCI	AL INFORMATION	N:	DUNS	\$ 066428400		
ORGN	DOCUMENT NO.	APPROPRIATIO		CAN NO.	NEW A		BLIG. NONFED %		
	06CH707404	75-16-153	3	G064120		701.00			
	06CH707404	75-16-153	6 6-	G064122	\$21,877,	450.00			

26. REMARKS: (Continued on separate sheets)

DATE:	28. SIGNATURE(S) CERTIFYING FUND AVAILABILITY				
01/28/2016	Mr. Omar Barrett	01/27/2016			
CIAL(S)	DATE:				
	01/27/2016				
	01/28/2016	01/28/2016 Mr. Omar Barrett CIAL(S) DATE:			

SAI NUMBER:

DEPARTMENT OF HEALTH AND HUMAN SERVICES ADMINISTRATION FOR CHILDREN AND FAMILIES NOTICE OF AWARD

PMS DOCUMENT NUMBER:

AWARDING OFFICE; Office of Head Start			2. ASSISTANCE TYPE Discretionary Grant	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	NARD NO.: 6CH7074-04-00	4. AMEND. NO.	
5. TYPE OF AWARD:		6. TYPE OF ACTION:		1 (22)	7. AWARD AUTHORITY:		
Service		Non-Competing Continuation			42 USC 9801 ET SEQ.		
8. BUDGET PERIOD:		9. PROJECT PERIOD:			10, CAT NO.:		
02/01/2016 THRU 01/31/2017		07/01/2013 THRU 06/30/2018			93.600		

CITY OF SAN ANTONIO

STANDARD TERMS

1. Paid by DHHS Payment Management System (PMS), see attached for payment information.

This award is subject to the requirements of the HHS Grants Policy Statement (HHS GPS) that are applicable to you based on your receipt type and the purpose of this award.

This includes requirements in Parts I and II (available at http://www.hbs.gov/grants/grants/policiesregulations/index.html of the HHS GPS.

Although consistent with the HHS GPS, any applicable statutory or regulatory requirements, including 45 CFR Part 75, directly apply to this award apart from any coverage in the HHS GPS.

This award is subject to requirements or limitations in any applicable Appropriations Act.

This award is subject to the requirements of Section 106 (g) of the trafficking Victims

Protection Act of 2000, as amended (22 U.S.C. 7104).

For the full text of the award term, go to http://www.acf.hhs.gov/discretionary-post-awardrequirements

This award is subject to the Federal Financial Accountability and Transparency Act (FFATA or

Transparency) of 2006 subaward and executive compensation reporting requirements.

For the full text of the award term, go to http://www.acf.hhs.gov/discretionary-post-awardrequirements

This award is subject to requirements as set forth in 2 CFR 25.110 Central Contractor

Registration (CCR) and DATA Universal Number System (DUNS).

For full text go to http://www.acf.hhs.gov/discretionary-post-award-requirements

This award is subject to the requirements as set forth in 45 CFR Part 87.

This grant is subject to the requirements as set forth in 45 CFR Part 75.

This award is subject to HHS regulations codified at 45 CFR 1301, 1302, 1303, 1304, 1305, 1306, 1307, 1308, 1309, and 1310.

Attached are terms and conditions, reporting requirements, and payment instructions.

Initial expenditure of funds by the grantee constitutes acceptance of this award.

Future support is anticipated.

** Reflects only federal share of approved budget."

ATTACHMENT V [INSERT HIPAA BAA HERE]

WITNESSETH:

HIPAA BUSINESS ASSOCIATE AGREEMENT

This HIPAA Business Associate Agreement is entered into by and between the City of San Antonio ("Covered Entity"), and the San Antonio Independent School District, a Business Associate ("BA").

WHEREAS, the City of San Antonio and BA have entered into an agreement to provide Head Start services ("Service Contract"), effective February 1, 2017; and

WHEREAS, Covered Entity and BA may need to use, disclose and/or make available certain information pursuant to the terms of the Service Contract, some of which may constitute Protected Health Information ("PHI"); and

WHEREAS, Covered Entity and BA intend to protect the privacy and provide for the security of PHI disclosed to each other pursuant to the Service Contract in compliance with the Health Insurance Portability and Accountability Act of 1996, Public Law 104-191 ("HIPAA") and regulations promulgated thereunder by the U.S. Department of Health and Human Services (the "HIPAA Regulations"), Health Information Technology for Economic and Clinical Health Act ("HITECH Act") and other applicable laws; and

WHEREAS, the purpose of this Agreement is to satisfy certain standards and requirements of HIPAA and the HIPAA Regulations, including, but not limited to, Title 45, Section 164.504(e) of the Code of Federal Regulations ("C.F.R."), as the same may be amended from time to time;

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, hereby agree as follows:

- A. <u>Definitions</u>. For the purposes of this Agreement, the following terms have the meanings ascribed to them:
 - (1) "Disclosure" with respect to PHI, shall mean the release, transfer, provision of access to or divulging in any other manner of PHI outside the entity holding the PHI.
 - (2) "Individual" shall have the same meaning as the term "Individual" in 45 C.F.R. 164.501 and shall include a person who qualifies as a personal representative in accordance with 45 C.F.R. 164.502(g).
 - (3) "Parties" shall mean Covered Entity and BA. "Business Associate" shall generally have the same meaning as the term "business associate" at 45 CFR 160.103. "Covered Entity" shall generally have the same meaning as the term "covered entity" at 45 CFR 160.103.
 - (4) "Privacy Rule" shall mean the Standards for Privacy of Individually Identifiable Health Information at 45 C.F.R. part 160 and Part 164, subparts A and E.
 - (5) "Security Rule" shall mean the HIPAA regulation that is codified at 45 C.F.R. Part 164.

- (6) "Protected Health Information" or "PHI" shall have the same meaning as the term "protected health information" in 45 C.F.R. 164.501, limited to the information created or received by BA from or on behalf of Covered Entity. PHI includes "Electronic Protected Health Information" or "EPHI" and shall have the meaning given to such term under the HIPAA Rule, including but not limited to 45 CFR Parts 160, 162, 164, and under HITECH.
- (6) "Required By Law" shall have the same meaning as the term "required by law" in 45 CFR § 164.501.
- (7) "Secretary" shall mean the Secretary of the Department of Health and Human Services ("HHS") or his designee.
- (8) "PHI Breach" shall mean an acquisition, access, use, or disclosure of PHI in a manner not permitted by the Privacy Rules and such action compromises the security or privacy of the PHI.
- (9) The Health Information Technology for Economic and Clinical Health ("HITECH") Act shall mean Division A, Title XII of the American Recovery and Reinvestment Act of 2009 (Pub. L. 111-5).

B. BA Obligations and Activities. BA agrees that it shall:

- (1) Not use or disclose the PHI other than as permitted or required by this Agreement or as Required by Law;
- (2) Establish and maintain appropriate administrative, physical, and technical safeguards that reasonably and appropriately protect, consistent with the services provided under this Agreement, the confidentiality, integrity, and availability of the electronic protected health information that it creates, receives, maintains, or transmits on behalf of covered entity;
- (3) Mitigate, to the extent practicable, any harmful effect that is known to BA of a use or disclosure of PHI by BA in violation of the requirements of this Agreement;
- (4) Report to Covered Entity any use or disclosure of PHI of which BA is aware or becomes aware that is not provided for or allowed by this Agreement as well as any security incident that BA becomes of aware of;
- (5) Ensure that a business associate agreement is in place with any of its agents or subcontractors with which BA does business and to whom it provides PHI received from, created or received by BA on behalf of Covered Entity are aware of and agree to the same restrictions and conditions that apply through this Agreement to BA with respect to such information, and further agree to implement reasonable and appropriate administrative, physical and technical safeguards in accordance with HHS guidance specifying the technologies and methodologies that render such PHI unusable, unreadable, or indecipherable to individuals unauthorized to acquire or otherwise have access to such PHI;

- (6) Provide access, at the request of Covered Entity, and in a reasonable time and manner as agreed by the Parties, to PHI in a Designated Record Set to Covered Entity or, as directed by Covered Entity, to an Individual in order to meet the requirements 45 C.F.R. §164.524;
- (7) Make any amendment(s) to PHI in a Designated Record Set that the Covered Entity directs or agrees to pursuant to 45 C.F.R. 164.526 at the request of the Covered Entity or an Individual, and in a reasonable time and manner agreed to by the Parties;
- (8) Make available to the Covered Entity or to the Secretary all internal practices, books and records, including policies and procedures, relating to the use and disclosure of PHI received from, or created or received by the BA on behalf of the Covered Entity, for purposes of the Secretary in determining Covered Entity's compliance with the Privacy Rule;
- (9) Document such disclosures of PHI, and information related to such disclosures, as would be required for Covered Entity to respond to a request from an Individual for an accounting of disclosures of PHI in accordance with 45 C.F.R. 164.528;
- (10) Provide Covered Entity or an Individual, in a reasonable time and manner as agreed to by the Parties, information collected in accordance with Section B(9) of this Agreement, to permit Covered Entity to respond to a request by an Individual for an accounting of disclosures of PHI in accordance with 45 C.F.R. 164.528;
- (11) Will immediately, and in no event later than three business days from discovery, notify Covered Entity of any breach of PHI, including ePHI, and will coordinate with Covered Entity to identify, record, investigate, and report to an affected individual and US Department of Health and Human Services, as required, any covered PHI breach. Breach notification to Covered Entity must include: names of individuals with contact information for those who were or may have been impacted by the HIPAA Breach; a brief description of the circumstances of the HIPAA Breach, including the date of the breach and date of discover; a description of the types of unsecured PHI involved in the breach; a brief description of what the BA has done or is doing to investigate the breach and mitigate harm. BA will appoint a breach liaison and provide contact information to provide information and answer questions Covered Entity may have concerning the breach;
- (12) Comply with all HIPAA Security Rule requirements;
- (11) To the extent the BA is to carry out one or more of Covered Entity's obligation(s) under Subpart E of 45 CFR Part 164, comply with the requirements of Subpart E that apply to the Covered Entity in the performance of such obligation(s);
- (12) Under no circumstances may BA sell PHI in such a way as to violate Texas Health and Safety Code, Chapter 181.153, effective September 1, 2012, nor shall BA use PHI for marketing purposes in such a manner as to violate Texas Health and Safety Code Section 181.152, or attempt to re-identify any information in violation of Texas Health and Safety Code Section 181.151, regardless of whether such action is on behalf of or permitted by the Covered Entity.

C. Permitted Uses and Disclosures by BA

- (1) Except as otherwise limited in this Agreement, BA may use or disclose PHI to perform functions, activities, or services for, or on behalf of, Covered Entity as specified in the Service Contract, provided that such use or disclosure would not violate the Privacy Rule if done by Covered Entity.
- (2) Except as otherwise limited in this Agreement, BA may disclose PHI for the proper management and administration of the BA, provided that disclosures are Required By Law, or BA obtains reasonable assurances from the person to whom the information is disclosed that it will remain confidential and used or further disclosed only as Required By Law or for the purpose for which it was disclosed to the person, and the person notifies the BA of any instances of which it is aware in which the confidentiality of the information has been breached.
- (3) Except as otherwise limited in this Agreement, BA may use PHI to provide Data Aggregation Services to Covered Entity as permitted by 45 C.F.R. 164.504(e)(2)(i)(B).
- (4) BA may use PHI to report violations of law to appropriate Federal and State authorities, consistent with 45 C.F.R. 502(j)(1).
- D. <u>Obligations of Covered Entity</u>. Covered Entity shall inform BA of its privacy practices and restrictions as follows. Covered Entity shall:
 - (1) notify BA of any limitations in its notice of privacy practices in accordance with 45 C.F.R. 164.520, to the extent that such limitation may affect BA's use or disclosure of PHI;
 - (2) notify BA of any changes in, or revocation of, permission by any Individual to use or disclose PHI, to the extent that such changes may affect BA's use or disclosure of PHI;
 - (3) notify BA of any restriction to the use or disclosure of PHI that Covered Entity has agreed to in accordance with 45 C.F.R. 164.522 to the extent that such changes may affect BA's use or disclosure of PHI.
 - (4) coordinate with BA regarding any PHI breach and make timely notification to affected individuals within 60 days of discovery.

E. Permissible Requests by Covered Entity.

Covered Entity shall not request BA to use or disclose PHI in any manner that would not be permissible under the Privacy Rule if done by Covered Entity, except that the BA may use or disclose PHI for data aggregation or management and administrative activities of the BA.

F. Term and Termination.

(1) The term of this Agreement shall commence on February 1, 2017. This Agreement shall terminate when all PHI encompassed by this Agreement is destroyed or returned to Covered Entity or, if it is infeasible to return or destroy

- the PHI, protections are extended to such information in accordance with the termination provisions in this Section.
- (2) Termination for Cause. Upon Covered Entity's knowledge of a material breach by BA, Covered Entity shall either (a) provide an opportunity for BA to cure the breach in accordance with the terms of the Service Contract or, if the BA does not cure the breach or end the violation within the time for cure specified in the Service Contract, end the violation and terminate this Agreement and the Contract; or (b) immediately terminate this Agreement and the Service Contract if BA has breached a material term of this Agreement and cure is not possible. If neither termination nor cure is feasible, Covered Entity shall report the violation to the Secretary of the U.S. Department of Health and Human Services.

(3) Effect of Termination.

- (a) Except as provided below in paragraph (b) of this Section F(3), upon termination of this Agreement for any reason, BA shall return or destroy all PHI received from the Covered Entity, or created or received by BA on behalf of Covered Entity. This provision shall apply to PHI that is in the possession of BA or its subcontractors or agents. BA shall not retain any copies of PHI, except as required by law.
- (b) In the event that BA determines that returning or destroying PHI is infeasible, BA shall provide to Covered Entity written notification of the condition that makes the return or destruction of PHI infeasible. Upon BA's conveyance of such written notification, BA shall extend the protections of this Agreement to such PHI and limit further uses and disclosures of such PHI to those purposes that make its return or destruction infeasible, for so long as BA maintains such PHI.
- G. Amendment to Comply with Law. The Parties agree to take written action as is necessary to amend this Agreement to comply with any federal and state HIPAA privacy requirements and HIPAA legal requirements for Covered Entity without the need for additional council action.
- H. <u>Survival</u>. The respective rights and obligations of the BA under Sections B, C (2) and (4), and F(3) shall survive the termination of this Agreement.
- I. <u>Interpretation</u>. Any ambiguity in this Agreement shall be interpreted to permit Covered Entity to comply with the Privacy Rule.
- J. <u>Regulatory References</u>. A reference in this Agreement to a section in the Privacy Rule means the section as in effect or amended.
- K. No Third Party Beneficiaries. Nothing express or implied in this Agreement is intended to confer, nor shall anything herein confer upon any person other than Covered Entity, BA, and their respective successors or assigns, any rights, remedies, obligations, or liabilities whatsoever.
- L. <u>LIMITED LIABILITY</u>. CONTRACTOR AND THE CITY ACKNOWLEDGE THEY ARE POLITICAL SUBDIVISIONS OF THE STATE OF TEXAS AND ARE SUBJECT TO.

AND COMPLY WITH THE APPLICABLE PROVISIONS OF THE TEXAS TORT CLAIMS ACT, AS SET OUT IN THE CIVIL PRACTICE AND REMEDIES CODE, SECTION 101.001, ET. SEQ., AND THE REMEDIES AUTHORIZED THEREIN REGARDING CLAIMS OR CAUSES OF ACTION THAT MAY BE ASSERTED BY THIRD PARTIES FOR ACCIDENT, INJURY OR DEATH.

- M. Reimbursement. BA will reimburse Covered Entity for reasonable costs incurred responding to a PHI breach by BA or any of BA's subcontractors.
- N. <u>Waiver</u>. No provision of this Agreement or any breach thereof shall be deemed waived unless such waiver is in writing and signed by the party claimed to have waived such provision or breach. No waiver of a breach shall constitute a waiver of or excuse any different or subsequent breach.
- O. <u>Assignment</u>. Neither party may assign (whether by operation or law or otherwise) any of its rights or delegate or subcontract any of its obligations under this Agreement without the prior written consent of the other party. Notwithstanding the foregoing, Covered Entity shall have the right to assign its rights and obligations hereunder to any entity that is an affiliate or successor of Covered Entity, without the prior approval of BA.
- P. Entire Agreement. This Agreement constitutes the complete agreement between BA and Covered Entity relating to the matters specified in this Agreement, and supersedes all prior representations or agreements, whether oral or written, with respect to such matters. In the event of any conflict between the terms of this Agreement and the terms of the Service Contracts or any such later agreement(s), the terms of this Agreement shall control unless the terms of such Service Contract comply with the federal law and regulations commonly referred to as the Privacy Standards and the Security Standards. No oral modification or waiver of any of the provisions of this Agreement shall be binding on either party. This Agreement is for the benefit of, and shall be binding upon the Parties, their affiliates and respective successors and assigns. No third party shall be considered a third-party beneficiary under this Agreement, nor shall any third party have any rights as a result of this Agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

Q. <u>Governing Law</u>. This Agreement shall be governed by and interpreted in accordance with the laws of the State of Texas.

EXECUTED to be effective on February 1, 2017.

COVERED ENTITY City of San Antonio	BUSINESS ASSOCIATE: San Antonio Independent School Distric			
	Kar			
Melody Woosley, Director Department of Human Services	Pedro Martinez, Superintendent			
APPROVED AS TO FORM:				
Assistant City Attorney				

ATTACHMENT VI [INSERT ADDITIONAL OMB PROVISIONS HERE]

Additional OMB Provisions

from Appendix II to Part 200—Contract Provisions for Non-Federal Entity Contracts Under Federal Awards

In addition to other provisions required by HHS or the City, all contracts made by the City under the Federal award must contain provisions covering the following, as applicable (2 C.F.R. 200, Appendix II).

Hereinafter in this Attachment VI, Center shall be referred to as "contractor."

Provision	Page Number
Equal Employment Opportunity	2
Davis Bacon Act	4
Contract Work Hours and Safety Standards Act	9

EQUAL EMPLOYMENT OPPORTUNITY provisions (60 C.F.R. 1.4(b).

During the performance of this contract, contractor agrees:

- (1) The contractor will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. The contractor will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.
- (2) The contractor will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive considerations for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.
- (3) The contractor will send to each labor union or representative of workers with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of the contractor's commitments under this section, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- (4) The contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- (5) The contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to CONTRACTOR's books, records, and accounts by the U.S. Department of Health and Human Services ("HHS") and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- (6) In the event of the contractor's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations, or orders, this contract may be canceled, terminated, or suspended in whole or in part and the contractor may be declared ineligible for further Government contracts or federally-assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law,

(7) The contractor will include the portion of the sentence immediately preceding paragraph (1) and the provisions of paragraphs (1) through (7) in every subcontract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The contractor will take such action with respect to any subcontract or purchase order as HHS may direct as a means of enforcing such provisions, including sanctions for noncompliance; provided, however, that in the event a contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the administering agency the contractor may request HHS to enter into such litigation to protect the interests of the United States.

DAVIS BACON ACT provisions (29 C.F.R. § 5.5(a))

For any contract or subcontract in excess of \$2,000 which is entered into for the actual construction, alteration and/or repair, including painting and decorating, of a public building or public work, or building or work financed in whole or in part from federal funds, and which is subject to the labor standards provisions of any of the acts listed in 29 C.F.R. §5.1, the following § 5.5(a) must be included and complied with:

- (1) Minimum wages.
- (i) All laborers and mechanics employed or working upon the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor, which is incorporated herein by reference, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of paragraph (a)(1)(iv) of this section; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in §5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein: Provided, that the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under paragraph (a)(1)(ii) of this section) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

(ii)

- (A) The contracting officer shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The contracting officer shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:
 - (1) The work to be performed by the classification requested is not performed by a classification in the wage determination; and
 - (2) The classification is utilized in the area by the construction industry; and
 - (3) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.
- (B) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the contracting officer agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the contracting officer to the Administrator of the Wage and Hour Division, Employment Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an

authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

- (C) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the contracting officer do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the contracting officer shall refer the questions, including the views of all interested parties and the recommendation of the contracting officer, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.
- (D) The wage rate (including fringe benefits where appropriate) determined pursuant to paragraphs (a)(1)(ii) (B) or (C) of this section, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.
- (iii) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.
- (iv) If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, Provided, That the Secretary of Labor has found, upon the written request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.
- (2) Withholding. HHS or the City shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld from the contractor under this contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), all or part of the wages required by the contract, HHS may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.
- (3) Payrolls and basic records.
- (i) Payrolls and basic records relating thereto shall be maintained by the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work (or under the United States Housing Act of 1937, or under the Housing Act of 1949, in the construction or development of the project). Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected,

and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

(ii)

- (A) The contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to HHS if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit the payrolls to the applicant, sponsor, or owner, as the case may be, for transmission to HHS. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on weekly transmittals. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from the Wage and Hour Division Web site at http://www.dol.gov/esa/whd/forms/wh347instr.htm or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to HHS if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit them to the applicant, sponsor, or owner, as the case may be, for transmission to HHS, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of this section for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the sponsoring government agency (or the applicant, sponsor, or owner).
- (B) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:
 - (1) That the payroll for the payroll period contains the information required to be provided under §5.5 (a)(3)(ii) of Regulations, 29 CFR part 5, the appropriate information is being maintained under §5.5 (a)(3)(i) of Regulations, 29 CFR part 5, and that such information is correct and complete;
 - (2) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in Regulations, 29 CFR part 3;
 - (3) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.
- (C) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH-347 shall satisfy the requirement for submission of the "Statement of Compliance" required by paragraph (a)(3)(ii)(B) of this section.
- (D) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under section 1001 of title 18 and section 231 of title 31 of the United States Code.
- (iii) The contractor or subcontractor shall make the records required under paragraph (a)(3)(i) of this section available for inspection, copying, or transcription by authorized representatives of HHS or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the Federal agency may, after written notice to the contractor, sponsor, applicant, or owner,

take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

- (4) Apprentices and trainees-
- (i) Apprentices. Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (ii) Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program,

- the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (iii) Equal employment opportunity. The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR part 30.
- (5) Compliance with Copeland Act requirements. The contractor shall comply with the requirements of 29 CFR part 3, which are incorporated by reference in this contract.
- (6) Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses contained in 29 CFR 5.5(a)(1) through (10) and such other clauses as the (write in the name of the Federal agency) may by appropriate instructions require, and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in 29 CFR 5.5.
- (7) Contract termination: debarment. A breach of the contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in 29 CFR 5.12.
- (8) Compliance with Davis-Bacon and Related Act requirements. All rulings and interpretations of the Davis-Bacon and Related Acts contained in 29 CFR parts 1, 3, and 5 are herein incorporated by reference in this contract.
- (9) Disputes concerning labor standards. Disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and the contracting agency, the U.S. Department of Labor, or the employees or their representatives.
- (10) Certification of eligibility.
- (i) By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (ii) No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (iii) The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

CONTRACT WORK HOURS AND SAFETY STANDARDS ACT provisions (29 C.F.R. § 5.5(b))

- (1) Overtime requirements.
 - No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek.
- (2) Violation; liability for unpaid wages; liquidated damages. In the event of any violation of the clause set forth in paragraph (b)(1) of this section the contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in paragraph (b)(1) of this section, in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in paragraph (b)(1) of this section.
- (3) Withholding for unpaid wages and liquidated damages. The U.S. Department of Health and Human Services or the City shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in paragraph (b)(2) of this section.
- (4) Subcontracts.
 - The contractor or subcontractor shall insert in any subcontracts the clauses set forth in paragraph (b)(1) through (4) of this section and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in paragraphs (b)(1) through (4) of this section.

	* .	
	**	
. *		
7		
,		
i i		

Contract No.	
Continue	

STATE OF TEXAS	§	AGREEMENT TO PROVIDE HEALTH
	8	SCREENINGS AND EXAMINATIONS FOR THE
COUNTY OF BEXAR	8	CITY OF SAN ANTONIO HEAD START PROGRAM

This Contract is entered into by and between the City of San Antonio (hereinafter referred to as "City"), a Texas Municipal Corporation, acting by and through its Director of the Department of Human Services pursuant to Ordinance No. _______ dated September 29, 2016, and the Bexar County Hospital District d/b/a University Health System, a hospital district created pursuant to Article IX Section 4 of the Texas Constitution and Chapter 281 of the Health and Safety Code, (hereinafter referred to as "Contractor") (individually "the Party" and collectively "the Parties") to set forth the objectives, understandings, and agreements between the Parties in connection with the use of Head Start grant funds as described herein. This Agreement is made and entered into by the Parties pursuant to the authority granted under the Interlocal Cooperation Act, Texas Gov't Code 791 et seq.

WITNESSETH:

WHEREAS, the City has received a grant pursuant to the Head Start Act (42 U.S.C. §9801 et seq., as amended) (the "Grant") for the purpose of providing Head Start services to children and families in the City of San Antonio and the Bexar County area; and

WHEREAS, the City is authorized by the U.S. Department of Health and Human Services ("HHS"), Administration for Children and Families ("ACF"), and desires, through its Department of Human Services ("DHS"), to execute an agreement with Contractor to provide Head Start services to children residing in the City of San Antonio and the Bexar County area (hereinafter referred to as the "Project" or "Program"); and

WHEREAS, Contractor is a hospital district established under Article IX, Section 4 of the Texas Constitution and Chapter 281 of the Texas Health and Safety Code, whose public purpose is providing medical and hospital care to the needy and indigent of the community; and

WHEREAS, City agrees to compensate Contractor for health screenings and physical examinations provided to children enrolled in its Head Start programs; and

NOW THEREFORE, in consideration of the mutual promises and covenants herein contained and intending to be legally bound hereby, the Parties agree as follows:

I. SCOPE OF WORK

- 1.1 The Contractor will provide, oversee, administer, and carry out all activities and services in a manner satisfactory to the City and in compliance with the Contractor's Scope of Work, attached hereto and incorporated herein for all purposes as Attachment I, this Contract, and the Terms of the Grant (hereinafter defined). If the terms of this Contract are inconsistent or in conflict with applicable Terms of the Grant, the applicable Terms of the Grant shall be controlling, unless the inconsistency or conflict results from more stringent requirements set forth in this Contract, in which case the terms imposing the most stringent requirements upon the Contractor shall control.
- 1.2 For purposes of this Contract, the terms listed below shall have the following meanings:
 - (A) "Terms of the Grant" shall mean all requirements of the Grant, whether contained in the Head Start Act, as amended by the Improving Head Start for School Readiness Act of 2007 (42 U.S.C. §9831, et seq.), or other applicable statutes, implementing regulations (e.g., 45 C.F.R. §1301 et seq. (the "Head Start Performance Standards" or "Performance Standards") and 45 C.F.R. Part 75, as amended), rules, Executive Orders, the award document from U.S. Department of Health and Human Services ("HHS")

- to the City, Relevant HHS Directives, or elsewhere, including, but not limited to circulars, Program Instructions, Information Memorandums and Policy Clarifications, the City's policies and procedures and the program design manual applicable to the Head Start Program, as such requirements exist as of the date of this Contract and as such requirements may be established or modified (by amendment, deletion, addition or otherwise) during the period of the Contract.
- (B) "Relevant HHS directives" shall mean regulations, manuals, guidelines, or other oral or written directives of HHS or any subdivision thereof, including the Administration for Children and Families, Head Start Bureau, the Program Operations Division and ACF Region VI, as such regulations, manuals, guidelines, or other oral or written directives shall be made applicable to the Grant or Grantee.
- 1.3 Contractor shall establish and implement policies and procedures governing personnel, financial management, and programmatic management, as specified more fully in 45 C.F.R Parts 1301 et. seq, and/or 45 C.F.R. Part 75 as applicable. Such policies and procedures shall be consistent with the Terms of the Grant, the policies and procedures approved by the Grantee's Policy Council and Governing Body, and content and service plans.
- 1.4 City retains the authority to contract with third-parties for the delivery of other Head Start services in San Antonio and Bexar County area. Contractor agrees to allow the City's other contractors access to the facilities leased and/or owned by Contractor in order to provide said services. Contractor agrees to cooperate with City and third-party Head Start contractors to establish, modify and comply with a set of policies and procedures and/or a program design manual governing the City's Head Start Program and the protocol for collaboration between Head Start service providers.

II. TERM

2.1 Except as otherwise provided for pursuant to the provisions hereof, this Contract shall begin on February 1, 2017 and shall terminate on June 2, 2017 The City shall have the option to renew this Agreement, with appropriate revisions to the budget and due dates in order to reflect the grant award and to the Scope of Work in Attachment I in order to best meet the needs of children and families through Head Start Grant activities, for one additional term of up to 6 months without the necessity of further City Council approval.

III. CONSIDERATION

- 3.1 In consideration of the services to be delivered by Contractor, the City will reimburse Contractor a total amount not to exceed \$20,000.00 ("the Federal Share") during the period in which this Contract is in effect for costs incurred in accordance with the Program Budget attached hereto and incorporated herein for all purposes as Attachment II. Contractor's Program Budget is comprised of the Federal Share and the Non-Federal Share. The Federal Share shall be no more than 80% of the total Program Budget. Should Contractor fail to raise all of the non-Federal Share funds (20% of the total Program Budget, or \$5,000.00) it is required to raise for the operation of its Program, City reserves the right to limit its reimbursements to Contractor proportionately. For instance, if Contractor succeeds in raising only eighty percent (80%) of its required non-Federal Share funds, City may limit its reimbursements to Contractor to eighty percent (80%) of City's total obligation to Contractor. To meet the requirements of this Contract, all claimed non-Federal Share must meet the requirements of 45 C.F.R. § 75.306, as applicable.
- 3.2 Prior to commencement of the Contract, Contractor shall submit to City for its approval a monthly budget by line item for the entire term of the Contract along with its program Budget, including detail by category alone. If the Contractor's budget is not submitted to the City for approval before the beginning of the Contract period, the City reserves the right to redirect Contractor's funding as necessary. Additionally, throughout the term of the Contract, Contractor shall submit on or before the last day of each month a forecast of the projected monthly expenses for each month remaining in the Contract so that the City may

review and compare actual expenses to projected expenditures and address issues associated with Contractor's expenditure rate (e.g., on or before March 31, 2017, Contractor shall submit the projected expenses by month for April, May, and so on until June 2, 2017). Contractor's budgeted development and administrative costs (as defined by 45 C.F.R. §1301.32) shall not exceed twelve percent (12%) of the Program Budget, unless the total Program Budget is modified in accordance with this Contract in which case this amount shall be reduced proportionately unless the Parties otherwise agree.

- 3.3 Approval required. Contractor shall seek and obtain City's (City's Head Start Program Administrator and DHS's Fiscal Administrator) prior written approval 30 calendar days before making budget modifications. City may make exceptions to the 30-day notice requirement on a case by case basis, but otherwise Contractor must make request in writing or via email to the City's Head Start Program Administrator. Contractor's written request must be accompanied by a justification for the change and indicate which lines items are affected by such change.
- 3.4. The funding level of this Contract is based on an allocation from the following funding sources:
 - U.S. Department of Health and Human Services (HHS) Head Start Funds Catalog of Federal Domestic Assistance # 93.600
 - Consequently, Contractor agrees to comply with the Terms of the Grant and the Special Provisions, affixed hereto and incorporated herein for all purposes as **Attachment III**.
- 3.5 It is expressly understood and agreed by the City and Contractor that the City's obligations under this Contract are contingent upon the actual receipt of adequate grant funds from HHS to meet City's liabilities hereunder. This Contract may be terminated by the City if HHS terminates the City as a grantee or reduces the amount granted to City, for any reason; provided that, if the reduction of grant funds does not result in complete unavailability of such funds, the Parties will use best efforts to amend this Contract accordingly. City will promptly notify Contractor of any such HHS action.

IV. PAYMENT

4.1 Contractor agrees that this is a cost reimbursement contract and that the City's liability hereunder is limited to making reimbursements for allowable costs incurred as a direct result of services provided by the Contractor in accordance with the terms of this Contract. Allowable costs are defined as those costs, which are necessary, reasonable and allowable under applicable federal, state, and local law, including but not limited to those laws referenced in Article XII hereof, for the proper administration and performance of the services to be provided under an agreement. . All requested reimbursed costs must be consistent with the terms and provisions of the approved budgeted line items described in Attachment II of this Contract, unless (a) a subsequent budget revision has been approved in accordance with the procedure set forth in Section 3.3 and signed by the Director of DHS in cases where the total Contract Budget remains the same, or (b) a Contract amendment has been approved and signed by the Director of DHS pursuant to Section 24.1 of this Contract in cases where there is an increase or decrease to the total Contract Budget. Approved budget revisions and Contract amendments modify the Budget attached hereto, and in such cases Contractor's requested reimbursed costs must be consistent with the last revised, approved budget. Approved budget revisions and Contract amendments supersede prior conflicting or inconsistent agreements with regard to the referenced Project Budget, and all references in the Contract to the budget shall mean the budget as revised through approved budget revisions or Contract amendments. Budget revision requests shall be submitted in advance of anticipated expense(s). The City will not accept budget revision requests submitted after May 2, 2017. In no event shall the City be liable for any cost of Contractor not eligible for reimbursement as defined within the Contract. Contractor shall remit to City within ten (10) business days after the City makes the request for remittance any funded amounts which were paid pursuant to this Article IV and used to cover disallowed costs. Any such amounts not remitted within ten (10) business days may, at City's option, be subject to offset against future funding obligations by City. For purposes of this Contract, the term, "business day" shall mean every day of the week except all Saturdays, Sundays and those scheduled holidays officially adopted and approved by the San Antonio

- City Council for City of San Antonio employees, except in those instances where this Contract expressly calls for "Contractor business days."
- 4.2 If specific circumstances require an advance payment on this Contract, Contractor must submit to the Director of DHS a written request for such advance payment, including the specific reason for such request in the form prescribed by City. Contractor agrees that the City shall not be obligated to pay for any advances requested. In those instances in which advance payments are authorized, the Director of DHS may, in the Director's sole discretion, approve an advance payment on this Contract. It is understood and agreed by the Parties hereto that (a) each request requires submission to the Director of DHS no less than ten (10) business days prior to the actual ostensible cash need; (b) each request will be considered by the Director of DHS on a case-by-case basis, and (c) the decision by the Director of DHS whether or not to approve an advance payment is final. For purposes of this Contract, the term "business day" shall mean every day of the week except all Saturdays, Sundays and those scheduled holidays officially adopted and approved by the San Antonio City Council for City of San Antonio employees. In those instances in which advance payments are authorized:
 - (A) Contractor's payments to its vendors using funds advanced by the City shall be remitted to the vendors in a prompt and timely manner so long as services have been performed by the subject vendor, defined as not later than (10) business days after the Contractor is notified that an advance payment check is available from the City.
 - (B) The Contractor must deposit Contract funds in an account in a bank insured with the Federal Deposit Insurance Corporation (FDIC). In those situations where Contractor's total deposits in said bank, including all Contract funds deposited with said bank, exceed the FDIC insurance limit, the Contractor must arrange with said bank to automatically have the excess collaterally secured. A written copy of the collateral agreement must be obtained by Contractor from the Contractor's banking institution, maintained on file and be available for City monitoring reviews and audits. Advanced funds that cause the Contractor's account balance to exceed the FDIC limit shall be deposited in a manner consistent with the Public Funds Investment Act (Chapter 2256 of the Texas Government Code) as amended. Contractor shall maintain the FDIC insured bank account in which Contract funds are deposited and its recordkeeping in a manner that will allow City to track expenditures made with Contract funds.
 - (C) The City may deduct from monthly reimbursements amounts necessary to offset the amount advanced based upon the number of months remaining in the Contract term, or from a single subsequent monthly reimbursement the full amount previously advanced to Contractor. The City may consider factors such as projected allowable costs and other indicators such as Contractor's financial stability. Contractor shall maintain a financial management system to account for periodic, or a lump sum, deduction from reimbursements.
- 4.3 Contractor shall submit to City not later than the fifteenth (15th) of every month a monthly Request for Payment in the form prescribed by City, which details the specific costs (by category and by program account number) Contractor expensed in the previous month for the services delivered as described in Article I herein, including supporting documentation of such costs as may be required by the Director of DHS. The Request for Payment shall also specify the Program Income (as defined herein) received or projected during the same time period. The Director of DHS may require the Contractor's submission of original or certified copies of invoices, cancelled checks, Contractor's general ledger and/or receipts to verify invoiced expenses.
- 4.4 City shall make reimbursement payments of eligible expenses to the Contractor of any undisputed amounts as determined by the Director of DHS in accordance with established procedures, so long as City receives a properly completed and documented Request for Payment. City shall make payment to Contractor within 30 calendar days of receiving a valid and approved Request for Payment.
- 4.5 The Contractor shall submit to City a full accounting of the Program Income and non-Federal Share funds received and total Program costs incurred, along with all requests for payment for the period February 1,

2017 through June 2, 2017 no later than July 17, 2017. In the event of early termination of this Contract, Contractor shall submit the information 45 calendar days from the early termination date of the Contract. These deadlines may be adjusted only if Contractor receives written authorization from the Director of DHS allowing Contractor to submit a request for payment at a later specified date.

- 4.6 Contractor agrees that the City shall not be obligated to any subcontractors or third party beneficiaries of the Contractor.
- 4.7 Contractor shall maintain a financial management system, and acceptable accounting records in accordance with 45 C.F.R. § 75.302 et seg, as applicable, and that provide for:
 - (A) accurate, current, and complete disclosure of financial support from each federal, state and locally sponsored project and program in accordance with the reporting requirements set forth in Article VIII of this Contract. If accrual basis reports are required, the Contractor shall develop accrual data for its reports based on an analysis of the documentation available;
 - (B) records that adequately identify the source and application of funds for City-sponsored activities. Such records shall contain information pertaining to City awards, authorizations, obligations, unobligated balances, assets, equity, outlays, and income;
 - (C) effective control over and accountability for all funds, property, and other assets. The Contractor shall adequately safeguard all such assets and shall ensure that they are used solely for authorized purposes. Contractor shall maintain a separate numbered account for all funds received and disbursed through this Contract;
 - (D) comparison of actual outlays with budget amounts for each award. Whenever appropriate or required by the City, financial information should be related to performance and unit cost data;
 - (E) procedures to minimize the time elapsing between the transfer of funds from the City and the disbursement of said funds by the Contractor;
 - (F) procedures for determining reasonable, allowable, and allocable costs in accordance with the provisions of any and all applicable cost principles, including but not limited to the cost principles referenced in Article XII hereof, and the terms of the award, grant, or contract, with the City;
 - (G) accounting records that are supported by source documentation (i.e., timesheets, employee benefits, professional services agreements, purchases, and other documentation as required by City). Contractor shall maintain records and shall meet necessary requirements under Generally Accepted Accounting Principles [GAAP]; and
 - (H) an accounting system based on generally acceptable accounting principles which accurately reflects all costs chargeable (paid and unpaid) to the Project. A Receipts and Disbursements Ledger must be maintained. A general ledger with an Income and Expense Account for each budgeted line item is necessary. Paid invoices revealing check number, date paid and evidence of goods or services received are to be filed according to the expense account to which they were charged.
- 4.8 Contractor agrees that Contractor costs or earnings claimed under this Contract may not be claimed under another contract or grant from another agency, organization, business entity or governmental entity.
- 4.9 Contractor shall establish, and abide by a cost allocation methodology and plan which ensures that the City is paying only its fair share of the costs for services, overhead, and staffing not solely devoted to the Project or funded pursuant to this Contract. The Cost Allocation Plan is a plan that identifies and distributes the cost of services provided by staff and/or departments or functions. It is the means to substantiate and support how the costs of a program are charged to a particular cost category or to the Program so as to assure Head Start Grant funds ("Grant Fund") provided hereunder do not subsidize other program(s).

Contractor must ensure that costs allocated and charged to the Head Start grant are not charged to other Federal, State or Local awards to overcome fund deficiencies, to avoid restrictions imposed by law or terms of the Federal awards, or for other reasons. Contractor shall provide to City prior to the start of the Contract term (i) a matrix identifying the shared use of such facilities and/or program services; and (ii) the Cost Allocation Plan and supporting documentation, along with its Budget, financial statements and audit that are applicable to the Contractor's Project. City shall have the right to approve the Cost Allocation Plan.

- 4.10 Contractor agrees to reimburse the City for any Contractor overpayment based upon reconciled adjustments resulting from Contractor's balance and/or Statement of Revenue and Expenditure sheet as of June 2, 2017, which balance or Statement sheet shall be due to the City no later than June 17, 2017. Reimbursement shall be made within 20 calendar days of written notification to the Contractor of the need for reimbursement.
- 4.11 Upon expiration or early termination of this Contract, or at any time during the term of this Contract, all unused funds, rebates, advances exceeding allowable costs, or credits on-hand or collected thereafter relating to the Project, shall be immediately returned by Contractor to the City.
- 4.12 Upon execution of this Contract or at any time during the term of this Contract, the City's Director of Finance, the City Auditor, or a person designated by the Director of DHS may review and approve all Contractor's systems of internal accounting and administrative controls prior to the release of funds hereunder.

V. PROGRAM INCOME

- 5.1 For purposes of this Contract, "program income" shall mean earnings of Contractor realized from activities resulting from this Contract or from Contractor's management of funding provided or received hereunder. Such earnings shall include, but shall not be limited to, interest income; usage or rental/lease fees; income produced from Contract-supported services of individuals or employees or from the use of equipment or facilities of Contractor provided as a result of this Contract, and if applicable, payments from clients or third parties for services rendered by Contractor pursuant to this Contract. At the sole option of the Director of DHS, Contractor will either (a) be required to return program income funds to City through DHS, or (b) upon prior written approval by the Director of DHS, Contractor may be permitted to retain such funds to be:
 - (A) added to the Project and used to further eligible Project objectives, in which case proposed expenditures must first be approved by the City; or
 - (B) deducted from the total Project cost for the purpose of determining the net cost reimbursed by the City.
- 5.2 In any case where Contractor is required to return program income to DHS, Contractor must return such program income to City within the timeframe that may be specified by the Director of DHS. If the Director of DHS does not specify a timeframe for Contractor to return program income to City, then Contractor must return such program income to City on the same date that Contractor submits its statement of expenditures and revenues to DHS set forth in Article V, Section 5.4 of this Contract. If the Director of DHS grants Contractor authority to retain program income, Contractor must submit all reports required by DHS within the timeframe specified in the Contract.
- 5.3 Contractor shall provide DHS with thirty (30) calendar days written notice prior to the activity that generates program income. Such notice shall detail the type of activity, time, and place of all activities that generate program income.
- 5.4 The Contractor shall fully disclose and be accountable to the City for all program income. Contractor must submit a statement of expenditures and revenues to DHS within thirty (30) calendar days of the activity that generates program income. The statement is subject to audit verification by DHS. Failure by Contractor to report program income as required is grounds for suspension, cancellation, or termination of this Contract.

- 5.5 Contractor is prohibited from charging fees or soliciting donations and is prohibited from inviting or contracting with vendors who shall charge fees or solicit donations from Head Start participants and their parents in any Contract-funded project without the prior written approval of the Director of DHS.
- 5.6 Contractor shall include this Article V, in its entirety, in all of its subcontracts involving income-producing services or activities.

VI. ADMINISTRATION OF CONTRACT

- 6.1 The Contractor agrees to comply with all the terms and conditions that the City must comply with within its award document from HHS. A copy of said award document is attached hereto and incorporated herein for all purposes as Attachment IV. From time to time, the award document may be amended or supplemented, and these shall be incorporated into the Contract collectively as Attachment IV.
- 6.2 In the event that any disagreement or dispute should arise between the Parties hereto pertaining to the interpretation or meaning of any part of this Contract or its governing rules, regulations, laws, codes or ordinances, the City Manager or the Director of DHS, as representatives of the City and the parties ultimately responsible for all matters of compliance with HHS and City rules and regulations, shall have the final authority to render or secure an interpretation.
- 6.3 Contractor shall not use funds awarded from this Contract as matching funds for any federal, state or local grant without the prior written approval of the Director of DHS.
- 6.4 The City shall have the authority during normal business hours to make physical inspections to all operating facilities occupied by Contractor for the administration of this Contract and to require such physical safeguarding devices as locks, alarms, security / surveillance systems, safes, fire extinguishers, sprinkler systems, etc. to safeguard property and/or equipment authorized by this Contract.
- 6.5 The Contractor Board of Managers, as applicable, and Contractor's management staff shall adopt and approve an Employee Integrity Policy and internal program management procedures, and require all staff to abide by these and the Head Start standards as established in the HHS regulations, to preclude theft, embezzlement, improper inducement, obstruction of investigation or other criminal action, and to prevent fraud and program abuse. These policies and procedures shall require repayment of stolen or erroneously received grant funds or property to the Contractor, or to the applicable Head Start service provider from whom the grant funds or property was received or stolen, if other than the Contractor, and shall specify any other consequences to Contractor's employees and vendors involved in such illegal activities to include but not be limited to termination and prosecution where necessary. Said policies and procedures shall be provided to DHS upon request by DHS. In the event that DHS finds the policies and procedures to be lacking, DHS may recommend revision.
- 6.6 Contractor agrees to comply with the following check writing and handling procedures:
 - (A) No blank checks are to be signed in advance;
 - (B) No checks are to be made payable to cash or bearer with the exception of those for petty cash reimbursement, not to exceed a \$100.00 maximum per check. Contractor agrees that the aggregate amount of petty cash reimbursement shall not exceed \$500.00 for any given calendar month during the term of this Contract unless Contractor receives prior written approval from DHS to exceed such limit. Such requests for petty cash must be supported by the submission to DHS of an original receipt.
 - (C) Checks issued by City to Contractor shall be deposited into the appropriate bank account immediately or by the next Contractor business day after Contractor's receipt of each such check, and shall never be cashed for purposes of receiving the face amount back. If such check(s) are not deposited within the next Contractor business day from the date of issue, such checks shall be investigated by City and stoppayment orders issued, as applicable. Upon cancellation of any outstanding check, if deemed

- appropriate by City, such check may be reissued to the Contractor or if deemed by City not to be a valid expense, such check shall be immediately returned to the City.
- (D) For checks other than petty cash reimbursement, Contractor shall adopt and comply with a policy requiring no less than two (2) signatures of authorized representatives of Contractor on each check. Contractor understands and agrees that City's reimbursement is subject to compliance with this provision of the Contract.
- 6.7 The use of gift cards to defray any expenses under this Contract by the Contractor is not permitted

VII. AUDIT

7.1 If Contractor expends \$750,000.00 or more of funds provided under this Agreement, or cumulative funds provided by or through City, and does not have to comply with the provisions of Section 7.2, then during the term of this Agreement, the Contractor shall have completed an independent audit and the reporting required submitted to the City within the earlier of 30 calendar days after receipt of the auditor's report(s), or nine (9) months after the end of Contractor's fiscal year, expiration or early termination of this Contract, whichever is earlier. Contractor understands and agrees to furnish DHS a copy of the audit report including the corrective action plan(s) on all audit findings, a summary schedule of prior audit findings, management letter and/or conduct of audit letter, within thirty (30) calendar days upon receipt of said report or upon submission of said corrective action plan to the auditor.

Contractor agrees and understands that upon notification from federal, state, or local entities that have conducted program reviews and/or audits of the Contractor or its programs of any findings about accounting deficiencies, or violations of Contractor's financial operations, a copy of the notification, review, investigation, and audit violations report must be forwarded to DHS within a period of ten (10) calendar days upon the Contractor's receipt of the report.

7.2 Contractor agrees that if Contractor receives or expends more than \$750,000.00 in federal funds from the City, an audit shall be made in accordance with the Single Audit Act Amendments of 1996, the State of Texas Single Audit Circular, and the U.S. Office of Management and Budget Circular (Uniform Guidance). Contractor shall also be required to submit copies of its annual independent audit report, and all related reports issued by the independent certified public accountant within the earlier of 30 days after receipt of the auditor's report(s), or nine months after the end of the audit period, unless a longer period is agreed to in advance by the Federal cognizant or oversight agency for audit to the Federal Audit Clearinghouse. A copy of this report must also be provided to City within this same time period. Contractor may submit reports through the following website: http://harvester.census.gov/sac/_and may also contact the Clearinghouse by telephone at (301) 763-1551 (local), 1-888-222-9907 or 1-800-253-0696 (toll free).

Per 2 C.F.R. 200.36, upon completion of Form SF-SAC, Contractor may submit the completed report by mail to:

Federal Audit Clearinghouse Bureau of the Census 1201 E. 10th Street Jeffersonville, Indiana 47132

Contractor agrees to reimburse the City or supplement any disallowed costs with eligible and allowable expenses based upon reconciled adjustments resulting from Contractor's Single Audit. Reimbursement shall be made within thirty (30) calendar days of written notification regarding the need for reimbursement.

7.3 If Contractor expends less than \$750,000.00 of City dollars during the term of this Contract, then the Contactor shall complete and submit an unaudited financial statement(s) within a period not to exceed nine months immediately succeeding the end of Contractor's fiscal year, expiration or early termination of this Contract, whichever is earlier. Said financial statement shall include a balance sheet and income statement

prepared by a bookkeeper and a cover letter signed by Contractor attesting to the correctness of said financial statement.

- 7.4 All financial statement(s) must include a schedule of receipts and disbursements by budgeted cost category for each program funded by or through the City.
- 7.5 The City reserves the right to conduct, or cause to be conducted an audit or review of all funds received under this Contract at any and all times deemed necessary by City. The City Internal Audit Staff, a Certified Public Accounting (CPA) firm, or other personnel as designated by the City, may perform such audit(s) or reviews. The City reserves the right to determine the scope of every audit. In accordance herewith, Contractor agrees to make available to City all accounting and Project records.

Contractor shall during normal business hours, and as often as deemed necessary by City and/or the applicable state or federal governing agency or any other auditing entity, make available the books, records, documents, reports, and evidence with respect to all matters covered by this Contract and shall continue to be so available for a minimum period of three (3) years or whatever period is determined necessary based on the Records Retention guidelines, established by applicable law for this Contract. Said records shall be maintained for the required period beginning immediately after Contract termination, save and except there is litigation or if the audit report covering such agreement has not been accepted, then the Contractor shall retain the records until the resolution of such issues has satisfactorily occurred. The auditing entity shall have the authority to audit, examine and make excerpts, transcripts, and copies from all such books, records, documents and evidence, including all books and records used by Contractor in accounting for expenses incurred under this Contract, all contracts, invoices, materials, payrolls, records of personnel, conditions of employment and other data relating to matters covered by this Contract.

The City may, in its sole and absolute discretion, require the Contractor to use any and all of the City's accounting or administrative procedures used in the planning, controlling, monitoring and reporting of all fiscal matters relating to this Contract, and the Contractor shall abide by such requirements.

7.6 When an audit or examination determines that the Contractor has expended funds or incurred costs which are questioned by the City and/or the applicable state or federal governing agency, the Contractor shall be notified and provided an opportunity to address the questioned expenditure or costs.

Should any expense or charge that has been reimbursed be subsequently disapproved or disallowed as a result of any site review or audit, the Contractor will promptly refund such amount to the City no later than ten (10) days from the date of notification of such disapproval or disallowance by the City. At its sole option, DHS may instead deduct such claims from subsequent reimbursements; however, in the absence of prior notice by City of the exercise of such option, Contractor shall provide to City a full refund of such amount no later than ten (10) days from the date of notification of such disapproval or disallowance by the City. If Contractor is obligated under the provision hereof to refund a disapproved or disallowed cost incurred, such refund shall be required and be made to City by check, cashier's check or money order. Should the City, at its sole discretion, deduct such claims from subsequent reimbursements, the Contractor is forbidden from reducing Project expenditures and Contractor must use its own funds to maintain the Project.

Contractor agrees and understands that all expenses, fees, fines and penalties associated with the collection of delinquent debts owed by Contractor shall be the sole responsibility of the Contractor and shall not be paid from any Project funds received by the Contractor under this Contract. Delinquent debts that would otherwise be identified as allowable costs may be paid with Project funds after written approval of DHS.

7.7 If the City determines, in its sole discretion, that Contractor is in violation of the above requirements, the City shall have the right to dispatch auditors of its choosing to conduct the required audit and to have the Contractor pay for such audit from non-City resources.

VIII. RECORDS, REPORTING, MONITORING AND INTELLECTUAL PROPERTY

- 8.1 In addition to those listed in this Contract, the Contractor shall submit to DHS any and all reports as may be required by HHS or the City. Contractor shall incorporate and use any City approved tracking or information system (e.g., ChildPlus) for the delivery of comprehensive Head Start Services and collect, input and update all data in accordance with the City's planned timeline to ensure the reporting of accurate and consistent information to HHS.
- 8.2 a) Additionally, Contractor shall maintain and furnish to City programmatic information and reports, in such forms as the City may prescribe, as required under the Head Start Act, as amended, and federal regulations such as 45 C.F.R. Part 74, 45 C.F.R. Part 92, or 45 C.F.R. Part 75, as applicable:
 - b) Contractor shall maintain all applicable and appropriate supporting documentation of costs, including but not limited to, payroll records, invoices, contracts or vouchers, and make these available to City upon request.
- 8.3 City reserves the right to request Contractor to provide additional records for travel expenses, long distance calls, faxes, internet service, cell phone calls, or other electronic communication devices charged to the budget associated with this Contract.
- 8.4 If applicable, Contractor shall report all notices served, violations found or complaints filed with regard to licensing, or lack thereof, of Contractor's centers within one (1) business day of receipt of the notice, violation or complaint.
- 8.5 Contractor shall comply with Head Start Performance Standards and regulations, the Head Start Act, and all applicable Federal, State and Local laws relating to child safety. Contractor must establish and implement policies and procedures to respond to medical, dental and other emergencies with which all staff should be familiar and trained. These policies and procedures must include, among other things, methods of notifying parents in the event of an emergency involving their child and established methods for handling cases of suspected or known child endangerment, abuse or neglect that are in compliance with applicable Federal, State, or Local laws. Contractor shall notify the City immediately, but no later than 24 hours, of any instances of actual or suspected cases of child endangerment, neglect or abuse, or in case of a program emergency or likely negative media coverage.
- 8.6 Within a period not to exceed forty-five (45) calendar days after the expiration or early termination date of the Contract, Contractor shall submit all final client reports and all required deliverables to City. Contractor understands and agrees that in conjunction with the submission of the final report, the Contractor shall execute and deliver to City a receipt for all sums received and a release of all claims against the Project.
- 8.7 Contractor shall maintain financial records, supporting documents, statistical records, and all other books, documents, papers or other records pertinent to this Contract or the Grant in accordance with the official records retention schedules established within the Local Government Records Act of 1989 and any amendments thereto, or for such period as may be specifically required by 45 C.F.R §75.361, as applicable, whichever is longer. Notwithstanding the foregoing, Contractor shall maintain all Contract and Grant related documents for no less than four (4) years from the date of City's submission of the annual financial report covering the funds awarded hereunder. If an audit, litigation, or other action involving the records has been initiated before the end of the four (4) year period, Contractor agrees to maintain the records until the end of the four (4) year period or until the audit, litigation, or other action is completed, whichever is later. Records for real property and equipment acquired with Head Start funds shall be retained for four (4) years after final disposition.
- 8.8 Contractor shall make available to City, HHS, or any of their duly authorized representatives, upon appropriate notice, such books, records, reports, documents, papers, policies and procedures as may be necessary for audit, examination, excerpt, transcription, and copy purposes, for as long as such records,

reports, books, documents, and papers are retained. This right also includes timely and reasonable access to Contractor's facility and to Contractor's personnel for the purpose of interview and discussion related to such documents. Contractor shall, upon request, transfer certain records to the custody of City or HHS when City or HHS determines that the records possess long-term retention value.

- 8.9 The Contractor agrees to incorporate and use any City approved tracking or information system for the delivery of comprehensive Head Start services. Contractor shall enter current, accurate and complete client data.
- B.10 DHS is assigned monitoring, fiscal control, and evaluation of certain projects funded by the City with General or Grant Funds, including the Project covered by this Contract. Therefore, Contractor agrees to permit City and/or HHS to evaluate, through monitoring, reviews, inspection or other means, the quality, appropriateness, and timeliness of services delivered under this Contract and to assess Contractor's compliance with applicable legal and programmatic requirements. At such times and in such form as may be required by DHS, the Contractor shall make available to DHS and the Grantor of the Grant Funds, if applicable, such statements, reports, records, personnel files (including evidence of criminal background check(s) as required by Head Start regulations), client files, data, all policies and procedures and information as may be requested by DHS and shall permit the City and Grantor of the Grant Funds, if applicable, to have interviews with its personnel, board members and program participants pertaining to the matters covered by this Contract. Contractor agrees that the failure of the City to monitor, evaluate, or provide guidance and direction shall not relieve the Contactor of any liability to the City for failure to comply with the Terms of the Grant or the terms of this Contract.
- 8.11 City may, at its discretion, conduct periodic, announced monitoring visits to ensure program and administrative compliance with Head Start Performance Standards, the Head Start Act, and with Program goals and objectives for the contract period. City reserves the right to make unannounced visits to Contractor Program sites when it is determined that such unannounced visits are in the interest of effective program management and service delivery.
- 8.12 Contractor understands that the City will inform Contractor in a timely manner of the findings of any such review or monitoring, specifically any default under the Contract or deficiencies in performance, and will also inform Contractor in writing of Program strengths and weaknesses and specify a deadline for corrective action when necessary. The City will assist Contractor in finding solutions for Program improvement if and as appropriate.
- 8.13 Unless otherwise provided herein, all reports, statements, records, data, policies and procedures or other information requested by DHS shall be submitted by Contractor to City within five (5) working days of the request. The Parties agree that a shorter time frame may be necessary for response in the case of the single audit and shall cooperate to meet deadlines necessary to comply with the single audit requirements. In the event that Contractor fails to deliver the required reports or information or delivers incomplete information within the prescribed time period, the City may, upon reasonable notice, suspend reimbursements to Contractor until such reports are delivered to City. Furthermore, the Contractor ensures that all information contained in all required reports or information submitted to City is accurate.
- 8.14 (A) Unless disclosure is authorized by the City, Contractor agrees to maintain in confidence all information pertaining to the Project or other information and materials prepared for, provided by, or obtained from City including, without limitation, reports, information, project evaluation, project designs, data, other related information (collectively, the "Confidential Information") and to use the Confidential Information for the sole purpose of performing its obligations pursuant to this Contract. Contractor shall protect the Confidential Information and shall take all reasonable steps to prevent the unauthorized disclosure, dissemination, or publication of the Confidential Information.
 - (B) However, if disclosure is permitted by law or required by order of a governmental agency or court of competent jurisdiction, Contractor shall give the Director of DHS prior written notice that such disclosure is required with a full and complete description regarding such requirement.

- (C) Contractor shall establish specific procedures designed to meet the obligations of this Article VIII, Section 8.14, including, but not limited to execution of confidential disclosure agreements, regarding the Confidential Information with Contractor's employees and subcontractors prior to any disclosure of the Confidential Information. This Article VIII, Section 8.14 shall not be construed to limit HHS's, the City's or its authorized representatives' right to obtain copies, review and audit records or other information, confidential or otherwise, under this Contract. Upon termination or expiration of this Contract, Contractor shall return to City all copies of materials related to the Project, including the Confidential Information. All confidential obligations contained herein (including those pertaining to information transmitted orally) shall survive the termination of this Contract. The Parties shall ensure that their respective employees, agents, and contractors are aware of and shall comply with the aforementioned obligations.
- 8.15 The Public Information Act, Government Code Section 552.021, requires the City and Contractor to make public information available to the public. Under Government Code Section 552,002(a), public information means information that is written, produced, collected, assembled or maintained under a law or ordinance or in connection with the transaction of official business: 1) by a governmental body; or 2) for a governmental body and the governmental body owns the information has a right of access to it, or has spent or contributed public money for the purpose of its writing, production, collection, assembly or maintenance.. Therefore, if Contractor or City receives inquiries regarding documents within its possession pursuant to this Contract, the Party receiving such request shall (a) within twenty-four (24) hours of receiving the requests forward such requests to the other Party for notification purposes and to afford the other Party the opportunity to assert any applicable arguments or protections necessary to protect its information, and (b) take action as authorized under the Public Information Act to protect information that may be confidential pursuant to State or Federal law. If the requested information is confidential pursuant to State or federal law, the Party receiving such request shall submit to the other Party the list of specific statutory authority mandating confidentiality no later than three (3) business days of Contractor's receipt of such request.

For the purposes of communicating and coordinating with regard to public information requests, all communications shall be made to the designated public information liaison for each Party. Each Party shall designate in writing to the other Party the public information liaison for its organization and notice of a change in the designated liaison shall be made promptly to the other Party.

8.16 In accordance with Texas law, Contractor acknowledges and agrees that all local government records as defined in Chapter 201, Section 201.003 (8) of the Texas Local Government Code created or received in the transaction of official business or the creation or maintenance of which were paid for with public funds are declared to be public property and subject to the provisions of Chapter 201 of the Texas Local Government Code and Subchapter J, Chapter 441 of the Texas Government Code. Thus, Contractor agrees that no such local government records produced by or on the behalf of Contractor pursuant to this Contract shall be the subject of any copyright or proprietary claim by Contractor.

Contractor acknowledges and agrees that all local government records, as described herein, produced in the course of the work required by this Contract, shall belong to and be the property of City and shall be made available to the City at any time. Contractor further agrees to turn over to City all such records upon termination of this Contract. Contractor agrees that it shall not, under any circumstances, release any records created during the course of performance of the Contract to any entity without the written permission of the Director of DHS, unless required to do so by a court of competent jurisdiction. DHS shall be notified of such request as set forth in Article VIII, Section 8.15 of this Contract.

8.17 Ownership of Intellectual Property. Contractor and City agree that the Project shall be and remain the sole and exclusive proprietary property of City. The Project shall be deemed a "work for hire" within the meaning of the copyright laws of the United States, and ownership of the Project and all rights therein shall be solely vested in City. Contractor hereby grants, sells, assigns, and conveys to City all rights in and to the Project and the tangible and intangible property rights relating to or arising out of the Project, including, without limitation, any and all copyright, patent and trade secret rights. All intellectual property rights including, without limitation, patent, copyright, trade secret, trademark, brand names, color schemes, designs, screens, displays, user interfaces, data structures, organization, sequences of operation, trade dress,

and other proprietary rights (the "Intellectual Property Rights") in the Project shall be solely vested in City. As owner of the tangible and intangible intellectual property, City shall have the right to reproduce, publish, authorize others to reproduce or publish, or otherwise use such material. Contractor agrees to execute all documents reasonably requested by City to perfect and establish City's right to the Intellectual Property Rights. In the event City shall be unable, after reasonable effort, to secure Contractor's signature on any documents relating to Intellectual Property Rights in the Project, including without limitation, any letters patent, copyright, or other protection relating to the Project, for any reason whatsoever, Contractor hereby irrevocably designates and appoints City and its duly authorized officers and agents as Contractor's agent and attorney-in-fact, to act for and in Contractor's behalf and stead to execute and file any such application or applications and to do all other lawfully permitted acts to further the prosecution and issuance of letters patent, copyright or other analogous protection thereon with the same legal force and effect as if executed by Contractor. Provided, however, nothing herein contained is intended nor shall it be construed to require Contractor to transfer any ownership interest in Contractor's best practice and benchmarking information to the City.

- 8.18 In the event that Contractor desires to copyright material or to permit any third-party to do so, Contractor must obtain City's prior written approval to do so and must appropriately acknowledge City's support in any such materials.
- 8.19 Subject to obligations to maintain confidentiality under the Health Insurance Portability and Accountability Act of 1996 ("HIPAA"), the HIPAA Business Associate Agreement (attached hereto and incorporated herein as Attachment V), and the limitations imposed by each regarding transfer of information, any and all writings, documents or information in whatsoever form and character produced by Contractor pursuant to the provisions of this Contract is the exclusive property of City; and no such writing, document or information shall be the subject of any copyright or proprietary claim by Contractor. Contractor understands and acknowledges that as the exclusive owner of any and all such writings, documents and information, City has the right to use all such writings, documents and information as City desires, without restriction.

IX. INSURANCE AND INDEMNIFICATION

9.1 Insurance

- (a) Contractor and City each maintain a self-insurance fund for general liability and worker's compensation claims and causes of action to meet their statutory obligations to each party's employees.
- (b) It is the stated policy of both City and Contractor not to acquire commercial general liability insurance for torts committed by employees of the state who are acting within the scope of their employment. Rather, Chapter 101 of the Civil Practice and Remedies Code states that a governmental unit in the state is liable for property damage, personal injury and death proximately caused by the wrongful act or omission or negligence of an employee acting within his scope of employment. Liability of the state government under this chapter is limited to money damages in a maximum amount of \$250,000.00 for each person and \$500,000.00 for each single occurrence for bodily injury or death and \$100,000.00 for each single occurrence for injury to or destruction of property. Employees of the City are provided Workers' Compensation coverage under a self-insuring, self-managed program as authorized by the Texas Labor Code, Chapter 503.

9.2 Indemnification.

Contractor acknowledges that is a political subdivisions of the State of Texas and is subject to, and complies with the applicable provisions of the Texas Tort Claims Act, as set out in the Civil Practices and Remedies Code, Section 101.001 et seq. and the remedies authorized therein regarding claims or causes of action that may be asserted by third parties, including but not limited to those resulting or arising from any and all injuries or death of any person or damage to any property arising from or related to this Contract. Each

Party shall promptly notify the other in writing of any claim or demands that become known against them in relation to or arising out of activities under this Contract.

9.3 Acts and Omissions.

City and Contractor both agree to accept and each entity is responsible for its own acts and omissions in providing services pursuant to this Contract as well as those acts or omissions of its employees and nothing in this Contract shall be construed to place any responsibility for such acts or omissions onto the other Party.

X. THIS SECTION INTENTIONALLY LEFT BLANK

XI, APPLICABLE LAWS

- 11.1 Contractor, and all of the work performed under this Contract, shall comply with all applicable laws, rules, regulations and codes of the United States and the State of Texas and with the charter, ordinances, bond ordinances, and rules and regulations of the City of San Antonio and Bexar County. Contractor agrees to abide by any and all future amendments or additions to such laws, rules, regulations, policies and procedures as they may be promulgated.
- 11.2 The Contractor understands that certain funds provided it pursuant to this Contract are funds which have been made available by the City's General Operating Budget and/or by Federal, State, or other granting entities. Consequently, Contractor agrees to comply with all laws, ordinances, codes, rules, regulations, policies, and procedures, including all licensing standards and all applicable accreditation standards, applicable to the funds received by Contractor hereunder as directed by the City or as required in this Contract, including but not limited to:
 - (A) The Head Start Act (42 U.S.C. §9801 et seq., as amended);
 - (B) 45 C.F.R. Part 1301 et seg.;
 - (C) The Terms of the HHS Grant;
 - (D) As applicable, 45 C.F.R. 75 ("Uniform Administrative Requirements, Cost Principles, and Audit Requirements for HHS Awards");
 - (E) Texas Child Care Licensing laws;
 - (F) The Office of Management and Budget (OMB) Circular at 2 C.F.R. 200 et. al. titled "Uniform Administrative Requirements, Cost Principles, and Audit Requirements for Federal Awards" (Uniform Guidance), as applicable;
 - (G) Official record retention schedules as established by the Local Government Records Act of 1989; and
 - (H) The Texas Public Information Act, at Chapter 552, The Texas Government Code.
- 11.3 Contractor further understands and agrees:
 - (A) to comply with all applicable standards, orders or regulations issued pursuant to the Clean Air Act (42 U.S.C. §§7401-7671q) and the Federal Water Pollution Control Act (33 U.S.C. §§1251-1387), as amended. Contractor agrees to report each violation to the City and understands that the City will, in turn, report each violation as required to the federal agency providing funds for this Contract and the appropriate EPA Regional Office. Additionally, Contractor agrees to include these requirements in each subcontract to this Contract exceeding \$150,000.00 financed in whole or in part with federal funds.

- (B) to make positive efforts to utilize small businesses, minority-owned firms and women's business enterprises in connection with the work performed hereunder, whenever possible.
- (C) to provide for the rights of the Federal Government in any invention resulting from the work performed hereunder, in accordance with 37 C.F.R. Part 401 and any applicable implementing regulations.
- (D) to include a provision requiring compliance with the Copeland "Anti-Kickback" Act (40 U.S.C 3145) and as supplemented by Department of Labor regulations at 29 C.F.R. Part 3, "Contractors and Subcontractors on Public Building or Public Work Financed in Whole or In Part by Loans or Grants from the United States."
- (E) to comply with the Davis-Bacon Act, as amended (40 U.S.C. 3141-3144, and 3146-3148), and as supplemented by Department of Labor regulations at 29 C.F.R. Part 5, implementing regulations, and the relevant Additional OMB Provisions attached hereto and incorporated herein for all purposes as Attachment VI, and to include a provision requiring compliance with the each in any construction contracts of more than \$2,000.00, and report all suspected or reported violations to HHS.
- (F) to comply with the certification and disclosure requirements of the Byrd Anti-Lobbying Amendment (31 U.S.C. § 1352), and any applicable implementing regulations. Contractor verifies it has tendered said Certificate to the City.
- (G) to comply with the applicable standards under the McKinney-Vento Homeless Assistance Act (42 U.S.C. §§11431 et seq.), and any applicable implementing regulations, as may be applicable.
- (H) to comply with the Contract Work Hours and Safety Standards Act (40 USC 3701-3708), as supplemented by Department of Labor regulations (29 CFR Part 5), relating to all contracts that involve the employment of mechanics or laborers, and the relevant provisions in Attachment VI, which provides, in part, that each contractor must be required to compute the wages of every mechanic and laborer on the basis of a standard work week of 40 hours, that work in excess of the standard work week be compensated at a rate at least one and a half times the basic rate of pay, and that no laborer or mechanic must be required to work under conditions which are unsanitary, hazardous or dangerous.
- (I) to comply with the prohibitions contained in the Pro-Children Act of 1994 (20 U.S.C §6081-84), relating to no smoking within any indoor facility (or portion thereof) owned or leased or contracted for by Contractor for the provision of regular or routine health care or day care or early childhood development (Head Start) services to children or for the use of the employees of the City or Contractor who provide such services.
- (J) that Contractor and its subcontractors shall comply with section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act, including, but not limited to, the regulatory provisions of 40 CFR Part 247, and Executive Order 12873, as they apply to the procurement of the items designated in Subpart B of 40 CFR Part 247.
- 11.4 The Contractor certifies that it will provide a drug-free workplace in compliance with the Drug-Free Workplace Act of 1988 (41 U.S.C. §§ 701-707 and 8101-8106, as amended). Failure to comply with the above-referenced law and regulations could subject the Contractor to suspension of payments, termination of Contract, and debarment and suspension actions.
- 11.5 Contractor shall comply with all federal, state, or local laws, rules, and orders prohibiting discrimination, and shall not engage in employment practices which have the effect of discriminating against any employee or applicant for employment. As a party to this Contract, Contractor understands and agrees to comply with the Non-Discrimination Policy of the City of San Antonio contained in Chapter 2, Article X of the City Code and further, shall not discriminate on the basis of race, color, religion, national origin, sex, sexual orientation, gender identity, veteran status, age or disability, unless exempted by state or federal law, or as

otherwise established herein. Consistent with the foregoing, Contractor agrees to comply with Executive Order 11246, entitled "Equal Employment Opportunity", as amended by Executive Order 11375, and as supplemented by regulations at 41 C.F.R. Part 60. Additionally, Contractor certifies that it will comply fully with the following nondiscrimination, minimum wage and equal opportunity provisions, including but not limited to:

- (A) Title VII of the Civil Rights Act of 1964, as amended;
- (B) Section 504 of the Rehabilitation Act of 1973, as amended;
- (C) The Age Discrimination Act of 1975, as amended;
- (D) Title IX of the Education Amendments of 1972, as amended; (Title 20 USC sections 1681-1688);
- (E) Fair Labor Standards Act of 1938, as amended;
- (F) Equal Pay Act of 1963, P.L. 88-38; and
- (G) All applicable regulations implementing the above laws.
- 12.5.1 Further, if Contractor engages in any contract that, except as otherwise provided under 41 C.F.R. Part 60, meets the definition of "federally assisted construction contract" in 41 C.F.R. Part 60-1.3, Contractor must comply with the Equal Employment Opportunity provisions in Attachment VI and all of the Executive Order and Code of Federal Regulations provisions previously cited in Section 12.5, and must include the provisions in any of its subcontracts.
- 11.6 The Contractor warrants that any and all taxes that the Contractor may be obligated for, including but not limited to, federal, state, and local taxes, fees, special assessments, Federal and State payroll and income taxes, personal property, real estate, sales and franchise taxes, are current, and paid to the fullest extent liable as of the execution date of the Contract. The Contractor shall comply with all applicable local, State, and Federal laws including, but not limited to, related to:
 - (A) worker's compensation;
 - (B) unemployment insurance;
 - (C) timely deposits of payroll deductions;
 - (D) filing of Information on Tax Return form 990 or 990T, Quarterly Tax Return Form 941, W-2's Form 1099 on individuals who received compensation other than wages, such as car allowance, Forms 1099 and 1096 for contract or consultant work, non-employee compensation, etc;
 - (E) Occupational Safety and Health Act regulations; and
 - (F) Employee Retirement Income Security Act of 1974, P.L. 93-406.
- 11.7 Contractor agrees to comply with the Americans with Disabilities Act of 1990, 42 U.S.C. 12101 et seq., and all regulations thereunder.
- 12.8 Contractor and its subcontractors shall comply with section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act, including, but not limited to, the regulatory provisions of 40 C.F.R. Part 247, and Executive Order 12873, as they relate to the procurement of recovered materials, as required by 2 CFR 200.322 and as they apply to the procurement of the items designated in Subpart B of 40 CFR Part 247.
- 11.9 All expenditures by the Contractor or any of its subcontractors must be made in accordance with all applicable federal, state and local laws, rules and regulations.
- 11.10 If applicable, Contractor shall submit to DHS its most recent form 990 or 990T and also submit any that are filed with the Internal Revenue Service subsequent to its last submission to the City if filed during the term of the Contract.

XII. NO SOLICITATION/CONFLICT OF INTEREST

- 12.1 The Contractor warrants that no person or selling agency or other organization has been employed or retained to solicit or secure this Contract upon a contract or understanding for a commission, percentage, brokerage, or contingent fee and further that no such understanding or agreement exists or has existed with any employee of the Contractor or the City. For breach or violation of this warrant, the City shall have the right to terminate this Contract without liability or, at its discretion, to deduct from the Contract or otherwise recover the full amount of such commission, percentage, brokerage, or contingent fee, or to seek such other remedies as legally may be available.
- 12.2 Contractor covenants that neither it nor any member of its governing body or of its staff presently has any interest, direct or indirect, which would conflict in any manner or degree with the performance of services required to be performed under this Contract. Contractor further covenants that in the performance of this Contract, no persons having such interest shall be employed or appointed as a member of its governing body or of its staff.
- 12.3 Contractor further covenants that no member of its governing body or of its staff shall possess any interest in, or use their position for, a purpose that is or gives the appearance of being motivated by desire for private gain for themselves or others, particularly those with which they have family, business, or other ties.
- 12.4 No member of City's governing body or of its staff who exercises any function or responsibility in the review or approval of the undertaking or carrying out of this Contract shall:
 - (A) Participate in any decision relating to this Contract which may affect his or her personal interest or the interest of any corporation, partnership, or association in which he or she has a direct or indirect interest; or
 - (B) Have any direct or indirect interest in this Contract or the proceeds thereof.
- 12.5 Contractor acknowledges that it is informed that Charter of the City of San Antonio and its Ethics Code prohibit a City officer or employee, as those terms are defined in Section 2-52 of the Ethics Code, from having a financial interest in any contract with the City. An officer or employee has "prohibited financial interest" in a contract with the City or in the sale to the City of land, materials, supplies or service, if any of the following individual(s) or entities is a party to the contract or sale: A City officer or employee; his parent, child or spouse; an entity in which the officer or employee, or his parent, child or spouse owns ten (10) percent or more of the voting stock or shares of the entity, or ten (10) percent or more of the fair market value of the entity; an entity in which any individual or entity above listed is subcontractor on a City contract, a partner or a parent or subsidiary business entity.
- 12.6 Contractor warrants and certifies, and this Contract is made in reliance thereon, that Contractor is a public entity and Contractor's representative further warrants and certifies that no City officer or employee nor any spouse, parent, child, sibling of first-degree relative of a City officer or employee owns ten (10) percent or more of the voting stock or shares of Contractor, or ten (10) percent or more of the fair market value of the Contractor. Contractor further warrants and certifies that is has tendered to the City a Discretionary Contracts Disclosure Statement in compliance with the City's Ethics Code.

XIII. TERMINATION

- 13.1 (a) Termination for Cause Upon written notice in accordance with Article XXVI, Official Communication, CITY may terminate this Contract as of the date provided in the notice in whole or in part, upon the occurrence of either:
 - (i) Contractor's failure to fulfill, in a timely and proper manner, obligations under this Contract to
 include performance standards established by the City, or if the Contractor should violate any of
 the covenants, conditions, or stipulations of the Contract, or

- (ii) Notification of any investigation, claim or charge by a local, state or federal agency involving fraud, theft or the commission of a felony by Contractor or Contractor's employee. Contractor shall have the opportunity to cure via the immediate termination and/or removal of the employee from the Head Start Program..
- (b) Termination for Convenience This Contract may be terminated in whole or in part upon providing notice in accordance with Article XXVI, Official Communication, which shall specify a date, not sooner than 120 days following the day on which notice is sent, unless earlier terminated under any other provision herein.
- 13.2 The Contractor shall be entitled to receive just and equitable compensation for any work satisfactorily completed prior to such termination date. The question of satisfactory completion of such work shall be determined by the City alone, and its decision shall be final. It is further expressly understood and agreed by the Parties that Contractor's performance upon which final payment is conditioned shall include, but not be limited to, the Contractor's complete and satisfactory performance, of its obligations for which final payment is sought.
- 13.3 Notwithstanding any other remedy contained herein or provided by law, the City may delay, suspend, limit, or cancel funds, rights or privileges herein given the Contractor for failure to comply with the terms and provisions of this Contract. Specifically, at the sole option of the City, the Contractor may be placed on probation during which time the City may withhold reimbursements in cases where it determines that the Contractor is not in compliance with this Contract. The Contractor shall not be relieved of liability to the City for damages sustained by the City by virtue of any breach of this Contract, and the City may withhold funds otherwise due as damages, in addition to retaining and utilizing any other remedies available to the City.
- 13.4 If an employee of Contractor is discharged or otherwise leaves employment with Contractor, then the Contractor shall pay in full to such employee all of such employee's earned salaries and wages, within the timeframe specified by law.
- 13.5 Should the Contractor be debarred by federal government or the City pursuant to a debarment policy currently existing or hereafter adopted, said debarment may be grounds for termination.
- 13.6 This Contract is subject to the availability of federal grant funds to City and may be terminated by the City if HHS terminates the City as a grantee or reduces the amount granted to City, for any reason; provided that, if the reduction of grant funds does not result in complete unavailability of such funds, the Parties will use best efforts to amend this Contract accordingly. City will promptly notify Contractor of any such HHS action.
- 13.7 In all instances of termination, Contractor shall not incur new obligations after the effective date of termination, and shall cancel as many outstanding obligations as possible. Contractor shall submit to City all required reports including a final financial statement which shall be a statement of all expenditures incurred by Contractor under this Contract. City shall pay Contractor the full cost of obligations that City determines were not subject to cancellation if such costs are properly documented, allowable, within the approved budget, and unavoidably incurred by Contractor prior to termination or expiration. The foregoing shall constitute full and complete reimbursement for all of Contractor's performance under this Contract.

XIV. PROHIBITION OF POLITICAL ACTIVITIES

- 14.1 Contractor agrees that no funds provided from or through the City shall be contributed or used to conduct political activities for the benefit of any candidate for elective public office, political party, organization or cause, whether partisan or non-partisan, nor shall the personnel involved in the administration of the Project provided for in this Contract be assigned to work for or on behalf of any partisan or non-partisan political activity.
- 14.2 Contractor agrees that no funds provided under this Contract may be used in any way to attempt to influence, in any manner, a member of Congress or any other State or local elected or appointed official.

- 14.3 The prohibitions set forth in Article XIV, Sections 14.1 and 14.2 of this Contract include, but are not limited to, the following:
 - (A) an activity to further the election or defeat of any candidate for public office or for any activity undertaken to influence the passage, defeat or final content of local, state or federal legislation;
 - (B) working or directing other personnel to work on any political activity during time paid for with City funds, including, but not limited to activities such as taking part in voter registration drives, voter transportation activities, lobbying, collecting contributions, making speeches, organizing or assisting at meetings or rallies, or distributing political literature;
 - (C) coercing personnel, whether directly or indirectly, to work on political activities on their personal time, including activities such as taking part in voter registration drives, voter transportation activities, lobbying, collecting contributions, making speeches, organizing or assisting at meetings or rallies, or distributing political literature; and
 - (D) using facilities or equipment paid for, in whole or in part with City funds for political purposes including physical facilities such as office space, office equipment or supplies, such as telephones, computers, fax machines, during and after regular business hours.
- 14.4 To ensure that the above policies are complied with, Contractor shall provide every member of its personnel paid out of Contract funds with a statement provided by Contractor of the above prohibitions and have each said individual sign a statement acknowledging receipt of the policy. Such statement shall include a paragraph that directs any staff person who has knowledge of violations or feels that he or she has been pressured to violate the above policies to call and report the same to DHS. Contractor shall list the name and number of a contact person from DHS on the statement that Contractor's personnel can call to report said violations.
- 14.5 Contractor agrees that in any instance where an investigation of the above is ongoing or has been confirmed, salaries paid to the Contractor under this Contract may, at the City's discretion, be withheld until the situation is resolved, or the appropriate member of the Contractor's personnel is terminated.
- 14.6 This Article shall not be construed to prohibit any person from exercising his or her right to express his or her opinion or to limit any individual's right to vote. Further, Contractor and staff members are not prohibited from participating in political activities on their own volition, if done during time not paid for with Contract funds.

XV. PERSONNEL

- 15.1 Contractor shall establish and maintain an organizational structure that supports the accomplishment of Program objectives, addresses the major functions and responsibilities assigned to each staff position and provides evidence of adequate mechanisms for staff supervision and support to ensure the effective oversight of the Head Start program operations. Contractor shall ensure that, at a minimum, the program management functions listed in the Scope of Work are assigned to and adopted by staff within the Program.
- 15.2 Contractor understands that the City shall periodically perform its own wage and salary comparison and shall issue such results to Contractor. Contractor understands and agrees that City shall have no obligation to reimburse Contractor in excess of wages to an employee that exceed the average rate of compensation paid to persons providing substantially comparable services in the area. For purposes of this Contract, the City shall accept the wage information set forth in the most recent study commissioned by, or issued by, the Texas Association of School Boards, in determining the rate of reimbursement. Although the City may consider factors such as training and experience as affecting compensation levels, the City shall have the sole and absolute authority to determine the rate of City's reimbursement under the Contract and its decision shall be final due to the City's obligation of ensuring that wage comparability studies meet the

- requirements of the Head Start Act and implementing regulations. Subject to the restriction set forth in 15.3, Contractor may compensate its employees above the rate the City will reimburse, so long as the additional compensation is not charged to the Contract budget.
- 15. 3 Contractor expressly understands and agrees that in accordance with 42 U.S.C. §9848, no portion of the Contract funds provided hereunder may be used to pay its employee if compensation (including non-federal funds) to that employee exceeds \$179,700.00, or the maximum authorized compensation as may be adjusted from time to time. Furthermore, Contractor agrees that all employees must devote to the Head Start Program the time proportionate to the percentage of their compensation funded through the Head Start program grant (e.g., employees who are one hundred percent (100%) funded through the Head Start grant must devote one hundred percent (100%) of their time and effort to support the Head Start program). Contractor agrees to submit employee certifications if requested by the City or HHS.
- 15.4 The Contractor agrees to establish internal procedures that assure employees of an established complaint and grievance process. The grievance process will include procedures to receive, investigate, and resolve complaints and grievances in an expeditious manner.
- 15.5 Contractor agrees to comply with all applicable federal regulations regarding the setting of, and maximum amount allowable for, salaries of Contractor's employees.
- 15.6 Contractor agrees that all copies of written job descriptions will be filed in all individual personnel folders for each position in the organization.
- 15.7 The Contractor agrees to provide the City with the names and license registration of any employees of Contractor regulated by State law whose activities contribute towards, facilitate, or coordinate the performance of this Contract.
- 15.8 At the sole discretion of the Director of DHS, Contractor may be reimbursed by City for the cost of pay granted to full time, permanent employees that is not chargeable to annual or personal leave only for the reasons listed below:
 - (A) To attend annual training in a branch of the Armed Services, not to exceed fifteen (15) business days during the term of this Contract;
 - (B) To serve as a juror;
 - (C) To attend the funeral of someone in the immediate family. Immediate family shall include father, step-father, father-in-law, mother, step-mother, mother-in-law, sister, step-sister brother, step-brother, spouse, child, and other relative who was the legal guardian of the employee or for whom the employee had legal guardianship. In such event, the Contractor may grant up to three (3) work days of leave with pay that is not chargeable to annual or personal leave; or
 - (D) To attend seminars or workshops.
- 15.9 Chief Executive Officers (CEOs), directors and other supervisory personnel of Contractor may not supervise a spouse, parents, children, brothers, sisters, and in-laws standing in the same relationship, (hereinafter referred to as "Relatives") who are involved in any capacity with program delivery supported through Contract funds. Relatives, however, may be co-workers in the same Project in a non-supervisory position.

XVI. ADVERSARIAL PROCEEDINGS

16.1 Except in circumstances where the following is in conflict with federal law or regulations pertaining to the Head Start Program, Contractor agrees to comply with the following special provisions:

- (A) Under no circumstances will the funds received under this Contract be used, either directly or indirectly, to pay costs or attorney fees incurred in any adversarial proceeding against the City or any other public entity; and
- (B) Contractor, at the City's option, could be ineligible for consideration to receive any future funding while any adversarial proceeding against the City remains unresolved.

XVII. FEDERAL AND CITY-SUPPORTED PROJECT

- This Section is applicable to all publicity, presentations, signs, press releases, public notices, flyers, brochures, marketing materials, and other informational material prepared, created, posted and/or disseminated during the term of the Agreement by Contractor. Contractor shall obtain City's prior approval of the language and logo, as applicable, to be used, and the Parties agree that all publicity regarding the establishment or operation of the Head Start Program affiliation between City and Contractor described herein shall be planned and implemented as mutually agreed to in advance by the Parties. Contractor agrees that all public notices and any publicity, signs and/or marketing materials regarding any program which is funded by this Contract shall provide a written statement in a form approved or provided by City acknowledging the role of the Federal funds provided by HHS through City hereunder. These public notices or signs include, but are not limited to, signs identifying the facilities from which these programs are provided, and electronic media. In addition, all publicity related to Contractor's Head Start services shall note that the Program is operated on a non-discriminatory basis.
- 17.2 Contractor further agrees to provide City with a copy of all proposed communications to the public, Head Start parents and employees as it may relate to the City's implementation of the City's Head Start Program model, the transition of Head Start contracts or transition of the Head Start Program, and to obtain the City's approval prior to dissemination.

XVIII. PROPERTY, EQUIPMENT AND SUPPLIES

- 18.1 The City retains ownership of all equipment/property purchased with funds received through the City and such equipment/property shall, at the City's sole option, revert to the City at Contract's termination, for whatever reason. The Contractor agrees to relinquish and transfer possession of and, if applicable, title to said property without the requirement of a court order upon termination of this Contract. Equipment that has reverted to the Contractor through a City-paid lease agreement with option to buy will be considered the same as though the equipment was purchased outright with Contract funds. It is understood that the terms, "equipment" and "property", as used herein, means tangible, non-expendable, personal property having a useful life of more than one year and an acquisition cost of \$5,000.00 or more per unit and shall include not only furniture and other durable property, but also vehicles, but shall not include supplies and consumables.
- 18.2 Contractor agrees that no equipment purchased with Contract funds may be disposed of without receiving prior written approval from DHS. In cases of theft and/or loss of equipment, it is the responsibility of the Contractor to replace it with like equipment. Contract funds cannot be used to replace equipment in those instances. All replacement equipment will be treated in the same manner as equipment purchased with Contract funds.
- 18.3 Contractor shall maintain accurate and complete records on all equipment and property obtained with Contract funds to include:
 - (A) A description of the equipment, including the model and serial number or other identification number, if applicable;
 - (B) The date of acquisition, cost and procurement source, purchase order number, and vendor number;
 - (C) An indication of whether the equipment is new or used;
 - (D) The vendor's name (or transferred from);
 - (E) The location of the property;

- (F) The property number shown on the property tag; and,
- (G) A list of disposed items and disposition
- 18.4 The Contractor is fully and solely responsible for the insuring, safeguarding, maintaining, and reporting of lost, stolen, missing, damaged, or destroyed equipment/property purchased or leased with Contract funds. All lost, stolen, missing, damaged and/or destroyed equipment/property shall be reported to the local Police Department and, if applicable, the Federal Bureau of Investigation (FBI) immediately. The Contractor shall make such reports immediately and shall notify and deliver a copy of the official report to DHS within seventy-two (72) hours from the date that Contractor discovers the lost, stolen, missing, damaged and/or destroyed equipment/property. The report submitted by the Contractor to DHS shall minimally include:
 - (A) A reasonably complete description of the missing, damaged or destroyed articles of property, including the cost and serial number and other pertinent information:
 - (B) A reasonably complete description of the circumstances surrounding the loss, theft, damage or destruction; and,
 - (C) A copy of the official written police report or, should the Police not make such copy available within 72 hours of the discovery and reporting of the loss, a summary of the report made to the Police, including the date the report was made and the name and badge number of the Police Officer who took the report. Contractor shall provide a copy of the official written report to the City within 72 hours of receiving the official written report from law enforcement.
- All equipment (as defined in 18.1 above) purchased under this Contract shall be fully insured against fire, loss and theft. Contractor shall, at a minimum, provide the equivalent insurance for real property and equipment acquired with Contract funds as provided to other property acquired or owned by the Contractor.
- 18.6 Upon request, the Contractor shall provide an annual inventory of assets purchased with funds received through the City to DHS.
- 18.7 Contractor shall fully comply with the property and equipment requirements of 45 C.F.R Part 74, including but not limited to Sections 75.316 through 75.323, related to the following:
 - (A) Insurance Coverage
 - (B) Real Property
 - (C) Federally-owned and exempt property
 - (D) Equipment
 - (E) Supplies
 - (F) Intangible property
 - (G) Property trust relationship
- 18.8 Relative to property, equipment and supplies purchased with Head Start grant funds, Contractor shall route all written correspondence through DHS for review, endorsement and processing. For equipment purchases in the amount of \$5,000.00 or greater or cumulative purchases in the amount of \$100,000.00 or greater Contractor shall obtain written approval from DHS prior to issuance of the bid or other procurement notice and prior to selection of the winning bid or proposal. Contractor shall not split the purchase of a line item greater than the preceding threshold(s) in order to avoid obtaining approval from DHS.
- 18.9 Contractor will maintain a system for tracking, on an ongoing basis, inventory of equipment and supplies purchased with Head Start funds that either (i) has a purchase price of \$5,000.00 or greater; or (ii) meets such other criteria as City may prescribe (and which City shall notify Contractor as appropriate). Upon request, Contractor will provide City a status report of the current inventory of equipment and supplies meeting these requirements. City shall have the right to review and approve Contractor's inventory tracking system.
- 18.10 City reserves the right to require transfer of property acquired with funds awarded under this Contract as provided in 45 C.F.R. Part 75, including but not limited to §75.316 et seq.

XIX. TRAVEL

- 19.1 The costs associated with budgeted travel for business, either in-town or out-of-town, are allowable costs provided documentation of expenses is present.
 - (A) Contractor agrees that mileage reimbursement paid to Contractor's employees shall be reimbursed at a rate no more liberal than the City's policy for mileage reimbursement, which is consistent with Internal Revenue Service (IRS) rules. Contractor further agrees that in order for its employees to be eligible for mileage reimbursement, the employees 1) shall be required to possess a valid Texas Driver's License and liability insurance as required by law, and 2) must record, on a daily basis, odometer readings before and after business use, showing total business miles driven each day and must keep such record in the vehicle. Mileage records are subject to spot-checks by City auditors and monitors. Contractor shall strongly encourage the participation by its employees in an approved defensive driving course. Evidence of the required driver's license and liability insurance must be kept on file with the Contractor.
 - (B) Contractor agrees that in order to obtain reimbursement of the costs associated with budgeted out of town travel for business in connection with this Contract, Contractor shall:
 - obtain City's prior approval and provide City with detailed documentation of such business travel expense(s),
 - ensure that any and all costs associated with out-of-town travel (including per diem rates) shall not be more liberal than the City's travel policies which conform with the reimbursement rates established by the United States General Services Administration,
 - 3) purchase all business travel at economy class rates and shall document such, and
 - maintain supporting documentation for conferences to include itineraries and documentation certifying conference attendance and provide such documentation to City upon request.

XX. NO USE OF FUNDS FOR RELIGIOUS ACTIVITIES

20.1 Contractor agrees that none of the performance rendered hereunder shall involve, and no portion of the funds received hereunder shall be used, directly or indirectly, for the construction, operations, maintenance or administration of any sectarian or religious facility or activity, nor shall said performance rendered or funds received be utilized so as to benefit, directly or indirectly, any such sectarian or religious facility or activity.

XXI. DEBARMENT

- 21.1 Contractor certifies that neither it nor its principals are presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from participation in any state or federal program.
- 21.2 Contractor shall provide immediate written notice to City, in accordance with the notice requirements of Article XXVI herein, if, at any time during the term of the contract, including any renewals hereof, Contractor learns that its certification was erroneous when made or have become erroneous by reason of changed circumstances.

XXII. ASSIGNMENT

22.1 Contractor shall not assign or transfer Contractor's interest in this Contract or any portion thereof without the written consent of the City, and if applicable, the Grantor of the Grant Funds. Any attempt to transfer, pledge or otherwise assign shall be void ab initio and shall confer no rights upon any third person or party.

XXIII. AMENDMENT

- 23.1 Any alterations, additions or deletions to the terms hereof shall be by amendment in writing executed by both City and Contractor and evidenced by passage of a subsequent City ordinance, as to City's approval; provided, however, the Director of DHS shall have the authority to execute an amendment of this Contract without the necessity of seeking any further approval by the City Council of the City of San Antonio if permitted by all applicable local, state and federal laws, and in the following circumstances:
 - A. An increase in funding (the cause of which is unrelated to the reason set forth in Section 23.1(E) below) of this Contract in an amount not exceeding (a) twenty-five percent (25%) of the total amount of this Contract or (b) \$25,000.00, whichever is the lesser amount; provided, however, that the cumulative total of all amendments executed without City Council approval pursuant to this subsection and increasing Contract funding during the term of this Contract shall not exceed the foregoing amount:
 - B. Modifications to the Scope of Work set forth in Attachment I hereto, so long as the terms of the amendment stay within the substantive parameters set forth in the original Scope of Work, also set forth in Attachment I hereto;
 - C. Budget line item shifts of funds, so long as the total dollar amount of the budget set forth in Section 3.1 of this Contract remains unchanged; (these modifications may be accomplished through Budget Revisions):
 - D. Modifications to the insurance provisions described in Article IX of this Contract that receive the prior written approval of the City of San Antonio's Risk Manager and the Director of DHS.
 - E. decreases (and increases if the City agrees to allocate additional enrollment slots to Contractor) in Contract funding based upon Head Start Program enrollment levels, and modifications to Contract terms related to enrollment; provided, however, that the cumulative total of all Head Start Program contracts, as amended, shall not exceed the City's total budget for the Head Start Program budget for the applicable grant year. Contractor shall execute any and all amendments to this Contract that are required as a result of a modification made pursuant to this Section 23.1(E).
- 23.2 Contractor further agrees that except when the terms of this Contract expressly provide otherwise, any alterations, additions or deletions to the terms hereof shall be by amendment in writing and approved by HHS.

XXIV. SUBCONTRACTING

- 24.1 None of the work or services covered by this Contract shall be sub-contracted without the prior written consent of the City and the Grantor of the Grant Funds, if so required by said Grantor.
- 24.2 Contractor must comply with all applicable local, State and Federal procurement standards, rules, regulations and laws in all its sub-contracts related to the work or funds herein. It is further agreed by the Parties hereto that the City has the authority to monitor, audit, examine, and make copies and transcripts of all sub-contracts, as often as deemed appropriate by the City. If, in the sole determination of the City, it is found that all applicable local, State and Federal procurement standards, rules, regulations and laws have not been met by Contractor with respect to any of its sub-contracts, then the Contractor will be deemed to be in default of this Contract, and as such, this Contract will be subject to termination in accordance with the provisions hereof.
- 24.3 Any work or services for sub-contracting hereunder, shall be sub-contracted only by written Contract, and unless specific waiver is granted in writing by City, shall be subject by its terms to each and every provision of this Contract. Compliance by sub-contractors with this Contract shall be the responsibility of Contractor. Contractor agrees that payment for services of any sub-contractor shall be submitted through Contractor, and Contractor shall be responsible for all payments to sub-contractors.
- 24.4 Contractor certifies that its subcontractors are not presently debarred, suspended or proposed for debarment, declared ineligible or voluntarily excluded from participation in any state or federal program.
- 25.5 Contractor understands and agrees that all subcontracts in excess of \$10,000.00 must address termination

for cause and for convenience, including the manner by which the termination will be effected and the basis for settlement.

XXV OFFICIAL COMMUNICATIONS

25.1 Except where the terms of this Contract expressly provide otherwise, any election, notice or communication required or permitted to be given under this Contract shall be in writing and deemed to have been duly given if and when delivered personally (with receipt acknowledged), or three (3) days after depositing same in the U.S. mail, first class, with proper postage prepaid, or upon receipt if sending the same by certified mail, return receipt requested, or upon receipt when sent by a commercial courier service (such as Federal Express or DHL Worldwide Express) for expedited delivery to be confirmed in writing by such courier, at the addresses set forth below or to such other address as either Party may from time to time designate in writing.

City:
Director
Department of Human Services
106 S. St. Mary's Street, Suite 700
San Antonio, Texas 78205

Contractor:
Theresa de la Haya, Senior Vice President
Community Health & Clinical Preventive Programs
University Health System
701 So. Zarzamora

With a copy to: Roberto Villarreal, MD, MPH Senior VP, Research & Information Management University Health System 4502 Medical Drive, MS 45-2 San Antonio, Texas 78229-4493

San Antonio, Texas 78207

Notices of changes of address by either Party must be made in writing delivered to the other Party's last known address within five (5) business days of the change.

XXVI. VENUE

26.1 Contractor and City agree that this Contract shall be governed by and construed in accordance with the laws of the State of Texas. Any action or proceeding brought to enforce the terms of this Contract or adjudicate any dispute arising out of this Contract shall be bought in a court of competent jurisdiction in San Antonio, Bexar County, Texas.

XXVII. GENDER

27.1 Words of any gender used in this Contract shall be held and construed to include any other gender, and words in the singular number shall be held to include the plural, unless the context otherwise requires.

XXIII. REPRESENTATIONS AND OTHER OBLIGATIONS

- 28.1 The signer of this Contract for Contractor represents, warrants, assures and guarantees that (s)he has full legal authority to execute this Contract on behalf of Contractor and to bind Contractor to all of the terms, conditions, provisions and obligations herein contained. Whether a non-profit or public entity, Contractor must be authorized to do business in the State of Texas and be formed under and operating in accordance with all applicable laws of the State of Texas. Contractor shall provide DHS verification of the foregoing requirements upon request by City.
- 28.2 This Contract is based on the representation of Contractor that it is financially viable, solvent, and accountable for its expenditures; that it has the continuing capability to furnish the non-Federal Share of the

cost of operating its approved Contractor Program; and that Program funds disbursed to Contractor will be expended only for allowable costs in the implementation of the Contractor Program which is the subject of this Contract. Contractor represents that there are no financial limitations or impediments that would make it not viable, solvent and accountable such that the flow of Program funds might be diverted away from the operation and maintenance of the Program which is the subject of this Contract.

28.3 In the event that circumstances arise which might result in interference with Contractor's ability to provide the services which are the subject of this Contract, Contractor agrees to inform City of those circumstances immediately upon their discovery. Contractor agrees that reimbursement to Contractor, upon reasonable notice, may be suspended by City until such financial circumstances giving rise to the possible interference with the operation of the Program have been eliminated, provided, however, that authorized expenditures made prior to the suspension, and approved by City shall be disbursed pursuant to the terms of this Contract.

XXIX. LICENSES AND TRAINING

29.1 Contractor warrants and certifies that Contractor's employees and its subcontractors have the requisite training, license or certification to provide said services, and meet all competence standards promulgated by all other authoritative bodies, as applicable to the services provided herein.

XXX. INDEPENDENT CONTRACTOR

- 30.1 It is expressly understood and agreed that the Contractor is and shall be deemed to be an independent contractor, responsible for its respective acts or omissions and that the City shall in no way be responsible therefor, and that neither Party hereto has authority to bind the other nor to hold out to third parties that it has the authority to bind the other.
- 30.2 Nothing contained herein shall be deemed or construed by the Parties hereto or by any third party as creating the relationship of employer-employee, principal-agent, partners, joint venture, or any other similar such relationship, between the Parties hereto.
- 30.3 Any and all of the employees of the Contractor, wherever located, while engaged in the performance of any work required by the City under this Contract shall be considered employees of the Contractor only, and not of the City, and any and all claims that may arise from the Workers' Compensation Act on behalf of said employees while so engaged shall be the sole obligation and responsibility of the Contractor.

XXXI. SEVERABILITY

31.1 If any clause or provision of this Contract is held invalid, illegal or unenforceable under present or future federal, state or local laws, including but not limited to the City Charter, City Code, or ordinances of City, then and in that event it is the intention of the Parties hereto that such invalidity, illegality or unenforceability shall not affect any other clause or provision hereof and that the remainder of this Contract shall be construed as if such invalid, illegal or unenforceable clause or provision was never contained herein; it is also the intention of the Parties hereto that in lieu of each clause or provision of this Contract that is invalid, illegal or unenforceable, there be added as a part of this Contract a clause or provision as similar in terms to such invalid, illegal or unenforceable clause or provision as may be possible, legal, valid and enforceable.

XXXII. NON-WAIVER OF PERFORMANCE

32.1 No waiver by City of a breach of any of the terms, conditions, covenants or guarantees of this Contract shall be construed or held to be a waiver of any succeeding or preceding breach of the same or any other term, condition, covenant or guarantee herein contained. Further, any failure of City to insist in any one or more cases upon the strict performance of any of the covenants of this Contract, or to exercise any option herein contained, shall in no event be construed as a waiver or relinquishment for the future of such covenant or option.

XXXIII. CONTRIBUTION PROHIBITIONS

- 33.1 Contractor acknowledges that City Code Section 2-309 provides that any person acting as a legal signatory to a proposed contractual relationship that applies for a "high profile" discretionary contract, as defined by the City of San Antonio Contracting Policy and Process Manual, may not make a campaign contribution to any councilmember or candidate at any time from the tenth business day after the Request for Proposal (RFP) or Request for Qualifications (RFQ) or other solicitation is released, or for a contract for which no competitive solicitation has been issued by the City from the time the City begins discussions or negotiations, and ending on the 30th calendar day following the contract award. Contractor understands that if the legal signatory entering the Contract has made such a contribution, the City may not award the Contract to that contributor or to that contributor's business entity. Any legal signatory for a proposed high-risk contract must be identified within the response to the RFP or RFQ, if the identity of the signatory will be different from the individual submitting the response.
- 33.2 Contractor acknowledges that the City has identified this Contract as high profile.
- 33.3 Contractor warrants and certifies, and this Contract is made in reliance thereon, that the individual signing this Contract has not made any contributions in violation of City Code section 2-309, and will not do so for 30 calendar days following the award of this Contract. Should the signor of this Contract violate this provision, the City Council may, in its discretion, declare the Contract void.

[The remainder of this page intentionally left blank]

XXXIV. ENTIRE CONTRACT

34.1 This Contract and its attachments, if any, constitute the entire and integrated Contract between the Parties hereto and contain all of the terms and conditions agreed upon, and supersede all prior negotiations, representations, or contracts, either oral or written. No such other negotiations or representations may be enforced by either Party nor may they be employed for interpretation purposes in any dispute involving this Contract.

In witness of which this Contract has been execu-	ted effective the day of
CITY OF SAN ANTONIO:	CONTRACTOR: Bexar County Hospital District d/b/a University Health System
Melody Woosley, Director Department of Human Services	George B. Hernandez, Jr. President/Chief Executive Officer
APPROVED AS TO PORM:	APPROVED AS TO FORM:
Assistant City Atterney	Associate General Counsel

ATTACHMENTS

Attachment I - Scope of Work

Attachment II - Program Budget

Attachment III - Special Provisions

Attachment IV - HHS Award Document

Attachment V - HIPAA Business Associate Agreement

Attachment VI - Additional OMB Provisions

SCOPE OF WORK

1. Summary

To promote healthy development and ensure optimal learning, Contractor will provide Head Start health services to children enrolled in the Head Start Program, as needed, in accordance with the Head Start Performance Standards and the terms of this Contract. Contractor will conduct Lead Screenings for all children who need it during the period of February 1, 2017 to June 2, 2017. Additionally during this period, Contractor agrees to serve as a medical provider and perform complete physical exams for children referred from Head Start service providers. Contractor will provide guidance and advice in the area of health to Head Start service providers when requested.

2. Program Services

- A. Contractor shall provide Head Start program services to eligible children without regard to age, race, creed, color, sex, or national origin.
- B. Contractor shall use Head Start funds for professional medical services when no other funding source is available, in accordance with Head Start Performance Standards. If funds are used for medical services, Contractor must have and provide to the City upon request written documentation of their efforts to access other available sources of funding.
- C. Contractor shall establish an ongoing monitoring system that includes an inventory of equipment and supplies, and conduct internal monitoring of operations throughout the program year, notifying the City when the Contractor identifies possible or actual lack of compliance with the Head Start Performance Standards, Head Start Act, City's program policies or terms of this contract. Contractor will make copies of monitoring reports available to the City upon request.
- D. Contractor shall establish and maintain an organizational structure that supports the accomplishment of Program objectives and supports Education Service Providers to ensure all the required Head Start health services are provided. At a minimum, management of medical services must be assigned to an individual serving in the role of a Medical Assistant (MA) or Patient Care Attendant (PCA). This individual and any supporting staff or consultants must be licensed or certified health professionals with experience and expertise in serving young children and their families.
- E. Contractor shall ensure that employees who are one hundred percent (100%) funded through the Head Start grant devote one hundred percent (100%) of their time and effort to support the Head Start Program. Contractor will provide administrative support to ensure that all Head Start activities are coordinated and billed accordingly.
- F. Contractor shall ensure that its Head Start Program shall be, and remain, in full compliance with the Head Start Performance Standards as provided in Head Start regulations, 45 CFR Part 1301 et seq. and with the Head Start Act, as amended, 42 U.S.C. 9801 et. seq.
- G. Contractor must strive to perform the following no later than 90 calendar days of the child's entry into the Program, and as needed throughout the program school year as new children enter the Program:

- 1. Contractor shall perform a complete unclothed physical examination for all children who are referred from Head Start service providers. This complete physical exam must be conducted in accordance with the Texas Health and Human Services Health Steps (THSteps) requirements. Contractor must provide copies of the physical exam documentation to Head Start service providers' Health Coordinators. Contractor must also verify or make a determination as to whether or not the child is eligible for health insurance, including Medicaid, Children's Health Insurance Program (CHIP), and Title V funds. Contractor shall ensure no other funding sources are available before using Head Start funds.
- 2. If Education Service Providers determine that an uninsured child has not received the required physical examination or that the physical examination is incomplete, Contractor will coordinate with Education Service Provider, the uninsured child's medical home or health service provider to have the physical examination completed or, if needed, Contractor must complete the physical examination. Before conducting said physical examination, Contractor shall ensure that the uninsured child has the appropriate consent forms and required documentation. Contractor must refer the uninsured child to the appropriate health care service provider if further medical attention is needed. Contractor shall also ensure no other funding sources are available before using Head Start funds.
- H. During the period of February 1, 2017 to June 2, 2017, Contractor, in coordination with the Education Service Provider, must strive to complete the following no later than 90 calendar days of the child's entry into the Program, and as needed throughout the program school year as new children enter the Program:
 - Contractor must conduct Lead Screenings for all children who need it at the appropriate
 Head Start Center. Contractor will provide the appropriate Consent and Release of
 Information form. Contractor will initiate the work as soon as the Education Service
 Provider completes the initial assessment and the Contractor receives a list of the
 children that require these screenings.
 - 2. If screenings indicate a lead level above normal, Contractor shall refer the child to a health professional for further evaluation and treatment and inform the appropriate Education Service Provider. Contractor will provide a referral letter to parents and Service Provider stating lead level, and will be responsible for explaining/counseling elevated lead levels to parents, when needed and as appropriate. All such parent follow up activities as described in this section shall be in accordance with the written procedures stipulated in Section 2(H)(3) of the Scope of Work.
 - Education Service Providers will make available the following information to the Contractor on all children that are screened for lead and report to the Department of State Health Services (with appropriate consent from parent) in accordance with Contractor's Policy and Procedures:
 - a.) Child's name
 - b.) Child's date of birth
 - c.) Current home address

- I. During the period of February 1, 2017 to June 2, 2017, Contractor must coordinate directly with the Education Service Providers' Health Coordinators to ensure that there is a system of appointments, document collection, follow-up and information sharing for the provision of Lead Screenings and Physical Exams for uninsured children in the Head Start Program to include the following:
 - Contractor's MA/PCA will carry out the following:
 - a) Report all lead results and document to the Texas Childhood Lead Poisoning Prevention Program – Texas Department of State Health Services in accordance with Contractor's Policy and Procedures.
 - All elevated lead results will be reported immediately to the Health Coordinator and the parents will be notified through a written referral given by the Family Service Worker.
 - Any direct contact between the UHS MA/PCA and parents of children with elevated lead results must be documented in ChildPlus within 7 days from the initial contact.
 - b) Contractor (RN, NP) must be available on an as needed basis to assess Physical Exams for health concern follow up and answer medical concerns related to the THSteps EPSDT. Findings will be reported to the City and Education Service Providers on a timely basis. All such parent follow up activities described in this section shall be in accordance with the written procedures stipulated in Section 2(H)(3) of the Scope of Work.
 - 2. During the period of February 1, 2017 to June 2, 2017, Contractor will carry out the following:
 - a) Provide City and Education Service Providers ChildPlus status reports on Lead Screenings of Head Start children on a monthly basis and as requested by the City. Reports must include number of campus visits, number of children screened for lead, and number of children indentified with abnormal lead results (referrals).
 - b) Be responsible for the ChildPlus data entry on health events of Lead Test. Contractor must create the Lead Test health event in ChildPlus, and input event date, status and results with the health services module for every Head Start child served. The information must be entered into ChildPlus no later than three (3) business days from the date of service. If appropriate, Contractor must check appropriate referral section on the health event.
 - c) Conduct a THSteps Medical Checkup, e.g. blood pressure, height and weight, etc., on Head Start children identified by the City and/or Education Service Provider on an as needed basis.
 - d) Reporting on testing updates during monthly Health Coordinators Monthly meetings and bi-monthly Head Start Directors meetings on an as needed basis.
 - 3. Contractor and City will collaboratively develop and mutually agree upon written procedures prior to the start of the 2016-2017 program school year that will govern the parent follow-up activities described in Section 2(G) and 2(H) of this Scope of Work. City acknowledges that Contractor is not required to conduct parent follow up activities as described in Section 2(G) and 2(H) of this Scope of Work until such written procedures are agreed upon by the parties. Such procedures must describe, at a minimum, the type of parent contact required, the required follow up timeline for each activity and the number of contacts required prior to referral to proper campus health coordinator or other Head Start staff member. Contractor acknowledges that such

procedures must be in compliance with the Head Start Performance Standards, Head Start Act and any other guidance or clarification from HHS as well as the City of San Antonio Head Start policies and guidance.

- J. Contractor must provide written referral or written correspondence to parent(s) of uninsured Head Start enrollees, who received a physical examination, explaining abnormal findings in accordance with the Health Insurance Portability and Accountability Act (HIPAA) and all appropriate state and federal statutes and regulations.
- K. Contractor will provide the City, upon request, with a list of Contractor's ambulatory facility hours and locations indicating the availability of child examinations and immunizations. Contractor will work with City to adjust the Contractor's hours of operation in order to meet the needs of parents and Head Start service providers.
- L. Contractor must establish, implement and maintain communication systems to ensure that timely, accurate and appropriate information is provided to parents, Health Coordinators and Education Service Providers.
- M. Contractor will coordinate with the City to make available University Health System's Pediatric Mobile Unit (PMU) to Head Start program sites on as needed basis to perform professional medical services in accordance with the Head Start Performance Standards. The City will facilitate the provision of Contractor's services by assisting with the following:
 - 1. City will obtain needed paperwork from Parent and/or legal guardian prior to appointment.
 - City will ensure Parent and/or legal guardian or other consented adult is present with the child during the duration of the THSteps EPSDT and/or immunizations
 - If the parent and/or legal guardian cannot be present during the time of the scheduled THSteps EPSDT and/or immunizations, the city will obtain special consent form.
 - 4. City will ensure City Health Coordinator and support staff is available to assist when needed during the time the PMU is present at the designated Head Start site.
 - Both City and Contractor will collaborate to ensure a maximum of children are seen during the time the PMU is at the designated Head Start site. The maximum number of children seen will be determined by the staffing availability of UHS.
 - Both City and Contractor will also collaborate to ensure all available PMU slots are scheduled in advance.
 - City and Contractor will collaborate to determine in advance the number of slots available for that PMU.
 - b. In the case that City cannot fill the PMU slots in advance, City may cancel scheduled PMU visit with one week's prior written notice to Contractor. Such cancellation will be at no additional cost to the City.
 - UHS will require at least 5 confirmed Head Start patients in order for the PMU to visit head start program sites.
 - d. City will ensure that all necessary paperwork required for UHS to perform a Texas Health Steps including a list of children who are scheduled to be seen will be emailed/faxed no later than one-week prior to scheduled visit. Failure to provide UHS the necessary paperwork prior to the visit will result in cancellation of the scheduled date to perform Texas Health Steps examinations.
 - e. If space is available, UHS agrees to provide services to any child City

- presents with proper documentation at PMU site
- f. Physical examinations may be performed utilizing the PMU at the Head Start Centers after hours or during the weekend.
- N. Contractor will provide the following professional medical services: THSteps EPSDT and immunizations. Contractor will provide parent and/or legal guardian with a copy of the THSteps EPSDT and/or proof of the immunization administered. Contractor will make every effort to provide services funded through Medicaid, CHIP, or private/3rd party health insurance. Contractor may invoice the City \$82.00 per unfunded exam for uninsured children. Contractor may also invoice the City \$22.00 for each immunization administered to uninsured children. If the child does not have insurance, Contractor will provide City with the child's name and contact information.

3. Program Governance

- A. Contractor's Governing Board shall be in full compliance with Head Start requirements regarding governance, management and programmatic operations applicable to recipients of Head Start grant funds, including those set forth at 45 CFR §1301 et. seq. Contractor's Governing Board members or representatives of the Board shall be offered the opportunity to participate in Board education activities arranged by City. Contractor shall also offer the Governing Board members and representatives of the Board the opportunity to engage in a cooperative strategic planning process with respect to the City's Head Start Program and shall submit any final strategic plans developed through such process to City for approval.
- B. Contractor shall assure that City is kept fully apprised of the composition and actions of Contractor's Governing Board to the extent such actions affect Contractor's provision of Head Start services.
- C. Contractor shall seek and obtain the City's written approval before making any material revisions in Contractor's Head Start services that conflicts or violates (i) the City's refunding application, as amended, to the U.S. Department of Health and Human Services (HHS), (ii) the Terms of the Grant, and (iii) the terms of this Contract.

4. Licensure/Staffing

Contractor shall obtain and maintain all necessary and appropriate State licenses, permits, certifications, and approvals required for the operation of Contractor's programs including those supported by this Contract. Upon commencement of the Contract, Contractor shall notify the City that it is in compliance with this provision. If at any time Contractor is out of compliance with this provision, Contractor shall notify the City within one (1) Contractor business day of receipt of written notice of violation or complaint from the state licensing, certifying or permit-issuing authority indicating lack of licensure, permitting or certification, as the case may be, and shall take all necessary steps to cure such violation. Contractor further agrees that all personnel, either employed or contracted, assigned by Contractor to provide the Head Start Program health services set forth above shall, as appropriate or required by law, be fully qualified and authorized under applicable law, to perform such Head Start Program Services.

5. Participation

Contractor shall make time and resources available to support: (i) participation by Contractor in meetings with City staff for community assessment, self-assessment, strategic planning, development of

training and technical assistance plan, communication and program development activities; (ii) participation in technical assistance trainings and service enhancements developed by City and the Head Start training and technical assistance service provider, as well as other Head Start trainings that may be developed by relevant federal or state agencies.

6. Records. Reporting, Monitoring and Intellectual Property

Additionally, Contractor shall maintain and furnish to City the following financial and programmatic information and reports, in such forms as the City may prescribe, as required under the Head Start Act, as amended, and federal regulations such as 45 C.F.R. Part 74, 45 C.F.R. Part 92, or 45 C.F.R. Part 75, as applicable:

- The total amount of public and private funds received by Contractor and the amount from each source;
- Financial reports showing all actual and/or projected costs of the Program, an explanation of budgetary expenditures, Program Income, non-Federal Share amounts;
- The results of the most recent financial audit;
- The number and percentage of enrolled children that received medical exams;
- Reports showing employee credentials and a list of personnel serving to satisfy Contractor's in-kind non-Federal Share requirement;
- f) Reports showing the wages of each employee;
- g) Head Start Service Providers report to the Head Start Policy Council, submitted on a monthly basis; and
- h) Any other information requested by City.

Contractor shall maintain all applicable and appropriate supporting documentation of costs, including but not limited to, payroll records, invoices, contracts or vouchers, and make these available to City upon request.

ATTACHMENT II [INSERT PROGRAM BUDGET HERE]

Contract #:		
Attachment III	- Special	Provisions

ATTACHMENT III SPECIAL PROVISIONS — Program Year 2017-2018

I. RESTRICTIONS ON USE OF FUNDS OR PROPERTY

In addition to the other applicable restrictions on the use of Head Start funds provided under this Contract, the Contractor is prohibited from:

- using or transferring funds provided under this Contract for purposes other than authorized Head Start activities;
- using, pledging, granting a security interest in, or otherwise encumbering any right under this Contract
 or any property acquired with funds provided under this Contract as collateral or security for any loan,
 note debenture, bond or any other debt instrument;
- using any funds provided under this Contract for payment of principal or interest on any loan, note, debenture, bond or any other debt instrument, other than those approved in the 45 C.F.R. Part 74 and by the City

II. REQUIREMENTS FOR PARTICIPATION IN CHILD PLUS DATA SYSTEM

2.01 Child Plus is an electronic case management system managed and licensed by the City. This system maintains child files and an overall wait list and streamlines the process for program entry, qualification, position reservation and referrals. Child Plus enhances performance and improves the overall efficiency of data processing and automation systems in support of Head Start initiatives and is used to compile the annual Program Information Report (PIR).

2.02 Contractor shall:

- a) maintain and support Child Plus Data System
- provide a data entry specialist for Child Plus Data System that will be responsible for entering all required data into the system and who will be the designated contact person with regard to data entries;
- attend meetings with the City's Child Plus vendor and City staff to ensure continuity and commitment to the this system;
- support all design, development, testing and implementation protocols as established by the City by carrying out and complying therewith;
- e) participate in preliminary and final testing of the system using City protocols;
- f) provide the technical detail required for matching Contractor's system with the Child Plus Data System environment;
- g) allow City and its vendor to install data encryption software on the Child Care System Database network; and
- provide City and its vendor with access to Confidential Data with parental permission, as defined in Article 3.01 below, which data is critical for the Head Start project.

2.03 Both Parties agree to:

- a) use best efforts to cooperate and exchange information regarding all aspects of the Head Start project and comply with all reasonable requests of the other Party with respect to information concerning the system.
- b) Parties agree that nothing herein shall be construed as to control or in any way limit the right of parents to choose a Head Start provider.

III. CONFIDENTIAL DATA

3.01 The Parties to this Contract shall have access to the following data ("Confidential Data"), with parental permission in the case of the child:

Contract #:		
Attachment III	- Special	Provisions

Parent's Information:

Case Number First Name Middle Initial Last Name Street Address City

Zip Code Telephone

Social Security Number (Optional)

Birth Date Gender Race Disability Yearly Income

Number of members in the Family

County of Residence

Employment and training status

Each child's Information:

Client Number First Name Middle Initial Last Name

Social Security Number (Optional)

Birth Date Gender Race Disability

- 3.02 Contractor understands that City intends to enter into additional agreements with other providers of child care services ("Additional Collaborators") in order to promote the success of the Head Start project. Confidential Data may be shared by City, Contractor, and any Additional Collaborator, except that all parties shall share such information in compliance with state and federal laws relating to confidentiality. All Additional Collaborators shall be required to enter into a written agreement with City containing the confidentiality requirements set forth in this Section III.
- 3.03 Each Party shall establish a method to secure the Confidential Data in accordance with the applicable federal, state, and local laws and regulations. This provision shall not be construed as limiting a Party to this Contract or an Additional Collaborator, or such Party's authorized representative's right of access to that Party's Confidential Data.
- Neither Party shall disclose or publish Confidential Data or public school education data to any individual or organization that is not a Party to this Contract or an Additional Collaborator, unless required by law or a lawful order of a court of competent jurisdiction. Each Party shall take measures within its organization to ensure that Confidential Data or public school education data is accessible only by those persons working on the Head Start project, or directly providing other public school education / child care services, and only for the purpose of performing or assisting with services required by the Head Start project or other specific public school education / child care services.
- 3.05 Either Party may disclose Confidential Data to a third party ("Third Party") under contract or affiliated with that Party for the sole purpose of performing or assisting with services required in relation to the Head Start project or other specific child care services, and in compliance with state and federal laws relating to confidentiality. Confidential Data provided to a Third Party shall remain confidential and written confirmation by such Third Party that the Third Party will conform to the requirements of this section shall be provided to the Party prior to delivery of any information to the Third Party.

ATTACHMENT IV [INSERT HHS AWARD DOCUMENT HERE]

WITNESSETH:

HIPAA BUSINESS ASSOCIATE AGREEMENT

This HIPAA Business Associate Agreement is entered into by and between the City of San Antonio ("Covered Entity"), and the Bexar County Hospital District d/b/a University Health System, a Business Associate ("BA").

WHEREAS, Covered Entity and BA have entered into a Head Start Agreement ("Service Contract") to provide health services, effective February 1, 2017; and

WHEREAS, Covered Entity and BA may need to use, disclose and/or make available certain information pursuant to the terms of the Service Contract, some of which may constitute Protected Health Information ("PHI"); and

WHEREAS, Covered Entity and BA intend to protect the privacy and provide for the security of PHI disclosed to each other pursuant to the Service Contract in compliance with the Health Insurance Portability and Accountability Act of 1996, Public Law 104-191 ("HIPAA") and regulations promulgated thereunder by the U.S. Department of Health and Human Services (the "HIPAA Regulations"), Health Information Technology for Economic and Clinical Health Act ("HITECH Act"), and other applicable laws; and

WHEREAS, the purpose of this Agreement is to satisfy certain standards and requirements of HIPAA and the HIPAA Regulations, including, but not limited to, Title 45, Section 164.504(e) of the Code of Federal Regulations ("C.F.R."), as the same may be amended from time to time;

NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, hereby agree as follows:

- A. <u>Definitions</u>. For the purposes of this Agreement, the following terms have the meanings ascribed to them:
 - (1) "Disclosure" with respect to PHI, shall mean the release, transfer, provision of access to or divulging in any other manner of PHI outside the entity holding the PHI.
 - (2) "Individual" shall have the same meaning as the term "Individual" in 45 C.F.R. 164.501 and shall include a person who qualifies as a personal representative in accordance with 45 C.F.R. 164.502(g).
 - (3) "Parties" shall mean Covered Entity and BA. "Business Associate" shall generally have the same meaning as the term "business associate" at 45 C.F.R. 160.103. "Covered Entity" shall generally have the same meaning as the term "covered entity" at 45 C.F.R. 160.103.
 - (4) "Privacy Rule" shall mean the Standards for Privacy of Individually Identifiable Health Information at 45 C.F.R. Part 160 and Part 164, subparts A and E.

- (5) "Security Rule" shall mean the HIPAA regulation that is codified at 45 C.F.R. Part 164.
- (6) "Protected Health Information" or "PHI" shall have the same meaning as the term "protected health information" in 45 C.F.R. 164.501, limited to the information created or received by BA from or on behalf of Covered Entity. PHI includes "Electronic Protected Health Information" or "EPHI" and shall have the meaning given to such term under the HIPAA Rule, including but not limited to 45 C.F.R. Parts 160, 162, 164, and under HITECH.
- (7) "Required By Law" shall have the same meaning as the term "required by law" in 45 C.F.R. 164.103.
- (8) "Secretary" shall mean the Secretary of the U.S. Department of Health and Human Services or his designee.
- (9) "PHI Breach" shall mean an acquisition, access, use, or disclosure of PHI in a manner not permitted by the Privacy Rules and such action compromises the security or privacy of the PHI.
- (10) The Health Information Technology for Economic and Clinical Health ("HITECH") Act shall mean Division A, Title XII of the American Recovery and Reinvestment Act of 2009 (Pub. L. 111-5).

B. BA Obligations and Activities. BA agrees that it shall:

- (1) Not use or disclose the PHI other than as permitted or required by this Agreement or as permitted by the HIPAA Regulations or as Required by Law;
- (2) Establish and maintain appropriate administrative, physical, and technical safeguards that reasonably and appropriately protect, consistent with the services provided under this Agreement, the confidentiality, integrity, and availability of the electronic protected health information that it creates, receives, maintains, or transmits on behalf of Covered Entity;
- (3) Mitigate, to the extent practicable, any harmful effect that is known to BA of a use or disclosure of PHI by BA in violation of the requirements of this Agreement;
- (4) Report to Covered Entity any use or disclosure of PHI of which BA is aware or becomes aware that is not provided for or allowed by this Agreement as well as any Security Incident as defined by 45 C.F.R. 164.304 that BA becomes of aware of;
- (5) Ensure that a business associate agreement is in place with any of its agents or subcontractors with which BA does business and to whom it provides PHI received from, created or received by BA on behalf of Covered Entity are aware of and agree to the same restrictions and conditions that apply through this Agreement to BA with respect to such information, and further agree to implement reasonable and appropriate administrative, physical and technical safeguards that render such PHI unusable,

unreadable, and indecipherable to individuals unauthorized to acquire or otherwise have access to such PHI;

- (6) Provide access, at the request of Covered Entity, and in a reasonable time and manner as agreed by the Parties, to PHI in a Designated Record Set to Covered Entity or, as directed by Covered Entity, to an Individual in order to meet the requirements 45 C.F.R. §164.524;
- (7) Make any amendment(s) to PHI in a Designated Record Set that the Covered Entity directs or agrees to pursuant to 45 C.F.R. 164.526 at the request of the Covered Entity or an Individual, and in a reasonable time and manner agreed to by the Parties;
- (8) Make available to the Covered Entity or to the Secretary all internal practices, books and records, including policies and procedures, relating to the use and disclosure of PHI received from, or created or received by the BA on behalf of the Covered Entity, for purposes of the Secretary in determining Covered Entity's compliance with the Privacy Rule;
- (9) Document such disclosures of PHI, and information related to such disclosures, as would be required for Covered Entity to respond to a request from an Individual for an accounting of disclosures of PHI in accordance with 45 C.F.R. 164.528;
- (10) Provide Covered Entity or an Individual, in a reasonable time and manner as agreed to by the Parties, information collected in accordance with Section B(9) of this Agreement, to permit Covered Entity to respond to a request by an Individual for an accounting of disclosures of PHI in accordance with 45 C.F.R. 164.528;
- (11) Will immediately, and in no event later than three days from discovery, notify Covered Entity of any breach of PHI, including EPHI, and will coordinate with Covered Entity to identify, record, investigate, and report to an affected individual and the U.S. Department of Health and Human Services, as required, any covered PHI breach. Breach notification to Covered Entity must include: names of individuals with contact information for those who were or may have been impacted by the HIPAA Breach; a brief description of the circumstances of the HIPAA Breach, including the date of the breach and date of discovery; a description of the types of unsecured PHI involved in the breach; a brief description of what the BA has done or is doing to investigate the breach and mitigate harm. BA will appoint a breach liaison and provide contact information to provide information and answer questions Covered Entity may have concerning the breach;
- (12) Comply with all HIPAA Security Rule requirements;
- (13) Comply with the provisions of HIPAA Privacy Rule for any obligation Covered Entity delegates to BA;
- (14) Under no circumstances may BA sell PHI in such a way as to violate Texas Health and Safety Code, Chapter 181.153, effective September 1, 2012, nor shall BA use PHI for marketing purposes in such a manner as to violate Texas Health and Safety Code

Section 181.152, or attempt to re-identify any information in violation of Texas Health and Safety Code Section 181.151, regardless of whether such action is on behalf of or permitted by the Covered Entity.

C. Permitted Uses and Disclosures by BA

- (1) Except as otherwise limited in this Agreement, BA may use or disclose PHI to perform functions, activities, or services for, or on behalf of, Covered Entity as specified in the Service Contract, provided that such use or disclosure would not violate the Privacy Rule if done by Covered Entity.
- (2) Except as otherwise limited in this Agreement, BA may disclose PHI for the proper management and administration of the BA, provided that disclosures are Required By Law, or BA obtains reasonable assurances from the person to whom the information is disclosed that it will remain confidential and used or further disclosed only as Required By Law or for the purpose for which it was disclosed to the person, and the person notifies the BA of any instances of which it is aware in which the confidentiality of the information has been breached.
- (3) Except as otherwise limited in this Agreement, BA may use PHI to provide Data Aggregation Services to Covered Entity as permitted by 45 C.F.R. 164.504(e)(2)(i)(B).
- (4) BA may use PHI to report violations of law to appropriate Federal and State authorities, consistent with 45 C.F.R. 164.502(j)(1).
- D. <u>Obligations of Covered Entity</u>. Covered Entity shall inform BA of its privacy practices and restrictions as follows. Covered Entity shall:
 - (1) notify BA of any limitations in its notice of privacy practices in accordance with 45 C.F.R. 164.520, to the extent that such limitation may affect BA's use or disclosure of PHI;
 - (2) notify BA of any changes in, or revocation of, permission by any Individual to use or disclose PHI, to the extent that such changes may affect BA's use or disclosure of PHI;
 - (3) notify BA of any restriction to the use or disclosure of PHI that Covered Entity has agreed to in accordance with 45 C.F.R. 164.522 to the extent that such changes may affect BA's use or disclosure of PHI;
 - (4) coordinate with BA regarding any PHI breach and make timely notification to affected individuals within 60 days of discovery.

E. Permissible Requests by Covered Entity.

Covered Entity shall not request BA to use or disclose PHI in any manner that would not be permissible under the Privacy Rule if done by Covered Entity, except that the BA may use or disclose PHI for data aggregation or management and administrative activities of the BA.

F. Term and Termination.

- (1) The term of this Agreement shall commence on February 1, 2017. This Agreement shall terminate when all PHI encompassed by this Agreement is destroyed or returned to Covered Entity or, if it is infeasible to return or destroy the PHI, protections are extended to such information in accordance with the termination provisions in this Section.
- (2) Termination for Cause. Upon Covered Entity's knowledge of a material breach by BA, Covered Entity shall either (a) provide an opportunity for BA to cure the breach in accordance with the terms of the Service Contract or, if the BA does not cure the breach or end the violation within the time for cure specified in the Service Contract, end the violation and terminate this Agreement and the Service Contract; or (b) immediately terminate this Agreement and the Service Contract if BA has breached a material term of this Agreement and cure is not possible. If neither termination nor cure is feasible, Covered Entity shall report the violation to the Secretary of the U.S. Department of Health and Human Services.

(3) Effect of Termination.

- (a) Except as provided below in paragraph (b) of this Section F(3), upon termination of this Agreement for any reason, BA shall return or destroy all PHI received from the Covered Entity, or created or received by BA on behalf of Covered Entity. This provision shall apply to PHI that is in the possession of BA or its subcontractors or agents. BA shall not retain any copies of PHI.
- (b) In the event that BA determines that returning or destroying PHI is infeasible, BA shall provide to Covered Entity written notification of the condition that makes the return or destruction of PHI infeasible. Upon BA's conveyance of such written notification, BA shall extend the protections of this Agreement to such PHI and limit further uses and disclosures of such PHI to those purposes that make its return or destruction infeasible, for so long as BA maintains such PHI.
- G. <u>Amendment to Comply with Law.</u> The Parties agree to take written action as is necessary to amend this Agreement to comply with any Privacy Rules and HIPAA legal requirements for Covered Entity without the need for additional council action.
- H. <u>Survival</u>. The respective rights and obligations of the BA under Sections B, C (2) and (4), and F(3) shall survive the termination of this Agreement.
- Interpretation. Any ambiguity in this Agreement shall be interpreted to permit Covered Entity to comply with the Privacy Rule.
- J. <u>Regulatory References</u>. A reference in this Agreement to a section in the Privacy Rule means the section as in effect or amended.

- K. <u>No Third Party Beneficiaries</u>. Nothing express or implied in this Agreement is intended to confer, nor shall anything herein confer upon any person other than Covered Entity, BA, and their respective successors or assigns, any rights, remedies, obligations, or liabilities whatsoever.
- L. <u>INDEMNIFICATION</u>. CONTRACTOR ACKNOWLEDGES THAT IS A POLITICAL SUBDIVISION OF THE STATE OF TEXAS AND IS SUBJECT TO, AND COMPLIES WITH THE APPLICABLE PROVISIONS OF THE TEXAS TORT CLAIMS ACT, AS SET OUT IN THE CIVIL PRACTICES AND REMEDIES CODE, SECTION 101.001 ET SEQ. AND THE REMEDIES AUTHORIZED THEREIN REGARDING CLAIMS OR CAUSES OF ACTION THAT MAY BE ASSERTED BY THIRD PARTIES, INCLUDING BUT NOT LIMITED TO THOSE RESULTING OR ARISING FROM ANY AND ALL INJURIES OR DEATH OF ANY PERSON OR DAMAGE TO ANY PROPERTY ARISING FROM OR RELATED TO THIS CONTRACT.
- M. <u>Reimbursement</u>. BA will reimburse Covered Entity for reasonable costs incurred responding to a PHI breach by BA or any of BA's subcontractors.
- N. <u>Waiver</u>. No provision of this Agreement or any breach thereof shall be deemed waived unless such waiver is in writing and signed by the party claimed to have waived such provision or breach. No waiver of a breach shall constitute a waiver of or excuse any different or subsequent breach.
- O. <u>Assignment</u>. Neither party may assign (whether by operation of law or otherwise) any of its rights or delegate or subcontract any of its obligations under this Agreement without the prior written consent of the other party. Notwithstanding the foregoing, Covered Entity shall have the right to assign its rights and obligations hereunder to any entity that is an affiliate or successor of Covered Entity, without the prior approval of BA.
- P. Entire Agreement. This Agreement constitutes the complete agreement between BA and Covered Entity relating to the matters specified in this Agreement, and supersedes all prior representations or agreements, whether oral or written, with respect to such matters. In the event of any conflict between the terms of this Agreement and the terms of the Service Contract or any such later agreement(s), the terms of this Agreement shall control unless the terms of such Service Contract comply with the federal law and regulations commonly referred to as the Privacy Standards and the Security Standards. No oral modification or waiver of any of the provisions of this Agreement shall be binding on either party. This Agreement is for the benefit of, and shall be binding upon the Parties, their affiliates and respective successors and assigns. No third party shall be considered a third-party beneficiary under this Agreement, nor shall any third party have any rights as a result of this Agreement.

COVERED ENTITY:	BUSINESS ASSOCIATE:
City of San Antonio, a Texas municipal corporation	Bexar County Hospital District d/b/a University Health System
Melody Woosley, Director Department of Human Services	Michael Hernandez, Chief Legal Officer
APPROVED AS TO FORM:	
Assistant enty Attorney	Associate General Counsel

Governing Law. This Agreement shall be governed by and interpreted in accordance

Q.

with the laws of the State of Texas.

EXECUTED to be effective February 1, 2017.

Additional OMB Provisions

from Appendix II to Part 200—Contract Provisions for Non-Federal Entity Contracts Under Federal Awards

In addition to other provisions required by HHS or the City, all contracts made by the City under the Federal award must contain provisions covering the following, as applicable (2 C.F.R. 200, Appendix II).

Hereinaster in this Attachment VI, Center shall be referred to as "contractor."

Provision	Page Number
Equal Employment Opportunity	2
Davis Bacon Act	4
Contract Work Hours and Safety Standards Act	9

EQUAL EMPLOYMENT OPPORTUNITY provisions (60 C.F.R. 1.4(b).

During the performance of this contract, contractor agrees:

- (1) The contractor will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. The contractor will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following: Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.
- (2) The contractor will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive considerations for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.
- (3) The contractor will send to each labor union or representative of workers with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of the contractor's commitments under this section, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.
- (4) The contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.
- (5) The contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to CONTRACTOR's books, records, and accounts by the U.S. Department of Health and Human Services ("HHS") and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.
- (6) In the event of the contractor's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations, or orders, this contract may be canceled, terminated, or suspended in whole or in part and the contractor may be declared ineligible for further Government contracts or federally-assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.

(7) The contractor will include the portion of the sentence immediately preceding paragraph (1) and the provisions of paragraphs (1) through (7) in every subcontract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The contractor will take such action with respect to any subcontract or purchase order as HHS may direct as a means of enforcing such provisions, including sanctions for noncompliance; provided, however, that in the event a contractor becomes involved in, or is threatened with, litigation with a subcontractor or vendor as a result of such direction by the administering agency the contractor may request HHS to enter into such litigation to protect the interests of the United States.

DAVIS BACON ACT provisions (29 C.F.R. § 5.5(a))

For any contract or subcontract in excess of \$2,000 which is entered into for the actual construction, alteration and/or repair, including painting and decorating, of a public building or public work, or building or work financed in whole or in part from federal funds, and which is subject to the labor standards provisions of any of the acts listed in 29 C.F.R. §5.1, the following § 5.5(a) must be included and complied with:

- (1) Minimum wages.
- (i) All laborers and mechanics employed or working upon the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), will be paid unconditionally and not less often than once a week, and without subsequent deduction or rebate on any account (except such payroll deductions as are permitted by regulations issued by the Secretary of Labor under the Copeland Act (29 CFR part 3)), the full amount of wages and bona fide fringe benefits (or cash equivalents thereof) due at time of payment computed at rates not less than those contained in the wage determination of the Secretary of Labor, which is incorporated herein by reference, regardless of any contractual relationship which may be alleged to exist between the contractor and such laborers and mechanics.

Contributions made or costs reasonably anticipated for bona fide fringe benefits under section 1(b)(2) of the Davis-Bacon Act on behalf of laborers or mechanics are considered wages paid to such laborers or mechanics, subject to the provisions of paragraph (a)(1)(iv) of this section; also, regular contributions made or costs incurred for more than a weekly period (but not less often than quarterly) under plans, funds, or programs which cover the particular weekly period, are deemed to be constructively made or incurred during such weekly period. Such laborers and mechanics shall be paid the appropriate wage rate and fringe benefits on the wage determination for the classification of work actually performed, without regard to skill, except as provided in §5.5(a)(4). Laborers or mechanics performing work in more than one classification may be compensated at the rate specified for each classification for the time actually worked therein: Provided, that the employer's payroll records accurately set forth the time spent in each classification in which work is performed. The wage determination (including any additional classification and wage rates conformed under paragraph (a)(1)(ii) of this section) and the Davis-Bacon poster (WH-1321) shall be posted at all times by the contractor and its subcontractors at the site of the work in a prominent and accessible place where it can be easily seen by the workers.

(ii)

- (A) The contracting officer shall require that any class of laborers or mechanics, including helpers, which is not listed in the wage determination and which is to be employed under the contract shall be classified in conformance with the wage determination. The contracting officer shall approve an additional classification and wage rate and fringe benefits therefore only when the following criteria have been met:
 - The work to be performed by the classification requested is not performed by a classification in the wage determination; and
 - (2) The classification is utilized in the area by the construction industry; and
 - (3) The proposed wage rate, including any bona fide fringe benefits, bears a reasonable relationship to the wage rates contained in the wage determination.
- (B) If the contractor and the laborers and mechanics to be employed in the classification (if known), or their representatives, and the contracting officer agree on the classification and wage rate (including the amount designated for fringe benefits where appropriate), a report of the action taken shall be sent by the contracting officer to the Administrator of the Wage and Hour Division, Employment Standards Administration, U.S. Department of Labor, Washington, DC 20210. The Administrator, or an

authorized representative, will approve, modify, or disapprove every additional classification action within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.

- (C) In the event the contractor, the laborers or mechanics to be employed in the classification or their representatives, and the contracting officer do not agree on the proposed classification and wage rate (including the amount designated for fringe benefits, where appropriate), the contracting officer shall refer the questions, including the views of all interested parties and the recommendation of the contracting officer, to the Administrator for determination. The Administrator, or an authorized representative, will issue a determination within 30 days of receipt and so advise the contracting officer or will notify the contracting officer within the 30-day period that additional time is necessary.
- (D) The wage rate (including fringe benefits where appropriate) determined pursuant to paragraphs (a)(1)(ii) (B) or (C) of this section, shall be paid to all workers performing work in the classification under this contract from the first day on which work is performed in the classification.
- (iii) Whenever the minimum wage rate prescribed in the contract for a class of laborers or mechanics includes a fringe benefit which is not expressed as an hourly rate, the contractor shall either pay the benefit as stated in the wage determination or shall pay another bona fide fringe benefit or an hourly cash equivalent thereof.
- (iv) If the contractor does not make payments to a trustee or other third person, the contractor may consider as part of the wages of any laborer or mechanic the amount of any costs reasonably anticipated in providing bona fide fringe benefits under a plan or program, Provided, That the Secretary of Labor has found, upon the written request of the contractor, that the applicable standards of the Davis-Bacon Act have been met. The Secretary of Labor may require the contractor to set aside in a separate account assets for the meeting of obligations under the plan or program.
- (2) Withholding. HHS or the City shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld from the contractor under this contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to Davis-Bacon prevailing wage requirements, which is held by the same prime contractor, so much of the accrued payments or advances as may be considered necessary to pay laborers and mechanics, including apprentices, trainees, and helpers, employed by the contractor or any subcontractor the full amount of wages required by the contract. In the event of failure to pay any laborer or mechanic, including any apprentice, trainee, or helper, employed or working on the site of the work (or under the United States Housing Act of 1937 or under the Housing Act of 1949 in the construction or development of the project), all or part of the wages required by the contract, HHS may, after written notice to the contractor, sponsor, applicant, or owner, take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds until such violations have ceased.
- (3) Payrolls and basic records.
- (i) Payrolls and basic records relating thereto shall be maintained by the contractor during the course of the work and preserved for a period of three years thereafter for all laborers and mechanics working at the site of the work (or under the United States Housing Act of 1937, or under the Housing Act of 1949, in the construction or development of the project). Such records shall contain the name, address, and social security number of each such worker, his or her correct classification, hourly rates of wages paid (including rates of contributions or costs anticipated for bona fide fringe benefits or cash equivalents thereof of the types described in section 1(b)(2)(B) of the Davis-Bacon Act), daily and weekly number of hours worked, deductions made and actual wages paid. Whenever the Secretary of Labor has found under 29 CFR 5.5(a)(1)(iv) that the wages of any laborer or mechanic include the amount of any costs reasonably anticipated in providing benefits under a plan or program described in section 1(b)(2)(B) of the Davis-Bacon Act, the contractor shall maintain records which show that the commitment to provide such benefits is enforceable, that the plan or program is financially responsible, and that the plan or program has been communicated in writing to the laborers or mechanics affected,

and records which show the costs anticipated or the actual cost incurred in providing such benefits. Contractors employing apprentices or trainees under approved programs shall maintain written evidence of the registration of apprenticeship programs and certification of trainee programs, the registration of the apprentices and trainees, and the ratios and wage rates prescribed in the applicable programs.

(ii)

- (A) The contractor shall submit weekly for each week in which any contract work is performed a copy of all payrolls to HHS if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit the payrolls to the applicant, sponsor, or owner, as the case may be, for transmission to HHS. The payrolls submitted shall set out accurately and completely all of the information required to be maintained under 29 CFR 5.5(a)(3)(i), except that full social security numbers and home addresses shall not be included on weekly transmittals. Instead the payrolls shall only need to include an individually identifying number for each employee (e.g., the last four digits of the employee's social security number). The required weekly payroll information may be submitted in any form desired. Optional Form WH-347 is available for this purpose from the Wage and Hour Division Web site at http://www.dol.gov/esa/whd/forms/wh347instr.htm or its successor site. The prime contractor is responsible for the submission of copies of payrolls by all subcontractors. Contractors and subcontractors shall maintain the full social security number and current address of each covered worker, and shall provide them upon request to HHS if the agency is a party to the contract, but if the agency is not such a party, the contractor will submit them to the applicant, sponsor, or owner, as the case may be, for transmission to HHS, the contractor, or the Wage and Hour Division of the Department of Labor for purposes of an investigation or audit of compliance with prevailing wage requirements. It is not a violation of this section for a prime contractor to require a subcontractor to provide addresses and social security numbers to the prime contractor for its own records, without weekly submission to the sponsoring government agency (or the applicant, sponsor, or owner).
- (B) Each payroll submitted shall be accompanied by a "Statement of Compliance," signed by the contractor or subcontractor or his or her agent who pays or supervises the payment of the persons employed under the contract and shall certify the following:
 - (1) That the payroll for the payroll period contains the information required to be provided under §5.5 (a)(3)(ii) of Regulations, 29 CFR part 5, the appropriate information is being maintained under §5.5 (a)(3)(i) of Regulations, 29 CFR part 5, and that such information is correct and complete;
 - (2) That each laborer or mechanic (including each helper, apprentice, and trainee) employed on the contract during the payroll period has been paid the full weekly wages earned, without rebate, either directly or indirectly, and that no deductions have been made either directly or indirectly from the full wages earned, other than permissible deductions as set forth in Regulations, 29 CFR part 3;
 - (3) That each laborer or mechanic has been paid not less than the applicable wage rates and fringe benefits or cash equivalents for the classification of work performed, as specified in the applicable wage determination incorporated into the contract.
- (C) The weekly submission of a properly executed certification set forth on the reverse side of Optional Form WH-347 shall satisfy the requirement for submission of the "Statement of Compliance" required by paragraph (a)(3)(ii)(B) of this section.
- (D) The falsification of any of the above certifications may subject the contractor or subcontractor to civil or criminal prosecution under section 1001 of title 18 and section 231 of title 31 of the United States Code.
- (iii) The contractor or subcontractor shall make the records required under paragraph (a)(3)(i) of this section available for inspection, copying, or transcription by authorized representatives of HHS or the Department of Labor, and shall permit such representatives to interview employees during working hours on the job. If the contractor or subcontractor fails to submit the required records or to make them available, the Federal agency may, after written notice to the contractor, sponsor, applicant, or owner,

take such action as may be necessary to cause the suspension of any further payment, advance, or guarantee of funds. Furthermore, failure to submit the required records upon request or to make such records available may be grounds for debarment action pursuant to 29 CFR 5.12.

(4) Apprentices and trainees-

- (i) Apprentices. Apprentices will be permitted to work at less than the predetermined rate for the work they performed when they are employed pursuant to and individually registered in a bona fide apprenticeship program registered with the U.S. Department of Labor, Employment and Training Administration, Office of Apprenticeship Training, Employer and Labor Services, or with a State Apprenticeship Agency recognized by the Office, or if a person is employed in his or her first 90 days of probationary employment as an apprentice in such an apprenticeship program, who is not individually registered in the program, but who has been certified by the Office of Apprenticeship Training, Employer and Labor Services or a State Apprenticeship Agency (where appropriate) to be eligible for probationary employment as an apprentice. The allowable ratio of apprentices to journeymen on the job site in any craft classification shall not be greater than the ratio permitted to the contractor as to the entire work force under the registered program. Any worker listed on a payroll at an apprentice wage rate, who is not registered or otherwise employed as stated above, shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any apprentice performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. Where a contractor is performing construction on a project in a locality other than that in which its program is registered, the ratios and wage rates (expressed in percentages of the journeyman's hourly rate) specified in the contractor's or subcontractor's registered program shall be observed. Every apprentice must be paid at not less than the rate specified in the registered program for the apprentice's level of progress, expressed as a percentage of the journeymen hourly rate specified in the applicable wage determination. Apprentices shall be paid fringe benefits in accordance with the provisions of the apprenticeship program. If the apprenticeship program does not specify fringe benefits, apprentices must be paid the full amount of fringe benefits listed on the wage determination for the applicable classification. If the Administrator determines that a different practice prevails for the applicable apprentice classification, fringes shall be paid in accordance with that determination. In the event the Office of Apprenticeship Training, Employer and Labor Services, or a State Apprenticeship Agency recognized by the Office, withdraws approval of an apprenticeship program, the contractor will no longer be permitted to utilize apprentices at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (ii) Trainees. Except as provided in 29 CFR 5.16, trainees will not be permitted to work at less than the predetermined rate for the work performed unless they are employed pursuant to and individually registered in a program which has received prior approval, evidenced by formal certification by the U.S. Department of Labor, Employment and Training Administration. The ratio of trainees to journeymen on the job site shall not be greater than permitted under the plan approved by the Employment and Training Administration. Every trainee must be paid at not less than the rate specified in the approved program for the trainee's level of progress, expressed as a percentage of the journeyman hourly rate specified in the applicable wage determination. Trainees shall be paid fringe benefits in accordance with the provisions of the trainee program. If the trainee program does not mention fringe benefits, trainees shall be paid the full amount of fringe benefits listed on the wage determination unless the Administrator of the Wage and Hour Division determines that there is an apprenticeship program associated with the corresponding journeyman wage rate on the wage determination which provides for less than full fringe benefits for apprentices. Any employee listed on the payroll at a trainee rate who is not registered and participating in a training plan approved by the Employment and Training Administration shall be paid not less than the applicable wage rate on the wage determination for the classification of work actually performed. In addition, any trainee performing work on the job site in excess of the ratio permitted under the registered program shall be paid not less than the applicable wage rate on the wage determination for the work actually performed. In the event the Employment and Training Administration withdraws approval of a training program,

- the contractor will no longer be permitted to utilize trainees at less than the applicable predetermined rate for the work performed until an acceptable program is approved.
- (iii) Equal employment opportunity. The utilization of apprentices, trainees and journeymen under this part shall be in conformity with the equal employment opportunity requirements of Executive Order 11246, as amended, and 29 CFR part 30.
- (5) Compliance with Copeland Act requirements. The contractor shall comply with the requirements of 29 CFR part 3, which are incorporated by reference in this contract.
- (6) Subcontracts. The contractor or subcontractor shall insert in any subcontracts the clauses contained in 29 CFR 5.5(a)(1) through (10) and such other clauses as the (write in the name of the Federal agency) may by appropriate instructions require, and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for the compliance by any subcontractor or lower tier subcontractor with all the contract clauses in 29 CFR 5.5.
- (7) Contract termination: debarment. A breach of the contract clauses in 29 CFR 5.5 may be grounds for termination of the contract, and for debarment as a contractor and a subcontractor as provided in 29 CFR 5.12.
- (8) Compliance with Davis-Bacon and Related Act requirements. All rulings and interpretations of the Davis-Bacon and Related Acts contained in 29 CFR parts 1, 3, and 5 are herein incorporated by reference in this contract.
- (9) Disputes concerning labor standards. Disputes arising out of the labor standards provisions of this contract shall not be subject to the general disputes clause of this contract. Such disputes shall be resolved in accordance with the procedures of the Department of Labor set forth in 29 CFR parts 5, 6, and 7. Disputes within the meaning of this clause include disputes between the contractor (or any of its subcontractors) and the contracting agency, the U.S. Department of Labor, or the employees or their representatives.
- (10) Certification of eligibility.
- (i) By entering into this contract, the contractor certifies that neither it (nor he or she) nor any person or firm who has an interest in the contractor's firm is a person or firm ineligible to be awarded Government contracts by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (ii) No part of this contract shall be subcontracted to any person or firm ineligible for award of a Government contract by virtue of section 3(a) of the Davis-Bacon Act or 29 CFR 5.12(a)(1).
- (iii) The penalty for making false statements is prescribed in the U.S. Criminal Code, 18 U.S.C. 1001.

CONTRACT WORK HOURS AND SAFETY STANDARDS ACT provisions (29 C.F.R. § 5.5(b))

(1) Overtime requirements.

No contractor or subcontractor contracting for any part of the contract work which may require or involve the employment of laborers or mechanics shall require or permit any such laborer or mechanic in any workweek in which he or she is employed on such work to work in excess of forty hours in such workweek unless such laborer or mechanic receives compensation at a rate not less than one and one-half times the basic rate of pay for all hours worked in excess of forty hours in such workweek.

(2) Violation; liability for unpaid wages; liquidated damages.

In the event of any violation of the clause set forth in paragraph (b)(1) of this section the contractor and any subcontractor responsible therefor shall be liable for the unpaid wages. In addition, such contractor and subcontractor shall be liable to the United States (in the case of work done under contract for the District of Columbia or a territory, to such District or to such territory), for liquidated damages. Such liquidated damages shall be computed with respect to each individual laborer or mechanic, including watchmen and guards, employed in violation of the clause set forth in paragraph (b)(1) of this section, in the sum of \$10 for each calendar day on which such individual was required or permitted to work in excess of the standard workweek of forty hours without payment of the overtime wages required by the clause set forth in paragraph (b)(1) of this section.

(3) Withholding for unpaid wages and liquidated damages.

The U.S. Department of Health and Human Services or the City shall upon its own action or upon written request of an authorized representative of the Department of Labor withhold or cause to be withheld, from any moneys payable on account of work performed by the contractor or subcontractor under any such contract or any other Federal contract with the same prime contractor, or any other federally-assisted contract subject to the Contract Work Hours and Safety Standards Act, which is held by the same prime contractor, such sums as may be determined to be necessary to satisfy any liabilities of such contractor or subcontractor for unpaid wages and liquidated damages as provided in the clause set forth in paragraph (b)(2) of this section.

(4) Subcontracts.

The contractor or subcontractor shall insert in any subcontracts the clauses set forth in paragraph (b)(1) through (4) of this section and also a clause requiring the subcontractors to include these clauses in any lower tier subcontracts. The prime contractor shall be responsible for compliance by any subcontractor or lower tier subcontractor with the clauses set forth in paragraphs (b)(1) through (4) of this section.

Attachment

2017-18 Head Start Program February 1, 2017-January 31, 2018

REVENUES:		Original Ordinance Budget	
p that i elephane has	4501100 Grants Federal - Operating	\$	22,483,945
	Subtotal Grant	\$	22,483,945
	6500000 In Kind Revenue	\$	5,620,987
	Subtotal (In Kind)	\$	5,620,987
	TOTAL REVENUES	\$	28,104,932

APPROPRIATIONS:

		Original
13800000xxxx 2017 Head Start COSA	Ord	inance Budget
5101010 Regular Salaries	\$	3,908,181
5101050 Language Skill Pay		15,600
5103005 FICA & Medicare Expense		298,977
5103010 Life Insurance		3,908
5103035 Personal Leave Buy Back		30,000
5103056 Transportation Allowance		1,080
5103105 Cell Phone Reimbursement		2,400
5105010 Retirement Exp		425,602
5170040 Civln Actv Healthcr		749,358
5202010 Temporary Services		25,663
5201040 Fees to Prof. Contractors		36,000
5202020 Contractual Services - COSA		108,000
5203040 Adv and Publications		5,000
5203050 Membership Dues		
5203060 Binding & Printing		20,000
5203070 Subs to Publications		2,000
5203090 Transportation Fees		12,000
5204010 Linen&Laundry		100
5204020 Maint & Rep-Comrcl		100
5204050 Maintenance -Buildings		48,779
5204090 Maint & Rep - Automotive		1,000
5205010 Mail and Parcel Post		200
5205020 Rental of Office Equipment		15,929
5207010 Official Travel (out of town)		15,000

5208530 Alarm and Security Services		45,079
5301020 M&R Parts Automotive		500
5302010 Office Supplies		32,605
5304010 Food		20,000
5304075 Computer Software		32,048
5304080 Other Commodities		-
5403040 Cellular Phone Service		13,000
5403060 Domain Names		200
5403510 Wireless Data Communications		6,400
5403545 Motor Fuel and Lubricants		300
5404520 Software Licenses		-
5404530 Gas and Electricity		28,834
5404540 Water and Sewer		6,505
5407032 DW Other		2,500
5501000 Cap <5000 - Computer Equipment		2,000
5501055 Cap <5000 - Mach & Equip Other		2,000
5501065 Cap <5000 - Mach & Equip Other		2,000
330 1003 Cap 13000 - Fulfillate & Fix	\$	5,914,848
13800000xxxx 2017 Head Start COSA - T&TA		
5201025 Education - Classes	\$	77,051
5201040 Fees to Prof. Contractors	\$	135,650
Total 13800000xxxx	\$	212,701
13800000xxxx 2017 Head Start - Edgewood ISD-Education		
5202020 Contractual Services	\$	4,171,411
Total 13800000xxxx	\$	4,171,411
, can recoverant	•	3,11.1,311
13800000xxxx 2017 Head Start - San Antonio ISD-Education		
5202020 Contractual Services	\$	12,053,245
Total 13800000xxxx	\$	12,053,245
13800000xxxx 2017 Head Start - Support Services		
5202020 Contractual Services-UHS	\$	20,000
5202020 Contractual Services-Metro Health	\$	111,740
ozozozo connaciaal convicco meno ricalin	\$	131,740
13800000xxxx 2017 Head Start - In Kind		
	•	F 600 007
6602025 In Kind Other Contractual Total 13800000xxxx	\$	5,620,987
10121 1/3000000000	\$	5,620,987
Total 13000000XXXX		

HEAD START PROGRAM FEBRUARY 1,2017 to JANUARY 31,2018 PERSONNEL COMPLEMENT

Positions 13800000xxxx 2017-2018 HEAD START COSA	Job Class	Current Number of Positions	Add/Delete		Budget FY 2017 Positions
ADMINISTRATIVE ASSISTANT I	0040	2			2
ADMINISTRATIVE ASSISTANT II	0041	0	1	*	1
MANAGEMENT ANALYST	0046	12	2	*	14
SPECIAL PROJECTS MANAGER	0866	3			3
SENIOR MANAGEMENT ANALYST	0999	6			6
HEAD START PROGRAM ADMINISTRATOR	2187	1			1
FISCAL MANAGER	2216	1			1
FISCAL ANALYST	2218	1			1
ACCOUNTANT	2220	1			1
FAMILY SUPPORT COORDINATOR	2290	1			1
FAMILY SUPPORT SUPERVISOR	2289	5			5
FAMILY SUPPORT WORKER	2283	57	(1)		56
CASE AIDE	0985	2			2
13800000xxxx 2017-2018 HEAD START COSA	-	92	2		94

TOTAL POSITIONS FOR HEAD START PROGRAM	92	2	94
TOTAL FOOTHOR FOR TEAD OTAK! FROOTHIN	- JL		34

^{*} Not new positions, 3 positions being moved from Early Head Start-Childcare Partner program to Head Start.